Sheffield Hallam University

Exploring the Need for Additional Nutritional Support in Adolescent Pregnancies

MARVIN-DOWLE, Katie

Available from the Sheffield Hallam University Research Archive (SHURA) at:

http://shura.shu.ac.uk/25121/

A Sheffield Hallam University thesis

This thesis is protected by copyright which belongs to the author.

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author.

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given.

Please visit http://shura.shu.ac.uk/25121/ and http://shura.shu.ac.uk/information.html for further details about copyright and re-use permissions.

Exploring the Need for Additional Nutritional Support in Adolescent Pregnancies

Katie Marvin-Dowle

A thesis submitted in partial fulfilment of the requirements of Sheffield Hallam University for the degree of Doctor of Philosophy

January 2019

Collaborating Organisations: Born in Bradford NIHR Collaboration for Leadership in Applied Health Research and Care Yorkshire and Humber University of Leeds

Candidate's declaration

I hereby declare that:

- 1. I have not been enrolled for another award of the University, or other academic or professional organisation, whilst undertaking my research degree.
- 2. None of the material contained in the thesis has been used in any other submission for an academic award.
- 3. I am aware of and understand the University's policy on plagiarism and certify that this thesis is my own work. The use of all published or other sources of material consulted have been properly and fully acknowledged.
- 4. The work undertaken towards the thesis has been conducted in accordance with the SHU Principles of Integrity in Research and the SHU Research Ethics Policy.
- 5. The word count of the thesis is approximately 40,000 plus approximately 20,000 additional words in included articles

Name	Katie Marvin-Dowle
Date	January 2019
Award	PhD
Faculty	Health and Wellbeing
Director(s) of Studies	Professor Hora Soltani

Acknowledgements

I would like to thank my research supervisors, Professor Hora Soltani and Dr Victoria Burley, and my statistical advisor Dr Karen Kilner, for their continued support, encouragement and vision. This program of work would not have been possible without their guidance, experience and expertise. I am grateful to Sheffield Hallam University for supporting this project and to the White Rose Scholarship program and the NIHR Collaboration for Leadership in Applied Health Research and Care (CLAHRC) Yorkshire and Humber, particularly the Born in Bradford team.

I appreciate the contributions of all of the participants, both in the Born in Bradford study and those who consented to be interviewed for this PhD. I am grateful to my friends and colleagues at Chestnut Court for sharing the journey with me, particularly to Dr Katie Shearn, Dr Helen Speak and Dr Martin Lamb for the support, debates, commiserations and celebrations, and to Rebecca Turner for keeping me sane during write up. To my friends Bryony O'Connell and Oonagh McClean, thank-you for always being ready with a shoulder and a glass of red.

A big thank-you to my parents, Gill and Glenn Marvin and my in-laws Sharon and Ken Dowle for their emotional and practical support, and, as always, for the babysitting.

Thank-you to my daughter Norah, whose arrival part way through this PhD gave me unique insight and renewed inspiration, and who keeps me going. Finally I am forever grateful to my wife, Emma, whose belief in me got me into this in the first place, as it has carried me through to the end.

3

Abstract

There is significant evidence to suggest that pregnancy outcomes among adolescent pregnant women may be less favourable compared to those of adult women. Incidence of preterm delivery, low birthweight and babies who are small for their gestational age have been identified as outcomes of particular concern for this population. One school of thought regarding this issue suggests that nutrition in adolescent pregnancies may have a role to play in reducing these differences and improving outcomes for adolescent mothers and their babies, however this area is in need of further research. The primary aim of this thesis therefore is to address the research question "*Is there a need for additional nutritional support in adolescent pregnancies?*"

This is a mixed methods study following a sequential explanatory design in which priority is given to the quantitative research phase. Following a systematic review of the academic literature, the program of research is carried out in two distinct phases and the findings from each integrated.

The quantitative research phase utilises secondary data from the Born in Bradford cohort study to investigate differences in maternal and neonatal outcomes and dietary patterns between adolescents (age ≤19 years) and an adult control group (age 20-34 years). Differences in outcomes were assessed using logistic regression models and differences in dietary pattern assessed using a combination of principle component analysis and regression models. The qualitative research phase was informed by the results of the quantitative phase and sought the perspectives of youth support professionals on supporting healthy eating in adolescent pregnancies through semi-structured interviews. Interviews were transcribed and analysed thematically. Findings from the program of research as a whole were then integrated narratively. The results of the systematic literature review confirmed that the nutritional status of pregnant adolescents is likely to be poor. The quantitative analyses identified significant differences in outcomes between adolescent and adult women, particularly a higher odds of delivering babies who were very (<32) weeks gestation) or extremely (<28 weeks gestation) pre-term (aOR 2.12, 95%) CI 1.06 to 4.25 and aOR 5.06, 95% CI 1.23 to 20.78, respectively), or at an extremely low birthweight (<1000g, aOR 4.13, 95% CI 1.41 to 12.11).

Results from the quantitative phase were used to inform the development of the qualitative interview guide following which the integration of findings from the program of research as a whole identified three overarching themes; social context, dietary pattern and need for support. Adolescent pregnant women were found to live in areas with higher levels of socio-economic deprivation, were more likely to be single parents and have unstable family backgrounds and housing insecurity. Adolescents were also found to consume higher levels of snack and processed foods, less fruit and vegetables, were likely to eat less regularly than older women and took fewer nutritional supplements. The need for additional social support, both in terms of the provision of services and facilitating young women to access existing services was also a key theme from the qualitative data.

The overarching themes identified in this program of work suggest that there are differences in diet quality between pregnant adolescents and adult pregnant women and that this may impact upon the health of the pregnancy. This research highlights that while additional support for eating healthily during pregnancy is important for this population, a more holistic approach which encompasses the range of complex issues faced by adolescent pregnant women is likely to have a greater impact on overall pregnancy health. This work makes an original contribution to knowledge by investigating an important large British cohort with unique characteristics to answer questions which have not previously been investigated in this data set. It has identified issues for pregnant adolescents which have not previously been documented and included the perspectives of a key professional group whose views have not previously been included in this debate.

Contents

Candidate's statement	2
Acknowledgements	3
Abstract	4
0.1 Article based thesis	10
0.2 Outputs, contribution statements and permissions	11
0.3 Abbreviations	14
0.4 List of tables	15
0.5 List of figures	15
Chapter 1 - Introduction	16
1.1 Introduction	16
1.2 Aims and Objectives	17
1.3 Overview of the programme of research	18
1.4 Thesis structure	19
1.5 Anticipated contribution of new knowledge	20
Chapter Two - Background	21
2.1 Introduction	21
2.2 Adolescent pregnancy outcomes	26
2.3 Role of nutrition in pregnancy	30
2.4 The nutritional status of women of childbearing age in the UK	36
2.5 Nutritional issues for pregnant adolescents	41
2.6 Summary	43

Chapter Three - Systematic Literature Review	44
3.1 Introduction	44
3.2 Publication and Impact	44
3.3 Published paper: Article A	44
3.4 Summary and Implications for Thesis	45
Chapter 4 - Methodology	46
4.1 Introduction	46
4.2 Research paradigms	46
4.3 Mixed methods research	48
4.4 Methods	48
4.5 Data sharing, consent and ethics	53
4.6 Summary	54
Chapter 5 - Maternal and neonatal outcomes	55
5.1 Introduction	55
5.2 Publication and Impact	55
5.3 Published paper: Article B	55
5.4 Summary and implications for thesis	56
Chapter 6 - Dietary patterns	57
6.1 Introduction	57
6.2 Publication and Impact	57
6.3 Published paper: Article C	57

6.4 Summary and implications for thesis	58
6.5 Development of qualitative research phase	58
Chapter 7 - Supporting Healthy Adolescent Pregnancies	59
7.1 Introduction	59
7.2 Publication and Impact	60
7.3 Published paper: Article D	60
7.4 Summary and implications for thesis	62
Chapter 8 - Integration of research Findings	93
8.1 Introduction	93
8.2 Integration of findings	93
8.2.1 Social context	
8.2.2 Dietary pattern	
8.2.3 Support	95
8.3 Summary	96
Chapter 9 - Discussion	97
9.1 Introduction	
9.2 Discussion of key findings	
9.3 Contributions of this research	108
9.4 Strengths and limitations	109
9.5 Recommendations for future work	110
9.6 Summary	112

Chapter 10 - Conclusions 113
10.1 Introduction 113
10.2 Review of aims and objectives 113
10.3 Impact to date 114
10.4 Summary 115
References 116
Appendix 1
Appendix 2 151
Appendix 3 152
Appendix 4 153
Appendix 5 154
Appendix 6 155
Appendix 7 156
Appendix 8 157
Appendix 9

0.1 Article based thesis

The research in this thesis is presented in an article-based format which is somewhat different to the traditional monograph Ph.D. presentation; however the regulations and assessment criteria for the award remain the same. The guidance produced by Sheffield Hallam University on article-based theses describes the approach as a thesis format in which a number of articles (between three and five) are produced by the Ph.D. candidate during their period of candidature. These articles will either already be published or will be accepted for publication in peer reviewed journals at the time of submission. An article-based thesis will usually comprise an introduction including an explanation of the research question(s), the research subject, relevant literature and methodology and a concluding chapter in which the results of the research are summarised and discussed.

This article-based thesis includes three papers which have been published in peer-reviewed journals and a fourth which is currently under review for publication. The articles report on the four studies which together comprise this program of research and are included within the thesis with an accompanying narrative.

Article A	
Title	Nutrient intakes and nutritional biomarkers in
	pregnant adolescents: a systematic review of studies
	in developed countries
Authors	Katie Marvin-Dowle, Dr Victoria Burley, Prof. Hora
	Soltani
Full Reference	Marvin-Dowle K, Burley VJ, Soltani H. Nutrient
	intakes and nutritional biomarkers in pregnant
	adolescents: a systematic review of studies in
	developed countries. BMC pregnancy and childbirth.
	2016 Dec;16(1):268.
Contribution Statement	This piece of work was led by Katie Marvin-Dowle
	who oversaw and was actively involved in all of the
	research processes at each stage
Permissions	The open access articles published in BMC's
	journals are made available under the Creative
	Commons Attribution (CC-BY) license, which means
	they are accessible online without any restrictions
	and can be re-used in any way, subject only to
	proper attribution (which, in an academic context,
	usually means citation).

0.2 Outputs, contribution statements and permissions

Article B	
Title	Impact of adolescent age on maternal and neonatal
	outcomes in the Born in Bradford cohort
Authors	Katie Marvin-Dowle, Dr Karen Kilner, Dr Victoria
	Burley, Prof. Hora Soltani
Full Reference	Marvin-Dowle K, Kilner K, Burley VJ, Soltani H.
	Impact of adolescent age on maternal and neonatal
	outcomes in the Born in Bradford cohort. BMJ open.
	2018 Mar 1;8(3):e016258.
Contribution Statement	This piece of work was led by Katie Marvin-Dowle
	who oversaw and was actively involved in all of the
	research processes at each stage
Permissions	The BMJ Author License allows authors to use their
	articles for their own non-commercial purposes
	without seeking permission from the BMJ – the only
	condition being that a full reference or link to the
	original is included.

Article C	
Title	Differences in dietary pattern by maternal age in
	the Born in Bradford cohort: A comparative analysis.
Authors	Katie Marvin-Dowle, Dr Karen Kilner, Dr Victoria
	Burley, Prof. Hora Soltani
Full Reference	Marvin-Dowle K, Kilner K, Burley V, Soltani H
	(2018) Differences in dietary pattern by maternal age
	in the Born in Bradford cohort: A comparative
	analysis. PLoS ONE 13(12): e0208879.
Contribution Statement	This piece of work was led by Katie Marvin-Dowle
	who oversaw and was actively involved in all of the
	research processes at each stage
Permissions	The open access articles published in PLoS One are
	made available under the Creative Commons
	Attribution (CC-BY) license, which means they are
	accessible online without any restrictions and can be
	re-used in any way, subject only to proper attribution
	(which, in an academic context, usually means
	citation).

0.3 Abbreviations

ACE	Adverse Childhood Experience
AOAC	Association of Official Analytical Chemists
APGAR	Appearance, Pulse, Grimace, Activity, Respiration
BIB	Born in Bradford
BMI	Body Mass Index
CASP	Critical Appraisal Skills Programme
CLAHRC	Collaboration for Leadership in Applied Health Research
	and Care
COMA	Committee on Medical Aspects of Food and Nutrition
DLW	Doubly Labelled Water
DQI	Diet Quality Index
DRV	Dietary Reference Value
EAR	Estimated Average Requirement
FFQ	Food Frequency Questionnaire
FNP	Family Nurse Practitioner/Family Nurse Partnership
HCP	Health Care Professional
HSE	Health Survey for England
IUGR	Intra-Uterine Growth Restriction
LRNI	Lower Reference Nutrient Intake
MFA	Maternal-Foetal Attachment
NATSAL	National Sexual Attitudes and Lifestyles Survey
NDNS	National Diet and Nutrition Survey
NHS	National Health Service
NICU	Neonatal Intensive Care Unit
NIHR	National Institute for Health Research
PCA	Principal Component Analysis
RDA	Recommended Daily Allowance
RNI	Reference Nutrient Intake
SACN	Scientific Advisory Committee on Nutrition
SCBU	Special Care Baby Unit
SES	Socio-Economic Status
TEE	Total Energy Expenditure

UK	United Kingdom
US	United States
WHO	World Health Organisation

0.4 List of tables

Table 1	
Recommended nutrient intake for micronutrients for women of childbearing age	
and pregnancy increments	35
Table 2	
Micronutrient intake from food	39

0.5 List of figures

Figure 1.	
Overview of the programme of research	18
Figure 2	
Live births to women aged 15-19 by mothers region of residence, 2015	21
Figure 3	
Live births to women aged 10-19 by mothers country of residence, 2016	22
Figure 4	
Under 18 conception rate in England and Wales, 1969 - 2016	23
Figure 5	
Method of derivation of dietary reference values	34
Figure 7	
Research design	49

Chapter 1 - Introduction

1.1 Introduction

Pregnancy and childbirth among adolescents in Western countries is often viewed as a social problem with women who have a child during the teenage years being more likely to suffer social isolation, poverty, lower levels of educational achievement and be unemployed or work in low paid jobs¹. In addition to the potential for adverse social outcomes the literature suggests that young maternal age may be an independent risk factor for adverse maternal and neonatal health outcomes. One systematic review² aiming to assess the relationship between early first childbirth and increased risk of poor pregnancy outcomes found that there was considerable evidence to suggest very young maternal age (<15 years or less than 2 years after menarche) had a negative effect on both maternal and foetal growth and infant survival. It is suggested that young women who are still themselves growing may compete with the foetus for nutrients, which may in turn impair foetal growth and result in low birth weight babies or babies who are small for their gestational age. The review also found a moderate relationship between young maternal age and anaemia, premature birth and neonatal mortality. These findings regarding foetal growth suggest maternal nutrition during adolescent pregnancy may impact on outcomes for both mother and child.

It has long been established that nutritional status in pregnancy has an important influence on birth outcomes³; particularly in terms of foetal growth and infant survival and that mothers need to consume adequate, yet not excessive, nutrition in order for their babies to thrive. There are two potential areas where nutritional differences may exist between adolescent and adult pregnancies. Firstly there is evidence to suggest that the general diet of adolescent girls may be of poorer quality than that of older women. The latest results of the National Diet and Nutrition Survey (years 7-8, 2014/15 - 2015/16)⁴ showed that girls aged 11-18 years consumed 2.8 portions of fruit and vegetables per day compared to 4.2 portions in women aged 19-64 years. A higher proportion of adolescent girls also had intakes of key vitamins and minerals below the lower

reference nutrient intake level than adult women, including vitamin A, riboflavin, vitamin B12, folate, iron, calcium, magnesium, potassium, zinc and iodine.

Secondly there is evidence to suggest that because some adolescent women will still themselves be growing during pregnancy they may compete with the developing embryo to satisfy their own growth needs. A study of a US based prospective cohort⁵ examining the relationship between maternal growth and outcomes in adolescent pregnancies found that continued maternal growth in adolescents affected nutrient partitioning between mother and child and that this, combined with poor nutritional status in pregnant adolescents, impacted negatively upon foetal growth and prematurity. This position is however contested by a UK based study⁶ which found that the average birthweight of babies born to adolescent women who were still growing was in fact higher than those of young women who had finished growing, and that gestational weight gain (rather than maternal growth in stature) was more significantly correlated with birthweight. Average infant birthweight in the non-growing adolescent group was however significantly lower than that in adult controls. This suggests that the relationship between maternal and foetal growth is very complex and further work to explore this relationship is needed.

1.2 Aims and Objectives

Therefore the primary aim of this programme of research is to address the research question:

Is there a need for additional nutritional support in adolescent pregnancies?

In order to consider this question the research will focus on three specific objectives:

- Establish what is known in the current research literature regarding the nutritional status of pregnant adolescents
- Examine differences in dietary intake and birth outcomes between adolescent and adult mothers in a large cohort data set
- Explore the perspectives of health and social care professionals of providing nutritional support for pregnant adolescents

1.3 Overview of the programme of research

Original research is presented in this thesis in an article based format. A collection of four research articles (three published and one that has been submitted for publication) in international, peer reviewed, scientific journals which are presented in their original forms. The overall findings of the research are drawn together and discussed, including discussion of the original contribution to knowledge offered by this work. Figure 1 shows an overview of the programme of research and how it is presented in theirs.



Figure 1. Overview of the programme of research

1.4 Thesis structure

Chapter 1 provides an introduction to the thesis, a brief rationale and the aims and objectives of the programme of research and an overview of the included studies.

Chapter 2 contains a comprehensive review of the relevant existing literature and identifies gaps therein.

Chapter 3 contains Article A "Nutrient intakes and nutritional biomarkers in pregnant adolescents: a systematic review of studies in developed countries" which was accepted for publication in BMC Pregnancy and Childbirth in September 2016.

Chapter 4 details the methodology that underpins the programme of research and discussed the philosophical approach and research design, and how this influenced the choice of methods

Chapter 5 contains Article B "Impact of adolescent age on maternal and neonatal outcomes in the Born in Bradford cohort" which was accepted for publication in BMJ Open in March 2018.

Chapter 6 contains Article C "Differences in dietary pattern by maternal age in the Born in Bradford cohort: a comparative analysis" which was accepted for publication in PLoS One in December 2018. This chapter then details how the findings from the two quantitative studies inform the qualitative enquiry.

Chapter 7 contains Article D "Perspectives of youth support professionals on supporting healthy eating in adolescent pregnancies" which at the time of submission was under review for publication in Evidence Based Midwifery.

Chapter 8 provides an integration of findings from the quantitative and qualitative phases of analysis and develops overarching themes from the research program.

Chapter 9 gives a summary of the key research findings in a discussion chapter, highlights the strengths and limitations of the thesis and discusses implications and recommendations for future research and practice development.

Chapter 10 concludes the thesis and reviews the aims and objectives set out in the introduction. The contribution of new knowledge made by this programme of research is clearly highlighted.

1.5 Anticipated contribution of new knowledge

It is anticipated that this programme of research will add to the body of literature regarding differential outcomes between adolescent and adult pregnancies by carrying out this analysis in a large, UK based cohort which has not previously been examined in this way. Furthermore this work will generate new knowledge regarding young women's patterns of eating during pregnancy and explore new perspectives on how young women can be supported to make positive dietary changes to improve the health of their pregnancy.

Chapter Two - Background

2.1 Introduction

Adolescent pregnancies are defined as pregnancies occurring in young women aged 19 years or under at the time of conception⁷. Adolescent pregnancy is a global issue with approximately 2 million girls under the age of 16 and 16 million between the ages of 15 and 19 becoming pregnant every year. There are significant disparities in adolescent fertility rates in different regions of the world with rates as high as 106 live births per 1,000 women aged 15-19 years in sub-Saharan Africa down to 6.2 births per 1,000 in Eastern Asia⁷. Figure 2 shows global fertility rates by mothers region of residence. Europe has the second lowest fertility rate amongst the global regions; there is however also significant variation within the region. Despite recent reductions, the UK still has the highest adolescent fertility rate in Western Europe, with only Bulgaria, Romania, Slovakia, Hungary, Latvia and Lithuania reporting higher rates⁸ (Figure 3).

Figure 2. Live births to women aged 15-19 by mothers region of residence, 2015



Figure 3. Live births to women aged 10-19 by mother's country of residence,

<u>2016</u>



Both globally and locally, adolescent pregnancies are associated with socioeconomic deprivation⁹. Those young women who are living in poverty, with low levels of education and in marginalised communities are most likely to become pregnant at an early age and continue to experience high levels of socioeconomic deprivation. In the UK context, evidence also suggests that those young women who do conceive and become mothers during adolescence are more likely to live in poverty¹⁰, be socially isolated¹¹ and experience poor mental health¹² compared to women who have their first child at age 20 or older, meaning they are more vulnerable to experiencing poorer outcomes post-partum.

There have been significant reductions in the rates of conceptions to young women in the UK which have fallen by more than 50% in the last 10 years¹³ with the under 18 conception rate now being at its lowest level since comparable records began in 1969, when the recorded rate was 18.9 conceptions per 1,000 female in the age group. Figure 4 shows the conception rate per 1,000 females aged 15-17 from 1969 to 2016. The data show that following a sharp decline in the early 1970's, coinciding with the wider availability of the contraceptive pill to unmarried women, rates remained largely static until 2007, where the current downward trend began its trajectory. While this is a positive trend, evidence from the National Survey of Sexual Attitudes and Lifestyles suggests that the majority of conceptions to young women aged 16-19 are still unplanned and that this is associated with less favourable behaviours and circumstances such as smoking, drug use and lower educational attainment¹⁴.



Figure 4. Under 18 conception rate in England and Wales, 1969 - 2016

23

In 1999 the UK government launched a national, ten year teenage pregnancy strategy with the aim to halve the under 18 conception rate¹⁵. The strategy was successful with small inroads being made each year until a steep decline beginning in 2007 seeing the 1998 baseline rate halved by 2014. The strategy was designed to be multifaceted and incorporate actions relating to multiple influences on young people's lives. The four central themes of the programme were:

- Joined up action nationally and locally
- Improved sex and relationships education and access to contraception
- A targeted communications campaign for young people and parents
- Coordinated support for young parents

Due to the complex nature of both the problem and the interventions it is not possible to quantify the causal mechanisms or to identify which elements of the interventions had the largest effect on conceptions. However in a paper discussing the implementation of the programme, Hadley et.al¹⁶ discuss the six key features contributing to the success of the strategy which are briefly outlined below.

Creating an opportunity for action

The teenage pregnancy strategy was an ambitious, long term plan which required cooperation between numerous agencies in all sectors. This was established from the start with the strategy described as being *'implemented with energy and enthusiasm, in an atmosphere of cooperation and consensus of those involved*¹⁷.

Developing evidence based strategy

The strategy was clearly based on the best available evidence, including evidence of why alternative approaches, such as abstinence only education, were unlikely to be successful.

Effective implementation

The strategy included a comprehensive strategic action plan giving clear implementation instructions to all key agencies.

Regularly reviewing programmes

Regular evaluations were carried out during the course of the strategy, including a 'deep dive' comparing high and low performing areas carried out half way through the ten year programme. Crucially, adjustments were then made in light of new evidence.

Embedding the strategy in wider government programmes

The strategy was integral to improving wider outcomes for children and young people. This was supported by legislative changes which obliged local government to work cooperatively with other agencies, therefore maximising the available resources to be utilised for common goals.

Providing leadership throughout the programme

In a long-term strategy in which results were likely to be slow to be seen, high level commitment was essential to ensure momentum was maintained to allow in impact of the strategy to become apparent.

Despite a successful strategy to reduce under 18 conceptions in the UK the rate remains the highest in Western Europe, meaning there are still a significant number of adolescent women becoming pregnant and delivering babies. It is therefore important that work to investigate the health status of young women and their children and how this may be improved is carried out. In this chapter an overview of the literature on differences in clinical maternal and neonatal outcomes between adolescent and adult pregnancies is given. Following this an overview of nutrition during adolescence and discussion of what is known about the role of nutrition in healthy pregnancies, with particular reference to specific issues in adolescent pregnancies will be presented.

2.2 Adolescent pregnancy outcomes

2.2.1 Foetal Growth and development

Foetal growth and development plays a major part in the health and wellbeing of individuals both in terms of immediate birth outcomes and infant survival as well as longer term measures. Babies born with extremely low birthweight and those who are extremely preterm are at significantly higher risk of dying within the first few months of life compared to those born larger and later¹⁸. Survival rates of babies born preterm (before 37 completed weeks of gestation) in the UK increase rapidly with each additional week of gestation¹⁹ meaning that understanding the causes of extremely preterm delivery is of paramount importance to reducing perinatal deaths. Longer term, outcomes for children born very preterm and/or with very low birthweight have been shown to include difficulties with behaviour and educational achievement²⁰ and low birthweight has been linked to a number of chronic conditions in adulthood such as ischaemic heart disease, hypertension and central adiposity²¹.

The literature exploring the impact of maternal age on preterm delivery, infant birthweight and intrauterine growth restriction (IUGR) has established that some important differences exist between babies born to adolescent and adult women. Numerous studies have reported mean birthweight in babies born to adolescent women to be significantly lower than those born to adult mothers^{22,23,24, 25}. Adolescent women have been shown to be at higher risk of pre-term delivery compared to adult women with the youngest adolescents being particularly vulnerable. Beeckman et.al²⁶ in a Belgian cohort consisting of 8,586 births found maternal age \leq 19 to be a predictive factor for pre-term delivery after controlling for confounding factors (aOR 2.15, CI 1.31 - 3.53) while a Swedish birth register study²⁷ with 798,674 participants found that young women aged <17 years had increased odds of pre-term delivery (aOR 1.46, CI 1.24 - 1.72) compared to women aged 25-29 years, whereas this was not the case for young women aged 17-19 years. The same study also detected differences in the odds of very (<32 completed weeks gestation) and extremely (<28 completed weeks gestation) pre-term deliveries with the youngest

adolescents (<17 years) having increased odds of all pre-term outcomes. The literature regarding low birthweight shows similar patterns with all adolescents being at increased risk compared to adults^{22,28}, and the youngest women being the most likely to deliver low or very low birthweight babies²⁹.

2.2.2 Perinatal mortality

Perinatal mortality is defined as death of the foetus or neonate between 22 completed weeks of gestation and 7 days after birth³⁰, with deaths up to 28 days following birth being termed neonatal death. There is some evidence to suggest that the risk of stillbirth (deaths which occur before birth or during delivery) is higher among adolescent mothers compared to adults^{31,32, 33}, however a number of studies have failed to detect any differences between groups on this variable^{34,35,36}. One large study (N= 5,874,203) based in the USA³⁷ found higher odds of stillbirth at both extremes of maternal age (≤19 years and ≥35 years) suggesting that the impact of young maternal age may be masked by the impact of older maternal age in studies which do not exclude older gravidas from the comparison group. This is further evidenced by a metaanalysis including data from 96 studies of factors associated with stillbirth in the five high income countries with the highest stillbirth rates (Australia, Canada, Netherlands, UK and USA). This review reported higher population-attributable risk (7-11%) in women aged over 35, this study did not however examine adolescent maternal age as a potential factor³⁸. A meta-analysis of outcomes by parity and maternal age in low and middle income countries found an association between nulliparous women age <18 and neonatal mortality (aOR 1.49, CI 1.13 to 1.97), however no similar meta-analyses have been carried out in high income countries.

There is also some evidence to suggest that perinatal and neonatal mortality may be higher among babies born to adolescent women not least in part due to the relationship between these types of death and pre-term, low birthweight babies, which it has already been established are more common in this population. Chen et.al³⁹ found that the odds of neonatal death were higher in all adolescent age groups studied (10-15 years, 16-17 years and 18-19 years) but that maternal age was no longer predictive of neonatal death once gestational age and birthweight were included in the regression model.

2.2.3 Neonatal morbidity

The most commonly used method of assessing the condition of a new born at birth is the APGAR score. The APGAR score is a value from 0 to 10 which is derived from the sum of scores out of 2 for each of the five components (Appearance, Pulse, Grimace, Activity, Respiration). Assessments are usually made at 1 and 5 minutes following birth and scores below 7 indicate cause for concern⁴⁰. Evidence for lower APGAR scores in babies born to adolescent mothers is mixed; while one large study reports greater relative risk of scores under 7 and under 4 in babies born to young women aged under 17³⁹, a number of other similar studies failed to detect a significant difference between groups for this variable^{41,42,43}.

A further clear indicator for neonatal morbidity is admission to a neonatal intensive care unit (NICU) or special care baby unit (SCBU). Babies born to adolescent women have been shown to be at higher risk of admission to NICU/SCBU^{42,44,45}however similarly to reports of perinatal and neonatal mortality this variable is intrinsically linked to pre-term and low birthweight deliveries.

2.2.4 Maternal mortality and morbidity

Maternal mortality in the UK is very low at less than 9 deaths per 100,000 women occurring during pregnancy or up to six weeks after the end of pregnancy. Two thirds of these deaths can be attributed to pre-existing physical or mental health problems⁴⁶. While maternal mortality is low, maternal morbidity due to conditions related to pregnancy is an important consideration, and one for which variations by maternal age can be observed.

Pre-eclamsia is a hypertensive disorder⁴⁷ which is a major cause of maternal morbidity and mortality worldwide⁴⁸. While mortality in developed countries is low, pre-eclampsia has been associated with severe maternal morbidity such as

strokes and adverse neonatal outcomes such as prematurity and intrauterine growth restriction⁴⁸. Pre-eclampsia is also indicated as a marker for increased risk of cardiovascular and metabolic diseases later in life⁴⁹. Risk of pre-eclampsia in adolescent mothers has been shown to be lower than in older women. A cohort study by De Vienne et.al⁵⁰ showed a clear linear association between maternal age and relative risk of pre-eclampsia reporting relative risk of 0.080 (95% CI 0.01 - 0.6) for 16 year olds and 0.44 (95% CI 0.23 - 0.8) for 18 year olds in the cohort compared to the reference group (age 20) and a relative risk of 1.51 (95% CI 0.99 - 2.3) at age 25.

Gestational diabetes is another potentially serious condition of pregnancy which is associated with both immediate and longer term complications for the mother and the infant. Macrosomia, shoulder dystocia and birth injuries are common short term complications with evidence suggesting increased risk of obesity and impairment of glucose tolerance (leading to type 2 diabetes) for both mother and child⁵¹ in the longer term. Gestational diabetes has also been shown to be positively correlated with maternal age; therefore adolescent women are at lower risk of developing the condition than older women⁵².

There is a significant body of evidence showing that these and other maternal conditions such as pregnancy induced hypertension⁵³ and post-partum haemorrhage⁵⁰ occur less frequently in adolescent pregnancies than for adult women. This suggests that while maternal conditions of pregnancy are important to consider, they are unlikely to be a major cause of higher rates of adverse neonatal outcomes in this population.

2.2.5 Mode of birth

Similarly to maternal morbidity, the risk of adverse obstetric outcomes such assisted birth is lower in adolescent mothers than older women. Caesarean delivery is associated with higher rates of post-natal complications and increased recovery time for the mother⁵⁴. Instrumental deliveries, while necessary to prevent serious neonatal complications, are associated with a higher prevalence of birth injuries and maternal rehospitalisation⁵⁵.

The literature presented regarding pregnancy outcomes shows a reasonable basis to consider that young maternal age may have a bearing on outcomes, with the strongest evidence concerning preterm deliveries and low birthweight infants. The studies in this area in high income countries are typically large cohorts or utilise birth registry data and therefore include a large number of participants, suggesting that the evidence is likely to be robust. The majority of studies also make efforts to control the analysis for potentially confounding factors, particularly the influence of socioeconomic deprivation. This said, very few studies have investigated the relationship between maternal age and extreme prematurity or extremely low birth weights and the evidence regarding the risk of stillbirth is less conclusive. It may also be the case that the inclusion of women aged 35 and over in comparator groups may mask some differences in outcomes between adolescents and women aged 20-34, for whom there is no evidence of age related complications.

2.3 Role of nutrition in pregnancy

2.3.1 The importance of optimal nutrition before and during pregnancy

It has long been established in the academic literature that individuals need to consume adequate but not excessive nutrition to support growth and repair and to provide energy for metabolic functions and everyday tasks. This is particularly important during pregnancy when dietary intake needs to not only support the mother but also the growth and development of the foetus³. The assessment of physiological demand for nutrients during pregnancy is complex due to the significant physiological and anatomical changes which take place during this period. The development and maintenance of placental and mammary tissues, body composition changes such as increases in fat storage, and energy mobilisation and the changing needs of the developing foetus mean that not only are the needs of pregnant women different to those who are not pregnant, physiological demand and therefore necessary nutrient intake change rapidly through the course of the pregnancy⁵⁶.

During the first trimester tissue patterns and organ systems are established through the process of embryogenesis; in the second trimester there is a significant increase in the size of the foetus and cells adapt to their specific functions, following which in the third trimester the size of the foetus continues to increase rapidly and organ systems mature ready for life in the outside world. Each of these developmental stages requires specific micronutrients to ensure optimum growth and development and reduce the incidence of adverse outcomes⁵⁷.

Micronutrient deficiency during pregnancy can have serious and long lasting impacts on the health of the child. The role of folic acid both pre-conception and during the first trimester in preventing neural tube birth defects has been well documented⁵⁸. In addition adequate levels of vitamin A have been shown to be essential for the development and maturation of the lungs⁵⁹ and insufficient intake of vitamin A has been linked to low birthweight. Selenium, alongside other dietary antioxidants (zinc, vitamin C, vitamin E, beta-carotene) has been shown to play a role in regulating the antioxidant status of the mother, reducing oxidative stress and inflammation and in turn reducing the likelihood of a number of adverse pregnancy outcomes including birth defects, pre-eclampsia and miscarriage⁶⁰. Both calcium and 25-hydroxy vitamin D have an important function in the development of the foetal skeleton and in maintaining bone health in the mother⁶¹. Calcium intake has also been linked to the incidence of pregnancy induced hypertensive disorders⁶² and 25-hydroxy vitamin D to preeclampsia⁶³ and gestational diabetes⁶⁴. Iron deficiency anaemia has been linked to a number of adverse pregnancy outcomes. Insufficient iron in early pregnancy impacts upon the size of the placenta which in turn is indicated in intrauterine growth restriction. Anaemia in the second trimester has also been associated with pre-term delivery and there is some evidence to suggest that iron has a crucial role in brain development and the prevention of cognitive and behavioural disorders in later life⁶⁵.

2.3.2 Foetal origins of adult disease

Epidemiological research of birth and death records has also led to the observation of correlations between environmental factors during gestation and health conditions in later life. These observations, which were initially made by Barker et.al⁶⁶ regarding links between undernutrition during pregnancy and ischaemic heart disease in the offspring, is referred to as the 'foetal origins of adult disease hypothesis'. This early hypothesis that states that *"an environment which produces poor fetal and infant growth is followed by an adult environment that determines high risk for ischemic heart disease"* (p. 579)⁶⁷ has since been developed into a field of study in its own right investigating the links between foetal environment in-utero and adverse outcomes later in life known as developmental origins of health and disease.

Early work^{68,69} in this field investigated cohorts which had been affected by population level problems with food access to make comparisons between babies conceived in times of significant food shortage, such as a Dutch cohort exposed to famine towards the end of World War II, to those conceived in the same area at a time not impacted by food shortage. The results of these studies showed that foetal undernutrition may have significant impacts on the development of organs and systems of the body, and that this was specific to the stage of pregnancy. Those who were conceived prior to the onset of famine and so were undernourished in late gestation were found to be at higher risk for impaired glucose tolerance and insulin resistance in adulthood. Those conceived during the famine however were at higher risk for cardiovascular issues in adulthood. These findings suggest a very complex relationship between foetal environment and how this interacts with genetic influences and postnatal environments to accentuate or mediate risk for adverse health outcomes. In more recent work on the developmental origins of health and disease the Southampton Women's Survey have prospectively collected preconception data from women followed by detailed foetal measures during pregnancy as well as infancy and childhood follow ups⁷⁰. Work from this cohort has identified mechanisms by which maternal central adiposity and diet quality affects the regulation of maternal blood flow from the liver to the developing brain, thus suggesting that the foetus may adapt their cardiovascular response

depending on the availability of nutrients⁷¹. In this research the participants were considered to have low risk pregnancies, meaning that women with higher risk pregnancies or mothers with higher nutritional needs may result in further foetal adaptations. It is therefore clear that nutrition during pregnancy is an important modifiable factor influencing both the immediate and long term health of the offspring. The rest of this chapter therefore considers nutritional recommendations for pregnancy and the nutritional status of women of childbearing age in the UK.

2.3.3 Dietary reference values in pregnancy

The nutritional needs of populations are difficult to estimate due to significant variations between individuals. Dietary reference values (DRVs) are therefore derived based on the estimated distribution of requirements among the individuals in a given population. In the UK the last full review of DRVs was conducted by the Committee on Medical Aspects of Food Policy (COMA) and published in 1991⁷²; however, guidance on some individual nutrients has since been updated. Figure 5, reproduced from the COMA report, illustrates the method of derivation of the three most commonly used DRVs. The actual requirements for nutrients in a population are assumed to be normally distributed therefore estimated average requirement (EAR) is the estimated population mean. The reference nutrient intake (RNI) is set at two standard deviations above the mean, meaning that this is a level which is likely to be adequate for approximately 95% of the population. The lower reference nutrient intake (LNRI) is set at two standard deviations below the mean, meaning this is a level of intake which is unlikely to meet the needs of 95% of the population. DRVs are derived for population groups based on age and sex with any increase in requirements to support pregnancy being reported as increments. These increments apply across the board with no account given for the age of the mother as any age related differences in requirement should be accounted for in the pre-pregnancy DRVs. RNIs for vitamins and minerals for adolescent and adult women and any pregnancy increments are shown in table 1.

Figure 5. Method of derivation of dietary reference values



Table 1. Recommended nutrient intake for micronutrients for women of childbearing age and pregnancy increments

	UK RNI per day		
	Females 15-18	Females 19-50	Pregnancy Increment
Vitamins			
Folate	200µg	200µg	100µg
Niacin	14mg	13mg	NA
Riboflavin	1.1mg	1.1mg	0.3mg
Thiamin	0.8mg	0.8mg	0.1mg
Vitamin A	600µg	600µg	100µg
Vitamin B12	1.5µg	1.5µg	NA
Vitamin B6	1.2mg	1.2mg	NA
Vitamin C	40mg	40mg	10mg
Vitamin D	10µg [†]	10µg [†]	NA
Minerals			
Calcium	800mg	700mg	NA
Copper	1mg	1.2mg	NA
Iodine	140µg	140µg	NA
Iron	14.8mg	14.8mg	NA
Magnesium	300mg	270mg	NA
Phospherous	625mg	550mg	NA
Potassium	3.5g	3.5g	NA
Selenium	60µg	60µg	NA
Sodium	1.6g	1.6g	NA
Zinc	7mg	7mg	NA
TUpdated in SACN Vitamin D and health report, 2016			

For the majority of nutrients there is no increment specified for pregnancy. This does not necessarily mean that there is no increase in requirement during pregnancy but that it is anticipated that any additional demand should be *"met by normal adaptation or increased efficiency of utilisation, or from stores of the nutrient" (pg.7)*. This is therefore an important consideration in cases where the pre-pregnancy nutrient intake may not be sufficient therefore meaning nutrient stores are also unlikely to be adequate. It is also important to consider that there is a level of uncertainty regarding estimates of nutrient requirement during pregnancy. This is due both to large individual differences in pre-pregnancy nutrient stores and body composition which alter need⁷³ and due to ethical and practical difficulties with collecting data from pregnant women.
The Scientific Advisory Committee for Nutrition (SACN) revised the DRVs for energy in 2011⁷⁴ and now reports the EAR for children and young people by single year of age. For young women the EAR for energy at age 11 is 2032kcal per day which rises steadily to 2462kcal per day at age 18; the EAR for adult women aged 19-34 is 2175 Kcal per day. There is an advised increment for pregnancy of 200kcal per day in the third trimester only. The EAR for protein for both adolescent and adult females is 45g per day with an increment of 6g for pregnancy. DRVs for fat and carbohydrate are expressed as percentages of total energy intake. It is recommended that calories from fat should make up no more than 35% of total calories, with no more than 11% being from saturated fatty acids. Total calories from carbohydrate are recommended to make up approximately 50% of energy intake, a figure which was maintained in a recent review of recommendations by SACN⁷⁵. In this review the recommendations for consumption of free sugars (that is sugars added to food or drinks plus those found naturally in honey, syrups and unsweetened fruit juices) was revised to no more than 5% of total energy and recommendations for dietary fibre were set at 30g per day for all adults and adolescents aged 16 to 18 years and 25g per day for younger adolescents aged 11-16 years. There are no recommended pregnancy increments for fat or carbohydrate intake.

2.4 The nutritional status of women of childbearing age in the UK

As previously discussed, the assessment of nutritional need in populations is difficult and imperfect due to its dependence on a very large number of individual factors, meaning that physiological need varies from one individual to another. The use of dietary reference intakes based on the average need of individuals in a given population is however considered acceptable for assessing nutritional adequacy in populations⁷⁶. For child and adolescent populations, and during pregnancy, the recommendations are therefore adjusted to account for growth needs, and taking into consideration variation in bioavailability.

In the UK, food consumption, nutrient intake and nutritional status of the general population is estimated from data collected by the National Diet and Nutrition Survey (NDNS). The NDNS is a continuous cross-sectional survey, now in its

36

eleventh year, which collects detailed data from a representative sample of approximately 1000 people aged 1.5 years and over. The NDNS therefore provides data on the nutrient intakes and nutritional status of adolescent and adult women in the UK which allow for estimates to be made of nutritional status in these populations preconception and in early pregnancy; pregnant women are however excluded from this survey. The data reported in this section are from the latest published results of the NDNS, years 7-8 (combined) 2014/15 -2015/16⁴. The data for adolescent and adult populations are reported in the age groups 11-18 years and 19-64 years, therefore these are the age groups referred to in this section unless otherwise stated.

2.4.1 Energy and types of food

The mean total energy intake reported by adolescent women in the NDNS was 1555 kcal per day compared to 1632 kcal for adult women. Values for both of these populations are below the population estimated average requirement (EAR) which for adolescent girls ranges from 2032 kcal at age 11 to 2462 kcal at age 18 and for women from 2175 kcal at ages 19-24 to 2079 kcal at ages 55-64. This suggests either that women at all ages are ingesting less energy than is likely to be required to meet their needs while maintaining body weight or that there is a degree of under-reporting of foods eaten or portion sizes in the survey. Under reporting of energy intake in the NDNS has been examined using the doubly labelled water technique (DLW) in which participants are asked to consume water enriched with two stable isotopes of hydrogen and oxygen. Excretion of these isotopes can then be measured and the mean daily rate of CO_2 production derived, which then allows for the calculation of total energy expenditure (TEE). This is a widely accepted and validated method of assessing energy expenditure in free-living individuals over a period of one to two weeks^{77,78}. It was concluded that energy intake was routinely under reported by participants in the NDNS, particularly in adults. This finding is consistent with other similar studies and highlights the common problem of misreporting in dietary studies⁷⁹.

The mean number of portions of fruit and vegetables consumed by adolescent women was 2.8 portions per day compared to 4.2 for adult women. Among

adult women 32% reported meeting the 5-a-day fruit and vegetable recommendations compared to only 9% of adolescents. It is also important to consider that social desirability bias may mean these values are over reported. Intakes of total fat and trans fatty acids were slightly under the recommended upper limit for both age groups however intake of saturated fatty acids was higher than the recommended population intake (no more than 11% of total energy from food) at 12.3% for both groups.

AOAC fibre intake was significantly below recommendations with only 2% of adolescent women and 4% of adult women meeting the SACN guideline. The mean intake for adolescents was 14.1g per day (recommended 25-30g/day) and for adults was 17.4g per day (recommended 30g/day).

SACN recommends that energy from intake of free sugars should not exceed 5% of total dietary energy. Adult women reported consuming more than double the recommended level of free sugars (11.2% of energy) and for adolescent women the reported intake was almost three times the recommendation (14.4% of energy). Adolescent women also reported consuming higher levels of sugar-sweetened soft drinks (183g/day) compared to adult women (100g/day).

2.4.2 Micronutrients

Table 2 shows the micronutrient intake from food (not including supplements) of women in both age groups; both the mean value expressed as a percentage of the reference nutrient intake and the proportion of the sample with intakes less than the lower reference nutrient intakes are included. With the exception of selenium, a higher proportion of adolescent women had intakes of all micronutrients below the LRNI compared to adult women. The mean micronutrient intake as a percentage of the RNI was also lower among adolescent women with mean intake only exceeding 100% of the RNI for riboflavin.

Micronutrient	Females 1	1-18 years	Females 19-64 years			
	Mean as %	% below	Mean as %	% below		
	RNI	LRNI	RNI	LRNI		
Vitamin A (µg/day)	83	24	138	10		
Riboflavin (mg/day)	108	26	129	14		
Folate (µg/day)	87	15	107	6		
Vitamin D (µg/day)	19	NA	25	NA		
Iron (mg/day)	56	54	76	27		
Calcium (mg/day)	83	22	107	11		
Magnesium (mg/day)	65	50	88	11		
Potassium (mg/day)	62	38	74	23		
lodine (µg/day)	75	27	100	15		
Selenium (mg/day)	70	45	73	47		
Zinc (mg/day)	80	27	109	8		

Table 2 micronutrient intake from food

2.4.3 Nutritional status

The NDNS also includes analysis of biomarkers of nutritional status. The results show cause for concern in markers of iron status, particularly among adolescent women with 9% being below the threshold for haemoglobin and 24% for plasma ferritin. The proportion of adult women below the threshold was similar for haemoglobin at 8% but only half the level among adolescent women at 12% for plasma ferritin. Levels of red blood cell folate were below the recommended level in 11% of adolescent and 7% of adult women. For serum folate 73% of adolescent and 45% of adult women were below the clinical threshold of 13nmol/L⁸⁰. Levels of plasma 25-hydroxyvitamin D were also below recommendations in a significant proportion of women, particularly adolescents where 39% of the sample tested below the recommended level of 25 nmol/L⁸¹. Urinary iodine data on women of childbearing age have recently been included in the NDNS. While the results show levels to be adequate for non-pregnant women the median urinary iodine concentration was below the recommended

level for pregnant and lactating women, suggesting that women's current intake of iodine would be insufficient to support a pregnancy⁸².

The results of the NDNS show some areas of concern. While the difference was greater among adolescents, intake and biomarkers of a number of key micronutrients was below recommendations for both adolescent and adult women. The mean intake of fruit and veg portions suggested that adult women were approaching achieving the recommended 5-a-day, but in fact only 32% consistently exceeded the thresholds, with levels even lower again among adolescents. High levels of free sugar and sugar sweetened beverage consumption and low intake of dietary fibre were also observed. These results suggest that the diet of adolescent women is nutritionally poorer compared to adult women, but also that there is significant room for improvement in the diets of all women of childbearing age.

The Health Survey for England (HSE)⁸³ reports data on body weight status of adults by age group and sex; that of women of childbearing age is shown in figure 6. The graph shows high levels of overweight and obesity in women of childbearing age with 26.5% of women being overweight and a further 19% obese, meaning almost half of women in this population are carrying excess body weight. There is a clear association between age and body mass index (BMI) with women in older age groups being more likely to be overweight or obese. However due to the cumulative effect of poor diet and excess energy intake on body weight this does not necessarily mean that older women's diet is currently poorer, rather that they are likely to have been consuming excess energy for a longer period of time. These findings also provide further evidence that energy intake is systematically under reported in the NDNS, as a population which routinely consumes fewer calories than needed for weight maintenance is unlikely to have high levels of overweight and obesity.



2.5 Nutritional issues for pregnant adolescents

As has previously been discussed, pregnancy is a life-stage at which the body's demand for nutrients is increased. In adolescent pregnancies this increased need may be compounded by the mothers' own growth needs.

There is some evidence to suggest that competition for nutrients between the mother and the foetus in growing adolescents may contribute to the higher risk of adverse outcomes. Early work on this topic published by Naeye in 1981⁸⁴ hypothesised that competition for nutrients may be an important cause for differences in birthweight, prematurity and mortality in young mothers compared to older ones based on measurement of urinary biomarkers associated with undernourishment in pregnancy. This work was expanded upon in a Peruvian study⁸⁵ which assessed nutritional status during pregnancy of adolescent and older mothers and compared neonatal outcomes. This study found that the risk of low birthweight, independent of prematurity, was higher in adolescent

mothers despite similar nutritional status to their older peers and that the availability of nutrients for the accumulation of calories in the foetus was lower in adolescents. The authors conclude therefore that the nutritional requirements of rapidly growing teenagers during pregnancy may be substantially greater than those of older women and that therefore the physiological growth needs of the mother compete with the growth needs of the foetus. More recent work by Scholl et.al⁵ has investigated differences in outcomes between growing adolescents, adolescents who were not actively growing (assessed by measuring changes in knee height/leg length) and adult women. They found that growing adolescents tended to give birth to smaller infants despite higher gestational weight gain. Further work⁸⁶ by this team to investigate the mechanisms behind this association found that growing adolescents experienced a surge in leptin concentrations in the third trimester which may increase the mother's use of glucose for energy by inhibiting breakdown of maternal fat stores. It is suggested that this partitioning of maternal energy sources could explain the lower infant birthweight despite higher gestational weight gain in growing adolescents.

Evidence⁸⁷ from the 1980's concerning the partitioning of nutrients between organs and tissues in the body assumed that nutrients were distributed depending on the varying metabolic rate of the tissue - organs with higher metabolic rates therefore demanding higher levels of nutrients. Applied to pregnancy the assumption being that because the metabolic rate of the foetalplacental unit was higher than any of the maternal organs (with the exception of the brain), the foetus would compete successfully for nutrients, even when these were limited. Both animal and human studies have however challenged this assumption with evidence suggesting that dietary restriction is correlated with impaired foetal growth well before dietary intake is restricted to the extreme. For example evidence from a study of dietary restriction in rats has shown that foetal weight is unaffected only when maternal intake is sufficient to support some maternal weight gain. Dietary restriction beyond this level resulted in reduction in tissue gains with the foetus being disproportionately more affected compared to the mother⁸⁸. A more recent review of published data on the effect of maternal feed restriction on prenatal development in rats

42

and rabbits⁸⁹ found that foetal growth was negatively affected by reduced maternal feed intake and that this impairment of growth was often accompanied by reduced skeletal ossification.

2.6 Summary

It has been demonstrated that babies born to adolescent women are at higher risk of adverse neonatal outcomes compared to those born to adult women. It has also been discussed that dietary intake and nutritional status are important modifiable factors which influence the outcome of pregnancies. While the diet of all women of childbearing age may be sub-optimal this is most pronounced in adolescent women, suggesting that there are likely to be problems with young women's nutritional status at conception, with any substantial dietary changes being unlikely and, even if changes were made, which are difficult to recover from as pregnancy progresses. It has also been discussed that the needs of adolescent women who are themselves still growing are likely to be substantially higher than women who have achieved their full adult stature, meaning that the nutrient availability to the foetus may be sacrificed for the mothers own growth needs. There is a lack of good quality systematic evidence regarding adolescent's nutritional status during pregnancy. In order to begin to address this gap in the current academic literature the following chapter presents a systematic review of nutrient intakes and nutritional biomarkers in pregnant adolescents in developed countries.

Chapter Three - Systematic Literature Review

3.1 Introduction

In this chapter the rationale for conducting a systematic literature review and its context is briefly summarised, followed by the full published version of the review. The chapter then concludes with a summary of key points, measurable impact of the work and discussion of its contribution to the thesis. As discussed in chapter two, there is a clear evidence base to suggest that the dietary intake and nutritional status of adolescent women may be insufficient to adequately support their own needs, and consequently would be inadequate to support a pregnancy. Two previous systematic reviews on this topic have addressed nutrient intakes⁹⁰ and nutritional biomarkers⁹¹ separately; in which the results were presented narratively. This review therefore aims to both update this previous work and to synthesise the results of the included papers to achieve a better understanding of the topic. This work addresses the first of the research objectives as set out in the introduction to "*Establish what is known in the current research literature regarding the nutritional status of pregnant adolescents.*"

3.2 Publication and Impact

This systematic review was accepted for publication in BMC Pregnancy and Childbirth in September 2016. BMC Pregnancy and Childbirth is a well-regarded open access, peer reviewed journal with a respectable impact factor of 2.9. Since publication online the article has been accessed 2,109 times and has achieved 6 citations (correct as of 20.01.19).

3.3 Published paper: Article A

The published paper entitled "Nutrient intakes and nutritional biomarkers in pregnant adolescents: a systematic review of studies in developed countries" is reproduced here in the format in which it was published online. Supplementary materials which were published alongside the article can be found in appendix 1.

RESEARCH ARTICLE

Open Access



Nutrient intakes and nutritional biomarkers in pregnant adolescents: a systematic review of studies in developed countries

Katie Marvin-Dowle¹, Victoria Jane Burley² and Hora Soltani^{1*}

Abstract

Background: Babies born to adolescent mothers have been shown to have poorer outcomes compared to those born to adults. Nutritional status may have an important role to play in improving the health of pregnant adolescents; however there is a lack of evidence regarding the adequacy of adolescent diets during pregnancy. This systematic review aims to examine what is known about the nutritional status of adolescent pregnant women.

Methods: A systematic search of the literature identified 21 studies which met the inclusion criteria for the review. Primary research papers using any methods were included where they were published in English between January 1995 and May 2015 and included measurements of nutrient intakes or biological markers of nutritional status in pregnant women aged 11–19 years. Individual study data was first summarised narratively before study means were pooled to give an estimate of nutritional status in the population.

Results: The results show that individual studies reported intakes of energy, fibre and a number of key micronutrients which were below recommended levels. Biological markers of iron and selenium status also showed cause for concern. Pooled analysis of individual means as a percentage of UK Dietary Reference Intakes showed intakes of vitamin D (34.8 % CI 0–83.1) to be significantly below recommendations (p = 0.05). Serum selenium levels were also found to be low (61.8 μ g/L, CI 39–84).

Conclusions: This review has identified a number of areas where the nutritional status of pregnant adolescents is sub-optimal, which may have implications for the health of adolescent mothers and their babies. It was not however possible to examine the impact of supplement use or socio-demographic characteristics which limits the interpretation these results. Further work is needed to establish the characteristics of those most at risk within this population, how this differs from adult pregnant women and the role of supplementation in achieving adequate nutrition.

Keywords: Adolescent, Pregnancy, Nutrition, Systematic review

Background

Pregnancy during adolescence is often viewed as a social problem with women who have a child during the teenage years being more likely to suffer social isolation, poverty, lower levels of educational achievement and be unemployed or work in low paid jobs [1]. Rates of teenage conceptions both in the UK and internationally have reduced over recent years;

¹Centre for Health and Social Care Research, Sheffield Hallam University, Collegiate Crescent, Sheffield S10 2BP, UK





© 2016 The Author(s). **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

^{*} Correspondence: H.Soltani@shu.ac.uk

Full list of author information is available at the end of the article

As well as the potential for adverse social outcomes associated with adolescent pregnancies there is evidence to suggest that health outcomes may be less favourable for younger mothers. A systematic review [3] aiming to assess the relationship between early first childbirth and increased risk of poor pregnancy outcomes found that very young maternal age (<15 years or less than 2 years after menarche) had a negative effect on both maternal and foetal growth and infant survival. It is suggested that young women who are still themselves growing may compete with the foetus for nutrients, which may in turn impair foetal growth and result in low birth weight babies or babies who are small for their gestational age. The review also found a moderate relationship between young maternal age and anaemia, premature birth and neonatal mortality.

It has long been established that good pregnancy nutrition has an important influence on birth outcomes, foetal growth and infant survival [4]. While specific nutritional issues may have changed since this early work, it is still maintained that mothers need to consume an adequate, yet not excessive diet in order to optimise pregnancy and birth outcomes [5]. Quantification of dietary adequacy in populations is difficult because individuals will have differing nutrient needs. This is especially true during phases of growth and physical change such as adolescence. However, the use of dietary reference intakes to estimate the adequacy of nutritional intakes has been established as acceptable [6] where the appropriate values for the age, sex and, in the case of pregnant women, stage of pregnancy are used. Evidence also suggests that nutritional needs change during the course of pregnancy with requirements for energy and several micronutrients increasing as pregnancy progresses [7].

Dietary habits of adolescent girls are often poorer than that of older women. The latest results of the UK National Diet and Nutrition Survey [8] showed that girls aged 11–18 years consumed 2.7 portions of fruit and vegetables per day compared to 4.1 portions in women aged 19–64, and adolescent girls also had some of the highest intakes of sugar-sweetened beverages within this dataset. A higher proportion of adolescent girls also had intakes of key vitamins and minerals below the lower reference nutrient intake level than adult women, including vitamin A, riboflavin, vitamin B_{12} , folate, iron, calcium, magnesium, potassium, zinc and iodine. Dietary patterns across highly developed countries have been shown to have substantial similarities [9], while the same cannot be said for less developed regions.

While the evidence presented above suggests that adolescent girls often have a poorer diet than adult women in the general population this may not also necessarily be the case in those who are pregnant. Two systematic reviews have previously been conducted [10, 11] which explored nutritional intake and biochemical markers in pregnant adolescents living in developed countries. It was acknowledged in these reviews that there was a lack of good quality evidence in relation to these topics. However the author concluded that there was some consensus in the available literature that pregnant adolescents had intakes of energy, iron, folate, calcium, vitamin E and magnesium which were below the dietary recommended intakes. The review of biochemical markers reported that indicators of anaemia and iron status were compromised in this population; however no further conclusions could be drawn from the limited available evidence. It is therefore important that the most recent evidence relating to the nutritional intake and status of pregnant adolescents is examined in order to establish what the particular issues may be for this group. The aim of his systematic review was therefore to investigate the nutritional status of pregnant adolescents living in developed countries.

Methods

Search strategy

The search strategy was developed using search terms detailed in Table 1 and applied across nine key electronic databases (AMED, ASSIA, CINAHL, Child Development and Adolescent Studies, Cochrane Library, Health Source: Nursing, Maternity and Infant Care, MEDLINE and MEDLINE in Process, SCOPUS). Reference lists of identified papers were hand searched, and reference and citation functions were used where available.

Table 1 search terms The main stages of the review including the number of references identified at each stage are illustrated in Fig. 1.

Inclusion criteria

Studies were limited to primary research papers using any methods published in English between 1995 and May 2015.

Studies were included where they provided data regarding either the nutritional intake or a biological marker of nutrient status of adolescents aged 11–19 years at any stage of pregnancy, from countries considered as having very high levels of human development by the United Nations Human Development Index [12]. This index has been selected as it provides a multidimensional model incorporating not only wealth but also health and education, and so provides a more reliable basis for assuming some commonality between countries of origin of the included studies. The definition of adolescence has been chosen to correspond with the World Health Organisation Growth Index [13].

Quality appraisal

The identified papers were assessed for risk of bias using the Critical Appraisal and Skills Program (CASP) checklist for

Table 1 Search terms

Theme	Nutritional Intake	Pregnancy	Age	Nutritional Status
Search Terms	nutrient	pregnan*	adolescen*	biomarker*
	nutrition*	gestatio*	teen	iron
	diet*	matern*	teenage*	folate
	eat*	mother*	youth	calcium
	food	gravid*		anaemi*
	nutrition assessment (MH)	Pregnancy in adolescence (MH)		anemi*
	Food habits (MH)			Biological markers (MH)
	Dietary surveys (MH)			

* indicates truncation of search tearm

systematic reviews [14] which was adapted to accommodate cross-sectional studies. Aspects of the studies giving an indication of methodological and interpretive rigor (e.g. research design, clear statement of research aims, recruitment of participants, consideration of confounding factors and reporting of results) were graded as either 'good' (+), 'adequate or unclear' (-/+) or poor (-), studies were then given an overall grade for quality.

Data extraction

Included studies were grouped depending on whether the study examined nutrient intake, biological markers of nutritional status or both. Information from the included studies was entered in to data extraction sheets using Microsoft Excel, one each for nutrient intakes and biological markers, and checked by a second reviewer.



Data synthesis

Reported data were first tabulated to explore patterns across the included studies and described narratively. Where 95 % confidence intervals were not reported they were calculated for the mean and used to assess the extent to which the study mean differed from the relevant reference value. Where the lower confidence limit was above the reference value the study mean was considered to be significantly higher, where the upper confidence limit was below the reference value the study mean was considered significantly lower, otherwise no significant difference was reported.

Analysis of micronutrient intakes was undertaken by comparing the reported data from included studies with the UK reference nutrient intake (RNI) [15] and US recommended daily allowance (RDA) [16] where available. Analysis has been undertaken using both UK and US thresholds as, while the majority of studies were undertaken in the USA, this review is also concerned with applying the results to the UK context. Energy intake was calculated by taking the mean of the single year of age estimated average requirements (EAR) for young women aged 11-19 with an increment applied in the third trimester; a population level EAR is only available in the UK [17]. For macronutrients an estimate of the percentage contribution to energy, if not provided, was calculated from the mean macronutrient intake and the mean energy intake.

In the case of biological markers, minimum thresholds for nutrient deficiency provided by WHO, UK and US authorities (where available) have been compared with reported study data. Nationally or internationally recognised cut offs for deficiency are not currently available for zinc, selenium, copper, magnesium and phosphorous, therefore these elements have been compared with suggested thresholds in academic literature [18–22].

Where data were available from two or more studies for a single nutrient or biological marker a pooled mean was calculated and weighted by the number of participants in each study. Mean measures of micronutrient and energy intake were expressed as a percentage of dietary reference values to allow for comparisons across different nutrients.

Sub-group analyses were performed by country of origin (USA only and UK only), stage of pregnancy (first, second and third trimester and reported average over the pregnancy) and age of adolescents (15 years and under and 16–19 years).

Results

A total of 4,084 unique papers were identified from the search of the literature with 78 studies remaining after title and abstract screening. Following examination of the full text of these papers a total of 21 papers were identified that met the review inclusion criteria. Details of the excluded studies are given in Additional file 1: Table S1. In brief, the main reasons for exclusion were not reporting appropriate data and the study population not meeting the inclusion criteria for age or pregnancy.

No studies were excluded for reasons of poor quality; after quality assessment 16 of the included studies were considered to be of good quality while the remaining five studies were of a satisfactory standard (Table 2).

Of the included studies, six provided information on dietary intakes only, 12 on biological markers only and three reported both types of information. Nutrient intakes from food sources were reported (therefore excluding any contribution from supplements) in all but one paper [23]. However, the majority (10 out of 15) of papers reporting biological markers also reported that participants were taking nutritional supplements, details of which along with other characteristics of the included studies are shown in Table 2. Due to inconsistencies in the type, dose, duration and compliance with supplement use it was not possible to quantify the impact of supplements on the results.

Of the 21 included studies 14 were carried out in the USA [24–37], one in the US territory of Guam [38], three in the UK [23, 39, 40] and one in each of Australia [41], Canada [42], Chile [43] and Poland [44]. Nutritional status was a primary outcome measure in all but one of the included studies where the primary outcomes were birth weight and prematurity [43].

The study designs of the included studies are listed within Table 2. The majority of the studies were crosssectional surveys. Five studies were randomised controlled trials, where baseline dietary assessments before randomisation, or data from the control group only, permitted the inclusion of nutritional intake or biomarker data cross-sectionally. One study was a retrospective cohort analysis and one was a retrospective chart review.

Participants were all aged between 12 and 20 years with the majority being aged 16 and over. The majority of studies selected participants using convenience samples; other sampling methods used were purposive [26], representative probability sample [32], stratified random sample [44] and a retrospective medical chart review including all eligible records [37].

The majority of studies reported a range of ethnicities in the sample with the exception of three studies where all participants were African American [31, 36, 37], one including only Mexican American participants [28] and one where all participants were White [26]. Where reported the majority of participants had a BMI in the healthy range. Six studies reported participant's weight gain from recruitment to delivery which ranged from 14 to 17 kg.

All but two of the studies reporting biological markers collected venous blood samples which were

Table 2 Characteristics of included studies

Study Information Participants				Supplements				Measuremen	Quality				
Study	Country	Study design	Study groups	Number	Age	Inclusion criteria	Exclusion criteria	Supplement type and dose	Number with available data	Estimated compliance: N (%)	Data collection method (s)	Time period(s)	Quality rating
Baker et.al (2009) [23]	UK	Cross- sectional	NA	500	14–18	Singleton pregnancy, age	Inability to provide informed consent,	Preconception folic acid	498	34 (6.9)	24 h recall (multiple)	Third trimester	+
		study				14–18, gestational age	preeclampsia, clotting disorders, HIV/AIDS, baemoglobinopathies	Folic acid in early pregnancy	498	220 (44.2)			
						< -20 weeks	diabetes, renal disease, hypertension, multiple	lron in early pregnancy	498	39 (7.8)			
							gestations, history of 3= > previous miscarriages	Multivitamins in early pregnancy	498	19 (3.8)			
								Folic acid only in 3rd trimester	290	5 (1.7)			
								Iron and folic acid in 3rd trimester	290	13 (4.5)			
								Iron only in 3rd trimester	290	42 (14.5)			
								Multivitamins in 3rd trimester	290	7 (2.4)			
Castillo- Duran et.al (2001) [43]	Chile	RCT	Zinc supplemented	249	16.4 (mean)	Beginning prenatal visits before 20 weeks	Adolescents whose background included chronic diseases, drug	20 mg Zinc sulphate daily	Intervention group - 249	At least 50 % compliance	24 h recall (multiple)	Second and third trimester	+
			Placebo	258		gestation, aged <19 at estimated due date	abuse, mental retardation, illiteracy or pregnancy due to incest or rape	40 mg iron sulphate	All participants - 507	·			
Chan et.al (2006) [24]	USA	RCT	Control group, 2 intervention groups excluded from review*	23	15–17	Enrolled before 20 weeks gestation	Hypertension, diabetes, renal or liver disease, alcohol or tobacco use, using medicines effecting Ca metabolism	No additional supplements reported			Food diary (weighing not mentioned)	Second and third trimester	-/+
Chang et.al (2003) [37]	USA	Retrospective chart review	NA	918	12–17	Self-reported racial group as African-American; singleton pregnancy	Birth results not available due to abortion, miscarriage or transfer of care	90 mg carbonyl iron daily plus additional up to 120 mg for those classified as anaemic	918	Not reported	Not reported	Second and third trimester	+

Dawson et.al (2000) [35]	USA	RCT	One-A-Day without Iron	20	16–20	Aged 16–20, less than 16 weeks gestation, no iron supplementation for previous 30 days	Hypertension; diabetes; other medical problems; haemoglobin <11 g/ dL; haematocrit <30 %	Vitamin A 5000 IU, vitamin D 400 IU, vitamin E 10 mg, vitamin C 60 mg, folic acid 0.4 mg, thiamine 1.5 mg, riboflavin 1.7 mg, niacin 20 mg, pyridoxine 2 mg, vitamin B- 6 mg, pantothenic acid 10 mg	20	Not reported	Venous blood sample	Second and third trimester and delivery	-/+
			One-A-Day with Iron	20				18 mg Iron, Vitamin A 5000 IU, vitamin D 400 IU, vitamin E 10 mg, vitamin C 60 mg, folic acid 0.4 mg, thiamine 1.5 mg, riboflavin 1.7 mg, niacin 20 mg, pyridoxine 2 mg, vitamin B- 6 mg, pantothenic acid 10 mg	20				
Derbyshire (2009) [39]	UK	Cross- sectional study	NA	20	15–19	Attending antenatal classes or community clinics	Incomplete diary	None reported			Food diary unweighed	Third trimester	+
Gadowsky et.al (1995)	Canada	Cross- sectional	NA	58	14–19	Not reported	Not reported	Folic Acid Mean 479 µg/day	58	82 %	Venous blood	Third trimester	+
[42]		study						Elemental Iron Mean 31.5 mg/ day			sample		
								Cyanocobalamin (B12) Mean 2.24 µg/d					
Giddens et.al (2000) [27]	USA	RCT (subset from a larger study)	NA	59	13–18	Singleton pregnancies, between 13 and 19 weeks gestation	Not reported	Reported that any contribution from supplements was not included in analysis			Food diary weighed	Mean over pregnancy	-/+

Table 2 Characteristics of included studies (Continued)

Ginde et.al (2010) [32]	USA	Secondary analysis of cross sectional survey	NA	84	13–19	Not reported	Not reported	Some participants taking vitamin D supplements	Not reported	Not reported	Venous blood sample	Mean over pregnancy	+
Gutierrez et.al (1999) [28]	USA	Cross- sectional study	NA	46	13–18	Self-identified ethnicity as Mexican American, primigravida, 13– 18 years of age	History of miscarriage or health problems, involved in competitive athletic performances or heavy exercise, planned to move away during study period	None reported			24 h recall (single)	Second and third trimester	+
lannotti et.al (2005) [36]	USA	Cross- sectional study	NA	80	13–18	Self-reported racial group as African-American; singleton pregnancy	Not reported	None reported			10 ml Venous blood sample	Second and third trimesters	+
Job et.al (1995) [41]	Australia	Cross- sectional study	NA	35	13–19	Not reported	Not reported	None reported			24 h recall	Mean over pregnancy	+
Lee et.al (2013) [25]	USA	Cross- sectional study	NA	156	13–18	Age 18 or under, carrying a single fetus, 12–30 weeks gestation at recruitment	Not reported	Reported that any contribution from supplements was not included in analysis			24 h recall (multiple)	<23 weeks gestation, 23–30 weeks gestation and Mean over pregnancy	+
McGuire et.al. (2010) [31]	USA	Cross- sectional study	NA	80	Under 18 (mean 16.5)	Self-reported ethnic group as African American, singleton pregnancy	Not reported	Routinely prescribed prenatal supplements containing 400 IU vitamin D		Data not available	Venous blood sample	Second and third trimester	+
Meier et.al (2002) [34]	USA	RCT	lron supplemented	20	15–18	Not reported	lron deficiency anaemia at recruitment	60 mg elemental iron & 1 mg folic acid	20		Venous blood sample	Second and third trimesters	+
			Placebo	17				1 mg folic acid	17				

Table 2 Characteristics of included studies (Continued)

Mistry et.al (2014) [40]	UK	Cross- sectional study	Appropriate for gestational age babies	107	14–18	Not reported	Inability to provide informed consent, pre-eclampsia, clot-
			Small for	9			ting disorders, HIV/ AIDS_Haemoglobin

Table 2 Characteristics of included studies (Continued)

Mistry et.al (2014) [40]	UK	Cross- sectional study	Appropriate for gestational age babies Small for gestational age babies	107 9	14–18	Not reported	Inability to provide informed consent, pre-eclampsia, clot- ting disorders, HIV/ AIDS, Haemoglobin- pathies, diabetes, renal disease, hyper- tension, multiple pregnancy and previ- ous miscarriage	None reported			30 ml venous blood sample	Third trimester	+
O'Brien et.al (2003) [33]	USA	Cross- sectional study	NA	23	Mean 16.5	First, singleton pregnancies; no medical problems; no medications known to influence calcium metabolism; non- smokers; no his- tory of drug or alcohol abuse	Not reported	Prenatal supplement including 5 mmol Ca	23	39 %	Not reported	Third trimester	+
Pobocik et.al (2003) [38]	Guam (USA Teritory)	Cross- sectional study	NA	434	14–20	Not reported	Not reported	Reported that any contribution from supplements was not included in analysis			24 h recall (single)	Mean over pregnancy	-/+
Rycel et.al (2009) [44]	Poland	Retro-spective cohort	NA	102	15–18	Not reported	Not reported	none reported			Venous blood sample	Before and after delivery	-/+
Young et.al (2010) [29]	USA	Cross- sectional study	NA	92	14–18	Healthy, singleton pregnancy	HIV, diabetes, pre- eclampsia, eating dis- orders, malabsorption diseases, self-reported drug use	Prenatal supplement including 27 mg iron	92	Not reported	Venous blood sample	Second trimester and delivery	+
Young et.al (2012) [30]	USA	Cross- sectional study	NA	171	Under 18 (mean 17.1)	Healthy, singleton pregnancy	HIV, diabetes, pre- eclampsia, eating dis- orders, malabsorption diseases	400 IU Vitamin D supplement given to participants found to be deficient	46 (estimated from reported percentages)	26.4 % - daily, 35.8 % at least twice per week	10 ml venous blood sample	Delivery	+

analysed in laboratories using standard testing procedures. One study which was a medical records review [37] did not provide details of how samples were collected. One study also assessed biological markers of calcium absorption using a 24 h urine collection followed by daily spot urine collections [33]. Data relating to participants nutrient intakes used a variety of data collection methods. Three studies used food diaries [24, 27, 39], one of which was weighed [27]. The remaining studies used single [28, 38, 41] or multiple [25, 43, 23] 24 h recalls. Four out of the nine studies reporting nutrient intakes stated that dietary assessments were carried out by a trained nutritionist or similar professional [25, 26, 43, 27].

Nutrient intakes

Energy intake was reported by nine studies with seven of these reporting intakes below the recommendations at one or more time point (Table 3). Four studies also reported gestational weight gain which ranged from 14 to 17 kg. Pooled analysis of the percentage of the EAR for energy in these 10 studies revealed wide confidence limits around the estimated mean, with an average intake 9 % lower than the UK EAR (mean % EAR, 91.2 %, CI 29.6-152.8 %). Analysis of energy intake by trimester and study country of origin did not show any significant differences (UK studies 89.1 % CI 39.2-139.1 %, US studies 100.4 %, CI 24.2-176.6 %). Analysis of those studies reporting gestational weight gain only showed young women to be achieving a higher percentage of the EAR for energy (99.1 %, CI 41.0-157.2 %) than those studies which did not report weight gain (90.2 %, CI 27.9–152.5 %) but this difference was not statistically significant.

Mean intakes of macronutrients are show in Table 4. Intakes of protein and total carbohydrate were roughly

in line with recommendations. There were too few studies reporting intakes of total fat, fat types or sugars to permit conclusions to be drawn. Three studies reported any measurements of dietary fibre, all of which were below recommended levels.

Tables 5 and 6 show the pattern of micronutrient intakes across the included studies compared to UK and US Dietary Reference Values (DRVs).

The individual study results show that the majority of reported nutrient intakes were significantly below both the UK RNI and US RDA for vitamin D, potassium and magnesium and below the US RDA for calcium, vitamin E, folate, phosphorous and iron. Zinc intakes reported as the mean intake over pregnancy were low whereas this was not the case in the studies reporting intakes in the second or third trimesters specifically.

Results of the pooled analyses however showed that only intake of vitamin D remained significantly below both the UK RNI and US RDA, and intakes of potassium below the US RDA. Sub-group analysis showed that micronutrient intake was lower in UK based studies than those based in the USA for all micronutrients with the exception of vitamin C, however vitamin D was the only micronutrient where the percentage of the DRV in UK based studies was below the UK RNI (21.4 %, CI 0–63.5 %) and US RDA (14.3 %, CI 0–42.3 %). Results of the pooled analysis of micronutrients are shown in Fig. 2. Detailed results of the sub-group analysis of nutrient intakes are available in Additional file 1: Table S2 and Additional file 1: Table S3.

Micronutrient analysis by stage of pregnancy showed that intake of vitamin D in the second (28 % of the UK RNI, CI 26–30 %) and third (31.2 % of the UK RNI, CI 0–47 %) trimesters was below recommendations; this was not the case for measures reported as a mean over pregnancy (54 % of the UK RNI, CI 0–114.3 %). In the

Fable 3 Mean energy inta	ake in individual studies	s compared to UK estimate	d average requirement
--------------------------	---------------------------	---------------------------	-----------------------

	Study	Seco	ond Trime	ster	Third Trimester				Mean over pregnancy			
		N=	N= Mean Confidence Interval N= Mean Confidence Interval		N=	Mean	Confidence Interval					
Energy kcal/day ^a	Baker (2009)				290	↓2147	2075, 2219					
	Castillo-Duran (2001) a	249	↓1887	1825, 1949	249	↓2030	1968, 2092					
	Castillo-Duran (2001) b	258	↓1863	1790, 1936	258	↓1982	1927, 2038					
	Chan (2006)	23	↔2223	2057, 2389	23	↓2276	2059, 2493					
	Derbyshire (2009)				20	↓2273	2052, 2494					
	Giddens (2000)							59	↓2342	2187, 2497		
	Gutierrez (1999)	46	↔2390	2150, 2630	46	↔2620	2389, 2851					
	Job (1995)							70	↓2134	1949, 2320		
	Lee (2013)				133	↓2303	2161, 2446	156	↔2273	2167, 2379		
	Pobocik (2003)							434	↑2487	2388, 2586		

^aComparison to UK EARs, First and Second Trimester requirement (Average UK EAR for Females aged 11–19) 2355 kcal/day, Third Trimester requirement 2546 kcal/day, Mean over pregnancy requirement (average of three trimester values) 2419 kcal/d, \uparrow Study mean higher than reference value (p < 0.05), \leftrightarrow Study mean not different to reference value (p < 0.05), \downarrow Study mean lower than reference value (p < 0.05)

		Com	nparison t	o UK Dietary f	Refere	nce Value	<u>.</u>				Cor	nparisor	n to US Dietary	/ Refe	rence V	'alue			
Nutrient	Study	Seco	ond Trime	ester	Thire	d Trimeste	er	Mea	n over pi	regnancy	Sec	ond Trir	nester	Thire	d Trime	ster	Mea	n over	pregnancy
		N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval
Protein grams/ day ^a	Castillo-Duran (2001) a	249	↑60	58, 62	249	↑ 62	59, 64				249	↓ 60	58, 62	249	↓ 62	59, 64			
	Castillo-Duran (2001) b	258	↑59	57, 61	258	↑ 60	58, 61				258	↓59	57, 61	258	↓60	58, 61			
	Chan (2006)	23	↑76	65, 87	23	↑76	65, 87				23	↔76	65, 87	23	↔76	65, 87			
	Derbyshire (2009)				20	↑72	65, 79							20	↔72	65, 79			
	Giddens (2000)							59	† 82	77, 87							59	182	77, 87
	Gutierrez (1999)	46	† 111	98, 124	46	† 118	105, 130				46	↑111	98, 124	46	† 118	105, 130			
	Job (1995)							70	↑73	73, 73							70	↑73	73, 73
	Lee (2013)				133	† 81	76, 87	156	↑70	65, 75				133	† 81	76, 87	156	↔70	65, 75
	Pobocik (2003)							434	↑ 99	95, 103							434	1199	95, 103
Englyst Fibre grams/day ^b	Derbyshire (2009)					↓12	11, 13												
AOAC Fibre grams/day ^c	Giddens (2000)																59	↓14	13, 15
	Lee (2013)													133	↓13	12, 14	156	↓14	13,14
		N=	% of Energy		N=	% of Energy		N=	% of Energy										
Total Carbohydrate	Derbyshire (2009)				20	54 %													
	Giddens (2000)							59	50 %										
	Lee (2013)				133	51 %		156	52 %										
	Gutierrez (1999)	46	56 %		46	56 %													
Total Fat	Chan (2006)	23	29 %		23	29 %													
	Gutierrez (1999)	46	29 %		46	27 %													

Table 4 Mean intakes of macronutrients (g/day or percent of energy) and dietary fibre (g/day) in individual studies compared to UK and US dietary reference values

Table 4 Mean intakes of macronutrients (g/day or perce	t of energy) and dietary fibre (g/day) in individual studies of	compared to UK and US dietary reference values (Continued)
--	---	--

Saturated Fat	Chan (2006) 23 9 %	23 10 %			
Total Sugars	Derbyshire (2009)	20 25 %			
Added Sugars	Lee (2013)	133 17 %	156 17 %		

^a Comparison to UK RNI 51 g/day and US RDA 71 g/day, ^b Comparison to UK RNI 18 g/day, ^c Comparison to US RDA 28 g/day, \uparrow Study mean higher than reference value (p < 0.05), \leftrightarrow Study mean not different to reference value (p < 0.05), \downarrow Study mean lower than reference value (p < 0.05)

		Con	Comparison to UK Dietary Reference Value								Comparison to US Dietary Reference Value								
Nutrient	Study	Seco	ond Trim	nester	Third	d Trimes	iter	Mea	n over p	pregnancy	Seco	ond Trim	nester	Third	d Trimeste	er	Mea	n over p	regnancy
		N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval
Calcium mg/	Baker (2009)				290	↔840	800, 880							290	↓840	800, 880			
day ^a	Chan (2006)	23	↔835	711, 959	23	↔862	714, 1010				23	↓835	711, 959	24	↓862	714, 1010			
	Derbyshire (2009)				20	↑ 1007	867, 1147							20	↓1007	867, 1147			
	Giddens (2000)							59	† 989	904, 1074							59	↓989	904, 1074
	Gutierrez (1999)	46	↑1561	1334, 1789	46	↑1655	1424, 1886				46	↑1561	1334, 1789	46	↑1655	1424, 1886			
	Job (1995)							70	↔923	756, 1090							70	↓923	756, 1090
	Lee (2013)				133	1€16	838, 995	156	1886	824, 948				133	↓916	838, 995	156	↓886	824, 948
	Pobocik (2003)							434	↓743	689, 797							434	↓743	689, 797
Phosphorous	Chan (2006)	23	1934	811, 1057	23	1¢961	812, 1110				23	↓934	811, 1057	24	↓961	812, 1110			
mg/day ⁵	Giddens (2000)							59	↑ 1340	1248, 1432							59	↔1340	1248, 1432
	Lee (2013)				133	↑1264	1182, 1347	156	↑1196	1131, 1261				133	↔1264	1182, 1347	156	↔1196	1131, 1261
	Pobocik (2003)							434	↑1338	1279, 1397							434	↑1338	1279, 1397
Iron mg/day $^{\circ}$	Baker (2009)				290	↑17	15, 19							290	↓17	15, 19			
	Castillo-Duran (2001) a	249	↑15.5	15, 16	249	↑16.8	16, 17				249	↓15.5	15, 16	249	↓16.8	16, 17			
	Castillo-Duran (2001) b	258	↑16.6	16, 17	258	↑16.8	16, 17				258	↓16.6	16, 17	258	↓16.8	16, 17			
	Chan (2006)	23	↑ 22	18, 26	23	↑25	20, 30				23	↓22	18, 26	23	↔25	20, 30			
	Derbyshire (2009)				20	↓12.6	11, 14							20	↔12.6	11, 14			
	Giddens (2000)							59	↔16	15, 17							59	↓16	15, 17
	Gutierrez (1999)	46	↑ 17.7	15, 20	46	↑22.7	17, 28				46	↓17.7	15, 20	46	↔22.7	17, 28			
	Job (1995)							70	↓11.2	10, 12							70	↓11.2	10, 12
	Lee (2013)				133	↑18.8	17, 20	156	↑18.6	17, 20				133	↓18.8	17, 20	156	↓18.6	17, 20
	Pobocik (2003)							434	120	19, 21							434	↓20	19, 21

Table 5 Intake of micronutrients in individual studies compared to UK and US dietary reference values - minerals

Magnesium	Baker (2009)				290	↓236	227, 245								↓236	227, 245			
mg/day ^a	Chan (2006)	23	↓263	230, 296	23	↓264	231, 297					↓263	230, 296		↓264	231, 297			
	Derbyshire (2009)				20	↓244	218, 270								↓244	218, 270			
	Giddens (2000)							59	↓252	234, 270								↓252	234, 270
	Lee (2013)				133	↓237	222, 253	156	↓231	218, 244					↓237	222, 253		↓231	218, 244
	Pobocik (2003)							434	↓270	258, 282								↓270	258, 282
Potassium mg/	Chan (2006)	23	↓2802	2512, 3092	23	↓2954	2635, 3273				23	↓2802	2512, 3092	23	↓2954	2635, 3273			
day ^c	Derbyshire (2009)				20	↓2948	2659, 3237							20	↓2948	2659, 3237			
Zinc mg/day ^f	Baker (2009)				290	† 8.1	7.8, 8.4							290	↓8.1	7.8, 8.4			
	Castillo-Duran (2001) a	249	↑7.4	7.1, 7.7	249	↑7.7	7.4, 8				249	↓7.4	7.1, 7.7	249	↓7.7	7.4, 8			
	Castillo-Duran (2001) b	258	↑7.4	7.1, 7.7	258	↑7.4	7.2, 7.6				258	↓7.4	7.1, 7.7	258	↓7.4	7.2, 7.6			
	Chan (2006)	23	16	11.9, 20.1	23	† 18	13.5, 22.5				23	↔16	11.9, 20.1	23	↑18	13.5, 22.5			
	Derbyshire (2009)				20	18.1	7.4, 8.9							20		7.4, 8.9			
	Giddens (2000)							59	↑11.6	10.5, 12.7							59	↑11.6	10.5, 12.7
	Gutierrez (1999)	46	↑14.5	12.6, 16.4	46	↑15.3	13.4, 17.2				46	↑14.5	12.6, 16.4	46	↑15.3	13.4, 17.2			
	Job (1995)							70	19.5	8.5, 10.5							70	19.5	8.5, 10.5
	Lee (2013)				133	† 12.8		156	↑ 12.6	11.4, 13.8				133	12.8		156	↑12.6	11.4, 13.8
	Pobocik (2003)							434	† 13	12.2, 13.8							434	↑13	12.2, 13.8
Sodium mg/	Chan (2006)	23	↑3316	2809, 3823	23	† 3323	2812, 3834					† 3316	2809, 3823		↑3323	2812, 3834			
day ⁹	Derbyshire (2009)				20	13089	2722, 3456								143089	2722, 3456			
Copper µg/ day ^h	Giddens (2000)							59	↑1200	1098, 1302							59	1200	1098, 1302
	Lee (2013)				133	† 1100	1015, 1185	156	↑1085	1021, 1151				133	↑1100	1015, 1185	156	↑1085	1021, 1151
Selenium µg/ day ⁱ	Giddens (2000)							59	↑ 116	109, 123							59	† 116	109, 123

Table 5 Intake of micronutrients in individual studies compared to UK and US dietary reference values - minerals (Continued)

^aComparison to UK RNI 800 mg/day and US RDA 1300 mg/day, ^bComparison to UK RNI 625 mg/day and US RDA 1250 mg/day, ^cComparison to UK RNI 14.8 mg/day and US RDA 27 mg/day, ^dComparison to UK RNI 300 mg/day and US RDA 400 mg/day, ^eComparison to UK RNI 3500 mg/day and US RDA 4700 mg/day, ^fComparison to UK RNI 7 mg/day and US RDA 12 mg/day, ^gComparison to UK RNI 1500 mg/day and US RDA 4700 mg/day, ^bComparison to UK RNI 7 mg/day and US RDA 12 mg/day, ^gComparison to UK RNI 1500 mg/day and US RDA 1600 mg/day, ^bComparison to UK RNI 7 mg/day and US RDA 12 mg/day, ^gComparison to UK RNI 1500 mg/day and US RDA 1600 mg/day, ^bComparison to UK RNI 60 µg/day and US RDA 60 µg/day, [†] Study mean higher than reference value (p < 0.05), \leftrightarrow Study mean not different to reference value (p < 0.05), \downarrow Study mean lower than reference value (p < 0.05)

		Cor	Comparison to UK Dietary Reference Value										Comparison to US Dietary Reference Value									
Nutrient	Study	Sec	ond Tri	mester	Thir	d Trime:	ster	Mea	n over	oregnancy	Sec	ond Trir	nester	Thir	d Trimes	ster	Mea	in over	pregnancy			
		N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval			
Vitamin D	Baker (2009)				290	↓2.1	2, 2.3							290	↓2.1	2, 2.3						
µg∕day ª	Chan (2006)	23	↓2.8	1.9, 3.7	23	↓3.1	2.2, 4				23	↓2.8	1.9, 3.7	23	↓3.1	2.2, 4						
	Derbyshire (2009)				20	↓2.0	1.5, 2.6							20	↓2.0	1.5, 2.6						
	Giddens (2000)							59	↓6.4	5.7, 7.1							59	↓6.4	5.7, 7.1			
	Lee (2013)				133	↓5.4	4.7, 6.1	156	↓5.1	4.6, 5.6				133	↓5.4	4.7, 6.1	156	↓5.1	4.6, 5.6			
Vitamin E $^{\rm b}$	Baker (2009)													290	↓8.9	8.3, 9.5						
	Derbyshire (2009)													20	↓7.7	6.5, 8.9						
	Gutierrez (1999)										46	↓10.7	10.7	46	↓11.2	8.2, 14.1						
	Lee (2013)													133	↓6.9	6.2, 7.6	156	↓6.9	6.4, 7.4			
	Pobocik (2003)																434	18	7.2, 8.8			
Vitamin C $^{\rm c}$	Baker (2009)				290	160	146, 174							290	160	146, 174						
	Derbyshire (2009)				20	† 138	111, 165							20	↑138	111, 165						
	Giddens (2000)							59	↑ 128	112, 144							59	↑ 128	112, 144			
	Gutierrez (1999)	46	↑252	208, 296	46	↑ 231	190, 271				46	↑252	208, 296	46	↑231	190, 271						
	Job (1995)							70	↑ 135	92, 178							70	↑135	92, 178			
	Lee (2013)				133	197	81, 113	156	106	94, 118				133	197	81, 113	156	106	94, 118			
	Pobocik (2003)							434	↑ 167	150, 184							434	↑167	150, 184			

Folate ^d	Baker (2009)				290	↔285	269, 301							290	↓285	269, 301			
	Derbyshire (2009)				20	↓227	205, 249							20	↓227	205, 249			
	Giddens (2000)							59	↔312	277, 347							59	↓312	277, 347
	Gutierrez (1999)	46	↑ 447	355, 540	46	1393	340, 445				46	↓447	355, 540	46	↓393	340, 445			
	Lee (2013)				133	1829	723, 935	156	↑849	645, 1053				133	↑829	723, 935	156	1849	645, 1053
	Pobocik (2003)							434	↔292	269, 315							434	↓ 292	269, 315
Riboflavin ^e	Chan (2006)	23	↑ 2.3	1.8, 2.8	23	↑ 2.4	2, 2.8				23	↑2.3	1.8, 2.8	23	↑2.4	2, 2.8			
	Giddens (2000)							59	↑ 2.3								59	↑ 2.3	2.1, 2.5
	Lee (2013)				133	↑2.5	2.3, 2.7	156	↑2.4					133	↑2.5	2.3, 2.7	156	12.4	2.2, 2.6
	Pobocik (2003)							434	† 2.1								434	† 2.1	2, 2.2
B12 ^f	Baker (2009)				290	↑5.3	4.7, 5.9							290	↑5.3	4.7, 5.9			
	Chan (2006)	23	↑5	3.8, 6.2	23	↑5.2	3.7, 6.7				23	∱5	3.8, 6.2	23	↑5.2	3.7, 6.7			
	Giddens (2000)							59	↑5.3	4.6, 6							59	↑5.3	4.6, 6
	Lee (2013)				133	15.6	5, 6.2	156	↑5.5	5, 6				133	15.6	5, 6.2	156	↑5.5	5, 6
	Pobocik (2003)							434	↑ 5.5	4.8, 6.2							434	↑5.5	4.8, 6.2
Thiamin ^g	Baker (2009)				290	1.6	1.5, 1.7							290	1.6	1.5, 1.7			
	Giddens (2000)							59	† 2.1	1.9, 2.3							59	↑ 2.1	1.9, 2.3
	Lee (2013)				133	↑ 2.1	1.9, 2.3	156	↑ 2.1	1.9, 2.3				133	↑ 2.1	1.9, 2.3	156	↑ 2.1	1.9, 2.3
	Pobocik (2003)							434	2.4	2.3, 2.5							434	2.4	2.3, 2.5
Niacin ^h	Baker (2009)				290	133	32, 35							290	14	32, 35			
	Giddens (2000)							59	1€1	22, 26							59	1€1	22, 26
	Lee (2013)				133	↑28	26, 30	156	↑26	24, 27				133	↑28	26, 30	156	↑26	24, 27
	Pobocik (2003)							434	1€1	29, 31							434	1€10	29, 31

Table 6 Intake of micronutrients in individual studies compared to UK and US dietary reference values - vitamins (Continued)

Table 6 Intake of micronutrients in individual studies compared to UK and US diet	ary reference values - vitamins (Continued)
--	---

B6 ⁱ	Baker (2009)	290		290	
	Giddens (2000)		59 ↑ 1.9 1.7, 2.1		59 11.9 1.7, 2.1
	Lee (2013)	133 ↑2.2 2, 2.4	156 †2.1 1.9, 2.3	133 †2.2 2, 2.4	156 †2.1 1.9, 2.3
	Pobocik (2003)		434 ↑2 1.9, 2.1		434 †2 1.9, 2.1
Vitamin A ^j	Baker (2009)	290 ↑759 651,867		290 ↑759 651,867	
	Derbyshire (2009)	20 1 1555 439, 671		20 ↓555 439, 671	
	Giddens (2000)	↑105	59 <u>↑</u> 1053 907, 1199	↑ 1053	59
	Gutierrez 46 †2492 1466, 3518 (1999)	46 197 1432, 2523	46 ↑2492 1466, 3518	46 1978 1432, 2523	
	Job (1995)		70 1973 802, 1144		70 ↑973 802, 1144
	Lee (2013)	133 ↑698 614, 783	156 ↑666 615, 717	133 1698 614, 783	156 ↓666 615, 717
	Pobocik (2003)		434 ↑109 944, 1242		434 11093 944, 1242
Vitamin K ^k	Lee (2013)			133 ↔70 57, 83	156 ↔70 59, 81

^a Comparison to UK RNI 10 µg/day and US RDA 15 µg/day, ^b Comparison to US RDA 15 mg/day, ^c Comparison to UK RNI 40 mg/day and US RDA 80 mg/day, ^d Comparison to UK RNI 200 µg/day and US RDA 600 µg/day, ^e Comparison to UK RNI 1.1 mg/day and US RDA 1.4 mg/day, ^f Comparison to UK RNI 1.5 µg/day and US RDA 2.6 µg/day, ^g Comparison to UK RNI 0.8 mg/day and US RDA 1.4 mg/day, ^h Comparison to UK RNI 1.4 mg/day and US RDA 1.4 mg/day, ^h Comparison to UK RNI 1.5 µg/day and US RDA 2.6 µg/day, ^g Comparison to UK RNI 0.8 mg/day and US RDA 1.4 mg/day, ^h Comparison to UK RNI 1.4 mg/day and US RDA 1.4 mg/day, ^h Comparison to UK RNI 1.5 µg/day and US RDA 2.6 µg/day, ^g Comparison to UK RNI 0.8 mg/day and US RDA 1.4 mg/day, ^h Comparison to UK RNI 1.4 mg/day, ^h Comparison to UK RNI 1.2 mg/day and US RDA 1.9 mg/day, ^j Comparison to UK RNI 600 µg/day and US RDA 750 µg/day, ^k Comparison to US AI 75 µg/day, [↑] Study mean higher than reference value (p < 0.05), \leftrightarrow Study mean not different to reference value (p < 0.05), \downarrow Study mean lower than reference value (p < 0.05)



third trimester the percentage of the US RDA represented by intakes of magnesium (19.9 %, CI 20.4–98.6 %) and potassium (62.8 %, CI 32.5–93.1 %) were also below recommendations.

Biological markers

Table 7 shows the pattern of biological markers reported across the included studies compared to WHO, UK and US minimum thresholds for deficiency where available. As the recommended cut off points given by all three authorities are consistent, the results are presented in one combined table for clarity. Other nutrients were considered compared to deficiency thresholds suggested in the academic literature as previously discussed.

The results show that the mean reported biomarker values in the majority of studies suggested that young women's nutritional status was sufficient, with the exception of markers of iron and selenium status. Results for haematocrit and plasma ferritin were mixed, with results being more likely to be below the cut off in the third trimester and at delivery. Measures of serum selenium were reported to be less than the cut off in the majority of studies.

Examination of pooled, weighted means showed that only mean selenium concentration was below the reference value. The weighted means for all biological markers where there were two or more valid results are shown in Table 8.

The sub-group analysis by country of origin was only possible by US vs. non-US studies as there was only one UK based study reporting biological markers. The analysis failed to detect any differences by study country of origin. Analysis by stage of pregnancy suggests a decline in iron status markers haemoglobin, haematocrit and ferritin as pregnancy progresses; the levels observed however do not necessarily reflect iron deficiency. Detailed results of the sub-group analysis of biological markers are available in Additional file 1: Table S4 and Additional file 1: Table S5.

Discussion

Compared with reviews of the nutritional status of pregnant adolescents published in 2007, this review identified a further 13 studies that reported data on nutritional intakes and biomarkers of status. The summary results show that there may be areas of concern in adolescent's nutritional intake during pregnancy, particularly compared to US recommendations, with regard to calcium, vitamin D, vitamin E, folate, potassium and magnesium. The evidence also suggests that overall energy intake may be lower than recommended.

	Study ID	First Trimes		ter	Seco	ond Trim	nester	Third	d Trimes	ter	Deliv	/ery	
		N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval	N=	Mean	Confidence Interval
Haemoglobin (g/L) $^{\rm a}$	Baker (2009)	404	↑122	121, 123				362	↓108	107, 109			
	Chang (2003)	445	↑121	121, 122	319	† 112	112, 114	836	↓107	107, 109			
	Dawson (2000) a				21	↑124	123, 125	21	↑ 115	114, 116	20	† 114	111, 112
	Dawson (2000) b				20	↑125	124, 126	20	† 119	118, 120	21	† 116	112, 114
	Dawson (2000) a							21	↑114	113, 115			
	Dawson (2000) b							20	↑ 120	119, 121			
	Gadowsky (1995)							50	† 119	115, 123			
	lannotti (2005)				35	↔111	107, 115	70	↔107	104, 110			
	McGuire Davies (2010)				78	↑118	118, 118	76	↓107	107, 108			
	Meier (2002) a				20	125	122, 129	15	↑ 123	119, 128			
	Meier (2002) b				17	↑128	124, 132	16	† 117	112, 123			
	Meier (2002) a				19	↑116	112, 120						
	Meier (2002) b				16	↑113	110, 116						
	Rycel (2009)										102	120	120, 229
	Rycel (2009)										102	↓103	103, 103
	Young (2010)				48	1113					62	↑117	113, 121
Ferritin (µg/L or ng/ ml) ^b	Gadowsky (1995)				50	↓7.4	5.7, 9.1						
	lannotti (2005)				44	133	26.8, 40.6	59	↔15	12.6, 17.8			
	Meier (2002) a				19	142	31.3, 52.8	15	↔25	14.6, 35.4			
	Meier (2002) b				17	↑57	36.5, 77.5	15	16.8	5.2, 8.5			
	Meier (2002) a				19	↑46	24, 68.8		·				
	Meier (2002) b				16	' 110	79.13						
	Young (2010)				81	↓16	12.9.20.1				88	↔17	149 203
Haematocrit ^{(g/L) c}	Chang (2003)	445	114	357 363	319	↔33	327 333	836	132	318 322	00		1 1107 2010
haematoene	Gadowsky (1995)	115	150	55.7, 56.5	515	.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	52.7, 55.5	50	¢32 ↑36	34.8, 37.2			
	lannotti (2005)				35	↔33	32. 34	70	1.32	31.3, 32.7			
	Rvcel (2009)						- , -		v	,	102	34	34.6. 34.8
	Bycel (2009)										102	131	31.4.31.6
Zinc µmol/L-1 ^d	Castillo-Duran (2001) a	249	† 11.9	11.7, 12.1				249	† 10.9	10.7, 11.1	102	* 51	51.1, 51.0
	Castillo-Duran (2001) b	258	↑ 11.7	11.5, 11.9				258	† 10.9	10.7, 11.1			
	Castillo-Duran (2001) a							249	↑ 10.5	10.3, 10.7			
	Castillo-Duran (2001) b							258	↑ 10.2	10, 10.4			
	Chan (2006)										23	↑26.3	23.2, 29.4
	Mistry (2014) a							107	11111111111111111111111111111111111111	8.8, 10.5			
	Mistry (2014) b							19	↑10.8	7.8, 13.9			

Table 7 Biological markers of nutritional status in individual studies compared to reference values

Table 7 Biological	markers of nutritiona	status in individual stuc	dies compared to reference	ce values (Continued)

Magnesium mmol/l	Chan (2006)							23	↑0.99	0.9, 1.1
Phosphorous mg/dl	Chan (2006)							23	∱5	4, 6
Copper µg/dl ^g	Mistry (2014) a				107	1206	1991, 2128			
	Mistry (2014) b				19	196	1712, 2207			
Selenium µg/L ^h	Dawson (2000) a	21	↓49	46, 52	21	↓55	52.9, 57.1	21	† 114	108.9, 119.1
	Dawson (2000) b	20	↓44	39.6, 48.4	20	↓53	50.8, 55.2	20	↓55	50.2, 59.8
	Dawson (2000) a	21	↓50	46.2, 53.8	21	↔85	76.9, 93.1			
	Dawson (2000) b	20	↓44	39.6, 48.4	20	↓ 62	57.2, 66.8			
	Mistry (2014) a				107	↓65	62.7, 67.5			
	Mistry (2014) b				19	↓49	45.9, 52.9			
Red blood cell folate (nmol/l) ⁱ	Baker (2009)				266	† 647	616, 680			
Serum folate (nmol/	Baker (2009)				291	↑12	12, 14			
))	Chan (2006)							23	↔13	10, 17
Vitamin A (µg/dL)	Chan (2006)							23	14	31, 45
B12 (pmol/l) ^k	Baker (2009)				293	† 177	169, 185			
	Chan (2006)							23	↑265	216, 315
	Gadowsky (1995)				50	↑ 170	146, 194			
Homocysteine	Baker (2009)				293	7.9	7.6, 8.2			
(µmol/L) '	Gadowsky (1995)				50	6.1	3, 9			
250HD (nmol/L) ^m	Baker (2009)				263	133	30.4, 35.8			
	Chan (2006)							23	∱57	48, 67
	Ginde (2010)									
	McGuire Davies (2010)	44	↑52	46, 58	36	↑ 56	50, 63			
	O'Brien (2003)				23	↑50	41, 60			
	Young (2012)							171	↑ 54	51, 59

^a Target value 110 g/L, ^b Target value 15 μ g/L, ^c Target value33 g/L, ^d Target value 6.12 μ mol/L-1, ^e Target value 0.9 mmol/L, ^f Target value 2.5 mg/dl, ^g Target value 63.7 μ g/dl, ^h Target value 90 μ g/L, ⁱ Target value 340 nmol/L, ^j Target value 20 μ g/dl, ^k Target value 150pmol/L, ¹ Target value less than 13 μ mol/L, ^m Target value 25 nmol/L, [↑] Study mean higher than reference value (p < 0.05), \leftrightarrow Study mean not different to reference value (p < 0.05), \downarrow Study mean lower than reference value (p < 0.05)

In terms of comparison with dietary reference values, combined analysis of the individual study means showed very few statistically significant results, with the exception of vitamin D. One possible explanation for this is that there was a high level of variance between participants in the majority of studies resulting in very wide confidence intervals. This suggests that there may be sub-groups of young women within the total population who are at higher risk of poor nutritional status which this review has failed to detect. Differences in micronutrient intake were observed between UK and USA based studies which may in part be explained by routine fortification of food products in the USA compared to the UK. Only one study reported that supplements were included in the reported intake values, and the compliance rates for supplements in this study were low, meaning that the impact of supplement use on intakes data is marginal.

Macronutrient contributions to energy were found to be roughly in line with recommendations, however there was a significant lack of data for carbohydrates (including sugars and fibre) and fats (including saturated fat)

Biological Marker	Pooled Mean	Confidence Interval	Target Value	Number of studies
Haemoglobin (g/dL)	112.4	95.4–133.8	110	9
25_OH_D (nmol/L)	55.9	6.2–105.7	25	6
Ferritin (µg/L)	18.2	0–48.3	15	4
Haematocrit (g/L)	32.5	28.0-39.4	33	4
Zinc (µmol/L-1)	11.5	5.9–17.1	6.12	3
B12 (pmol/L)	181.6	0-271.1	150	3
Serum Folate (nmol/L)	12.8	0–26.8	10	2
Selenium (µg/L)	61.8*	39.2-84.4	90	2
Homocystine (µmol/L)	7.6	0–17.0	Less than 13	2

 Table 8 Pooled weighted means of biological markers

* Significant at the *p*<0.005 level

meaning these results should be interpreted with caution. Further research into macronutrient intakes in this population, particularly with regard to types of carbohydrates and fats, is needed.

The methods used to assess dietary intake varied across the included studies. The two methods reported by the included studies were 24 h dietary recalls (single and multiple) and food diaries. While these are validated and accepted methods of nutritional surveillance [45], it is acknowledged that underreporting biases may exist [46] and so results should be considered with this in mind.

Four of the studies reporting energy intake also reported mean gestational weight gain which ranged from 14 kg to 17 kg, consistently higher than the required pregnancy weight gain [47]. Mean percentage intake of the EAR for energy was higher in the studies reporting weight gain than those which did not report this measure, but not significantly so. This is potentially contradictory of the finding that energy intake was low in individual studies and suggests further work is needed regarding the potential level of under reporting in this population and the relationship between dietary patterns, overall energy consumption and gestational weight gain.

Inadequacies in nutrient intakes did not necessarily translate to systemic deficiencies as measured by mean values of biological markers, with the possible exceptions of markers of iron and selenium status. One possible explanation for this is that food intake may have been under reported therefore suggesting that intake was insufficient when this was not the case. A further possible explanation is that measures of biological markers were elevated by dietary supplements. Details regarding the type, dose, duration and number of participants taking supplements were inconsistent in the included papers meaning that detailed analysis of the impact of supplement use on nutritional status was not possible, however 10 out of the 15 included studies reporting biomarkers did report some level of supplementation. This finding does suggest that supplements may play an important role in ensuring young women do not experience nutrient insufficiency, however attention to clear reporting of supplement use in research papers is essential to allow a better understanding of the impact of supplementation on nutritional status.

The participants in all of the included studies where supplements were provided may also have been more compliant with taking supplements due to the very fact that they were taking part in a research study than might be expected outside of a study environment. A systematic review [48] of the effect of dietary interventions in adolescent pregnancies found some evidence to suggest that nutritional supplements may reduce the likelihood of low birth weight; however the review also reported a serious lack of good quality research papers in this area. Further work to establish the extent to which pregnant young women in the general population suffer more from nutrient deficiencies and the impact of supplement use would be advantageous.

There is significant evidence in the literature regarding the role of nutrition in supporting healthy pregnancies and allowing the foetus to achieve its full potential. Adolescent girls are at particular risk of iron deficiency anaemia due in part to rapid growth during adolescence [49] combined with the onset of menarche. This coupled with the increased demand for iron in pregnancy for expansion of maternal tissues and foetal growth, makes pregnant adolescents a particularly vulnerable group. There is some evidence to suggest that iron deficiency is implicated in the risk of adverse birth outcomes such as prematurity and low birth weight [50], meaning this is potentially an important factor in improving maternal and infant health. While analysis of mean values for markers of iron deficiency in this review did not indicate a significant issue, consideration of the reported

prevalence of iron deficiency anaemia in the included studies suggests that this may be a concern for this population.

Vitamin D and calcium have an essential role in the mineralization of the developing foetal skeleton and insufficient intake of these nutrients may impact on foetal bone growth. The interaction between these two nutrients has been shown to be key to maximising foetal bone growth in pregnant adolescents and that growth is adversely affected when either of the two nutrients were lacking [30]. The pooled mean for vitamin D status as reflected in blood 25 (OH)D in this review was significantly below recommended levels across all trimesters. It was not possible to conduct analysis of vitamin D status by ethnicity or exposure to sunlight, however, which are factors known to have a significant impact on vitamin D status [51–53].

The role of folate in the prevention of neural tube defects in early pregnancy has been well documented [54]. There is a lack of data collected in early pregnancy in the papers included in the current review, however the observed failure to meet recommendations for folate intake in later pregnancy reported may suggest that the participants were unlikely to have been meeting recommendations prior to taking part in a research study. A systematic review of the impact of folate intake over the course of pregnancy [55] also found a significant effect in the second and third trimesters on infant birth weight, suggesting that the importance of folate for a healthy pregnancy extends beyond the first trimester.

Selenium status has been identified as a potential area of concern in this review. Selenium is a trace element which has an anti-oxidative effect and protects cell membranes [56]. The target value used to assess selenium status was based on the intake necessary for maximisation of plasma glutathione peroxidase activity, which is the criteria used in the derivation of the US RDA [18]. Selenium status has been shown to be associated with a number of adverse outcomes for both mother and child including neural tube defects [57], lower birth weights [58], cholestasis [59] and gestational diabetes [60].

While demographic characteristics of participants were reported in the included studies data was generally reported for the study population as a whole, meaning that sub-group analysis was not possible. This is significant in that evidence suggests that younger adolescents, those who smoke and those from more deprived backgrounds may be at higher risk of nutritional issues [3, 61, 62].

There is also evidence to suggest that the nutritional status of adult pregnant women may raise similar concerns to those identified within this review. A systematic review of micronutrient intakes in pregnancy found that intakes of folate, vitamin D and iron were sub-optimal [63]. A further review focusing on energy and macronutrient intake in this population found that intakes of energy and fibre were also below recommendations [64]. These results are consistent with the findings of this review suggesting that maternal age alone may not be the most important factor in sub-optimal nutritional status during pregnancy. Further work to identify the characteristics of those most at risk, particularly within the adolescent population, and the nature of that risk is needed.

Limitations

There were some significant limitations which impact on the conclusions of this review. The majority of the included papers used convenience samples meaning that there is likely to be an element of bias in the reported outcomes. The majority of participants in the included studies were aged 16 and over meaning that the results may not be generalisable to younger adolescents, who may also be at greater nutritional risk compared to older adolescents due to competing growth needs [3]. The lack of detail regarding participant's supplement use meant that it was not possible to evaluate the impact of supplements on biological markers of nutritional status. It is therefore likely that these results may have been biased by supplement use in some participants.

There was significant heterogeneity in the included papers in terms of study design. Measurements of dietary intake differed between papers however the majority of studies used 24 h recall methods to assess nutrient intake. This has been shown to have limitations in terms of both participants reporting their intake accurately and the likelihood that the recorded intake is representative of the usual diet, particularly in adolescents [46]. Three studies used multiple 24 h recalls in order to produce more reliable estimates of intakes; however this approach was not consistent across the included studies using this method.

There were also considerable differences in the number of nutritional indicators represented. Pooled means were calculated wherever two or more data points were available in order to maximise the results available from the review. This however means that some estimates will be more robust than others depending on the number of data points on which they are based. There was a large degree of variation in the amount of data available for different nutrients, for example assessment of serum selenium was based on data from only two papers whereas nine independent studies contributed data on haemoglobin concentration.

The pooling of study means gives a useful indication of potential inadequacies across the population as a whole; however this approach lacks the sensitivity to draw conclusions regarding the prevalence of nutrient deficiencies in the population. Examination of the reported prevalence of deficiency in the included studies shows results which are inconsistent with the analysis based on study means. The prevalence of iron deficiency anaemia measured by haemoglobin concentration reportedly ranged from 1.2 to 63.5 %, with prevalence in the third trimester ranging from 29 to 63.5 %. Other markers of iron status followed a similar pattern with higher prevalence of deficiency occurring in the third trimester. The prevalence of vitamin D deficiency in many of these studies was also higher than suggested by the analysis of study means. This suggests, similarly to the data regarding nutrient intakes, that there are substantial skews within the data and sub-groups of young women who may be more at risk of deficiencies.

Conclusion

This review identifies some concerns in the nutritional status of pregnant adolescents which may impact on maternal and infant outcomes. Intake of vitamin D and serum selenium status were identified as being significantly low in pooled analysis of included studies. Fibre intake was also below recommendations.. This said there are some significant limitations meaning these results should be interpreted with caution. No analysis of the effect of demographic characteristics on either nutritional intake or biological markers was possible, nor was it possible to examine the impact of supplement use on biological markers.

Patterns in this population are also similar to those reported in the adult population during pregnancy. These findings suggest that targeted work to identify those most at risk, and the nature of that risk, is needed. Recommendations for other areas of further research include the macronutrient composition of adolescent's diet during pregnancy, the relationship between nutrient intakes and birth outcomes and the role of nutritional supplements in this population.

Additional file

Additional file 1: Table S1. Characteristics of Excluded Studies. Table S2. Pooled, Weighted Mean Nutrient Intakes Expressed as Percentages of Dietary Reference Values by Stage of Pregnancy. Table S3. Pooled, Weighted Mean Nutrient Intakes Expressed as Percentages of Dietary Reference Values by Study Country of Origin. Table S4. Pooled, Weighted Mean Biological Markers of Nutritional Status by Stage of Pregnancy. Table S5. Pooled, Weighted Mean Biological Markers of Nutritional Status by Study Country of Origin. (DOCX 57 kb)

Abbreviations

CASP: Critical appraisal and skills program; DRV: Dietary reference value; EAR: Estimated average requirement; RDA: Recommended daily allowance; RNI: Reference nutrient intake; WHO: World Health Organisation

Acknowledgments

The authors would like to thank Dr Karen Kilner for her assistance with the statistical methods used in this review.

We would also like to thank the National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care for Yorkshire and Humber (NIHR CLAHRC YH) for supporting us in conducing this review. Further details about the new NIHR CLAHRC YH can be found at www.clahrc-yh.nihr.ac.uk. The views and opinions expressed are those of the authors, and not necessarily those of the NHS, the NIHR or the Department of Health.

Funding

This review has been carried out as part of a PhD project supported by Sheffield Hallam University and the National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care for Yorkshire and Humber. Therefore no additional funding was required for this work.

Availability of data and materials

All data on which the paper is based has been made available in either the main tables or Additional file 1: Tables S1-S5.

Authors' contributions

KM-D Primary responsibility for development of research question and study design, extraction and analysis of data and writing the main body of the article. VB Advisory role in development of research question and study design, revisions to and approval of final article. HS Advisory role in development of research question and study design, cross-checking of all extracted data, revisions to and approval of final article. All authors read and approved the final manuscript.

Competing interests

The authors declare that they have no competing interests.

Consent for publication

Not applicable.

Ethics approval and consent to participate

Ethical approval was not required for this systematic review due to only studies previously published in the academic literature being eligible for inclusion.

Author details

¹Centre for Health and Social Care Research, Sheffield Hallam University, Collegiate Crescent, Sheffield S10 2BP, UK. ²School of Food Sciences and Nutrition, University of Leeds, Leeds LS2 9JT, UK.

Received: 23 April 2016 Accepted: 6 September 2016 Published online: 15 September 2016

References

- Cook SM, Cameron ST. Social issues of teenage pregnancy. Obstetrics Gynaecol Rep Med. 2015;25(9):243–8.
- Office for National Statistics. Census data. http://www.ons.gov.uk/ons/guidemethod/census/2011/census-data/index.html. Updated 2014. Accessed 1 Mar 2016.
- Gibbs CM, Wendt A, Peters S, et al. The impact of early age at first childbirth on maternal and infant health. Paediatr Perinat Epidemiol. 2012;26(s1):259–84.
- Burke BS, Harding W, Stuart HC. Nutrition studies during pregnancy: IV. Relation of protein content of mother's diet during pregnancy to birth length, birth weight, and condition of infant at birth. J Pediatr. 1943;23(5):506–15.
- Oliver MH, Jaquiery AL, Bloomfield FH, et al. The effects of maternal nutrition around the time of conception on the health of the offspring. Soc Rep Fertil suppl. 2007;64:397.
- Murphy SP, Guenther PM, Kretsch MJ. Using the dietary reference intakes to assess intakes of groups: pitfalls to avoid. J Am Diet Assoc. 2006;106(10):1550–3.
- Picciano MF. Pregnancy and lactation: physiological adjustments, nutritional requirements and the role of dietary supplements. J Nutr. 2003;133(6):1997S–2002S.
- 8. Bates, B., Lennox, A., Prentice, A., et.al (Eds.). National Diet and Nutrition Survey Results from Years 1, 2, 3 and 4 (combined) of the Rolling

Programme (2008/2009-2011/2012): A Survey Carried Out on Behalf of Public Health England and the Food Standards Agency. 2014

- Kearney J. Food consumption trends and drivers. Philo Trans Royal Soc London B. 2010;365(1554):2793–807.
- Moran VH. A systematic review of dietary assessments of pregnant adolescents in industrialised countries. Br J Nutr. 2007;97(03):411–25.
- 11. Hall Moran V. Nutritional status in pregnant adolescents: a systematic review of biochemical markers. Maternal child nutri. 2007;3(2):74–93.
- Malik K. Human development report 2014. New York: New York United Nations Development Programme; 2014.
- Onis MD, Onyango AW, Borghi E, et al. Development of a WHO growth reference for school-aged children and adolescents. Bull World Health Organ. 2007;85(9):660–7.
- 14. Critical Appraisal Skills Programme. CASP checklists. http://www.casp-uk.net/ #lcasp-tools-checklists/c18f8. Updated 2013. Accessed 7 Dec 2015.
- Department of Health (DoH). Dietary reference values for food energy and nutrients for the United Kingdom. Committee on Medical Aspects of Food Policy. Report on Health and Social Subjects. 1991;41:xxii-xxiv.
- Otten JJ, Hellwig JP, Meyers LD. (Eds.). Dietary reference intakes: the essential guide to nutrient requirements. Washington DC: National Academies Press; 2006.
- 17. Scientific Advisory Committee for Nutrition. Dietary reference values for energy. London: HMSO; 2011.
- Duffield AJ, Thomson CD, Hill KE, et al. An estimation of selenium requirements for New Zealanders. Am J Clin Nutr. 1999;70(5):896–903.
- 19. Leffler D. BMJ Best Practice Zinc deficiency. 2015 http://bestpractice.bmj. com Accessed 1 Jun 2016
- Uauy R, Olivares M, Gonzalez M. Essentiality of copper in humans. Am J Clin Nutr. 1998;67(5):9525–95.
- 21. Romani A. Assessment of magnesium deficiency. 2015 http://bestpractice. bmj.com Accessed 1 Jun 2016
- Walker HK, Hall WD, Hurst JW, et al. Chapter 198. In: Walker HK, Hall WD, Hurst JW, editors. Serum inorganic phosphorus in Clinical Methods: The History, Physical, and Laboratory Examinations. 3rd ed. Boston: Butterworths; 1990.
- 23. Baker PN, Wheeler SJ, Sanders TA, et al. A prospective study of micronutrient status in adolescent pregnancy. Am J Clin Nutr. 2009;89(4):1114–24.
- Chan GM, McElligott K, McNaught T, et al. Effects of dietary calcium intervention on adolescent mothers and newborns: a randomized controlled trial. Obstet Gynecol. 2006;108(3, Part 1):565–71.
- Lee, S., Young, B. E., Cooper, E. M., et.al. Nutrient inadequacy is prevalent in pregnant adolescents, and prenatal supplement use may not fully compensate for dietary deficiencies. ICAN: Infant, Child, & Adolescent Nutrition. 2014. 1941406414525993
- Pope JF, Skinner JD, Carruth BR. Adolescents' self-reported motivations for dietary changes during pregnancy. J Nutr Educ. 1997;29(3):137–44.
- Giddens JB, Krug SK, Tsang RC, et al. Pregnant adolescent and adult women have similarly low intakes of selected nutrients. J Am Diet Assoc. 2000;100(11):1334–40.
- Gutierrez YM. Cultural factors affecting diet and pregnancy outcome of Mexican American adolescents. J Adolesc Health. 1999;25(3):227–37.
- Young MF, Pressman E, Foehr ML, et al. Impact of maternal and neonatal iron status on placental transferrin receptor expression in pregnant adolescents. Placenta. 2010;31(11):1010–4.
- Young BE, McNanley TJ, Cooper EM, et al. Maternal vitamin D status and calcium intake interact to affect fetal skeletal growth in utero in pregnant adolescents. Am J Clin Nutr. 2012;95(5):1103–12.
- McGuire Davis L, Chang SC, Mancini J, et al. Vitamin D insufficiency is prevalent among pregnant African American adolescents. J Pediatr Adolesc Gynecol. 2010;23(1):45–52.
- Ginde AA, Sullivan AF, Mansbach JM, et al. Vitamin D insufficiency in pregnant and nonpregnant women of childbearing age in the United States. Am J Obstet Gynecol. 2010;202(5):436–e1.
- O'Brien KO, Nathanson MS, Mancini J, et al. Calcium absorption is significantly higher in adolescents during pregnancy than in the early postpartum period. Am J Clin Nutr. 2003;78(6):1188–93.
- Meier PR, Nickerson HJ, Olson KA, et al. Prevention of iron deficiency anemia in adolescent and adult pregnancies. Clin Med Res. 2003;1(1):29–36.
- Dawson EB, Albers JH, McGanity WJ. The apparent effect of iron supplementation on serum selenium levels in teenage pregnancy. Biol Trace Elem Res. 2000;77(3):209–17.

- Iannotti LL, O'Brien KO, Chang SC, et al. Iron deficiency anemia and depleted body iron reserves are prevalent among pregnant African-American adolescents. J Nutr. 2005;135(11):2572–7.
- Chang SC, O'Brien KO, Nathanson MS, et al. Hemoglobin concentrations influence birth outcomes in pregnant African-American adolescents. J Nutr. 2003;133(7):2348–55.
- Pobocik RS, Benavente JC, Boudreau NS, et al. Pregnant adolescents in Guam consume diets low in calcium and other micronutrients. J Am Diet Assoc. 2003;103(5):611–4.
- Derbyshire E. Dietary habits of pregnant and non-pregnant adolescents. Nutri Food Sci. 2009;39(5):490–8.
- Mistry HD, Kurlak LO, Young SD, et al. Maternal selenium, copper and zinc concentrations in pregnancy associated with small-for-gestational-age infants. Maternal child nutri. 2014;10(3):327–34.
- Job, J., Capra, S., & Ash, S. Nutritional assessment of pregnant teenagers attending a metropolitan, public, and maternity hospital in Brisbane. I. Nutrient intakes. Australian j nutri diet. 1995
- Gadowsky SL, Gale K, Wolfe SA, et al. Biochemical folate, B 12, and iron status of a group of pregnant adolescents accessed through the Public Health System in Southern Ontario. J Adolesc Health. 1995;16(6):465–74.
- Castillo-Durán C, Marín VB, Alcázar LS, et al. Controlled trial of zinc supplementation in Chilean pregnant adolescents. Nutr Res. 2001;21(5):715–24.
- Rycel M, Sobala W, Wilczynski J, et al. Hemoglobin and hematocrit concentrations influence birth outcome in pregnant polish adolescents. Arch Perinatal Med. 2009;15(2):101–5.
- Rockett HR, Colditz GA. Assessing diets of children and adolescents. Am J Clin Nutr. 1997;65(4):11165–225.
- Livingstone MBE, Robson PJ, Wallace JMW. Issues in dietary intake assessment of children and adolescents. Br J Nutr. 2004;92(S2):S213–22.
- 47. Luke B. Nutrition During Pregnancy: Part I, Weight Gain; Part II, Nutrient Supplements. JAMA. 1991;265(2):281–2.
- Soltani H, Duxbury A, Rundle R, et al. A systematic review of the effects of dietary interventions on neonatal outcomes in adolescent pregnancy. Evidence Based Midwifery. 2015;13(1):29–34.
- Scientific Advisory Committee for Nutrition. Iron and health. London: HMSO; 2010.
- Scholl TO, Hediger ML, Fischer RL, et al. Anemia vs iron deficiency: increased risk of preterm delivery in a prospective study. Am J Clin Nutr. 1992;55(5):985–8.
- Webb AR, Kline L, Holick MF. Influence of Season and Latitude on the Cutaneous Synthesis of Vitamin D3: Exposure to Winter Sunlight in Boston and Edmonton Will Not Promote Vitamin D3 Synthesis in Human Skin*. j clin endocrinol metabol. 1988;67(2):373–8.
- Van Der Meer IM, Karamali NS, Boeke AJP, et al. High prevalence of vitamin D deficiency in pregnant non-Western women in The Hague, Netherlands. Am J Clin Nutr. 2006;84(2):350–3.
- 53. Scientific Advisory Committee for Nutrition. Draft vitamin D and health report. London: HMSO; 2015.
- MRC Vitamin Study Research Group. Prevention of neural tube defects: results of the Medical Research Council Vitamin Study. Lancet. 1991;338(8760):131–7.
- 55. Fekete K, Berti C, Trovato M, et al. Effect of folate intake on health outcomes in pregnancy: a systematic review and meta-analysis on birth weight, placental weight and length of gestation. Nutr J. 2012;11(1):1.
- 56. Pieczyńska J, Grajeta H. The role of selenium in human conception and pregnancy. J Trace Elem Med Biol. 2015;29:31–8.
- Cengiz B, Söylemez F, Öztürk E, Çavdar AO. Serum zinc, selenium, copper, and lead levels in women with second-trimester induced abortion resulting from neural tube defects. Biol Trace Elem Res. 2004;97(3):225–35.
- Bogden JD, Kemp FW, Chen X, Stagnaro-Green A, Stein TP, Scholl TO. Lownormal serum selenium early in human pregnancy predicts lower birth weight. Nutr Res. 2006;26(10):497–502.
- Reyes H, Báez ME, González MC, Hernández I, Palma J, Ribalta J, Zapata R. Selenium, zinc and copper plasma levels in intrahepatic cholestasis of pregnancy, in normal pregnancies and in healthy individuals, in Chile. J Hepatol. 2000;32(4):542–9.
- Al-Saleh E, Nandakumaran M, Al-Shammari M, Makhseed M, Sadan T, Harouny A. Maternal-fetal status of copper, iron, molybdenum, selenium and zinc in insulin-dependent diabetic pregnancies. Arch Gynecol Obstet. 2005;271(3):212–7.
- 61. Golley RK, Hendrie GA, McNaughton SA. Scores on the dietary guideline index for children and adolescents are associated with

nutrient intake and socio-economic position but not adiposity. J Nutr. 2011;141(7):1340–7.

- 62. Giannakopoulos G, Panagiotakos D, Mihas C, et al. Adolescent smoking and health-related behaviours: Interrelations in a Greek school-based sample. Child Care Health Dev. 2009;35(2):164–70.
- Blumfield ML, Hure AJ, Macdonald-Wicks L, et al. A systematic review and meta-analysis of micronutrient intakes during pregnancy in developed countries. Nutr Rev. 2013;71(2):118–32.
- Blumfield ML, Hure AJ, Macdonald-Wicks L, et al. Systematic review and meta-analysis of energy and macronutrient intakes during pregnancy in developed countries. Nutr Rev. 2012;70(6):322–36.

Submit your next manuscript to BioMed Central and we will help you at every step:

- We accept pre-submission inquiries
- Our selector tool helps you to find the most relevant journal
- We provide round the clock customer support
- Convenient online submission
- Thorough peer review
- Inclusion in PubMed and all major indexing services
- Maximum visibility for your research

Submit your manuscript at www.biomedcentral.com/submit



3.4 Summary and Implications for Thesis

This review was carried out in a robust and transparent manner, utilising appropriate, validated methods and reported in line with current best practice reporting standards for systematic reviews. The review met the research objective which it was designed to address by creating a comprehensive and structured review of the available literature regarding adolescent women's nutrient intake and nutritional status during pregnancy.

The review also identifies significant remaining gaps in the current research evidence on this topic, particularly in regard to identifying the characteristics of young women at greatest risk of nutritional deficiencies and how these differ from older women.

The following chapter sets out the proposed methodology for this programme of research and details the approach to exploring further these identified gaps in the literature.

Chapter 4 - Methodology

4.1 Introduction

This chapter gives an overview of relevant research paradigms and discusses the theoretical approach underpinning this programme of research. This chapter also explains and justifies the decision to use a mixed method, sequential exploratory design. A brief description of the methods employed at each stage of the research process is also given to complement the methods described in the published papers.

4.2 Research paradigms

A research paradigm has been defined as *"the set of common beliefs and agreements shared between scientists about how problems should be understood and addressed"⁹².* These are characterised by a set of assumptions regarding the nature of reality (ontology), ideas about how we can discover this reality (epistemology) and what methods and procedures we can use to go about acquiring new knowledge (methodology)⁹³.

Traditionally the dominant paradigms have been regarded as holding diametrically opposed views regarding the nature of reality and therefore the appropriate approaches to acquiring new knowledge. The positivist perspective posits that there is a single, objective reality which exists independently regardless of any individual's beliefs or understandings. This single reality can be known and measured using objective and value-free enquiry⁹⁴. Positivists therefore favour quantitative methods and validated tools with which to gather empirical data.

Conversely the constructivist or interpretivist paradigm contends that there is no single reality or truth, rather there are multiple realities which are socially constructed and interpreted by individuals⁹⁵. Therefore in order to learn about the social world it is necessary to collect data, usually using qualitative methods, which helps the researcher to understand these realities from the perspective of the individuals being studied.

The approach to this research is informed by a pragmatic research philosophy. Pragmatic research philosophy asserts that there are numerous different ways in which the natural world may be interpreted and therefore that no single point of view can ever give the full picture⁹⁶. Therefore, research is strongly driven by the research question or questions and the methods employed are those which are judged to fit the question most appropriately. This has often resulted in the reduction of pragmatism as a research paradigm to a 'what works' approach or a purely practical way of selecting research methods with a focus on the 'how' research should be conducted, while neglecting the questions of 'why' research is to be carried out.

The importance of these issues is summarised by Denzin⁹⁷ who advances that classical pragmatism is a theory of truth as opposed to a methodology per se. The central argument being that the meaning of an event cannot be given in advance of experience and that therefore the focus of enquiry should be on the consequences and meanings of an action or event.

Pragmatism is strongly influenced by the work of John Dewey⁹⁸ which sought to reorient philosophy of knowledge away from abstract concepts and towards a focus on human experience. Human experience is described as being built upon two questions which are perpetually linked in a cycle: what are the sources of our beliefs? And, what are the meanings of our actions? In which our beliefs inform our actions and the outcomes of our actions subsequently inform our beliefs. Dewey categorises experiences as those which happen relatively automatically with little conscious thought termed habits, and self-conscious decision making described as inquiry. Conscious inquiry occurs when an element of an experience becomes problematic or has not been encountered before. Central to these ideas about human experience is also the impact of individual emotions or feelings and the influence of others on beliefs and actions. Human beings are inescapably social and emotional creatures and it is this focus on human experience which separates pragmatists from other thinkers who focus on metaphysical discussions about the nature of reality and truth.
4.3 Mixed methods research

As described above, a pragmatic philosophical approach to research is primarily directed by the research question, meaning that the choice of methods is focused on those which address the research question most fully. For this reason this is the paradigm most commonly associated with mixed methods research, where researchers employ both quantitative and qualitative methods which are integrated to address the research problem⁹⁹.

Creswell and Plano Clark¹⁰⁰ describe the core characteristics of mixed methods research as follows:

- 1. It collects and analyses rigorously both qualitative and quantitative data
- 2. It mixes the two forms of data
- 3. It gives priority to one or both forms of data
- 4. It can be a single phase or in multiple phases of a programme of study
- 5. It frames the procedures within philosophical worldviews
- 6. It combines procedures into specific research designs that direct the plan for conducting the study

Use of mixed methods research has been gathering significant momentum over recent years and has significant strengths as a means to enhance understanding of complex issues¹⁰¹. As described in previous chapters, the health of adolescent women during pregnancy and the extent to which this may be influenced by nutritional status is a complex problem requiring a nuanced research approach which can help to address this complexity.

4.4 Methods

This thesis is presented in an article based format meaning that the methods used are reported in the published papers. This section therefore includes more detailed information on the methods employed by the Born in Bradford cohort from which data for the present study is drawn. It also gives a brief overview of the rationale for the analysis methods used and details of ethics and consent procedures.

This programme of research follows a sequential explanatory mixed methods design which, as discussed in the previous section, encompasses two distinct

phases. The first involves quantitative analysis of numerical data which is followed by a qualitative phase in which data is collected to help to explain or elaborate on the findings of the quantitative phase¹⁰². This research design is being employed in this thesis to first establish quantitative differences in birth outcomes and dietary patterns between adolescent pregnant women and older gravidas, following which potential explanations for the anticipated differences will be explored in more depth using qualitative enquiry. In this design, priority is given to the quantitative phases with the qualitative phase employed to expand upon the findings and explore the implications in more detail. The results of the study will then be reported in two distinct phases before being drawn together in a final discussion of findings¹⁰⁰. An overview of the research design is illustrated in figure 7.

Figure 7. Research design



4.4.1 Born in Bradford

The Born in Bradford (BiB) study was set up in response to high rates of childhood morbidity and mortality compared to other similar cities in the UK. Infant mortality had been consistently higher than the national average in the years leading up to the formation of the cohort and rates of congenital abnormalities and childhood disability were among the highest in the UK. Bradford is a large city in the north of England with high levels of deprivation and a large South Asian community, the majority of who are from Pakistan. Due to the differential age and fertility rates between the Pakistani and the majority white British populations almost half of the babies born in Bradford have parents of Pakistani origin, despite making up only around 20% of the total population of the city.

The BiB cohort was therefore established to examine how genetic, nutritional, environmental, behavioural and social factors impact upon children's health and their development. It is also intended that the cohort be followed up into adult life to examine the relationships between birth factors and health outcomes in later life. Pregnant women were recruited to take part in the study at approximately 28 weeks gestation; between 2007 and 2011 a total of 12,453 women with 13,776 pregnancies were recruited to take part. Detailed questionnaire data including information on demographic characteristics, family histories and environmental risk factors was collected from all of the women taking part, a copy of the baseline questionnaire is shown in appendix 2. Anthropometric and blood biomarker measures were also recorded from mothers at recruitment and babies at birth. The study has a biobank containing over 250,000 samples of DNA, maternal blood, urine, cord blood and paternal saliva. The full published cohort profile¹⁰³ and further information regarding the Born in Bradford cohort is available at https://borninbradford.nhs.uk/.

Data from the baseline questionnaire has also been linked to hospital records providing information regarding the birth. It is this linked data which has been utilised in this programme of research.

4.4.2 Statistical analysis

The details of how specific methods were used in each phase of the data analysis are given in the published papers and therefore will not be repeated here. This section therefore briefly states the statistical analysis methods selected and gives a rationale for their use.

4.4.2.1 Rationale for use of binary logistic regression

The first of the published original research papers *"Impact of adolescent age on maternal and neonatal outcomes in the Born in Bradford cohort"* uses binary logistic regression as the main analysis method to assess the extent to which maternal age group influences maternal and neonatal outcome. This method was selected as the primary interest of this programme of work is to assess the needs of adolescent pregnant women; therefore by grouping the participants by age it was possible to assess to what extent adolescent maternal age is a predictor of adverse maternal and neonatal outcomes in comparison to a reference group¹⁰⁴.

Potentially confounding variables were identified based on the academic literature and considered for inclusion in the regression model. In the final model only two confounding variables were included, index of multiple deprivation score and ethnicity due high levels of correlation between other potentially confounding variables. The analysis was also restricted to primiparous women delivering a singleton in order to mediate the effects of parity and multiple births on the outcomes.

Due to the nature of the study being a secondary analysis of an existing data set no sample size calculations were carried out and rather all of the eligible women in the cohort were included in the analysis. A post-hoc power analysis has not been carried out as it is not appropriate. Observed power is directly derived from the p-value of the statistical test, therefore providing a post-hoc power calculation would be superfluous; where the tests produce a statistically non-significant result it can be assumed that the study is insufficiently powered to detect an effect of the size obtained in the study¹⁰⁵.

4.4.2.2 Rationale for use of principle component analysis

The second of the published original research papers "Differences in dietary pattern by maternal age in the Born in Bradford cohort: A comparative analysis" uses principle component analysis (PCA) to identify patterns in the types of foods consumed by individuals, the results of which then being compared between age groups.

PCA was selected as it allows for a larger number of variables to be reduced and underlying factors exposed. Portions per day of food groups were entered into PCA in SPSS using 'DIMENSION REDUCTION' -> 'FACTOR' functions. The PCA was conducted based on the correlation matrix in order to control for differences in variances among the variables.

Rotation was used in order to maximise the loading of each variable on to one of the extracted factors whilst minimising the loading on all other factors. Because there is a reasonable theoretical rationale for expecting dietary intake variables to be correlated with one another an oblique rotation (direct oblimin) was selected.

Factor scores were retained as new variables; the regression method was selected due to correlations between the factor scores being considered to be acceptable.

Several combinations of food grouping were explored to see which gave the most sensitive output. Further tests were carried out to assess the suitability of the data for carrying out a PCA. The overall Kaiser-Meyer-Olkin (KMO) Measure of Sampling Adequacy test returned a result of 0.677 which is higher than the suggested minimum of 0.5^{106} . The anti-image correlation matrix was also examined to ensure that the KMO statistic for each individual variable was adequate. Bartlett's Test of Sphericity was also undertaken to confirm that the correlation matrix was significantly different to an identity matrix (in which all of the included variables are perfectly independent) which returned a statistically significant result (P=0.000).

4.5 Data sharing, consent and ethics

Ethical considerations in research are of paramount importance to ensure that research is of good quality and of benefit to the communities it is designed to answer questions about. The key principles of social research ethics are to minimise any risk of harm to individuals, including avoiding deceptive practices, to obtain informed consent from participants including informing them of their right to withdraw from the study and to protect the anonymity and confidentiality of individuals wherever possible¹⁰⁷.

Different types of research therefore require different actions in order to ensure that these basic principles are upheld. These are discussed below along with details of the formal processes undertaken to ensure the project adheres to ethical guidelines.

4.5.1 Born in Bradford

The use of secondary data in research has a number of significant advantages including saving time, money and other resources and ensuring that populations are not unnecessarily overburdened by repeated research¹⁰⁸. The main ethical concerns with the use of secondary data are to do with avoiding potential harm to participants and the issue of consent for specific use of personal data beyond the original research project. The original consent procedure for the Born in Bradford study included obtaining consent for anonymised data to be used for future research projects, meaning that explicit consent for the use of data for this study is not required. Any potential harm to participants is minimised as individuals are not identifiable from the data provided for this study, meaning that their confidentiality is protected. This said participants still have the right to withdraw consent and for their data to be removed from the analysis up until the time that the research is published. Use of the anonymised Born in Bradford data set was agreed subject to a data transfer agreement (appendix 3). Ethical approval for the main Born in Bradford cohort study was granted in April 2008 and a copy of the approval letter is included in appendix 4. Further to this, ethical approval for the specific use of the data for the current program of research was granted by Sheffield Hallam University Ethics Committee, a copy of which is shown in appendix 5.

4.5.2 Qualitative interviews

The ethical considerations associated with the qualitative interviews were similar to those addressed in the quantitative phase in that the same ethical principles are to be adhered to. Participants were self-selected by responding to recruitment materials circulated in their place of work. The purpose and scope of the project was explained to participants both at the time of recruitment and immediately prior to the interview, ensuring that the consent they provided was fully informed. Participants were given a participant information sheet (appendix and the opportunity to ask any questions at least 24 hours ahead of being asked to sign a consent form (appendix 7). Interviews were audio recorded and transcribed verbatim with any identifying information removed, following which the audio files were destroyed. This process ensures that individuals are not identifiable from the data and their confidentiality is maintained. Consent forms containing participant's personal details have been stored securely and will be destroyed when they are no longer needed. Ethical approval for the qualitative phase of this programme of research was granted by Sheffield Hallam University Ethics Committee in July 2016, a copy of the approval letter is included in appendix 8.

4.6 Summary

In this chapter the concept of philosophical research paradigms has been discussed and a rationale given for the adoption of a pragmatic approach to this programme of research. The research design has been described with reference to the philosophical underpinnings. The cohort from which the participants for the quantitative phase of enquiry are drawn has been described and the statistical methods used in the analysis, which are presented in the published papers, have been elaborated on with rationale given for the choice of methods.

The following three chapters present the three main studies which make up this programme of research. The studies are presented as published research articles in their original form.

Chapter 5 - Maternal and neonatal outcomes

5.1 Introduction

This chapter briefly summarises the rationale for the first study in this programme of research assessing the impact of age on maternal and neonatal outcomes in the Born in Bradford cohort, followed by a copy of the full published research paper in its original form. The chapter concludes with a summary of key points, measurable impact of the work and discussion of its contribution to the thesis. As discussed in chapter two there is a body of evidence suggesting that the rates of some adverse neonatal outcomes such as low birthweight and pre-term delivery are higher among babies born to adolescent mothers despite lower rates of adverse maternal outcomes such as pre-eclampsia. This study aims to contribute to this body of evidence by comparing outcomes in adolescent women to an adult reference group (20-34 years) in the Born in Bradford cohort. This large data set, the characteristics of which were described comprehensively in the methods chapter, has not previously been interrogated in this way, providing the opportunity to contribute original knowledge to the field of research. This study contributes to addressing the second research objective as set out in chapter one to "Examine differences in dietary intake and birth outcomes between adolescent and adult mothers in a large cohort data set".

5.2 Publication and Impact

The manuscript describing this study was accepted for publication in BMJ Open in March 2018. BMJ Open is a medical journal which publishes open access papers in the fields of clinical medicine, public health and epidemiology. With an impact factor of 2.413 the journal is internationally well regarded. Since publication online the full text of the article has been accessed 1220 times and has achieved 3 citations (correct as of 20.01.19).

5.3 Published paper: Article B

The published paper entitled "Impact of adolescent age on maternal and neonatal outcomes in the Born in Bradford cohort" is reproduced here in the format in which it was published online.

BMJ Open Impact of adolescent age on maternal and neonatal outcomes in the Born in Bradford cohort

Katie Marvin-Dowle,¹ Karen Kilner,¹ Victoria Jane Burley,² Hora Soltani¹

To cite: Marvin-Dowle K, Kilner K, Burley VJ, *et al.* Impact of adolescent age on maternal and neonatal outcomes in the Born in Bradford cohort. *BMJ Open* 2018;**8**:e016258. doi:10.1136/ bmjopen-2017-016258

Prepublication history for this paper is available online. To view these files, please visit the journal online (http://dx.doi. org/10.1136/bmjopen-2017-016258).

Received 3 February 2017 Revised 31 January 2018 Accepted 2 March 2018

ABSTRACT

Objectives Explore associations between maternal and neonatal outcomes and maternal age, with particular reference to adolescent women.

Design Population-based cohort study.

Setting Maternity department of a large hospital in Northern England.

Participants Primiparous women delivering a singleton at Bradford Royal Infirmary between March 2007 and December 2010 aged \leq 19 years (n=640) or 20–34 years (n=3951). Subgroup analysis was performed using women aged \leq 16 years (n=68). Women aged 20–34 years were used as the reference group.

Primary outcome measures Maternal and neonatal outcomes.

Results The odds of extremely low birth weight (<1000 g) were significantly higher in the adolescent group (≤19 years) compared with the reference group (adjusted OR (aOR) 4.13, 95% CI 1.41 to 12.11). The odds of very (<32 weeks) and extremely (<28 weeks) preterm delivery were also higher in the adolescent group (aOR 2.12, 95% Cl 1.06 to 4.25 and aOR 5.06, 95% Cl 1.23 to 20.78, respectively). Women in the adolescent group had lower odds of gestational diabetes (aOR 0.35, 95% CI 0.20 to 0.62), caesarean delivery (aOR 0.53, 95% CI 0.42 to 0.67 and instrumental delivery (aOR 0.53, 95% CI 0.41 to 0.67). **Conclusions** This study identifies important differences in maternal and neonatal outcomes between women by age group. These findings could help in identifying at-risk groups for additional support and tailored interventions to minimise the risk of adverse outcomes for these vulnerable groups. Further work is needed to identify the causal mechanisms linking age with outcomes in adolescent women where significant gaps in the literature exist.

Check for updates

¹Centre for Health and Social Care Research, Sheffield Hallam University, Sheffield, UK ²School of Food Science and Nutrition, University of Leeds, Leeds, UK

Correspondence to

Katie Marvin-Dowle; k.marvin-dowle@shu.ac.uk

INTRODUCTION

Pregnancy during adolescence is often associated with less favourable outcomes for both mother and child. Childbearing in adolescence is associated with social problems such as isolation, poverty, low levels of education and unemployment.¹

The impact of maternal age on obstetric and neonatal outcomes has been studied in various parts of the world and with variable results. A WHO multicountry study including

Strengths and limitations of this study

- A particular strength of this work is that it uses well-established, ethnically diverse, UK-based cohort data in a way that is unique to this study.
- A further strength is in the large number of participants available for analysis that enables robust conclusions to be drawn.
- Despite the large number of participants, however, this study is limited by small numbers of occurrences of some rare outcomes, particularly in subgroup analyses.
- It should also be considered that the generalisability of this study to contexts that are very different in terms of socioeconomic and demographic characteristics is limited.

29 low-income and middle-income countries² found adolescent mothers were at higher risk of several adverse outcomes including low birth weight, preterm delivery eclampsia and infections compared with mothers aged 20–24 years.

Similarly in higher income countries, there is evidence to suggest that health outcomes may be less favourable for younger mothers. Babies born to adolescent mothers have been shown to be at higher risk of preterm birth and low birth weight,^{3 4} and higher rates of stillbirth and neonatal mortality have also been reported.⁵ Adolescents have, however, been consistently shown to experience lower rates of caesarean and instrumental delivery⁶ and therefore are at lower risk of complications associated with assisted births. It is not currently clear from the available literature, however, to what extent differences in birth outcomes between adolescent and adult mothers are predicted by age alone.

A systematic review⁷ aiming to assess the relationship between early first childbirth and increased risk of poor pregnancy outcomes found that there was considerable evidence to suggest that very young maternal age (<15 years or less than 2 years after menarche) had a negative effect on both maternal and

BMJ

fetal growth and infant survival. It is suggested that young women who are still themselves growing may compete with the fetus for nutrients, which may in turn impair fetal growth and result in low birthweight babies or babies who are small for their gestational age. The review also found a moderately increased risk of anaemia, premature birth and neonatal mortality associated with young maternal age. Advanced maternal age (35+ years) has also previously been shown to be an independent risk factor for adverse maternal and neonatal outcomes.⁸ This suggests that women aged 20–34 years could reasonably be considered as the population less likely to suffer age-related pregnancy complications.

Differences in outcomes have also been associated with demographic and behavioural characteristics. Lifestyle and sociodemographic factors such as smoking,⁹ alcohol use¹⁰ and deprivation¹¹ have all been shown to contribute to less favourable birth outcomes. It is also established that adolescent mothers in high-income countries are at higher risk of exhibiting these characteristics.¹²

The Born in Bradford study is a cohort of approximately 13500 children born at Bradford Royal Infirmary between March 2007 and December 2010. The cohort reflects the diversity of the population in Bradford and as such is a largely biethnic sample with high levels of socioeconomic deprivation, which presents a unique opportunity to explore any differences in birth outcomes between adolescent and adult women and the factors that contribute to these differences. A detailed profile of the cohort has been previously published.¹³

Some work has already been carried out looking at maternal and neonatal outcomes in the Born in Bradford cohort, particularly with reference to maternal ethnicity^{14 15}; however, this cohort has not previously been examined with reference to maternal age.

While these studies have shown some interesting associations between maternal and neonatal outcomes and maternal ethnicity, the impact of maternal age on outcomes is yet to be explored in this cohort. The size and diversity of this cohort allow for detailed analysis to be carried out and factors known to impact on maternal and neonatal outcomes to be controlled for, making this study unique in a UK context. For these reasons, the primary aim of this investigation is to explore the relationship between maternal and neonatal outcomes and maternal age in the Born in Bradford cohort.

METHODS

Born in Bradford is a prospective cohort study for which participants were recruited during pregnancy. The cohort was originally established in response to concerns regarding the high rates of morbidity and mortality in the city. All women booked for delivery at Bradford Royal Infirmary are offered an oral glucose tolerance test at 26–28 weeks' gestation. Women were invited to participate in the Born in Bradford study when attending this appointment or when attending other antenatal appointments. Informed consent was obtained, and women were asked to complete a baseline questionnaire providing data on maternal characteristics. Blood and urine samples were also collected from the mothers as well as cord blood samples collected at birth. Recruitment took place between March 2007 and December 2010, and over 80% of women eligible in this period agreed to take part, which represents approximately 64% of the births occurring in Bradford during this period.¹³ This study uses baseline questionnaire data and hospital maternity data collected by Born in Bradford to examine maternal and neonatal outcomes. The youngest women recruited to the cohort were 15 years old; therefore, data for this study were limited to primiparous women aged 15-34 years at delivery who had a singleton pregnancy; data relating to 4591 pregnancies were available for this analysis. A flow chart describing the Born in Bradford cohort and the subset used for this study is shown in figure 1.

Outcome variables

The binary neonatal outcome variables studied were low birth weight (below 2500 g), very low birth weight (below 1500g), extremely low birth weight (below 1000 g), macrosomia (birth weight over 4000 g), small for gestational age (birth weight lower than the 10th percentile for the sample),¹⁶ large for gestational age (birth weight higher than the 90th percentile for the sample),¹⁶ preterm birth (<37 completed weeks gestation), very preterm birth (<32 completed weeks' gestation), extremely preterm birth (<28 completed weeks' gestation), outcome of birth (live birth or stillbirth) and Apgar score at 1 min and 5 min (analysed as two groups: <7 and 7–10). Low, very low and extremely low birthweight and macrosomic infants were compared with infants born weighing 2500-4000 g, small and large for gestational age infants were compared with appropriate for gestational age infants and those born preterm or very or extremely preterm to those born \geq 37 completed weeks' gestation. Birth weight and gestational age at delivery were also considered as continuous variables. The maternal outcome variables included in this analysis were diagnosis of pre-eclampsia (diagnosis in this cohort was made when proteinuria is >0.3 mg and blood pressure is $\geq 140/90$ mmHg on more than one occasion), diagnosis of gestational diabetes (defined as a 2-hour postglucose load plasma glucose level of 7.8 mmol/L or a fasting plasma glucose level of 6.1 mmol/L)¹⁴ and mode of birth (normal vaginal, instrumental (including both forceps and ventouse deliveries) or caesarean section). Distinction between elective and emergency caesarean sections was not available. The outcome variables were collected in the process of routine maternity care and were made available for this analysis via data linkage to questionnaire data.



Figure 1 Details of the Born in Bradford cohort and subset used for the present study. This shows participants recruited to the main Born in Bradford cohort study and the subset of these participants whose data is used in the present study.

Statistical analysis

Outcomes in women aged ≤ 19 years were compared with outcomes for women in the reference group (20–34 years). Age group of 20–34 years was selected as the reference group as this group is the least likely to suffer age-related complications as discussed in the introduction.

Characteristics of the sample were described, presenting categorical variables as percentages and continuous variables as means and SD. This analysis was carried out both for demographic characteristics and for maternal and neonatal outcome variables. Differences between maternal age groups were explored using $\chi 2$ for categorical data and Student's t-test for continuous data.

Simple linear regression was calculated to predict both birth weight and gestation to last completed week at delivery based on maternal age at delivery.

Logistic regression analyses were used to compare the rate of each of the binary outcome variables for adolescents and the reference group and differences between groups estimated using ORs.

Multivariate logistic regression models were then used to adjust these comparisons for confounding variables. Crude and adjusted ORs (OR and aOR) are therefore presented with 95% CIs. Index of multiple deprivation (IMD) score and maternal ethnicity (white British, Pakistani or any other ethnicity) were included as covariates in the adjusted analysis. IMD is the official measure of relative deprivation for small areas in England and combines information from seven domains of deprivation (income,

Marvin-Dowle K, et al. BMJ Open 2018;8:e016258. doi:10.1136/bmjopen-2017-016258

employment, education, health, crime, housing and environment) to give a deprivation score.¹⁷

In the multivariate logistic regression model for this study, there is no clear logical or theoretical basis for assuming any variable to be prior to any other, either in terms of its relevance to the research goal of explaining phenomena or in terms of a hypothetical causal structure of the data. For this reason, a simultaneous model of including independent variables in the multivariate logistic regression model was considered to be most appropriate.

Further subgroup analysis was also undertaken to examine the maternal and neonatal outcomes for young women aged ≤ 16 years compared with the reference group and reported in the same way as the main analysis. Statistical analysis was undertaken using SPSS V.24.

RESULTS

Characteristics of the sample

Data were available for 4591 pregnancies for this analysis; characteristics of the participants included in the study are shown in table 1. The majority of participants in the cohort were aged 20–34 years (86.1%) with 13.9% aged 19 years or under. The sample overall was made up of 37.7% Pakistani women, 44.4% white British woman and 17.6% women of other ethnicities. Among women aged 19 years and under only 16.7% were of Pakistani ethnicity and 70% were white British. Women in the adolescent group were also more likely to have been born in the UK

Open Access

6

Table 1 Characteristics of the	aracteristics of the sample by maternal age								
	≤19		20–34		Total		Missin	g	
	n	%	n	%	n	%	n	%	P values
Whole cohort	640	13.9	3951	86.1	4591	100			
Ethnicity							14	0.3	
Pakistani	107	16.7	1623	41.1	1730	37.7			<0.001
White British	448	70.0	1590	40.2	2038	44.4			
Any other ethnicity	85	13.3	724	18.3	809	17.6			
Mother's country of birth							1	0.0	
UK and Ireland	564	88.1	2588	65.5	3152	68.7			<0.001
Southeast Asia	41	6.4	984	24.9	1025	22.3			
Eastern Europe	15	2.3	135	3.4	150	3.3			
Other/unknown	20	3.1	243	6.2	263	5.7			
Marital status							9	0.2	
Married	87	13.6	2445	61.9	2532	55.2			<0.001
Not married – living with	147	23.0	841	21.3	988	21.5			
Single	406	63.4	656	16.6	1062	23.1			
Parents related other than by							3	0.1	
marriage									
Yes	76	11.9	988	25.0	1064	23.2			<0.001
No	564	88.1	2960	74.9	3524	76.8			
Highest level of education							14	0.3	
Less than 5 GCSEs grades A–C or equivalent	231	36.1	553	14.0	784	17.1			<0.001
5 GCSEs grades A–C or equivalent	298	46.6	1121	28.4	1419	30.9			
A-levels or higher	60	9.4	1971	49.9	2031	44.2			
Other/unknown	50	7.8	293	7.4	343	7.5			
Smoked during pregnancy							7	0.2	
Yes	302	47.2	608	15.4	910	19.8			<0.001
No	338	52.8	3336	84.4	3674	80.0			
Drunk alcohol in the first 3 months of pregnancy							2862	62.3	
Yes	185	28.9	698	17.7	883	19.2			0.068
No	140	21.9	702	17.8	842	18.3			
Don't know	1	0.2	3	0.1	4	0.1			
Drunk alcohol since the fourth month of pregnancy							2872	62.6	
Yes	89	13.9	478	12.1	567	12.4			0.06
No	233	36.4	916	23.2	1149	25.0			
Don't know	1	0.2	2	0.1	3	0.1			
Used recreational drugs during pregnancy	I						771	16.8	
Yes	29	4.5	47	1.2	76	1.7			<0.001
No	509	79.5	3235	81.9	3744	81.6			
Used any vitamins or iron supplements in the last 4 weeks							16	0.3	
									Continued

•								-	
Table 1 Continued									
	≤19		20–34		Total		Missin	g	
	n	%	n	%	n	%	n	%	P values
Yes	152	23.8	1610	40.7	1762	38.4			<0.001
No	487	76.1	2326	58.9	2813	61.3			
BMI category							413	9	
Underweight (below 18.5)	59	9.2	199	5.0	258	5.6			<0.001
Healthy weight (18.5–24.9)	368	57.5	1853	46.9	2221	48.4			
Overweight (25–29.9)	113	17.7	955	24.2	1068	23.3			
Obese (30 or higher)	46	7.2	585	14.8	631	13.7			
	n	Mean (SD)	n	Mean (SD)	n	Mean (SD)	n	%	P values
BMI at booking appointment	594	23.3 (4.6)	3641	25.1 (5.4)	4235	24.8 (5.3)	356	7.8	<0.001
IMD score	640	44.7 (18.0)	3948	41.6 (17.9)	4588	41.8 (17.9)	3	0.1	<0.001
Number of weeks' gestation at booking appointment	640	12.1 (5.0)	3951	11.4 (4.3)	4246	12.4 (3.1)	345	7.5	<0.001

BMI, body mass index (kg/m²); GCSE, General certificate of secondary education; IMD, Index of multiple deprivation.

or Ireland (88.1%) compared with the reference group (65.5%). There were other significant variations in the characteristics of the sample by maternal age. Women in the adolescent age group were more likely to not be married or living with a partner, to be expecting their first child and to have completed lower levels of education compared with older women. Women in the adolescent age groups were also more likely to have smoked or used recreational drugs during pregnancy; they were also more likely to have drunk alcohol in the first trimester. Women in the reference group were more likely to be overweight or obese, while adolescent women were found to have higher prevalence of underweight. Older women were also more likely to have taken nutritional supplements in the 4weeks before questionnaire completion compared with younger women. Analysis of continuous variables showed that IMD score decreased as maternal age increased suggesting adolescent women lived in areas of higher deprivation. Adolescent women also booked with a midwife for antenatal care later than older women; there was a mean difference of 1 week between the two groups.

Descriptive analysis relating to maternal and neonatal outcomes is shown in table 2. This analysis suggests that there are several outcome variables that show significant variation by maternal age group. Among the neonatal outcomes, the results show babies born to adolescent women were significantly more likely to have extremely low birth weights or to be born very or extremely preterm. Among the maternal outcomes, lower rates of gestational diabetes, caesarean delivery and instrumental birth were associated with adolescent age.

Linear regression models

A simple linear regression was carried out to assess the relationship between birth weight and maternal age. A statistically significant relationship was found (P=0.044).

The slope coefficient for maternal age was 3.749, meaning that for each 1-year increase in maternal age, birth weight increases by 3.749g. The R^2 value was 0.001, meaning that only 0.1% of the variation in birth weight can be explained by the model containing only maternal age.

Open Access

Similarly, a simple linear regression to assess the relationship between gestation at delivery to last completed week and maternal age found a significant relationship (P=0.011). The slope coefficient for maternal age was -0.016, meaning that for each 1-year increase in maternal age gestation at delivery decreases by 0.016 weeks. The R² value for this regression was also 0.001, meaning that only 0.1% of the variation in gestation at delivery can be explained by the model containing only maternal age.

Logistic regression analysis

The crude and aORs for maternal and neonatal outcomes by maternal age group are shown in table 3.

Women in the adolescent age group were found to have a significantly higher odds of delivering extremely low birthweight babies (<1000 g) compared with the reference group (aOR 4.13, 95% CI 1.41 to 12.11) and delivering extremely preterm (<28 weeks) (aOR 5.06, 95% CI 1.23 to 20.78). Adolescent pregnant women experienced lower odds of being diagnosed with gestational diabetes than the reference group (aOR 0.35, 95% CI 0.20 to 0.62). The odds of women in this age group delivering by caesarean section were decreased (aOR 0.53, 95% CI 0.42 to 0.67), as were the odds of having an instrumental delivery (aOR 0.53, 95% CI 0.41 to 0.69) compared with the reference group.

Subgroup analysis

For some outcomes, the number of events occurring in the subgroup aged ≤ 16 years, was either very small or no events took place. This resulted in either the regression model failing to produce a valid result or the aOR being **Open Access**

Table 2 Descriptive analysis of mate	ernal	and ne	onatal ou	itcomes t	by maternal age	e					
		≤19		20-34	4	Total		Miss	ing		
		n	%	n	%	n	%	n	%		P values
Whole cohort		640	13.9	3951	86.1	4591	100				
Neonatal outcomes											
Low birth weight (<2500g)		56	9.3	349	9.4	405	9.3	0	0.	0	0.933
Very low birth weight (<1500g)		9	1.6	36	1.1	45	1.1	0	0.	0	0.248
Extremely low birth weight (<1000	g)	6	1.1	10	0.3	16	0.4	0	0.	0	0.007
Macrosomia (birth weight >4000g)		35	6.0	223	6.2	258	6.2	0	0.	0	0.852
Small for gestational age		81	14.0	576	16.3	657	16.0	0	0.	0	0.153
Large for gestational age		61	10.9	426	12.6	487	12.4	0	0.	0	0.256
Preterm delivery (<37 weeks)		44	6.9	236	6.0	280	6.1	0	0.	0	0.376
Very preterm delivery (<32 weeks)		12	2.0	35	0.9	47	1.1	0	0.	0	0.021
Extremely preterm delivery (<28 we	eks)	4	0.7	5	0.1	9	0.2	0	0.	0	0.008
Stillborn		5	0.8	26	0.7	31	0.7	0	0.	0	0.724
Apgar score <7 at 1 min		75	11.7	456	11.5	531	11.6	0	0.	0	0.896
Apgar score <7 at 5 min		24	3.8	136	3.4	160	3.5	0	0.	0	0.694
r	า	Mean	(SD)	n	Mean (SD)	n	Mean (SD)	n	%	P values
Birth weight (g) 6	640	3167.6	(580.6)	3950	3183.1 (556.3)	4590	3180.9	(559.7)	1	0.0	0.919
Gestation to last completed 6 week	640	39.2 (2	.2)	3951	39.2 (1.9)	4591	39.2 (1.	9)	0	0.0	0.516
Maternal outcomes											
Pre-eclampsia 1	19	3.0		146	3.7	165	3.6		0	0.0	0.36
Gestational diabetes 1	13	2.0		264	6.7	277	6.0		0	0.0	<0.001
Caesarean delivery S	93	14.5		990	25.1	1083	23.6		0	0.0	< 0.001
Instrumental birth* 7	78	14.3		706	23.9	784	22.4		5	0.1	<0.001

*Vaginal deliveries only, included both forceps and ventouse deliveries.

subject to extremely wide CIs. The results presented do however provide a useful indication of the outcomes that may be important for further investigation. Results of the subgroup analysis are shown in table 4. The only variable to return a significant result in this analysis was for incidence of caesarean section where the odds were lower for women in the ≤ 16 subgroup (aOR 0.31, 95% CI 0.13 to 0.72).

DISCUSSION

Analysis of maternal and neonatal outcomes in the Born in Bradford cohort in this study has found some important differences between women in different age groups.

Adolescent women in the sample were found to be at significantly increased risk of delivering babies extremely preterm and with extremely low birth weights after adjustment for confounding factors. Identifying the risk of delivering babies with an extremely low birth weight is of particular importance due to its association with neonatal mortality and morbidity. Babies with extremely low birth weight are more likely to die in the first few months of life¹⁸ and are more likely to have long lasting physical and cognitive developmental issues¹⁹ compared with babies

born at higher weights. Extreme low birth weight and extreme preterm delivery are intrinsically linked, and thus morbidity and mortality in extremely preterm infants is similar to those with extremely low birth weights.²⁰

Preterm deliveries may be clinically indicated due to medical factors such as intrauterine growth restriction or spontaneous. Both spontaneous preterm delivery²⁰ and intrauterine growth restriction²¹ have been shown to be associated with maternal under nutrition, and the links between intrauterine growth restriction and maternal smoking during pregnancy are well established.^{20,22,23} This study has identified a higher prevalence of both maternal underweight and smoking during pregnancy among the adolescent group compared with controls, suggesting that these may be important mechanisms for further investigation in examining the causes of poorer outcomes in adolescent pregnancies.

In the UK, survival rates for babies born extremely preterm increase rapidly with each additional week the fetus remains in the womb from close to 0 at 22 weeks' gestation to 92% at 28 completed weeks,²⁴ meaning that neonatal death is a significant concern for babies born in this time period. Mortality data were not available for this

	_	_			_	_	_	_	_
			1		 - 1	_			
				_				-	-
<u> </u>					 _	<u> </u>	-	· · ·	-

Table 3 Neonatal and maternal outcomes for adolescent women									
	n	Crude OR (95% CI)	aOR (95% CI)*						
Neonatal outcomes									
Low birth weight (<2500g)	4332	0.99 (0.73 to 1.33)	1.10 (0.81 to 1.50)						
Very low birth weight (<1500g)	3972	1.54 (0.74 to 3.21)	1.59 (0.74 to 3.42)						
Extremely low birth weight (<1000g)	3943	3.69 (1.34 to 10.20)	4.13 (1.41 to 12.11)						
Macrosomia (birth weight >4000g)	4185	0.97 (0.67 to 1.40)	0.78 (0.54 to 1.14)						
Small for gestational age	4104	0.83 (0.65 to 1.07)	1.05 (0.81 to 1.37)						
Large for gestational age	3934	0.85 (0.64 to 1.13)	0.74 (0.55 to 0.99)						
Preterm delivery (<37 weeks)	4591	1.16 (0.83 to 1.62)	1.10 (0.78 to 1.56)						
Very preterm delivery (<32 weeks)	4358	2.14 (1.10 to 4.14)	2.12 (1.06 to 4.25)						
Extremely preterm delivery (<28 weeks)	4320	4.99 (1.34 to 18.62)	5.06 (1.23 to 20.78)						
Stillborn	4591	1.19 (0.46 to 3.11)	1.39 (0.51 to 3.80)						
Apgar score <7 at 1 min	4591	1.02 (0.79 to 1.32)	0.95 (0.73 to 1.25)						
Apgar score <7 at 5 min	4591	1.09 (0.70 to 1.70)	1.11 (0.70 to 1.76)						
Maternal outcomes									
Pre-eclampsia	4591	0.80 (0.49 to 1.30)	0.84 (0.51 to 1.39)						
Gestational diabetes	4591	0.29 (0.17 to 0.51)	0.35 (0.20 to 0.62)						
Caesarean delivery	4591	0.51 (0.40 to 0.64)	0.53 (0.42 to 0.67)						
Instrumental birth†	3503	0.53 (0.41 to 0.69)	0.53 (0.41 to 0.69)						

Reference group: maternal age 20-34 years.

*Adjusted for IMD score and ethnicity.

†Vaginal deliveries only, included both forceps and ventouse deliveries.

aOR, adjusted OR; IMD, Index of multiple deprivation.

study for infants who were born alive; this would be an important area for further study to assess how mortality rates in preterm infants born to adolescent mothers compare with those born to older women.

The linear regression analysis of both birth weight and gestation at delivery showed statistically significant results. This said, the R^2 value for both of these analyses showed that maternal age accounted for only 0.1% of the variation in the analysis, meaning that the clinical importance of this finding is limited. It is likely that there are a number of variables that were either not measured in this study or that are currently unknown in the research literature that contribute to these outcomes.

Adolescent women were also found to be at significantly lower risk of caesarean and instrumental delivery in this analysis. Caesarean delivery is associated with higher rates of postnatal complications and increased recovery time for the mother.²⁵ Instrumental deliveries, while necessary to prevent serious neonatal complications, are associated with a higher prevalence of birth injuries and maternal rehospitalisation.²⁶ These results are consistent with a large body of existing work where these outcomes have been found to be associated with maternal age.^{27 28} It is not known whether these differences are due to biological differences between younger and older women or whether the reasons are more likely to be social or cultural. Further investigation regarding the reasons for difference in mode of birth in women of different ages would be advantageous. The results of this study are consistent with a number of previous similar studies. Results from a study looking at differences in outcomes between adolescent mothers and an older reference group from the North Western Perinatal Survey²⁹ found an increased risk of low birth weight and preterm delivery among adolescent mothers. This study also measured the effect of parity on these outcomes and reported and increased effect in the second pregnancies of adolescents. Analysis in the present study was limited to primiparous mothers only in order to control for the impact of parity in comparison with the control group. There were insufficient numbers of multiparous women in the adolescent group to allow for analysis of these as a separate group in this study; however, the results of this previous study suggest that by excluding second and subsequent pregnancies, the extent of low birth weight and preterm delivery may have been underestimated.

A further study³⁰ comparing adolescent pregnancy outcomes with those of older women found a decreased risk of caesarean section and instrumental delivery in the adolescent group, which is consistent with the findings of this study. This study did however fail to find any association with low birth weight or preterm delivery after adjusting for confounding variables. This analysis did not however look at extreme low birth weight or extreme preterm delivery, which is where the present study has detected differences between groups. Comparison of the **Open Access**

Table 4	Subgroup	analysis c	of neonatal	and materna	l outcomes
---------	----------	------------	-------------	-------------	------------

			Total valid		
	n ≤16	n 20–34	n	Crude OR (95% CI)	aOR (95% CI)*
	68	3951			
Neonatal outcomes					
Low birth weight (<2500g)	5	349	3792	0.81 (0.32 to 2.02)	0.83 (0.32 to 2.13)
Very low birth weight (<1500g)	2	36	3476	3.13 (0.74 to 13.29)	3.00 (0.66 to 13.59)
Extremely low birth weight (<1000g)	1	10	3449	5.63 (0.71 to 44.68)	5.90 (0.67 to 51.85)
Macrosomia (birth weight >4000g)	3	223	3664	0.76 (0.24 to 2.43)	0.62 (0.19 to 2.02)
Small for gestational age	6	576	3585	0.57 (0.24 to 1.33)	0.74 (0.31 to 1.77)
Large for gestational age	8	426	3437	1.03 (0.49 to 2.17)	0.91 (0.42 to 1.95)
Preterm delivery (<37 weeks)	5	236	4019	1.25 (0.50 to 3.14)	1.08 (0.42 to 2.76)
Very preterm delivery (<32 weeks)	1	35	3814	1.69 (0.23 to 12.49)	1.66 (0.21 to 12.88)
Extremely preterm delivery (<28 weeks)	1	5	3784	11.79 (1.36 to 102.41)	6.24 (0.61 to 64.20)
Stillborn	0	26	4019	†	†
Apgar score <7 at 1 min	9	456	4019	1.17 (0.58 to 2.37)	1.02 (0.50 to 2.11)
Apgar score <7 at 5 min	2	136	4019	0.85 (0.21 to 3.51)	0.85 (0.20 to 3.60)
Maternal outcomes					
Pre-eclampsia	4	146	4019	1.63 (0.59 to 4.53)	1.71 (0.59 to 4.91)
Gestational diabetes	0	264	4019	†	†
Caesarean delivery	6	990	4019	0.29 (0.13 to 0.67)	0.31 (0.13 to 0.72)
Instrumental birth‡	14	706	3025	0.83 (0.46 to 1.50)	0.87 (0.47 to 1.60)

Reference group: maternal age 20–34 years.

*Adjusted for IMD score and ethnicity.

†No valid result available due to small numbers.

‡Vaginal deliveries only, included both forceps and ventouse deliveries.

aOR, adjusted OR; IMD, Index of multiple deprivation.

results of this study to key indicators published by Public Health England's Child and Maternal Health Intelligence Network³¹ suggests that despite the uniqueness of this cohort, the results are generalisable to other areas of the UK. Reported national rates for smoking in pregnancy, low birth weight and stillbirth are similar both among the adolescent population and the population as a whole to those reported in this study.

The results of this study contribute to the wider understanding of neonatal and maternal morbidity and mortality both in a UK context and internationally. This study identifies important differences in the risk of adverse outcomes by maternal age, which align with the United Nations sustainable development goals³² and the targets outlined in the Every Woman, Every Child Global Strategy.³³ Preterm births and low birth weights are a major cause of neonatal death and cause more than 1 million deaths globally per year.³⁴ In addition to this, the second leading cause of death for young women aged 15-19 years is complications during pregnancy and childbirth.³⁵ Identifying characteristics that put individuals at higher risk of these complications will help in targeting interventions to populations that are appropriate to their setting.

A significant strength of this study is that it uses a large cohort study, meaning that the majority of statistical analyses do not suffer from problems due to small numbers and the population recruited the cohort is largely representative of the population as a whole. There are however some small difference between the populations recruited and not recruited that should be acknowledged. A lower proportion of mothers aged 20–24 years were recruited compared with those not in the cohort and a higher proportion of South Asian and primiparous women. A lower proportion of mothers at the lower end of the control group may therefore have had some bearing on the prevalence of some outcomes in that group, which is a limitation of this study.

Attempts were made to control for the effect of confounding variables in the multivariate logistic regression model by including a measure of socioeconomic deprivation and ethnicity in the model and by restricting the analysis to primiparous women delivering a singleton. These variables were selected due to their independent association with the outcome variables. Other variables were not included in the model due to a high degree of correlation between variables. There still exists, however, the possibility that the effect sizes detected in this study 6

are influenced by unmeasured or residual confounding variables.

Despite the large numbers overall, there was still only a relatively small number of adolescent women in the cohort, particularly in the subgroup analysis. Stillbirth, premature deliveries and very and extremely low birth weights were also relatively rare events, meaning that this study may have failed to detect differences in outcomes between groups due to being insufficiently powered.

The availability of routine hospital data linked to the cohort data was also a significant strength of this study. The use of this data did however also present limitations in that the analysis was restricted to the variables collected routinely, and there was no opportunity to recover missing data.

CONCLUSIONS

This study identifies some important variations in obstetric and perinatal outcomes by maternal age. Extremely low birth weight and extremely preterm delivery were concerns for adolescent mothers. Findings relating to maternal outcomes were also consistent with the existing literature showing lower risk of gestational diabetes, caesarean delivery and instrumental birth. Further work to establish the causal mechanisms behind the links between maternal age and maternal and neonatal outcomes would be advantageous, particularly for adolescent mothers where there are significant gaps in the existing literature.

Acknowledgements Born in Bradford is only possible because of the enthusiasm and commitment of the children and parents in BiB. We are grateful to all the participants, practitioners and researchers who have made Born in Bradford happen.

Contributors KM-D: completion of data analysis and responsible for writing the manuscript. KK: providing specialist input on statistical methods. VJB: providing specialist input on methods and structure, providing comments and making amendments to the manuscript. HS: providing specialist input on methods and structure, providing comments to the manuscript.

Funding The research wasfunded by the NIHR CLAHRC Yorkshire and Humber through the White Rose PhD studentship network. The views expressedare those of the author(s), and not necessarily those of the NHS, the NIHR orthe Department of Health and Social Care. The Born in Bradford study presents independent research commissioned by the National Institute for Health Research Collaboration for Applied Health Research and Care (NIHR CLAHRC) and the Programme Grants for Applied Research funding scheme (RP-PG-0407-10044).

Disclaimer The views and opinions expressed are those of the authors, and not necessarily those of the NHS, the NIHR or the Department of Health.

Competing interests None declared.

Patient consent Detail has been removed from this case description/these case descriptions to ensure anonymity. The editors and reviewers have seen the detailed information available and are satisfied that the information backs up the case the authors are making.

Ethics approval Ethical approval for the study was granted by Bradford Research Ethics Committee (ref no. 07/H1302/112).

Provenance and peer review Not commissioned; externally peer reviewed.

Data sharing statement Requests for access to data should be addressed to the corresponding author or to the Born in Bradford programme manager rosie. mceachan@bthft.nhs.uk.

Open Access This is an Open Access article distributed in accordance with the Creative Commons Attribution Non Commercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited and the use is non-commercial. See: http://creativecommons.org/ licenses/by-nc/4.0/

© Article author(s) (or their employer(s) unless otherwise stated in the text of the article) 2018. All rights reserved. No commercial use is permitted unless otherwise expressly granted.

REFERENCES

- Cook SMC, Cameron ST. Social issues of teenage pregnancy. Obstetrics, Gynaecology & Reproductive Medicine 2015;25:243–8.
- Ganchimeg T, Ota E, Morisaki N, et al. Pregnancy and childbirth outcomes among adolescent mothers: a World Health Organization multicountry study. BJOG 2014;121 Suppl 1:40–8.
- Gilbert W, Jandial D, Field N, et al. Birth outcomes in teenage pregnancies. J Matern Fetal Neonatal Med 2004;16:265–70.
- Tyrberg RB, Blomberg M, Kjølhede P. Deliveries among teenage women - with emphasis on incidence and mode of delivery: a Swedish national survey from 1973 to 2010. *BMC Pregnancy Childbirth* 2013;13:1.
- Mohsin M, Bauman AE, Jalaludin B. The influence of antenatal and maternal factors on stillbirths and neonatal deaths in New South Wales, Australia. *J Biosoc Sci* 2006;38:643–57.
- Blomberg M, Birch Tyrberg R, Kjølhede P. Impact of maternal age on obstetric and neonatal outcome with emphasis on primiparous adolescents and older women: A Swedish Medical Birth Register Study. *BMJ Open* 2014;4:e005840.
- Gibbs CM, Wendt A, Peters S, et al. The impact of early age at first childbirth on maternal and infant health. *Paediatr Perinat Epidemiol* 2012;26 Suppl 1:259–84.
- Kenny LC, Lavender T, McNamee R, et al. Advanced maternal age and adverse pregnancy outcome: evidence from a large contemporary cohort,. *PLoS One* 2013;8:e56583.
- Pollack H, Lantz PM, Frohna JG. Maternal smoking and adverse birth outcomes among singletons and twins. *Am J Public Health* 2000;90:395.
- Jaddoe VW, Bakker R, Hofman A, et al. Moderate alcohol consumption during pregnancy and the risk of low birth weight and preterm birth. The generation R study. *Ann Epidemiol* 2007;17:834–40.
- Blumenshine P, Egerter S, Barclay CJ, et al. Socioeconomic disparities in adverse birth outcomes: a systematic review. Am J Prev Med 2010;39:263–72.
- 12. East PL, Felice ME. Adolescent pregnancy and parenting: findings from a racially diverse sample. *Psychology Press* 2014.
- Wright J, Small N, Raynor P, et al. Cohort profile: the Born in Bradford multi-ethnic family cohort study. Int J Epidemiol 2013;42:978–91.
- 14. West J, Lawlor DA, Fairley L, et al. Differences in socioeconomic position, lifestyle and health-related pregnancy characteristics between Pakistani and White British women in the Born in Bradford prospective cohort study: the influence of the woman's, her partner's and their parents' place of birth. *BMJ Open* 2014;4:e004805.
- Fairley L, Petherick ES, Howe LD, et al. Describing differences in weight and length growth trajectories between white and Pakistani infants in the UK: analysis of the Born in Bradford birth cohort study using multilevel linear spline models. *Arch Dis Child* 2013;98:274–9. archdischild-2012.
- Gardosi J, Chang A, Kalyan B, et al. Customised antenatal growth charts. *The Lancet* 1992;339:283–7.
- Office for National Statistics, English indices of deprivation 2015. https://www.gov.uk/government/statistics/english-indices-ofdeprivation-2015(accessed 17 Nov 2017).
- Saugstad OD, Aune D. Optimal oxygenation of extremely low birth weight infants: a meta-analysis and systematic review of the oxygen saturation target studies. *Neonatology* 2014;105:55–63.
- Luttikhuizen dos Santos ES, de Kieviet JF, Königs M, et al. Predictive value of the Bayley scales of infant development on development of very preterm/very low birth weight children: a meta-analysis. *Early Hum Dev* 2013;89:487–96.
- Goldenberg RL, Culhane JF, Iams JD, et al. Epidemiology and causes of preterm birth. *The Lancet* 2008;371:75–84.
- Valsamakis G, Kanaka-Gantenbein C, Malamitsi-Puchner A, et al. Causes of intrauterine growth restriction and the postnatal development of the metabolic syndrome. Ann N Y Acad Sci 2006;1092:138–47.

Open Access

- Horta BL, Victora CG, Menezes AM, et al. Low birthweight, preterm births and intrauterine growth retardation in relation to maternal smoking. *Paediatr Perinat Epidemiol* 1997;11:140–51.
- Nordentoft M, Lou HC, Hansen D, et al. Intrauterine growth retardation and premature delivery: the influence of maternal smoking and psychosocial factors. Am J Public Health 1996;86:347–54.
- Tommy's. Premature Birth Statistics, [online]. https://www.tommys. org/our-organisation/why-we-exist/premature-birth-statistics (accessed 17 Nov 2017).
- 25. van Ham MA, van Dongen PW, Mulder J. Maternal consequences of caesarean section. A retrospective study of intra-operative and postoperative maternal complications of caesarean section during a 10-year period. *Eur J Obstet Gynecol Reprod Biol* 1997;74:1–6.
- Lydon-Rochelle M, Holt VL, Martin DP, et al. Association between method of delivery and maternal rehospitalization. JAMA 2000;283:2411–6.
- Jolly M, Sebire N, Harris J, et al. The risks associated with pregnancy in women aged 35 years or older. Hum Reprod 2000;15:2433–7.
- Jacobsson B, Ladfors L, Milsom I. Advanced maternal age and adverse perinatal outcome. *Obstet Gynecol* 2004;104:727–33.

- Freinkel N, Metzger BE, Phelps RL, et al. Gestational diabetes mellitus: heterogeneity of maternal age, weight, insulin secretion, HLA antigens, and islet cell antibodies and the impact of maternal metabolism on pancreatic B-cell and somatic development in the offspring. *Diabetes* 1985;34(Suppl 2):1–7.
- Khashan AS, Baker PN, Kenny LC. Preterm birth and reduced birthweight in first and second teenage pregnancies: a register-based cohort study. *BMC Pregnancy Childbirth* 2010;10:36.
- de Vienne CM, Creveuil C, Dreyfus M. Does young maternal age increase the risk of adverse obstetric, fetal and neonatal outcomes: a cohort study. *Eur J Obstet Gynecol Reprod Biol* 2009;147:151–6.
- Public Health England, Teenage Parent Outcomes Modelling Tool. http://www.chimat.org.uk/teenconceptions/chimattools(accessed 12 Dec 2016).
- United Nations, Sustainable Development Goals.http://www. undp.org/content/undp/en/home/sustainable-development-goals. html(accessed 12 Dec 2016).
- 34. Child EW. Global strategy for women's, children's and adolescents' health. New York, NY: Every Woman Every Child, 2015.
- 35. World Health Organisation. Adolescent pregnancy fact sheet. http:// www.who.int/mediacentre/factsheets/fs364/en/

5.4 Summary and implications for thesis

This study provides a robust comparative analysis of how rates of important health outcomes differ between adolescent women and a reference group aged 20-34 years. The study has identified associations between young maternal age and extremely low birthweight and very and extremely preterm delivery which have not previously been reported in UK based studies. This work also concurs with and contributes to the body of evidence showing that adolescent women are less likely to experience gestational diabetes or have assisted deliveries. This study met the element of the research objective it was designed to address by examining differences in birth outcomes between adolescent and adult mothers in a large cohort data set; this objective also aimed to examine differences in dietary patterns between the two age groups which is presented in the following chapter.

Chapter 6 - Dietary patterns

6.1 Introduction

This chapter briefly summarises the rationale for the second study in this programme of research assessing the relationship between maternal age and dietary patterns in the Born in Bradford cohort, followed by a copy of the full published research paper in its original form. The chapter also includes a summary of key points, measurable impact of the work and discussion of its contribution to the thesis.

The systematic literature review presented in chapter three showed that there were some areas of concern regarding adolescent women's nutrient intake and nutritional biomarker status during pregnancy. This review also identified that there were gaps in the existing literature regarding the characteristics of the young women at highest risk of nutritional deficiencies and the extent to which the intake of older pregnant women differed to that of adolescents. This study therefore aims to begin to address these identified gaps in the literature and to contribute to addressing the second research objective as set out in chapter one to "Examine differences in dietary intake and birth outcomes between adolescent and adult mothers in a large cohort data set".

6.2 Publication and Impact

The manuscript describing this study was accepted for publication in PLoS One in December 2018. PLoS One is a multidisciplinary open access journal with a focus on scientific rigour regardless of subject area. The journal has an impact factor of 2.766 and an international readership. Since publication online the full text of the article has been accessed 109 times (correct as of 20.01.19).

6.3 Published paper: Article C

The published paper entitled " *Differences in dietary pattern by maternal age in the Born in Bradford cohort: A comparative analysis*" is reproduced here in the format in which it was published online.



Citation: Marvin-Dowle K, Kilner K, Burley V, Soltani H (2018) Differences in dietary pattern by maternal age in the Born in Bradford cohort: A comparative analysis. PLoS ONE 13(12): e0208879. https://doi.org/10.1371/journal. pone.0208879

Editor: Matteo Rota, Universita degli Studi di Milano, ITALY

Received: November 20, 2017

Accepted: November 27, 2018

Published: December 13, 2018

Copyright: © 2018 Marvin-Dowle et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data on which this study is based have not been uploaded because they belong to a third party. Data requests can be made directly to Born in Bradford by completing an expression of interest form available from https://borninbradford.nhs.uk/research/howto-access-data/ and submitting it to the BiB Programme Director (rosie.mceachan@bthft.nhs. uk).

Funding: This study has been carried out as part of a White Rose University Consortium PhD project

RESEARCH ARTICLE

Differences in dietary pattern by maternal age in the Born in Bradford cohort: A comparative analysis

Katie Marvin-Dowle^{1*}, Karen Kilner¹, Victoria Burley², Hora Soltani¹

1 Centre for Health and Social Care Research, Sheffield Hallam University, Sheffield, United Kingdom,

2 School of Food Sciences and Nutrition, University of Leeds, Leeds, United Kingdom

So These authors contributed equally to this work.

* k.marvin-dowle@shu.ac.uk

Abstract

Objective

Explore associations between dietary patterns and maternal age

Design

Population based cohort study

Setting

Maternity department of a large hospital in northern England

Sample

Women delivering a singleton at Bradford Royal Infirmary between March 2007 and December 2010 (N = 5,083).

Methods

Survey data including maternal dietary patterns derived from food frequency questionnaire data using principal component analysis (PCA) were compared by maternal age using one-way ANOVA and chi-squared as appropriate.

Main outcome measures

Dietary pattern PCA scores, supplement use, familiarity and compliance with 5-a-day fruit and vegetable recommendations, consumption of cola, maternal BMI.

Results

Three distinct dietary patterns were derived from the data; snack and processed foods, meat and fish and grains and starches. Mean PCA score for snack and processed foods was higher among women aged ≤ 19 (0.6, Cl 0.4 to 0.8) than women aged 20–34 (-0.02, Cl -0.1 to 0.01) and those aged $35 \geq$ (-0.3, Cl -0.4 to -0.2). Women aged $35 \geq$ had a significantly



supported by Sheffield Hallam University and the National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care for Yorkshire and Humber. Therefore the authors received no specific funding for this work. The Born in Bradford study presents independent research commissioned by the National Institute for Health Research Collaboration for Applied Health Research and Care (NIHR CLAHRC YH) and the Programme Grants for Applied Research funding scheme (RP-PG-0407-10044).

Competing interests: The authors have declared that no competing interests exist.

higher mean PCA score for the grains and starches dietary pattern (0.1, Cl 0.03 to 0.3) compared to both the 20–34 years (-0.01, Cl -0.05 to 0.02) and the \leq 19 (-0.04, Cl -0.2 to 0.1) groups. No differences were observed between groups in mean PCA scores for the meat and fish dietary pattern. Adolescent women also had higher intakes of sugar sweetened cola (0.9 cups per day, Cl 0.7 to 1.1) and reported lower levels of fruit and vegetable and supplement intake. Women aged 35 \geq had a higher mean BMI (28.0, Cl 27.5 to 28.4) and higher prevalence of overweight (36.8%) and obesity (29.6%, p<0.001).

Conclusions

Significant differences were observed between age groups both in terms of diet quality and BMI. Interventions targeted by age group may be advantageous in improving maternal nutrition and contribute to healthy pregnancies.

Introduction

Nutrition during pregnancy is a well-established modifiable factor which has the potential to impact upon the health and well-being of both mother and child[1]. Study of nutritional intake and nutrient sufficiency in populations is difficult owing to the fact that individuals will have differing nutritional needs due to differences in environmental and genetic factors such as energy expenditure, body composition and metabolism. This is particularly true when investigating the relationship between nutrition and maternal and neonatal health as this is complicated by other biological, demographic and social factors, which vary substantially between different populations.

One such factor is the age of the mother during pregnancy. Adolescence is a time of substantial physiological change during which the nutritional needs of adolescent women are likely to differ from those of older women. Previous work examining the impact of maternal growth on outcomes in adolescent pregnancies suggested that nutrient partition-ing between mother and child in growing adolescents negatively affected foetal growth and prematurity[2]. A further study[3] also found mean birthweight in babies born to adolescent women to be lower than those born to adult women; however this was not related to maternal growth in the adolescent participants. This suggests that the relationship between maternal age, maternal growth and foetal growth is complex and warrants further investigation, given that low birthweights are associated with short and long term adverse health outcomes[4].

A recent systematic review of nutritional status of pregnant adolescents in developed countries found that intakes of energy, fibre and a number of micronutrients were below recommended levels in this population[5]. The study also found that there was some cause for concern with regard to biological markers of iron and selenium status.

Similar results were however reported in two reviews which were not confined to adolescent populations[6–7]. This suggests that while there is evidence that diet quality may vary by maternal age, more work to examine the nature of these differences would be advantageous.

These previous works suggest that there may be nutritional issues that may impact on the health of both mother and child which vary by maternal age. This study therefore aims to assess differences in dietary pattern by maternal age in the Born in Bradford cohort.

Methods

Born in Bradford is a largely bi-ethnic cohort of approximately 13,500 children with high levels of socio-economic deprivation. Pregnant women who were booked to deliver at Bradford Royal Infirmary were invited to join the cohort between March 2007 and December 2010. A detailed profile of the cohort has been published elsewhere[8]. This study uses data collected in phase 2 of the Born in Bradford cohort study[8] to examine dietary patterns of women delivering singletons who took part in the study. The Born in Bradford study is a prospective cohort study for which participants were recruited during pregnancy. All women booked for delivery at Bradford Royal Infirmary are offered an oral glucose tolerance test (OGTT) at 26-28 weeks gestation. Women were invited to participate in the Born in Bradford study when attending this appointment. Informed consent was obtained from all participants women were asked to complete a baseline questionnaire. All participants including those aged under 18 were considered competent to consent on their own behalf. Documents including participant information, consent forms and questionnaires are available on the Born in Bradford website [9] Recruitment took place between March 2007 and December 2010 and over 80% of women eligible in this period agreed to take part. A sub-set of the whole cohort completed a food frequency questionnaire (FFQ) as part of the baseline data collection; therefore data relating to 5,083 pregnancies was available for this analysis. Ethical approval for the study was granted by Bradford Research Ethics Committee (ref no. 07/H1302/112).

Variables in the analysis

The following demographic characteristics of the sample reported in the baseline questionnaire were explored; ethnicity, marital status, parity, education, smoking during pregnancy, alcohol and illegal drug use during pregnancy and index of multiple deprivation (IMD) score (IMD is the official measure of relative deprivation for small areas in England and combines information from seven domains of deprivation; income, employment, education, health, crime, housing and environment)[10]. Body mass index and number of weeks gestation at the booking appointment (usually before 12 weeks of pregnancy) were obtained from medical records. Variables related to nutrition examined were: use of any vitamins or iron supplements in the last 4 weeks, use of Pregnacare multivitamins at least twice per week, familiarity with 5-aday fruit and vegetable recommendations, consumption of 5 fruit and vegetable portions per day and sources of advice about healthy eating, number of cups of regular (sugar sweetened) and diet (artificially sweetened) cola consumed per day. Dietary patterns were derived from FFQ data using principal component analysis as described below.

Statistical analysis

Due to the focus of this study being on detecting differences between age groups the study population was categorised into three groups for analysis according to maternal age; \leq 19, 20–34 and 35 \geq years, with women aged 20–34 being considered as the reference group. Statistical analysis was undertaken using SPSS 24.

Principal component analysis

In order to evaluate nutrition in this population a series of statistical analyses were conducted using responses to the FFQ and applying principal component analysis (PCA) to identify distinct patterns of food types consumed. The FFQ was completed at 26–28 weeks gestation and refers to food intake during the preceding 4 weeks. There were a total of 37 items included in the FFQ, details of which are given with the results of the PCA. Since the original objective of

the FFQ was to characterise the intakes of key food items only, it was not appropriate to derive an estimate of total energy intake. Responses to the food frequency questionnaire were given on either an eight (1 = Rarely or never, 8 = 5+ times per day) or five (1 = Rarely or never, 5 = 7 + times per week) point scale, these responses were therefore transformed to give an estimation of frequency per day in order to standardise questions asked in different formats.

Individual items on the FFQ were then grouped by type and total frequency of consumption of different food groups estimated. The grouped nutritional variables were then used as the basis for a PCA in order to examine the dietary patterns which exist within the data set. PCA was selected as it allows for a larger number of variables to be reduced and underlying factors exposed.

Analysis of differences between age groups

Differences between maternal age groups in both the demographic and nutrition related variables were explored using Chi-Square for categorical data and one-way ANOVA for continuous data; categorical variables are presented as percentages and continuous variables as means and 95% confidence intervals. Logistic regression analyses were used to compare the rate of each of the categorical nutrition related variables by age group and differences between groups estimated using odds ratios. Multivariate logistic regression models were then used to adjust these comparisons for confounding variables.

Crude and adjusted odds ratios (OR and aOR) are therefore presented with 95% confidence intervals. Index of multiple deprivation (IMD) score and maternal ethnicity (white British, Pakistani or any other ethnicity) were included as covariates in the adjusted analysis, due to the association of these variables with dietary pattern identified in the literature[11–12]. Where dependent variables were dichotomous, binary logistic regression models were produced. There were also two variables with more than two categorical outcomes for which multinomial logistic regression models were produced. In order to examine the continuous variables regarding consumption of cola linear regression models were produced and dummy variables created to enable the inclusion of categorical independent variables in the model.

In the multivariate regression models for this study there is no clear logical or theoretical basis for assuming any variable to be prior to any other, either in terms of its relevance to the research goal of explaining phenomena, or in terms of a hypothetical causal structure of the data. For this reason a simultaneous method of including independent variables in the multivariate regression models was considered to be most appropriate, as opposed to a stepwise method.

Due to the assumptions required to produce statistical models not being met for the dietary pattern scores (the distribution of the dietary pattern scores is significantly skewed, even following attempts to transform the data) it was not possible to adjust the analysis of these variables for potentially confounding variables.

Results

Characteristics of the sample

Data were available for 5,083 pregnancies for this analysis; demographic characteristics of the participants included in the study are shown in Table 1. Most participants in the cohort were aged 20–34 (84%) with 9.5% aged 35 or over and 6.5% aged 19 or under. The sample has a diverse ethnic mix consisting of 46.8% women describing themselves as of Pakistani origin, 38.4% white British and 14.8% of other ethnicities; this distribution of ethnic groups was roughly consistent across the age groups with the exception of the adolescent group which was significantly different. Among women aged 19 and under only 17.6% were of Pakistani

Table 1. Demographic characteristics.

		≤19		20-34		35≥		Total	
	N	%	N	%	N	%	N	%	p =
Whole Cohort	330	6.5	4269	84.0	484	9.5	5083	100.0	
Ethnicity									
Pakistani	58	17.6	2052	50.0	266	41.9	2376	46.8	< 0.001
White British	229	69.4	1454	35.4	264	41.6	1947	38.4	1
Any other ethnicity	43	13.0	602	14.7	105	16.5	750	14.8	1
Marital Status									
Married	48	14.5	2869	69.9	493	77.9	3410	67.3	< 0.001
Not married—living with partner	84	25.5	704	17.1	85	13.4	873	17.2]
Single	198	60.0	533	13.0	55	8.7	786	15.5]
Parity									
0	295	90.2	1571	39.0	112	17.9	1978	39.7	< 0.001
1	29	8.9	1296	32.2	164	26.2	1489	29.9	
2	3	0.9	723	18.0	121	19.4	847	17.0	
3 or more	0	0.0	435	10.8	228	36.5	663	13.3	
Highest level of education									
Less than 5 GCSEs* grade A-C or equivalent	129	39.1	825	20.0	184	29.0	1138	22.4	< 0.001
5 or more GCSEs grade A-C or equivalent	152	46.1	1307	31.7	132	20.8	1591	31.3]
A-levels** or higher	21	6.4	1696	41.2	248	39.1	1965	38.7	
Other/unknown	28	8.5	290	7.0	71	11.2	389	7.7	
Smoked during pregnancy									
Yes	159	48.2	644	15.7	62	9.8	865	17.1	< 0.001
No	171	51.8	3461	84.3	573	90.2	4205	90.2	
Drunk alcohol in the first three months of pregnancy									
Yes	102	58.3	574	47.8	119	52.2	795	49.6	0.59
No	73	41.7	621	51.7	109	47.8	803	50.1	
Drunk alcohol since the fourth month of pregnancy									
Yes	43	24.4	459	38.4	105	47.1	607	38.1	< 0.001
No	133	75.6	733	61.4	118	52.9	984	61.8	
Used recreational drugs during pregnancy									
Yes	18	5.5	42	1.0	1	0.2	61	1.2	< 0.001
No	312	94.5	4062	99.0	632	99.8	5006	98.8	
BMI Category									
Underweight (Below 18.5)	23	7.1	183	4.6	5	0.8	211	4.3	< 0.001
Healthy weight (18.5–24.9)	200	61.7	1837	46.4	200	32.8	2237	45.7	
Overweight (25–29.9)	63	19.4	1142	28.8	224	36.8	1429	29.2	
Obese (30 or higher)	38	11.7	798	20.2	180	29.6	1016	20.8	
	N	Mean (95% CI)	N	Mean (95% CI)	N	Mean (95% CI)	N	Mean (95% CI)	p =
BMI at booking appointment	324	23.7 (23.2 to 24.3)	3972	25.9 (25.7 to 26.1)	609	28.0 (27.5 to 28.4)	4905	26.0 (25.9 to 26.2)	< 0.001
IMD Score	330	45.1 (43.1 to 47.0)	4117	43.0 (42.5 to 43.6)	635	37.3 (35.8 to 38.7)	5082	42.4 (42.0 to 42.9)	< 0.001
Number of weeks gestation at booking appointment	311	13.9 (13.5 to 14.4)	3905	12.7 (12.6 to 12.8)	602	13.1 (12.8 to 13.3)	4818	12.8 (12.7 to 12.9)	< 0.001

*General Certificate of Secondary Education—subject specific examinations taken at the end of compulsory education at age 16 in England ** Advanced Level—non-compulsory subject specific examinations taken at age 18 in England

https://doi.org/10.1371/journal.pone.0208879.t001

ethnicity and 69.4% were white British, the proportion belonging to other ethnic groups was similar to other age groups. There were other significant variations in the characteristics of the sample by maternal age. Women in the adolescent age group were more likely to be unmarried or living with a partner, to be expecting their first child and to have completed lower levels of education compared to older women. Women in the adolescent age groups were also more likely to have smoked or used recreational drugs during pregnancy, however there was no difference in reported alcohol use in the first trimester between age groups, and adolescents were less likely to have used alcohol since the fourth month of their pregnancy compared to older women. Women in the oldest age category were most likely to be overweight or obese while adolescent women were found to have higher prevalence of underweight. IMD score decreased as maternal age increased suggesting adolescent women lived in areas of higher deprivation. Adolescent women also booked with a midwife for antenatal care later than older women.

Nutrition related variables

Analysis of the nutrition related variables (Table 2) showed that young women aged 19 and under were less likely to have used any vitamins or iron supplements in the 4 weeks preceding completing the questionnaire compared to other age groups. Adolescents were also the least likely group to have used Pregnacare multi-vitamins at least twice per week during their pregnancy. While there was no differences between age groups in awareness of the recommendations to eat at least five portions of fruit and vegetables a day the adolescent group were less likely than older women to achieve this recommendation. The adolescent group consumed significantly more sugar sweetened cola compared to older women; there were no statistically significant differences in consumption of artificially sweetened cola between age groups.

Adjustment for ethnicity and IMD score in the regression analysis produced similar results as shown in Table 3.

Results of the principal component analysis

Several combinations of food groupings were explored to see which gave the most sensitive output in the PCA. These included loading all of the food items in separately, separating whole grains from other grains and starches, analysing all proteins as one group and separating proteins by type (red meat, poultry and fish). The groupings outlined in <u>Table 4</u> explained the highest amount of variance; therefore these were the input variables which were selected. Loading all of the foods individually returned very low levels of correlation between the variables, meaning that a very large number of factors (11) were returned. This suggested that grouping the variables to form new variables would improve the correlation of intake between food types and therefore make the PCA more meaningful.

When the analysis was completed without stipulating the number of factors to be retained two factors were retained which explained 43.8% of the total variance. On closer examination of the output, including examination of the inflection point of the scree plot, it was considered that a third and fourth factor with Eigen values of 0.971 and 0.939 respectively should be considered for inclusion. The analysis was therefore run with both 3 and 4 factors stipulated and the results compared with the initial analysis. The inclusion of a third factor presented a distinct dietary pattern which added to the logical understanding and interpretation of the data. The inclusion of a fourth factor did not however add to the understanding of the data. The decision was therefore made to retain 3 factors from the analysis, the inclusion of these 3 factors would then account for 55.9% of the total variance. Details of the retained factors and factor loadings are given in Table 5, factor loadings were considered to be very good where they were 0.6 or above and moderate where they were 0.45–0.59[13]. The PCA therefore identified

Table 2. Maternal age and nutrition related variables.

		<u>≤</u> 19	20-34		35≥		Total		
	N	%	N	%	N	%	N	%	p =
Used any vitamins or iron supplements in the last 4 weeks									
Yes	86	26.1	1606	39.1	318	50.1	2010	39.6	< 0.001
No	244	73.9	2500	60.9	317	49.9	3061	60.4	
Used Pregnacare multivitamins at least twice per week									
Yes	21	6.4	553	13.4	114	18.0	688	13.5	< 0.001
No	309	93.6	3565	86.6	521	82.0	4395	86.5	
Are you familiar with the 5 a day recommendations for fruit and vegetables?									
Yes	303	95.6	3279	95.5	500	97.1	4082	95.7	0.238
No	14	4.4	156	4.5	15	2.9	185	4.3	
Do you consume 5 portions of fruit and vegetables per day?									
Always	28	8.8	586	17.0	126	24.5	740	17.3	< 0.001
Sometimes	259	81.7	2616	76.0	365	70.9	3240	75.8	
Never	30	9.5	238	6.9	24	4.7	292	6.8	
Where does most of your advice about healthy eating during pregnancy come from?									
Family members	124	37.6	845	20.5	83	13.1	1052	20.7	< 0.001
Midwives/Health visitors	119	36.1	1175	28.5	155	24.4	1449	28.5	
Magazines/Newspapers	31	9.4	490	11.9	99	15.6	620	12.2	
GPs/Doctors	16	4.8	126	3.1	13	2.0	155	3.0	
Books	11	3.3	395	9.6	85	13.4	491	9.7	
Other	11	3.3	310	7.5	66	10.4	387	7.6	
Friends	5	1.5	96	2.3	13	2.0	114	2.2	
	N	Mean (95% CI)	N	Mean (95% CI)	N	Mean (95% CI)	N	Mean (95% CI)	p =
Cups of regular (sugar sweetened) cola per day	330	0.9 (0.7 to 1.1)	4269	0.4 (0.4 to 0.4)	484	0.2 (0.1 to 0.2)	5083	0.4 (0.4 to 0.4)	< 0.001
Cups of diet (artificially sweetened) cola per day	330	0.2 (0.1 to 0.2)	4269	0.1 (0.1 to 0.1)	484	0.1 (0.1 to 0.2)	5083	0.1 (0.1 to 0.1)	0.674

https://doi.org/10.1371/journal.pone.0208879.t002

three independent variables whose relationship to other variables in the data set could be considered.

Association between dietary pattern and maternal age

The analysis showed that there was an association between both the snacks and processed foods pattern and the grains and starches pattern and maternal age group (Table 6). The mean PCA score for snack and processed foods was highest in the adolescent group. The opposite was observed in the case of the grains and starches dietary pattern where women aged 35 and over had the highest mean PCA score. There were no statistically significant differences between groups relating to the meat and fish dietary pattern. The results of these analyses are shown in Table 6.

Discussion

The results show an association between two of the dietary patterns in this analysis and maternal age. Strongest associations were seen in the snack and processed foods pattern where

Table 3. Dietary behaviour by maternal age.

	Crude OR (95% CI)	aOR (95% CI)*	Crude OR (95% CI)	aOR (95% CI)*
	\leq 19 Years ($n = 641)^{\dagger}$	35≥ Years (n	= 1427) [†]
Used any vitamins or iron supplements in the last 4 weeks				
Yes	0.54 (0.42 to 0.70)	0.71 (0.55 to 0.92)	1.53 (1.27 to 1.85)	1.56 (1.29 to 1.89)
No	Ref	Ref	Ref	Ref
Used Pregnacare multivitamins at least twice per week				
Yes	0.44 (0.28 to 0.68)	0.44 (0.28 to 0.70)	1.49 (1.16 to 1.90)	1.33 (1.04 to 1.71)
No	Ref	Ref	Ref	Ref
Familiar with the 5 a day recommendations for fruit and vegetables				
Yes	1.01 (0.58 to 1.77)	0.62 (0.35 to 1.12)	1.46 (0.80 to 2.65)	1.26 (0.67 to 2.24)
No	Ref	Ref	Ref	Ref
Consume 5 portions of fruit and vegetables per day				
Always	0.37 (0.22 to 0.63)	0.35 (0.20 to 0.61)	1.96 (1.18 to 3.23)	1.68 (1.01 to 2.80)
Sometimes	0.77 (0.52 to 1.15)	0.68 (0.45 to 1.03)	1.20 (1.18 to 3.23)	1.11 (0.69 to 1.78)
Never	Ref	Ref	Ref	Ref
Where does most of your advice about healthy eating during pregnancy come from?				
Family members	4.26 (2.27 to 8.00)	5.65 (3.14 to 10.18)	0.49 (0.33 to 0.72)	0.56 (0.36 to 0.88)
Midwives/Health visitors	2.94 (1.56 to 5.51)	2.77 (1.54 to 4.99)	0.69 (0.49 to 0.99)	0.78 (0.53 to 1.16)
Magazines/Newspapers	1.76 (0.87 to 3.55)	1.70 (0.86 to 3.36)	0.84 (0.57 to 1.25)	0.83 (0.53 to 1.29)
GPs/Doctors	3.66 (1.65 to 8.10)	3.92 (1.86 to 8.25)	0.46 (0.22 to 0.97)	0.62 (0.31 to 1.24)
Books	0.79 (0.34 to 1.84)	1.09 (0.51 to 2.35)	1.05 (0.70 to 1.56)	1.03 (0.66 to 1.61)
Friends	1.49 (0.50 to 4.38)	1.61 (0.64 to 4.05)	0.60 (0.29 to 1.27)	0.71 (0.36 to 1.42)
Other	Ref	Ref	Ref	Ref
	Unadjusted β (95%CI)	Adjusted β (95%CI)*	Unadjusted β (95%CI)	Adjusted β (95%CI)*
Cups of regular (sugar sweetened) cola per day	0.49 (0.36 to 0.62)	0.46 (0.33 to 0.59)	-0.22 (-0.33 to -0.12)	-0.20 (-0.31 to -0.09)
Cups of diet (artificially sweetened) cola per day	0.03 (-0.04 to 0.09)	-0.02 (-0.09 to 0.05)	0.01 (-0.05 to 0.07)	0.94 (-0.06 to 0.05)

*Adjusted for index of multiple deprivation score and maternal ethnicity †comparison to reference category women aged 20–34 years

https://doi.org/10.1371/journal.pone.0208879.t003

women aged \leq 19 were found to have the highest mean PCA score reflecting a higher intake of foods that are typically viewed as snack items (including crisps, chocolate, biscuits and cake) and moderate intake of processed meat. Adolescent women also reported consuming higher levels of sugar sweetened cola compared to older women; no differences were observed in levels of consumption of artificially sweetened cola.

There were no differences by age group in women's awareness of recommendations to eat five portions of fruit and vegetables per day; however adolescent women were less likely to say

Potatoes	Grains and Starches	Whole meat and fish	Meat in Sauce	Processed meat and fish	Cured Pork Products	Savoury Snacks	Sweet Snacks
Chips	Fibre or bran-rich wheat breakfast cereal	Beef	Chicken or turkey in sauce	Beef burgers	Bacon	Potato Crisps	Cakes, buns, gateaux, doughnuts, muffins
Roast or fried potatoes	Oat cereals	Lamb	Beef, lamb or goat in sauce	Kebabs	Ham	Other salted savoury snacks	Sweet pastries
	Crispbread	Pork	Pork in sauce	Meat pies and pastries	Cured sausage		Chocolate bars and chocolate coated biscuits
	Other breakfast cereals	Chicken or turkey	Gravy made with pan or meat juices (not instant)	Sausages			Sweet biscuits
	Pasta or noodles	White fish		Hotdogs			
	Savouries like Yorkshire pudding, pakoras etc.	Tinned tuna		Chicken or turkey nuggets			
		Fresh or tinned oily fish		White fish in batter or breadcrumbs			
		Smoked fish					
		Salted or dried fish					

Table 4. Food groupings for principal components analysis.

https://doi.org/10.1371/journal.pone.0208879.t004

they achieve this recommendation. Adolescent women in the sample also reported the lowest use of nutritional supplements.

Adolescent women had a higher prevalence of underweight (BMI <18.5) and older women a higher prevalence of overweight or obesity (BMI >25).

These results are largely consistent with findings from the latest wave of the National Diet and Nutrition Survey (NDNS)[14]. Results from the 2012/13–2013/14 surveys found significantly fewer adolescent girls reported meeting 5-a-day recommendations and had higher intakes of sugar sweetened beverages compared to adult women. The prevalence of overweight and obesity in the NDNS was also in line with this cohort (58% among adult women and 38% among girls aged 11–19). Whilst the observation that a population with an overall poorer quality diet would have lower levels of overweight and obesity is counterintuitive, however the impact of poor diet on BMI status is cumulative meaning that the effects of poor diet on the adolescent population will be evident in time if dietary changes are not made[15]. These similarities suggest that the findings of the current study are likely to be reflective of the UK

	Component					
	l Snack and processed foods	2 Meat and fish	3 Grains and starches			
Savoury Snacks	0.756					
Potatoes	0.691					
Sweet Snacks	0.666					
Processed meat and fish	0.510	0.480				
Whole meat and fish		0.794				
Meat in Sauce		0.729				
Cured pork products		0.470				
Grains and Starches			0.888			

Table 5. Dietary patterns identified by principal component analysis.

https://doi.org/10.1371/journal.pone.0208879.t005

		≤ 19		20-34		$35\geq$	Total		
	N	Mean PCA Score (95% CI)	N	Mean PCA Score (95% CI)	N	Mean PCA Score (95% CI)	N	Mean PCA Score (95% CI)	p =
Snack and Processed Foods	277	0.6 (0.4 to 0.8)	2949	-0.02 (-0.1 to 0.01)	320	-0.3 (-0.4 to-0.2)	3546	0.0 (-0.03 to 0.03)	< 0.001
Meat and Fish	277	0.04 (-0.1 to 0.2)	2949	-0.01 (-0.04 to 0.03)	320	0.04 (-0.05 to 0.1)	3546	0.0 (-0.03 to 0.03)	0.55
Grains and Starches	277	-0.04 (-0.2 to 0.1)	2949	-0.01 (-0.05 to 0.02)	320	0.1 (0.03 to 0.3)	3546	0.0 (-0.03 to 0.03)	0.03

Table 6. Differences in dietary pattern scores[†] by maternal age.

† Dietary pattern scores derived from Principal Component Analysis indicating the extent to which participants adhere to the dietary pattern

https://doi.org/10.1371/journal.pone.0208879.t006

population. Dietary patterns during pregnancy have been previously examined using similar methods. One UK study[16] found an association between dietary pattern and maternal age with the 'health conscious' pattern being associated with increasing age and the opposite being true of the 'processed' pattern identified in the study. A further study conducted in New Zea-land[17] similarly found a positive association between increasing maternal age and 'health conscious' and 'fusion/protein' dietary patterns and the opposite association with 'junk' and 'traditional/white bread' patterns.

The findings of these studies are consistent with those of the present study however neither specifically addressed adolescent women or older women as distinct groups. As discussed in the introduction, the nutritional needs of adolescent women in particular may differ from those of older women due to the continued growth of the mother meaning that understanding the nutritional intake of young women during pregnancy is important. This study helps us to start to understand the dietary patterns associated with this group, however further work based on more robust dietary surveys is needed in order to improve understanding of this complex topic. A particular strength of this work is that it utilises well-established, ethnically diverse, UK based cohort data in a way which is unique to this study. This cohort includes a large number of participants meaning that analysis by maternal age including an adolescent group is both possible and robust.

There were limitations in the dietary data available for this study. The number of items included in the FFQ was limited meaning that there is potential for the dietary patterns identified to be lacking in detail, or that additional patterns which might be present in the population could have been missed. This issue also resulted in the distribution of dietary pattern scores being significantly skewed, meaning that it was not possible to use statistical modelling to predict the extent to which maternal age influences dietary pattern, only that there appears to be an association. There was also no definition of portion size in the collection of data meaning there is likely to be significant variation between participants self-defined portion sizes. The FFQ was completed at 26–28 weeks of pregnancy and related to the preceding 4 weeks, meaning that the dietary patterns identified are only reliably relevant to the second trimester of pregnancy. Women's diet may change over the course of pregnancy[18] meaning that these results may not be generalisable to the first and third trimesters.

This study assesses dietary patterns ascertained by principal component analysis. Dietary patterns have been defined as "foods that are actually consumed in various characteristic combinations" [19]. The study of dietary patterns as opposed to single foods or nutrients has the advantages of allowing for interactions between nutrients to be accounted for and reflecting more accurately that people do not eat single foods or nutrients, they eat meals comprising of many food types. Dietary patterns can be derived either theoretically as in the case of the Diet

Quality Index[20] where foods are ranked based on what current knowledge defines as healthy or less healthy or empirically using statistical methods to reduce collected dietary data into distinct patterns. A systematic review of the use of these methods in nutritional epidemiology has concluded that empirically derived eating patterns may improve our understanding of eating behaviour and therefore provide a stronger evidence base from which to provide dietary advice[21]. For these reasons this method of assessing dietary patterns is considered to be appropriate for this study.

The results suggest that there may be some cause for concern regarding the quality of adolescent pregnant women's diet, despite reporting similar levels of knowledge of recommendations to older women. This suggests that factors other than knowledge may be important in developing behaviour change interventions. There are a number of different theoretical models discussing the process of translating knowledge into action in public health. For example the health belief model[22] is based on the premise that individuals will make an assessment as to whether the benefits of a change in behaviour outweigh the perceived costs. This includes an assessment of both the risks associated with not changing behaviour, their belief in the potential benefits and the perceived barriers to taking action. Applied to this data therefore there are several avenues for further investigation with the aim of encouraging positive dietary changes.

The adolescent women in the sample also reported less use of nutritional supplements compared to the reference group. This suggests that any gaps in the nutritional profile of young women during pregnancy as a result of a sub-optimal diet may not be being filled by supplementary vitamins and minerals and may therefore leave women in this group with nutrient deficiencies.

Consumption of sugar and artificially sweetened cola in this cohort has previously been examined[23] with results showing that high intakes of sugar sweetened cola were associated with higher odds of preterm delivery in the cohort as a whole. This suggests sugar sweetened cola intake in adolescent women may be a cause for concern.

Energy intake was not available for analysis in this dataset meaning that maternal BMI at booking appointment is a useful indicator. Individual differences in total energy intake are largely determined by a combination of body size, physical activity and metabolic efficiency. Measurements of energy intake and physical activity in epidemiology are often crude and suffer from under or over reporting, and metabolic efficiency is essentially impossible to measure in this setting. For these reasons height and weight measurements are often considered a suitable alternative to measures of energy intake [24]. The results of this study suggest that for some adolescent women their overall energy intake during pregnancy may be insufficient while an association with overweight and obesity in older women suggests an excessive energy intake.

The results of this study are consistent with previous work assessing maternal obesity in the UK[25]. This work found that levels of maternal obesity (measured during the first trimester) increased as maternal age increased. Increases in maternal obesity impact significantly on the health of mothers and babies and increase the need for specialist and high dependency care, therefore having both health and economic implications.

Conclusions

There may be some cause for concern with regard to diet quality in adolescent women and obesity among older women during pregnancy. Further work using more comprehensive methods of dietary data collection would be advantageous in assessing the nature of these relationships.

The impact of dietary pattern on birth outcomes is also an important area for further work, particularly in light of evidence presented previously that there is a higher risk of extremely low birthweight babies and extremely pre-term delivery among adolescent women in this cohort[26].

Interventions targeted to specific age groups to facilitate behaviour change may be useful due to differences identified in the types of nutritional issues affecting pregnant women at different ages. There is a need to further understand the barriers to healthy eating during pregnancy in order to develop successful interventions.

Acknowledgments

Born in Bradford is only possible because of the enthusiasm and commitment of the Children and Parents in BiB. We are grateful to all the participants, practitioners and researchers who have made Born in Bradford happen.

The authors would also like to thank the National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care for Yorkshire and Humber (NIHR CLAHRC YH) for supporting us in conducing this work. Further details about the new NIHR CLAHRC YH can be found at www.clahrc-yh.nihr.ac.uk. The views and opinions expressed are those of the authors, and not necessarily those of the NHS, the NIHR or the Department of Health.

The authors would also like to thank Mr Derek Burns for his advice regarding statistical analysis.

Author Contributions

Conceptualization: Katie Marvin-Dowle, Victoria Burley, Hora Soltani.

Data curation: Katie Marvin-Dowle.

Formal analysis: Katie Marvin-Dowle, Karen Kilner.

Methodology: Katie Marvin-Dowle, Karen Kilner, Victoria Burley, Hora Soltani.

Project administration: Katie Marvin-Dowle.

Supervision: Victoria Burley, Hora Soltani.

Validation: Katie Marvin-Dowle.

Visualization: Katie Marvin-Dowle.

Writing – original draft: Katie Marvin-Dowle.

Writing - review & editing: Victoria Burley, Hora Soltani.

References

- 1. Burke BS, Harding VV, Stuart HC. Nutrition studies during pregnancy: IV. Relation of protein content of mother's diet during pregnancy to birth length, birth weight, and condition of infant at birth. The Journal of Pediatrics. 1943 Nov 1; 23(5):506–15.
- Scholl TO, Hediger ML. Weight gain, nutrition, and pregnancy outcome: findings from the Camden study of teenage and minority gravidas. In Seminars in Perinatology 1995 Jun 30; 19(3):171–81
- Jones RL, Cederberg HM, Wheeler SJ, Poston L, Hutchinson CJ, Seed PT, et al. Relationship between maternal growth, infant birthweight and nutrient partitioning in teenage pregnancies. BJOG: An International Journal of Obstetrics & Gynaecology. 2010 Jan 1; 117(2):200–11.
- 4. Hack M, Klein NK, Taylor HG. Long-term developmental outcomes of low birth weight infants. The future of children. 1995 Apr 1:176–96.

- Marvin-Dowle K, Burley VJ, Soltani H. Nutrient intakes and nutritional biomarkers in pregnant adolescents: a systematic review of studies in developed countries. BMC Pregnancy and Childbirth. 2016 Sep 15; 16(1):268.
- Blumfield ML, Hure AJ, Macdonald-Wicks L, Smith R, Collins CE. A systematic review and meta-analysis of micronutrient intakes during pregnancy in developed countries. Nutrition reviews. 2013 Feb 1; 71 (2):118–32. https://doi.org/10.1111/nure.12003 PMID: 23356639
- Blumfield ML, Hure AJ, Macdonald-Wicks L, Smith R, Collins CE. Systematic review and meta-analysis of energy and macronutrient intakes during pregnancy in developed countries. Nutrition reviews. 2012 Jun 1; 70(6):322–36. https://doi.org/10.1111/j.1753-4887.2012.00481.x PMID: 22646126
- Wright J, Small N, Raynor P, Tuffnell D, Bhopal R, Cameron N, et al. Cohort profile: the Born in Bradford multi-ethnic family cohort study. International journal of epidemiology. 2013 Aug 1; 42(4):978–91. https://doi.org/10.1093/ije/dys112 PMID: 23064411
- Born in Bradford, Documents and Data, https://borninbradford.nhs.uk/research/documents-data/, [accessed 28.03.18]
- Office for National Statistics, English indices of deprivation 2015.https://www.gov.uk/government/ statistics/english-indices-of-deprivation-2015 (accessed 28.03.18)
- Moore L. V., Diez Roux A. V., Nettleton J. A., & Jacobs D. R. Jr (2008). Associations of the local food environment with diet quality—a comparison of assessments based on surveys and geographic information systems: the multi-ethnic study of atherosclerosis. American journal of epidemiology, 167(8), 917–924. https://doi.org/10.1093/aje/kwm394 PMID: 18304960
- Neumark-Sztainer D., Hannan P. J., Story M., Croll J., & Perry C. (2003). Family meal patterns: associations with sociodemographic characteristics and improved dietary intake among adolescents. Journal of the american dietetic association, 103(3), 317–322. https://doi.org/10.1053/jada.2003.50048 PMID: 12616252
- **13.** Comrey AL and Lee HB (1992) A first course in factor analysis (2nd edition). Hillsdale,NJ: Lawrence Erlbaum Associates.
- Public Health England, National Diet and Nutrition Survey. Results from Years 5–6 (combined) of the Rolling Programme (2012/13–2013/14) [Online] available from https://www.gov.uk/government/ statistics/ndns-results-from-years-5-and-6-combined accessed 02.03.17
- 15. Darnton-Hill I, Nishida C, James WP. A life course approach to diet, nutrition and the prevention of chronic diseases. Public health nutrition. 2004 Feb; 7(1a):101–21. PMID: 14972056
- Northstone K, Emmett P, Rogers I. Dietary patterns in pregnancy and associations with socio-demographic and lifestyle factors. European Journal of Clinical Nutrition. 2008 Apr; 62(4):471. <u>https://doi.org/ 10.1038/sj.ejcn.1602741</u> PMID: 17375108
- Wall CR, Gammon CS, Bandara DK, Grant CC, Atatoa Carr PE, Morton S. Dietary patterns in pregnancy in New Zealand—Influence of maternal socio-demographic, health and lifestyle factors. Nutrients. 2016 May 19; 8(5):300.
- Devine CM, Bove CF, Olson CM. Continuity and change in women's weight orientations and lifestyle practices through pregnancy and the postpartum period: the influence of life course trajectories and transitional events. Social Science & Medicine. 2000 Feb 29; 50(4):567–82.
- Schwerin HS, Stanton JL, Smith JL, Riley AM Jr, Brett BE. Food, eating habits, and health: a further examination of the relationship between food eating patterns and nutritional health. Am J Clin Nutr. 1982; 35:1319–1325. https://doi.org/10.1093/ajcn/35.5.1319 PMID: 7081115
- Haines PS, Siega-Riz AM, Popkin BM. The Diet Quality Index revised: a measurement instrument for populations. J Am Diet Assoc. 1999; 99:697–704. <u>https://doi.org/10.1016/S0002-8223(99)00168-6</u> PMID: 10361532
- Newby PK, Tucker KL. Empirically derived eating patterns using factor or cluster analysis: a review. Nutrition reviews. 2004 May 1; 62(5):177–203. PMID: 15212319
- Green EC, Murphy E. Health belief model. The Wiley Blackwell Encyclopedia of Health, Illness, Behavior, and Society. 2014 Feb.
- 23. Petherick ES, Goran MI, Wright J. Relationship between artificially sweetened and sugar-sweetened cola beverage consumption during pregnancy and preterm delivery in a multi-ethnic cohort: analysis of the Born in Bradford cohort study. European journal of clinical nutrition. 2014 Mar 1; 68(3):404–7. https://doi.org/10.1038/ejcn.2013.267 PMID: 24398641
- Willett W. Nutritional epidemiology. Oxford University Press; 2012 Nov 7. https://doi.org/10.1097/EDE. 0b013e31825afb0b PMID: 22833016
- Heslehurst N, Ells LJ, Simpson H, Batterham A, Wilkinson J, Summerbell CD. Trends in maternal obesity incidence rates, demographic predictors, and health inequalities in 36 821 women over a 15-year period. BJOG: An International Journal of Obstetrics & Gynaecology. 2007 Feb 1; 114(2):187–94.

26. Marvin-Dowle K, Kilner K, Burley VJ, et al Impact of adolescent age on maternal and neonatal outcomes in the Born in Bradford cohort, BMJ Open 2018; 8:e016258. https://doi.org/10.1136/bmjopen-2017-016258 PMID: 29549196

6.4 Summary and implications for thesis

The results of this study show significant differences between adolescents and adult women with regard to their diet. Adolescent women had higher intakes of snack and processed foods and sugar sweetened cola and lower intakes of fruit and vegetables compared to older women. This is an important finding as it clearly demonstrates differences by age group which are likely to result in differences in nutritional status which may impact upon the health of the pregnancy.

6.5 Development of qualitative research phase

The final study in this programme of work uses qualitative methods to gain a deeper understanding of the factors which contribute to the observed differences in outcomes and diet in pregnant adolescents. The qualitative phase uses semi-structured interviews following a topic guide, the formation of which being informed by the results of the quantitative analyses presented in chapters 5 and 6. The topic guide (appendix 9) therefore focuses on gathering information regarding what the participants believe to be the most important issues are for young women to have healthy pregnancies before asking questions which are specific to nutrition. This approach allows for data regarding other health issues to be collected before leading the participants to focus on nutrition as this is the primary focus of this program of research. The topic guide was also designed to investigate how youth support professionals viewed their own role in supporting healthy adolescent pregnancies. The findings of the qualitative research phase are presented in the following chapter.

Chapter 7 - Supporting Healthy Adolescent Pregnancies

7.1 Introduction

This chapter presents the third and final study in this programme of research which is a qualitative investigation into how adolescents may be supported to achieve a healthier diet during pregnancy from the perspectives of professionals who provide support for young women. A brief summary of the rationale for and key findings of this study is then followed by a copy of the full published research paper in its original form. The chapter also includes discussion of the contribution of this study to the thesis as a whole.

The work so far presented in this thesis has shown that there are differences between pregnant adolescents and older pregnant women, both in terms of their diet and the outcomes of their pregnancies for both mother and neonate. The quantitative data analysis does not however contextualise these findings by considering the factors which may affect young women's ability to eat well during pregnancy.

The views of young women and health care professionals on this topic have previously been reported¹⁰⁹ in a study which interviewed 34 young women (aged 16-19) and 20 healthcare professionals. The findings of this study suggested that young women did make small changed to their diet during pregnancy despite significant barriers such as lack of knowledge, financial concerns and social constraints. Young women reported that their primary consern was to have a healthy baby and so wanted to understand the benefits of dietary changes in those terms. Healthcare professionals reported that while they did provide information and advice regarding nutrition they often lacked the time and resources to focus on this message. This study therefore aims to explore the perspectives of youth support workers with a wider remit to investigate how they might contribute to improving the health and wellbeing of pregnant young women. The development of the interview guide (appendix 9) was informed by the results of the quantitative analysis. In this sequential exploratory mixed methods design research this process is the first point of integration between the two phases of the study⁹⁹.
7.2 Publication and Impact

The manuscript describing this study is currently under review for publication in Evidence Based Midwifery.

7.3 Article D

The manuscript entitled "Perspectives of youth support professionals on supporting healthy eating in adolescent pregnancies" is reproduced here in the format in which it has been submitted for publication.

Title

Perspectives of youth support professionals on supporting healthy adolescent pregnancies

Authors

1. Katie Marvin-Dowle,

Centre for Health and Social Care Research, Sheffield Hallam University, Collegiate Crescent, Sheffield, S10 2BP,

k.marvin-dowle@shu.ac.uk

2. Hora Soltani,

Centre for Health and Social Care Research, Sheffield Hallam University, Collegiate Crescent, Sheffield, S10 2BP,

H.Soltani@shu.ac.uk

3. Victoria Jane Burley,

School of Food Sciences and Nutrition, University of Leeds, Leeds, LS2 9JT,

v.j.burley@leeds.ac.uk

Corresponding Author:

Katie Marvin-Dowle,

Centre for Health and Social Care Research, Sheffield Hallam University, Collegiate Crescent, Sheffield, S10 2BP,

k.marvin-dowle@shu.ac.uk

Perspectives of youth support professionals on supporting healthy eating in adolescent pregnancies

Abstract

Background: Nutrition during pregnancy has been identified as an important modifiable factor to reduce adverse outcomes in adolescent pregnancies. Young women are supported during their pregnancies by a variety of professionals with both clinical and non-clinical roles. Professionals with a non-clinical support role provide practical and emotional support for young women and often have longer lasting professional relationships with their clients. For this reason this study aims to explore the perspectives of these professionals on how young women can be encouraged to improve their diet during pregnancy.

Methods: This exploratory, qualitative study purposefully sampled participants to take part in semi-structured interviews. Interviews were audio recoded and transcribed. Interview transcripts were then analysis thematically and emerging themes identified.

Results: Five overarching themes were identified from the data: Perceptions of dietary pattern; Connection with baby; Family and social stability; Building relationships and Service availability. Youth support professionals felt that young women often encountered numerous complex barriers to eating healthily during pregnancy. Their lives are often chaotic and lack stable partner and family relationships. They suggested that young women often needed specific practical support, such as being accompanied to health appointments, in order to make improvements. There was also some concern that increasing cuts to

services for this group would make it more difficult for vulnerable young women to access help.

Conclusions: A higher level of consistent, holistic support delivered by joined up networks of professionals is needed in order to help young women achieve healthier pregnancies. Further research is necessary to understand the context of young women's lives, how this relates to their experiences of pregnancy and what type of interventions or resources would have the biggest impact in supporting health behaviours.

Keywords: Adolescent; Pregnancy; Nutrition; Healthy eating; Youth work

Introduction

Pregnancy during the adolescent years has often been associated with higher rates of adverse outcomes for mother and child compared to women aged 20-35. This is the case both in terms of social issues such as isolation and poverty (Cook and Cameron, 2015) and clinical outcomes such as low birthweight and prematurity (Gilbert, Jandial and Field, 2004; Tyrberg, Blomberg and Kjolhede, 2013).

Work to assess the mechanisms by which early pregnancy increases the risk of poor outcomes has suggested that maternal nutrition may be an important modifiable factor. A systematic review by Gibbs et.al (2012) looking at the relationship between adolescent pregnancy and pregnancy outcomes found that very young maternal age had a negative effect on foetal growth and infant survival. The authors suggested that there may be competition between the foetus and the mother who is still herself growing resulting in babies with low birthweight or who are small for their gestational age.

A further systematic review (Marvin-Dowle, 2016) of nutrient intakes and nutritional biomarkers in adolescent pregnancies found that intakes of energy, fibre and a number of key micronutrients were below recommended levels. Assessment of nutritional biomarkers also suggested there may be some cause for concern with regard to iron and selenium status.

Assessment of the dietary patterns of pregnant adolescents in the Born in Bradford cohort (Marvin-Dowle, 2018) has found that young women had higher intakes of snack and processed foods which were high in salt, sugar and saturated fats compared to older pregnant women. The study also found that adolescents had higher intakes of sugar sweetened cola and lower intakes of fruit and vegetables and nutritional supplements.

The evidence therefore suggests that pregnant adolescents are likely to have a diet which is nutritionally poor and also that this may have an impact on the health of their developing child. It is therefore important to understand more about the nature of young women's diets and how they can be supported to make positive changes during pregnancy.

One survey of young women and health care professionals regarding dietary habits and supplementation practices during pregnancy (Soltani et.al, 2017) found that young women did report making some positive changes to their diet during pregnancy. Participants reported increasing intake of fruit, vegetables and milk and the majority also reported taking nutritional supplements at some point during pregnancy. The young women taking part in the survey reported

that their reasoning for making improvements to their diet was the impact on foetal development. There were however also areas of concern in that young women reported consuming less red meat, eggs, oily fish and soft cheese, suggesting that there may be some confusion over foods which should be avoided during pregnancy. The same study also collected responses from health care professionals (HCPs) including midwives, family nurse practitioners and health visitors regarding their discussions with young women around diet and nutrition during pregnancy. Family nurse practitioners were more likely to discuss nutrition in greater depth with young women due to time constraints on midwife and health visitor appointments. Young women stated that they mostly prefer receiving advice and information from HCPs, with mixed responses for other sources including websites and apps, leaflets, family members or parenting classes. HCPs reported that young women often struggled to access Healthy Start vouchers or vitamins but that those who were most motivated would purchase supplements for themselves. It should be recognised however that these findings are restricted to those young women who have chosen to complete a survey on pregnancy nutrition, and so are likely to represent a more motivated group compared to the population as a whole.

Pregnant teenagers with otherwise uncomplicated pregnancies in the United Kingdom (UK) will receive standard National Health Service (NHS) antenatal care as a minimum. Additional services for this population are commissioned locally and therefore vary amongst NHS Trusts. The area in which this research has been conducted has previously had both teenage pregnancy specialist midwives and family nurse partnership services; however these services have been decommissioned at the time of writing.

HCPs are not the only group of professionals who have a role in supporting young women during pregnancy in the UK. Pregnant adolescents are at increased risk of being socially isolated or experiencing difficulties with housing or access to education and training (Cook and Cameron, 2015). They are also more likely to have experienced being cared for by a local authority (Craine et.al, 2014). This means that a significant proportion of young women who become pregnant will have contact with professionals such as youth workers and support advisors who are likely to spend more time with young women than HCPs, whose appointment times are limited. These relationships are largely voluntary on the part of young people and some young women will have difficulties accessing support services, meaning that this type of additional support is far from universal. This said professionals who work in these roles have a unique insight into the experiences of young women during pregnancy and how they may be best supported to have healthy pregnancies. It is therefore this group who are the target for this study.

Aims

The research questions this study aims to address are:

What do youth support professionals perceive to be the barriers and facilitators to healthy pregnancies for young women?

How do youth support professionals think young women can be best supported to have a healthy diet during pregnancy?

Methods

Design

This exploratory, qualitative study used semi-structured interviews to gather information from professionals with a role in providing social or practical support to young women during pregnancy. Professionals whose main role was clinical such as midwives and health visitors were excluded, as the views of these professionals, alongside those of young women, have been sought previously (Rundle, Soltani and Duxbury, 2018). All participants worked in Sheffield which is a large city in Northern England. The interviews took place locally at a time and location convenient to the participants.

Characteristics of the research team

This study was undertaken as part of a PhD programme of work by Katie Marvin-Dowle (KMD), supervised by Prof. Hora Soltani (HS) and Dr Victoria Burley (VB). All of the recruitment and interviews were conducted by KMD during February 2017. KMD worked as a youth worker in sexual health and teenage pregnancy for a number of years before moving into public health epidemiology, she holds a Master's degree in Public Health and a post-graduate certificate in research methods as well as experience of working on numerous research projects and service evaluations. The interviewer was herself pregnant at the time of the research, however every effort was made to conceal the pregnancy until after interviews were completed in order not to influence the responses of participants.

Participants and recruitment

Participants were recruited following a purposive sampling strategy via email invitations sent to relevant agencies, physical flyers distributed to services, social media and word of mouth. As this was intended as an exploratory piece of work the planned sample size was 8-10 participants which was considered adequate to assess the issues considered to be most important by participants. Individuals were eligible to take part if they had a professional role supporting young people and had some experience of working with young women (aged ≤19) during pregnancy. Participants were contacted prior to the study after having expressed interest in taking part in order to arrange appointments to meet. Participants were aware that the study was part of a PhD programme and that the focus was on supporting healthy pregnancies in adolescents.

Information sheets and consent forms were sent out to individuals who expressed an interest in taking part and informed consent obtained before interviews took place. Ethical approval for the study was provided by Sheffield Hallam University Ethics Committee.

Data Collection

Semi-structured interviews were carried out either on a one to one basis or with two participants together who worked in the same agency (two interviews). Participants were also asked to complete a brief demographic questionnaire.

A topic guide was developed which was informed by previous quantitative work carried out during the PhD study. This work has been published elsewhere (Marvin-Dowle et.al 2018a, Marvin-Dowle et.al 2018b, Marvin-Dowle 2016) but

in brief the results suggested that the dietary patterns and nutritional status of pregnant adolescents is likely to be poorer than that of adult women. They are also more likely to experience higher levels of poverty, lower education and their babies may be at higher risk of adverse outcomes. The topic guide for this study therefore covered these main topic areas; subjects explored included issues faced by pregnant young women, perceptions of young women's attitudes to health behaviours (including, but not limited to, dietary behaviours), barriers for young women to making changes and facilitators for healthier pregnancies. Interviews were audio recorded with participants' consent.

Data analysis

Audio recordings of interview data were transcribed verbatim and transcripts anonymised with participants being allocated a participant number for identification purposes. Transcripts were loaded into NVivo 11 software to facilitate analysis. Data analysis was carried out inductively using thematic analysis which involved familiarisation with the data, generation of initial codes, assigning coded data into related categories and developing overarching themes (Braun and Clarke, 2008). Initial analysis was conducted by one researcher and codes and themes verified by a second researcher to enhance inter-reliability.

Results

Participant characteristics

A total of 8 participants took part in 6 interviews; four of the interviews were individual and two interviews took place with two colleagues from the same agency at the same time. No participants withdrew from the study after agreeing to take part; interviews lasted between 20 and 40 minutes. Characteristics of the participants are shown in table 1.

Table 1. Participant characteristics

	Job role	Gender	Age	Highest	Number of	Individual or
			group	qualification	children	shared
						interview
P1	Parenting support	Female	26-35	A Level or	0	Shared
	worker			equivalent		
P2	Parenting support	Male	Not	Not reported	Not reported	Shared
	worker		reported			
P3	Youth support worker	Female	26-35	Degree or	1	Shared
				equivalent		
P4	Education officer	Female	36-45	Degree or	2	Shared
				equivalent		
P5	Youth project	Female	46-55	Degree or	4	Individual
	manager			equivalent		
P6	Youth support	Female	36-45	Degree or	2	Individual
	advisor			equivalent		
P7	Youth support worker	Female	36-45	A Level or	1	Individual
				equivalent		
P8	Perinatal support co-	Female	36-45	A Level or	3	Individual
	ordinator			equivalent		

Analysis of interview transcripts identified 5 over-arching themes within the data: Perceptions of dietary pattern; Connection with baby; Family and social stability; Building relationships and Service availability.

Perceptions of dietary pattern

Concerns over poor diet

All participants discussed the content of young women's diet during pregnancy and expressed concern over the dietary patterns they observed among their client group. Participants perceived the majority of young women were eating a large amount of fast/junk food, snacks that are high in salt, sugar and fat and sugary and/or highly caffeinated drinks and very little fresh produce or home cooked meals.

"And they're overeating, every pregnant girl we see has just got energy drinks and chocolate and you know their diet is so phenomenally bad, McDonalds and KFC and there's no proper food, even a jacket potato with cheese and beans is better than you know, McDonalds and energy drinks and rubbish like that"

Concern over poor dietary patterns was a common theme across all of the interviews. Some participants did speak about young women who did eat more healthily or made a particular effort to change their eating habits, however these young women were singled out as unusual or the exception to the rule.

"I've got my lady who just had a baby a few weeks ago took the health of herself and the unborn child really seriously, she had low iron so made sure she went and bought really high iron foods, and really got that link, so it's really a bit hit and miss to be honest with you."

Lack of sufficient knowledge or information

One participant mentioned that young women were often confused regarding which foods should be avoided during pregnancy, and that they did not have adequate knowledge or reliable information to be able to provide advice regarding this.

"then when they do come to places like this and they will ask, you know they're asking questions like can I eat mayonnaise? and I'm like I really don't know, you're going to have to ask your midwife, and they'll be like my midwife says I can after 38 weeks, and my midwife says something else and it's like, what?! There's still confusion"

There was a sense that young women were less knowledgeable and less prepared for the pregnancy than older women in terms of diet and nutritional supplements to support the pregnancy. This was presented almost as a dichotomy by one participant who suggested the older women would have significantly more knowledge than adolescents.

"it's not like for older women getting pregnant and knowing exactly what you're supposed to be eating, what you're not supposed to be eating, take your folic acid and all of that"

Connection with baby

All participants talked about their perceptions of young women's readiness or motivation to make changes to improve their own health and that of their baby. A key theme related to behaviour change was the importance young women gave to healthy eating. The majority of participants reported that they didn't think the young women they worked with felt their nutrition was important.

"I don't feel I'm discussing with them that they feel that's [diet] important at all"

There was a regular theme across the data which suggested that the participants felt that some young women struggled to see their baby as really there before the birth. This meant that these young women did not necessarily connect their behaviours with the health of their unborn child, particularly with reference to diet.

"It's a hard one cause we've come across mums who don't actually see baby as being there until the baby is actually in their arms, and it doesn't matter what I'm eating because they might not see it as it does affect the growing baby"

This was linked to ideas put forward by several participants that the young women they worked with were not always emotionally mature enough to make decisive changes or to accept the significant changes that were happening to them as a result of the pregnancy. Similarly to the comments on diet content however there were exceptions where participants spoke of young women they had worked with who had made positive changes, but these were much less common than references to young women who either did not try to make changes or who struggled to do so.

"There's some 17, 18 year old mums that we've worked with and I've worked with where you're like she's really switched on she's amazing and she could teach some of the older mums and people her own age sometimes where you think they could really learn something, but I think sometimes that's a big barrier to a healthy pregnancy is willingness and acceptance that things have to change."

Family and social stability

The majority of young women the participants were involved with face significant challenges, both in terms of their historical family situations and current living arrangements.

Family support and growing up in care

A number of the participants spoke about the difficult family backgrounds their clients had come from including high levels of poverty, unstable families and that a significant proportion of the young women they worked with had spent time being looked after in local authority care.

"A lot of the issues for our young people are because they've not had any positive parenting in their formative years, so they've not grown up with good routines, they've come from abused backgrounds, so they don't always understand actually what a good parent is and what they should do."

It was also suggested that experiences of being removed from families and being placed in local authority care may influence young women's willingness to seek or accept support, particularly from social services, for fear that their own children may be removed from their care.

"Remember these young people have been removed from their families, their automatic fear is that you're there to do the same"

It was also discussed how unstable family situations may have contributed to a lack of knowledge and skills around healthy eating and cooking, in that young women may not have had role models available to them or have been taught how to prepare and cook healthier meals.

"They've probably not built those skills up to care for themselves ... in that kind of way so in a heathy eating sort of way"

Housing and poverty

Young women's current living situation was also a common theme with the majority of participants talking about young women living in poverty and experiencing housing issues. Being able to provide a safe and appropriate home for their baby was discussed as being clearly important to young women and a significant source of stress.

"I've just had a lady who gave birth about 4 weeks ago and she moved house 2 weeks before she gave birth, so I mean I was there 'til 7, 8 o'clock at night helping her get everything ready, and as soon as, she'd literally just moved in and got everything sorted and baby arrived"

The situation of living in poverty was also considered a barrier to eating well by some participants, as healthy food was considered to be prohibitively expensive.

"How are they supposed to eat healthily if they've got no extra money, have you seen the price of fruit? You know, it's ridiculously overpriced."

One participant also suggested that promotions on less healthy foods made it difficult for young women to make healthier choices.

"Yeah our teenage parents will come in with all kinds of rubbish, donuts and you think geez, but when they're offering things like 4 donuts for a pound why wouldn't you? I do it, you know, there's a nice healthy sandwich or 4 donuts for a pound"

Social support and isolation

Difficult personal relationships, particularly with partners/baby's fathers were discussed by all of the participants. Young women were described as being very heavily influenced by their partners. Young fathers were described as either a negative, controlling influence or as absent.

"Their boyfriends are a massive influence on them, on the decisions they make about breastfeeding, stopping smoking, getting their figure back, starting to have sex again, you know, they're very dominated by their partners."

Similar issues were described with young women losing friends after having a baby as young women's priorities changed, leading young women to become isolated and without support.

"It's the age old you know, he's not going to go out with his friends anymore, he's going to stay in with me and the baby and all my friends are going to come round and visit, you know after 6 weeks it's, I could 76 write it down on a piece of paper and say there, 6 weeks' time I can guarantee your friends will disappear and so will your boyfriend, and they nearly always do, 99% of the time they do."

Isolation and mental health issues, compounded by this lack of support, were very prominent themes identified from the data. Some participants spoke about how many of their clients had mental health issues which were present before pregnancy and which clearly impacted on their wellbeing and ability to cope with a pregnancy.

"Most actually had complex needs in terms of their health and mental health already that weren't being addressed"

For others the stress and pressure of the pregnancy and parenthood, connected to the difficult living situations the young women were in meant that mental health issues were almost inevitable:

"so a lot of the ladies that I work with are single mums with no family, no support and are very isolated in the community on their own basically, and it's really difficult, really challenging, to think if you are pregnant at 16, 17, 18, 19 it's quite scary, I think, not having anybody there."

There was a sense that young women actually do incredibly well to cope with pregnancy and motherhood in the context of chaotic lives, and that often more is expected of young parents than is possible given the reality of their situation.

"I think we expect these young people that have got post-traumatic stress disorder or, you know, lots of really significant abuse in their life, to just function as an everyday parent in life. And a lot of them do don't get me wrong, I've got loads of success stories, but sometimes they're tired and just need a break"

Building Relationships

The participants spoke extensively about their own roles in supporting young women, how they worked with other agencies and where they felt services were lacking or falling short of providing what is necessary. The participants generally saw their role as separate to that of clinical health professionals, but that didn't mean they were not concerned with the health of the pregnancy.

"We're not social workers, we're not midwives, we're youth workers and we come from a completely different perspective, but they know and we know that we've got their health and their baby's health at the forefront of everything we do"

Building relationships with young women was considered central to their work, and it was suggested that this may be more challenging due to young women's backgrounds.

"cause they've got to build up that trust in professionals and people that are trying to support them, whereas they might be starting two steps behind someone else whose on a more secure base that can kind of go ok I've got to speak to A_____ about this and I know that I can trust A____, whereas someone who's never had that ability to have a trusting relationship with an adult, you've got to build that trust."

Most of the participants also spoke about providing practical support, particularly in facilitating young women to attend health appointments which were often missed, leading to potential consequences for the health of the pregnancy.

"We make assumptions that they've got access to services therefore they access them, so they've got midwives and social workers some of the young people but actually often it's about relationships, about hand holding, about transition, about actually getting to places, so some of those health needs we assume are being addressed but they're not because just on a basic level they're not accessing appointments, so they miss lots of appointments"

Service Availability

There was also a significant amount of discussion in every interview about the need for more resources and the impact of cuts to existing services. All of the participants considered that cuts to services had had a detrimental effect on young women's health during pregnancy.

"They used to have, years ago, specialists that once a girl found out she was pregnant was put straight on to her so she would be given all the information and then directed to midwives and doctors and everything else. First round of cuts that was the first thing that went"

When asked about whether additional resources would be helpful the majority agreed that an information resource which gave universal information and also provided local signposting would be useful, and that this would need to be provided in multiple formats. "I think you'd need to cover all of those [formats], I mean definitely use technology, something they could easily download on their phone or they could come in here and have somebody sit and go through with them like a physical paper pack."

Overall, the overwhelming message from the data was that the participants recognised the importance of supporting healthy eating as part of a healthy pregnancy and supported young women to do their best in often particularly difficult circumstances. There are significant concerns, particularly around young women's mental health, and dedicated, specialised services are needed for any significant improvements to be made.

"Yeah if you could just ask to get a new service for pregnant teenagers back that'd be great, that's all we want, that'd be really good"

Discussion

This study is novel in that it seeks to gather the perspectives of a professional group which have not before been consulted on the topic of improving health in adolescent pregnancies. Integrated models of care which bring together social support services and primary health care in the UK are rare; one example of such a service is the Well Centre in South London (Hagell and Lamb, 2016), which is designed to be a 'one-stop-shop' specifically for adolescents. Evaluation has shown that young people were accessing the services who were not engaged with any other service, particularly those with mental health needs. This suggests that collaborative work between primary care services and youth

work providers may facilitate the most vulnerable young people to access and engage with the services they need, however the effectiveness of such services require robust independent evaluations.

In this research, nutrition was a topic that was generally considered to be important by youth support professionals, however they perceived that this view was not necessarily shared by the young women they worked with. Participants reported that the young women in their care ate large amounts of take away and processed food and sugary drinks and little fruit and vegetables. This is largely consistent with existing literature suggesting that adolescent's diet during pregnancy is often poor (Marvin-Dowle et.al, 2016; Marvin-Dowle et.al, 2018; Northstone, Emmett and Rogers, 2008; Scholl and Hediger, 1995). These findings are also consistent with evidence from the latest wave of the National Diet and Nutrition Survey (NDNS) produced by Public Health England (2018). Results from the 2014/15 - 2015/16 surveys found adolescent girls reported consuming an average of 2.8 portions of fruit and vegetables a day compared to 4.2 portions for adult women. Young women also consumed almost twice the amount of sugar sweetened beverages compared to adult women (183g and 100g respectively). This suggests that it is not just pregnant adolescents but in fact the majority of young women who have a poor diet. While the immediate consequences may be more apparent for young women who are pregnant this does suggest population wide interventions are needed to improve nutrition in all adolescents.

Reasons suggested for poor diet in pregnant young women were varied and included lack of interest or belief that nutrition makes a difference, lack of

knowledge and skills and the price of healthier choices. Participants spoke about how they believed young women often didn't make the connection between themselves and their baby until after the baby was born, particularly with reference to how what they ate might affect the developing foetus. Previous research has shown that a mother's emotional connection with her child during pregnancy can have a significant effect on her health behaviours. The theory of maternal-fetal attachment (MFA) proposed by Cranley (1981) is defined as "the extent to which women engage in behaviours that represent an affiliation and interaction with their unborn child" and the validated tool has been used to explore connections between attachment and maternal health behaviours. One study exploring correlations between MFA and comprehensive measure of pregnancy health behaviours (including balance of rest and exercise, safety measures, nutrition, avoiding use of harmful substances, obtaining health care, and obtaining information) in low-income urban women (Alhusen et.al, 2012) found an overall positive association between women's scores on the two measures. This suggests that improving women's emotional connection to the developing foetus may in turn improve health behaviours. There is little evidence in the existing literature regarding differences in MFA scores between adolescent and adult mothers, suggesting that this may be an important area for further study.

There is also a significant lack of evidence in the literature evaluating whether young women make connections between their diet during pregnancy and the health of their child, suggesting that this is an important area for further research. One small study (Whisner, Bruening and O'Brien, 2016) reporting survey data reported that the majority of pregnant teenagers recognised that

diet during pregnancy was important, but that this did not necessarily translate into making positive changes. A further recent study (Rundle, Soltani and Duxbury, 2018) found that the desire to deliver a healthy baby was a prime motivator for pregnant young women and therefore that the benefits of making dietary changes should be framed in those terms. This study also reports young women making some small positive changes to their diet despite barriers to doing so. Further insight into young women's understanding of how their diet impacts on the health of their baby and the factors which motivate positive changes is needed.

Psychosocial stability, lack of social support and isolation were themes which emerged clearly from this data set. Mental health needs in this population are likely to be high as these are associated with neglect or abuse during childhood (Mills et.al, 2013) (a likely contributor to young women being looked after in local authority care) and pregnant and parenting adolescents being at higher risk of experiencing mental health problems compared to older mothers (Siegel and Brandon, 2014). This is an important observation in the context of this study due to the impact that poor mental health may have on young women's ability to eat well and look after their own wellbeing. Poor diet has also been correlated with poor mental health in adolescents (Kulkarni, Swinburn and Utter, 2015; O'Neil et.al, 2014) meaning that improvements to young women's mental health may in turn help them to improve their diet.

The price of healthy food such as fruit was cited as a significant barrier to eating well and that young women favoured cheap, high energy density snack foods which were often subject to price promotions. There is significant evidence to suggest that socioeconomic deprivation is associated with poorer nutritional status, particularly in adolescents. One study looking both at the attitudes and behaviours of adolescents towards healthy eating found that while participants had similar attitudes to healthy eating regardless of socio-economic position those from more deprived areas were more likely to report higher intakes of fast foods and sugar sweetened soft drinks (Utter et.al, 2011).

The main message regarding improving the health and wellbeing of young women during pregnancy gained from this study was that more targeted support services are needed to meet the needs of a vulnerable population. While physical resources were considered a useful tool for health promotion, having somebody to talk to with a holistic view of health and wellbeing was considered to be the key to making improvements. This, along with greater levels of consistency in the messages that young women receive regarding how to make positive changes to their diet, are key in developing practical strategies to support young women. There is a wealth of research addressing approaches to sexual health promotion and preventing teenage pregnancies (Mezey e.al, 2017; Oyedele, Wright and Maja, 2015; Sorhaindo et.al, 2016), including the prevention of second pregnancies (Aslam et.al, 2015), however evidence looking at strategies to support the health of young women during their pregnancy in the academic literature is severely lacking.

This study is limited somewhat by the small number of participants; however the use of qualitative methods means that the depth of the data is central to the research design and the intended number of participants was reached. Saturation of the data with regard to practical support to improve dietary habits

was achieved, however it is likely that longer interviews with more participants would yield more detailed insights regarding the myriad difficulties faced by young women during pregnancy Further work to include the perspectives of other professionals working with young people such as teachers and social workers would also be useful in order to develop a more complete picture of the available support for young women.

This study also does not include the perspective of pregnant young women themselves. This study was designed to complement a program of work in which the opinions of pregnant young women and young mothers have been explored (this has been reported elsewhere (Soltani et.al, 2016; Rundle, Soltani and Duxbury, 2018)⁻ Further, more in depth work to examine how young women view health during pregnancy and the factors which would support or motivate them is essential to our understanding of how to improve outcomes for young mothers and their children.

Conclusion

This study adds to the body of work exploring how young women's pregnancy outcomes could be improved through dietary changes. The findings do however suggest that for many young women in this population there are numerous, complex factors which impact on their ability to have healthier pregnancies. A higher level of consistent, holistic support delivered by joined up networks of professionals is needed in order to help young women achieve healthier pregnancies. Further research is necessary to understand the context of young women's lives, how this relates to their experiences of pregnancy and what type

of interventions or resources would have the biggest impact in supporting health behaviours.

List of abbreviations

- FNP Family Nurse Practitioner
- HCP Health Care Professional
- MFA Maternal-Foetal Attachment
- NDNS National Diet and Nutrition Survey
- NHS National Health Service
- UK United Kingdom

Declarations

Ethics approval and consent to participate

All participants gave their informed, written consent to participate in the study, ethical approval was granted by Sheffield Hallam University Ethics Committee on 19.07.16, proposal number 2015-16/HWB/HSC/35.

Consent for publication

Not applicable

Availability of data and materials

The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

Competing interests

The authors declare that they have no competing interests

Funding

This research has been carried out as part of a PhD project supported by Sheffield Hallam University and the National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care for Yorkshire and Humber. Therefore no additional funding was required for this work.

Authors' Contributions

KMD - Primary responsibility for development of research question and study design, collection and analysis of data and writing the main body of the article

HS - Data analysis including verification of codes, advisory role in development of research question and study design, revisions to and approval of final article

VB - Advisory role in development of research question and study design, revisions to and approval of final article

Acknowledgements

The authors would like to thank the participants of this study for their time and valuable contributions to this work.

References

Alhusen JL, Gross D, Hayat MJ, Sharps PW. (2012) The influence of maternal– fetal attachment and health practices on neonatal outcomes in low-income, urban women. Research in nursing & health;35(2):112-20. Aslam RH, Hendry M, Carter B, Noyes J, Rycroft Malone J, Booth A, Pasterfield D, Charles JM, Craine N, Tudor Edwards R, Ntambwe LI. (2015) Interventions for preventing unintended repeat pregnancies among adolescents. The Cochrane Library.

Cook SMC, Cameron ST. (2015) Social issues of teenage pregnancy. Obstetrics, Gynaecology & Reproductive Medicine; 25(9):243-8

Craine, N., Midgley, C., Zou, L., Evans, H., Whitaker, R. & Lyons, M. (2014) Elevated teenage conception rates amongst looked after children: a national audit. Public Health, 128, 668–670.

Cranley, M. S. (1981). Development of a tool for the measurement of maternal attachment during pregnancy. *Nursing Research*, **30**, 281–284.

Gibbs, C.M., Wendt, A., Peters, S. and Hogue, C.J., (2012). The impact of early age at first childbirth on maternal and infant health. *Paediatric and perinatal epidemiology*, *26*, pp.259-284.

Gilbert WM. Jandial D. Field NT. (2004) Birth outcomes in teenage pregnancies. *J Matern Fetal Neonatal Med*;16(5):265-270.

Hagell A, Lamb S. (2016) Developing an integrated primary health care and youth work service for young people in Lambeth: learning from the Well Centre. Journal of Children's Services;11(3):233-43.

Kulkarni AA, Swinburn BA, Utter J. (2015) Associations between diet quality and mental health in socially disadvantaged New Zealand adolescents. European Journal of Clinical Nutrition;69(1):79. Marvin-Dowle K, Burley VJ, Soltani H. (2016) Nutrient intakes and nutritional biomarkers in pregnant adolescents: a systematic review of studies in developed countries. BMC pregnancy and childbirth;16(1):268. Marvin-Dowle, K., Kilner, K., Burley, V. J., & Soltani, H. (2018a). Impact of adolescent age on maternal and neonatal outcomes in the Born in Bradford cohort. BMJ open, 8(3), e016258.

Marvin-Dowle, K., Kilner, K., Burley, V. and Soltani, H., (2018b). Differences in dietary pattern by maternal age in the Born in Bradford cohort: A comparative analysis. *PloS one*, *13*(12), p.e0208879.

Mezey G, Robinson F, Gillard S, Mantovani N, Meyer D, White S, Bonell C. (2017) Tackling the problem of teenage pregnancy in looked-after children: a peer mentoring approach. Child & Family Social Work;22(1):527-36.

Mills R, Scott J, Alati R, O'Callaghan M, Najman JM, Strathearn L. (2013) Child maltreatment and adolescent mental health problems in a large birth cohort. Child abuse & neglect;37(5):292-302.

Northstone K, Emmett P, Rogers I. (2008) Dietary patterns in pregnancy and associations with socio-demographic and lifestyle factors. European Journal of Clinical Nutrition;62(4):471.

O'Neil A, Quirk SE, Housden S, Brennan SL, Williams LJ, Pasco JA, Berk M, Jacka FN. (2014) Relationship between diet and mental health in children and adolescents: a systematic review. American journal of public health;104(10):e31-42.

Oyedele OA, Wright SC, Maja TM. (2015) Community participation in teenage pregnancy prevention programmes: A systematic review. Journal of Research in Nursing and Midwifery;4(2):24-36.

Public Health England, National Diet and Nutrition Survey. Results from Years 7-8 (combined) of the Rolling Programme (2014/15 - 2015/16) [Online] available from https://www.gov.uk/government/statistics/ndns-results-from-years-7-and-8-combined accessed 23.06.18

Rundle, R., Soltani, H. and Duxbury, A., (2018). Exploring the views of young women and their healthcare professionals on dietary habits and supplementation practices in adolescent pregnancy: a qualitative study. *BMC Nutrition*, *4*(1), p.45.

Scholl TO, Hediger ML. (1995) Weight gain, nutrition, and pregnancy outcome: findings from the Camden study of teenage and minority gravidas. In Seminars in Perinatology; 19(3):171-81

Siegel RS, Brandon AR. (2014) Adolescents, pregnancy, and mental health. Journal of pediatric and adolescent gynecology.;27(3):138-50.

Soltani H, Duxbury A, Rundle R, Marvin-Dowle K. (2017) Dietary habits and supplementation practices of young women during pregnancy: an online cross-sectional survey of young mothers and health care professionals. BMC Nutrition;3(1):19.

Sorhaindo A, Mitchell K, Fletcher A, Jessiman P, Keogh P, Bonell C. (2016) Young women's lived experience of participating in a positive youth development programme: The "Teens & Toddlers" pregnancy prevention intervention. Health Education;116(4):356-71. Tyrberg RB, Blomberg M, Kjolhede P. (2013) Deliveries among teenage women - with emphasis on incidence and mode of delivery: A swedish national survey from 1973 to 2010. *BMC Pregnancy Childbirth*; 9;13(1):1

Utter J, Denny S, Crengle S, Ameratunga S, Clark T, Maddison R, Percival T. (2011) Socio-economic differences in eating-related attitudes, behaviours and environments of adolescents. Public health nutrition;14(4):629-34.

Virginia Braun & Victoria Clarke (2008) Using thematic analysis in psychology,

Qualitative Research in Psychology, 3:2, 77-101, DOI:

10.1191/1478088706qp063oa

Whisner, C. M., Bruening, M., & O'Brien, K. O. (2016). A Brief Survey of Dietary Beliefs and Behaviors of Pregnant Adolescents. *Journal of pediatric and adolescent gynecology*, *29*(5), 476-481.

7.4 Summary and implications for thesis

The results of this study showed that there were numerous, complex factors which professionals perceive to impact negatively on young women's ability to achieve and maintain a healthy diet during pregnancy. The included both individual factors and social context and the availability of appropriate services. This work contributes to the narrative regarding how adolescent women can be best supported to have healthier pregnancies a potentially begin to reduce the disparities seen in adverse neonatal outcomes.

The following chapter synthesises the findings of all three studies in this programme of work and discusses the contribution of this thesis in the context of the existing field of academic literature. Recommendations for future work arising from this thesis will also be made and the strengths and limitations of the research discussed.

Chapter 8 - Integration of research Findings

8.1 Introduction

This thesis has presented a systematic review of current evidence followed by a programme of research composed of three original research studies. This program of research was undertaken as a mixed methods research project with sequential exploratory design. A key feature of this research design is the synthesis of research findings from the constituent parts of the programme of work to develop overarching themes. This chapter therefore completes this synthesis of findings from the quantitative and qualitative phases of data collection and analysis.

8.2 Integration of findings

As described in chapter four this thesis follows a sequential explanatory mixed methods design. As priority in the design was given to the quantitative phase, and this analysis was undertaken first in the sequence, the findings of the quantitative phases are used to inform the development of the qualitative phase which aims to explain and expand upon the previous findings¹⁰². The integration of methods therefore occurred at two points across the study; in the development of study materials for the qualitative phase, as described in chapter 7, and in this section where the findings of both quantitative and qualitative analysis will be integrated.

The key quantitative findings in this study show that pregnant adolescents are likely to not meet recommendations for intake of key nutrients to support pregnancy and eat higher quantities of nutrient poor snack and processed foods compared to older pregnant women. The results also show that babies born to adolescents have higher odds of being born extremely premature or at extremely low birth weights. The qualitative phase of the study therefore aimed to elaborate on these findings by gathering more in-depth information regarding possible influential factors on difficulties with eating well during pregnancy and disparities in neonatal health outcomes, by striving to understand the context of young women's lives. This section brings together the findings across the whole study in a series of overarching themes.

8.2.1 Social context

The social context of the population of young women who become pregnant during adolescence is an important consideration which impacts on both the health of the pregnancy and future life chances for mother and child. In this first overarching theme, results from the quantitative phase indicated that there were differences in social, cultural and economic indicators between age groups. Women in the adolescent group were, on average, from areas of higher socioeconomic deprivation (as measured by index of multiple deprivation score), were more likely to be single parents and to have smoked or used recreational drugs during pregnancy. They were also more likely to describe their ethnicity as white British. The qualitative analysis phase supported and elaborated upon these findings; participants discussed young women often having experienced significant poverty and having difficulties with stable housing and maintaining relationships with the baby's father.

Together the findings of this study therefore suggest that the social situation of pregnant adolescents is likely to be less favourable than that of adult women, which is an important overarching theme from this research.

8.2.2 Dietary pattern

The second overarching theme identified from the research is regarding evidence of difference in dietary patterns between adolescents and adult women during pregnancy. Results from the quantitative analysis phase suggested that adolescent women showed higher adherence to the dietary pattern characterised by snack and processed foods which are high in fat, sugar and salt and low in vitamins and minerals. In addition, adolescent women reported consuming less fruit and vegetables and more sugar sweetened cola compared to adult women. This finding was supported by the results of the qualitative analysis phase which identified a perception that adolescent pregnant women consumed larger quantities of 'junk food' than was considered to be healthy and often did not consume enough of healthier alternatives such as fruit and vegetables or home cooked, balanced meals.

There were differences in BMI between adolescents and adult women identified in the quantitative analysis with adolescent women being more likely to be underweight and less likely to be overweight or obese. This finding may seem to contradict the finding that adolescents consumed larger amounts of foods which are high in fat and sugar, and therefore with greater energy density. Evidence from the qualitative phase however suggests that young women in this population may have irregular eating patterns and be more likely to skip meals or eat less frequently than older women which may contribute to the lower BMI status among adolescents. Reasons for this include insufficient access to food due to cost or lack of skill in food preparation, which is maintained by the differences in socio-economic status observed in both phases of data analysis. Lower use of nutritional supplements was also seen amongst adolescent women in the quantitative data analysis. While specific discussion of supplement use was not a major theme from the qualitative work, significant issues with access to services for young women were reported. This, coupled with the cost of buying supplements from retailers, could in part explain why adolescents are less likely to access and use dietary supplements to support their nutrition in pregnancy.

8.2.3 Support

Need for, and ability to access support was highlighted as a significant issue in both phases of data collection. The quantitative results showed that adolescent women booked with a midwife significantly later in their pregnancy than adult women. The results also showed that adolescent women were likely to have higher support needs for issues likely to affect the health of the pregnancy such as smoking and poor nutritional status. It is also important to consider that the cohort utilised for the quantitative analysis includes only those women who were engaged with services during the second trimester of pregnancy, therefore potentially the most vulnerable women in terms of access to support are excluded from the analysis by virtue of not being engaged with services. The qualitative analysis found significant concerns with young women's ability to access services for a number of reasons including existence of suitable services, knowledge of services, ability to travel to services and emotional capacity to engage with services. The data also revealed a perceived high level of mental health problems among pregnant adolescents and young mothers. A lack of family and social support was also recognised; adolescent women being more likely to be single parents was reflected across the data collection phases.
8.3 Summary

As described above, the findings from this programme of research have been integrated and three overarching themes have been identified. The following discussion chapter expands upon these key findings and situates them within the existing research literature. It also includes identification of the main strengths and any limitations of this programme of research and makes suggestions for future developments.

Chapter 9 - Discussion

9.1 Introduction

The primary research question in this thesis was '*Is there a need for additional nutritional support in adolescent pregnancies?*' which has been addressed through a program of mixed methods research. The research has identified three overarching themes which are presented in chapter 8. These overarching themes emerging from this program of research provide insight into the specific issues faced by adolescent women during pregnancy and the impacts these might have on their own health and that of their babies.

In this chapter the key findings are discussed, with reference to the existing academic literature, and the contribution to knowledge made by these new findings is explored. This chapter will also discuss the strengths and limitations of this program of work and present suggested avenues for future development.

9.2 Discussion of key findings

9.2.1 Social context

This study has identified that social context is of paramount importance when considering the health and wellbeing of adolescents during pregnancy and outcomes for their babies. The association between socio-economic status (SES) and the prevalence of adolescent pregnancy has been well documented with young women living in socially disadvantaged areas reported as being more likely to become pregnant as teenagers. The most recent under 18 conceptions data for England shows that the rate of conceptions among young women living in the most deprived decile (according to the 2015 index of multiple deprivation) was 24.8 per 1,000 young women aged 15-17 compared to 11.9 per 1,000 in the least deprived decile, meaning that the difference in conception rate due to deprivation is more than doubled¹¹⁰. There is little evidence to suggest that this situation is changing over time, rather inequalities

in the number of conceptions due to deprivation may indeed be widening. A Scottish study¹¹¹ assessing the social determinants of teenage pregnancy over a 60 year period found that despite a significant decrease in the overall teenage conception rate during the study period, the rate ratio between the most and least deprived areas increased significantly. They conclude that socio-economic deprivation is a significant and enduring predictor of teenage pregnancy. It has also been established that socio-economic deprivation is a risk factor for both physical and mental health problems in populations, including among children and young people.

Particularly relevant here is the association between socio-economic status and a number of adverse neonatal outcomes including low birthweight and pre-term delivery, with women living in areas of higher deprivation being at increased risk. One review¹¹² of the research evidence suggested that these associations may be due at least in part to differences in cigarette smoking and lower gestational weight gain, issues which are also independently associated with young maternal age. The review also acknowledged that there was some evidence that psychosocial factors may have an important role to play but that more research in this area was needed.

It is also important to consider that the Born in Bradford cohort utilised for the quantitative analysis in this study is somewhat unique in terms of its demographic make-up. Born in Bradford was established in 2007 in response to concerns about the high rates of childhood morbidity and mortality in the city of Bradford with the aim of examining how genetic, nutritional, environmental, behavioural and social factors may influence health and development through childhood and into adult life. Approximately 20% of the population of Bradford are South Asian, of whom approximately 90% are of Pakistani origin. Due to the relative youth and high fertility rates among this population almost half of the babies born in Bradford also has significant areas of socio-economic deprivation with approximately 60% of the babies born in Bradford being born into the poorest 20% of the population of England and Wales. Further details regarding features of the cohort which were not relevant to the work presented in this thesis are available in the cohort profile¹⁰³.

As discussed in chapter 5, the population of women aged under 20 years in the cohort is demographically significantly different to the cohort as a whole. The analysis in this study was restricted to primiparous women with singleton pregnancies in order to control for the impact of parity and multiple births on outcomes, particularly as the average parity increased with maternal age. While over 37% of women in the study population as a whole were of Pakistani ethnicity the proportion was only just over 16% among the adolescent women in the sample. The adolescent women in the sample were also more likely to have smoked during pregnancy, be single parents, have booked for antenatal care later in their pregnancy and live in more deprived areas. While recent national statistics regarding the ethnicity of pregnant adolescents are not available, the demographic profile of the young women included in this study is reflected elsewhere in the literature^{111,113, 114}.

9.2.2 Maternal weight status

Pre-pregnancy BMI and gestational weight gain and their association with maternal age and socio-economic status are important considerations in this study. As discussed in chapter 2, evidence of BMI status among women in England showed that, while incidence of overweight and obesity in the UK overall is high, younger women were more likely to be underweight and less likely to be overweight or obese than older women and this pattern was also reflected in the results of the present study. There is also evidence to suggest that pre-pregnancy BMI is associated with gestational weight gain with adolescent women with a lower BMI gaining less during their pregnancy¹¹⁵. Guidelines for gestational weight gain from the institute of medicine vary by prepregnancy BMI with recommendations stating that women with lower BMIs should aim to gain more weight during pregnancy than their heavier counterparts¹¹⁶. A recent systematic review¹¹⁷ aiming to assess the association of adverse outcomes with gestational weight gain either above or below these recommendations confirmed their efficacy. The results showed that weight gain below recommendations to be associated with small for gestational age and pre-term birth while weight gain above recommendations was associated with

large for gestational age, macrosomia and caesarean section. A further study¹¹⁸ has however highlighted a potential issue with the use of these guidelines in adolescent pregnancies as BMI category is assessed based on adult criteria as opposed to centile based growth charts as would otherwise be used for assessing adolescent BMI categories. This BMI classification disparity may therefore lead to weight gain recommendations for pregnant adolescents which are in fact inappropriate.

Low gestational weight gain is associated with pre-term delivery and low birthweight¹¹⁹, outcomes which were also observed to be associated with young maternal age in the analysis of the Born in Bradford data presented in chapter 5. Unfortunately it was not possible to examine the impact of gestational weight gain on outcomes in this cohort as no further information on gestational weight changes was available for the women at the end of their pregnancy. This may therefore be an important avenue for consideration in exploring the causal mechanisms of poorer outcomes among adolescent women. High prepregnancy BMI and high gestational weight gain have also been associated with adverse outcomes including increased risk of caesarean section, macrosomia and large for gestational age infants¹²⁰. While high BMI and high gestational weight gain are not necessarily associated with young maternal age, maternal obesity is associated with low socio-economic status¹²¹ meaning that this could be a consideration for women in this population in future pregnancies, especially in the context of poor diets.

9.2.3 Mental health

Mental health was identified in the qualitative enquiry presented in chapter 7 as a key factor influencing the health and wellbeing of pregnant young women. Mental health has been reported to account for a large proportion of the disease burden among young people globally, and is strongly related to other health issues including poor sexual and reproductive health¹²². A systematic review addressing the association between socio-economic inequalities and mental health problems in children and adolescents (aged 4-18 years) concluded that children and adolescents with low SES are at higher risk of mental health problems. The review included examination of 55 published studies, all of which reported at least one marker of SES and at least one identified mental health problem using validated instruments. Further to this an American study¹²³ assessing internalising behaviour problems (depression, anxiety, withdrawal and somatic problems) in almost 3,000 children aged 5-11 years found a linear relationship between neighbourhood socio-economic status and the percentage of children above the clinical threshold for diagnosis. The study also found that areas of concentrated disadvantage were associated with higher prevalence of mental health problems among children. Having established that socioeconomic deprivation is a significant predictor of both adolescent pregnancy and mental health issues, it is perhaps no surprise that poor mental health and unstable family backgrounds were identified in this study as significant issues for pregnant adolescents, as reported in chapter 7.

These findings are further supported in the existing academic literature; One USA based study¹²⁴ which looked specifically at the association between adverse childhood experiences (ACEs) and teenage pregnancy found that as ACE scores on a validated tool increased so did the likelihood of becoming pregnant during adolescence. The same study also found that while teenage pregnancy was not independently associated with foetal death, ACE score was associated with increased foetal death after first pregnancy.

Mental health issues have also been independently associated with less healthy diets in adolescents. A study of almost 3,000 socially deprived young people in East London¹²⁵ found that poor diet quality was significantly associated with mental health difficulties as assessed using two validated assessment tools (strengths and difficulties questionnaire and the short mood and feelings questionnaire). Nutrition has also been suggested as an important modifiable factor in the treatment and management of mental health conditions. One review¹²⁶ of the impact of nutrition on depression in women of childbearing age found significant support for dietary interventions including fish oil and folic acid supplementation in supporting management of major depressive disorders and that deficiencies in folate, vitamin B12, iron, zinc and selenium tend to be more prevalent among those suffering with depression. The direction of the relationship between mental health difficulties and poor diet is not clear, and it is likely that it is somewhat cyclical. This suggests that further work is needed to

explore the complex relationships between adverse childhood experiences, mental health issues, socio-economic deprivation, nutritional deficiencies conceptions to young women and outcomes for pregnant adolescents and their children.

9.2.4 Diet quality

There is also a significant body of evidence supporting an association between socio-economic status and poorer quality diets in various populations. One review¹²⁷ of the family and social-determinants of children's eating patterns and diet quality found that the amount and type of foods eaten by children are dependent on both the physical and social environment, much of which is dictated by the behaviours, attitudes and feeding styles of the adults with whom they live. A more quantitative approach assessed the nutrient intake of 250 pregnant women in Sheffield¹²⁸ and found that participants living in the 40% most deprived wards had significantly lower nutrient intakes compared to more affluent women in the study.

The findings of the research presented in this thesis suggest that adolescent age may be an independent predictor of poorer quality diets during pregnancy when compared to those of adult pregnant women. The systematic review of the available evidence presented in chapter 3 found that pregnant adolescents were not meeting the recommended intakes for energy, fibre and a number of key micronutrients. As discussed previously, under reporting of energy intake is common in epidemiological studies and this is unlikely to be of serious concern given the adequate gestational weight gain also reported in the studies included in the systematic review. Micronutrient intake and subsequent status is however more likely to be an issue for this population given the evidence presented both by the systematic review and the analysis of dietary patterns presented in chapter 6.

The team involved with supervising and advising on this program of research have recently published two studies which are closely aligned with the work presented in this thesis. These papers seek to explore the perspectives of young women and health care professionals through the use of an online survey and qualitative interviews. The results of the survey phase of this project showed that there were some difficulties with access to nutritional supplements and many young women reported low compliance, largely due to forgetting to take them. Young women did report some positive dietary changes but also reported unnecessary food avoidances which may be due to misunderstanding of recommendations.

The qualitative phase of this study¹⁰⁹ interviewed young women who were aged 16-19 during pregnancy and who were either currently pregnant or had given birth in the preceding 6 months. The study interviewed 34 young women with a focus on dietary habits and supplementation practices. Young women in this study reported high consumption of processed snack foods and take away meals and often irregular meal patterns with long gaps between eating. The study also reported that young women's social circumstances and living arrangements had an impact on food availability and food choices, with those who lived alone being least likely to cook for themselves. The study also identified that young women's understanding of the impact of nutrition on their baby was key to making changes. Small changes in diet during pregnancy were reported by some young women, some healthier changes and some for other reasons such as cravings or aversions to certain foods or beliefs about food safety. Young women in this study also reported confusion over the role of nutritional supplements in pregnancy. Despite this, most reported taking some sort of supplements at some point in their pregnancy but reported poor adherence, often simply because they forgot to take them. The results showing poorer quality diet and lower adherence to supplements, accentuated by lower socio-economic status are consistent across both the previously published work and the research presented in this thesis. Due to the close relationship between these papers and the work presented in this thesis the published manuscripts are available in full in appendices 10 and 11.

While the work presented in this thesis uses a mixture of established methods in order to evaluate nutritional issues in adolescent pregnancies it is also accepted that accurate dietary assessment of populations is incredibly difficult to achieve, particularly in marginalised and hard to reach populations such as this. There is a significant risk of underestimation of portion sizes in dietary recall methods, particularly among adolescents¹²⁹. As discussed in chapter 2, assessment of under reporting in the national diet and nutrition survey using the doubly labelled water technique revealed a significant level of under reporting of energy intake across the sample^{77,78}. A systematic review which also included studies using the doubly labelled water method of assessing energy intake to assess the validity of dietary assessment methods in children¹³⁰ found that the accuracy of data collection methods altered by the age of children and adolescents and concluded that for young people aged 16-18 diet history taking provided a better estimate than other methods.

While acknowledging that dietary assessment in pregnant adolescents is difficult, particularly in terms of collecting data which is accurate on an individual level, the use of dietary assessment can give important insights into the nutritional status of populations. As reported in the systematic review included in chapter 3 of this thesis, a number of studies have successfully used 24hr dietary recalls^{131, 133,134,135, 136} or food diaries^{137,138, 139} to assess dietary intake in this population. In addition to this, assessment of nutritional biomarker status also provides important information in terms of assessing the risk of harm due to nutrient deficiencies^{61,140,141, 142,143, 144,145,146,147}.

It has been shown that improving nutritional status in adolescent women during pregnancy is likely to be beneficial to both the young woman and her child. One review looking at interventions to improve diet and weight gain among pregnant adolescents¹⁴⁸ found that positive effects on birth outcomes were evident in interventions which focused on multidisciplinary teams who supported the psychosocial needs of young women including the provision of individual education and counselling. These findings are consistent with the results of this program of research suggesting that interventions which are multifaceted and consider young women's lives in context are most likely to be effective.

Increasing micronutrient intake by supplementation is also a useful method of improving nutritional status in populations. A further review focusing on the impact of nutritional interventions in adolescent pregnancies¹⁴⁹ found that micronutrient supplementation had a positive effect on mean birthweight and the rates of low birthweight and preterm births. There is a lack of research regarding barriers to supplementation in pregnant adolescents and the efficacy of interventions to improve adherence. As discussed previously the results of the study by Rundle et.al¹⁰⁹ show that limited access to appropriate supplements and not remembering to take them were reasons given for low adherence, suggesting these may be avenues for further research into potential interventions. In the UK women who are pregnant and under 18 or in receipt of certain benefits can claim Healthy Start vouchers, which can be exchanged for fruit and vegetables, and vitamins to help support their nutrition. An evaluation¹⁵⁰ of this scheme has found that women reported that receiving health start vouchers improved the quality and diversity of fruit and vegetables consumed and they believed their diet quality to be improved. There were however issues with access to the scheme including complex eligibility criteria and low levels of awareness among families and practitioners. These issues were echoed in a survey of pregnant adolescents who reported that access to healthy start vitamins was patchy and ease of access was dependent on geographical area¹⁵¹. Systems which offer reminders for behaviours such as supplement use may also be useful in improving adherence in this population. Text messaging reminder services to mobile phones have been shown to be effective in paediatric and adolescent populations¹⁵² for improving healthy behaviours across demographic groups, however evidence for more complex mobile based intervention requires further research. There is also some evidence to suggest that the use of mobile technology for nutritional behaviour change interventions may be useful^{153,154}. To date these have largely focused on weight loss interventions so further work to explore the feasibility and efficacy for mobile application based interventions for improving diet quality and adherence to supplement use would be useful.

There is also evidence to suggest that the routine fortification of grain based products with folic acid may have a significant effect on the prevalence of neural tube defects. A review of data from 19 population-based birth defects surveillance programs since the introduction of mandatory folic acid fortification in the USA¹⁵⁵ concluded that this is an effective public health strategy in the prevention of neural tube defects. At the time of writing, ministers in the UK have very recently approved plans to begin mandatory fortification of flour, with a public consultation on the proposals due to launch in early 2019¹⁵⁶.

9.2.5 Supporting adolescent pregnancies

It has been established both in the findings of this program of research and the relevant research literature that the young women most vulnerable to adolescent pregnancy are also at higher risk of other issues which affect their health and wellbeing such as poverty, unstable living arrangements and poor mental health. The research findings also highlighted a concern that these young women may also be struggling to access appropriate support to improve their situation either due to lack of knowledge of services and how to access them or because services are over-subscribed or simply don't exist. Many teenage pregnancy services are focused on prevention, with clinical services encouraging long-acting reversible contraceptives, as these have been shown to be significantly associated with a reduction in conceptions¹⁵⁷. Less is available to support young women once they become pregnant, particularly since the decommissioning of family nurse partnerships (FNP) in many areas across the country. Under the FNP services first-time young parents (aged 24 and under) would receive personal support from a dedicated family nurse who would visit regularly from early pregnancy until their child's second birthday. Many local authorities have however decommissioned this service and the offer of support for young parents varies significantly depending on geographical location. In many areas, at the time of writing, there is no specialist support for pregnant teenagers or young parents, neither is there any additional support to facilitate access to universal midwifery and health visiting services.

As identified in this study, adolescent women on average book with a midwife later than older women meaning that they are less likely to be supported well through the first trimester. These findings are consistent with evidence from other parts of the world in that pregnant adolescents often receive a lower level of prenatal care which is relative to local standards. In a report by the World Health Organisation it is noted that evidence from the USA suggests that among unmarried adolescents 16.4% receive inadequate care compared to only 2% of "non-poor", married, white, adult women¹⁵⁸. The pattern of care is also similar in developing countries, which is a particular concern where the national standard of prenatal care is already insufficient¹⁵⁹. Cost of care¹⁶⁰ and being unmarried were reported as significant barriers to care in this context. This is particularly important as there is robust evidence presented by a Cochrane systematic review¹⁶¹ to show midwife-led continuity models of care have beneficial effects in improving clinical outcomes as well as enhancing women's experiences. The most significant and relevant findings from this review is that the risk of preterm birth and foetal/neonatal mortality was lower in mothers who received midwife-led continuity care models compared to other (non-midwife led) models of care. While the benefits of this specific intervention have not been investigated in this particular group of women, given the observed higher risks of extremely preterm birth for adolescent pregnancies in this cohort, further investigation of the benefits of applying the principles identified in this review including building trusting relationships and providing compassionate care with full respect and dignity in this population would be advantageous.

In addition to the advantages of midwife-led continuity of care there is also evidence to suggest that lay health workers may have a role to play in improving social support and ultimately improving outcomes for marginalised groups. A randomised control trial¹⁶² comparing the addition of pregnancy outreach workers to usual maternity care in disadvantaged women in the West Midlands found that women with two or more risk factors saw improvements in mean Edinburgh post-natal depression scores and that mother-to-infant bonding was improved among all women receiving the intervention. Further work arising from this trial has sought to theorise lay health worker interventions and presents the concept of 'synthetic social support'¹⁶³. This is support which is provided by either paid workers or peer or volunteer supporters and encompasses emotional, appraisal, instrumental and informational support. This differs from other types of social support in that the support provided is not reciprocal and is time limited, meaning that some of the benefits of social support previously identified may not apply. This said, this work provides an important framework for future investigation into the role of youth support professionals in supporting adolescent pregnancies, given that the value of the support offered by pregnancy support workers to women in other marginalised groups has already been evidenced.

9.3 Contributions of this research

There are a number of ways in which this program of research makes positive contributions to the field of study. Firstly, this research provides a unique analysis of an important existing cohort therefore increasing the breadth of what is known about pregnancy health in the UK. The Born in Bradford cohort has produced a wide variety of important and informative research; however the data had not previously been examined for differences due to maternal age. This is also the first time that the maternal food frequency questionnaire data has been used to investigate dietary patterns.

The identification of significant differences in extremely pre-term and extremely low birthweight babies predicted by maternal adolescent age is also an important and novel finding for a relatively recent, UK based study.

Previous to this program of research there was a severe lack of evidence regarding the nature of adolescent women's diet during pregnancy; this study has provided both a systematic review and an original research study in order to begin to bridge this gap.

In addition to the new quantitative analysis, the qualitative phase of this study sought to gain the perspectives of a professional group who have not previously been consulted on this topic. As previously discussed, there is existing work documenting the views of young women and health care professionals regarding supporting healthy eating in pregnancy, however professionals whose role is to provide a wider range of support to young people encompassing psychosocial support and practical assistance have been overlooked. Taking into account the perspectives of these professionals is important as they often have enduring relationships with young people which last beyond the pregnancy period, and are in a unique position to be able to offer support.

9.4 Strengths and limitations

A significant strength of this program of research was the ability to utilise a large and diverse existing cohort to create new knowledge. The Born in Bradford cohort is unique in terms of its diverse population, meaning it presents the opportunity for new research which would not be possible elsewhere. The large number of participants made for a robust analysis and made it possible to investigate relatively rare events; however the study was limited by very low numbers of younger adolescents (aged under 16 in particular) and low occurrences of some rarer outcomes such as stillbirth. There were also a relatively small number of adolescent south Asian women in the cohort, meaning that the potential for examining the impact of ethnicity on outcomes was limited.

It should also be considered that the sample for this study consisted only of women who were engaged with services during the second trimester, meaning that those young women accessing services much later in their pregnancy are not included in the cohort. Adolescent age was identified as a key determinant of late presentation to antenatal care and poor foetal outcomes in a retrospective study of almost 60,000 records in a large UK based maternity hospital¹⁶⁴, suggesting that this cohort may not include some of the most hard to reach young women.

The availability of data from a food frequency questionnaire was also a major strength of this study as it enabled investigation of the types of food eaten in the assessment of dietary patterns, which is more useful than assessment of nutrients in terms of real world applicability. This said the food frequency questionnaire included in the Born in Bradford baseline questionnaire is limited by brevity and inability to characterise whole diets. This is due to the original intention of the FFQ being to categorise women according to their intakes of a small range of agents with potential genotoxic and/or immunotoxic properties, including polycyclic aromatic hydrocarbons, heterocyclic amines, nitrosamines,

109

acrylamide, the mycotoxin deoxynivalenol, dioxin and polychlorinated biphenyls, alcohol and DNA-reactive aldehydesotential¹⁶⁵. This therefore means that more robust analysis may have been possible had a questionnaire been developed specifically for this study with the intention of characterising the whole diets of pregnant women. This was a pragmatic decision to investigate the limits of the existing available data and provide a baseline for future developments. The qualitative data in this program of research was collected from professionals with a wide variety of roles; however all had the common feature of providing social, emotional and practical support to young women. This is a strength of the study in that this strategy allowed for a wide range of perspectives to be heard. The study was however limited by a small overall number of participants, meaning that further insights may be gained by additional work with professionals in similar roles.

9.5 Recommendations for future work

9.5.1 Research Recommendations

This study has highlighted a number of important avenues for the development of future research. Mental health and social factors have emerged as an important theme impacting on the health and wellbeing of pregnant adolescents. Variables relating to these topics have yet to be evaluated in the Born in Bradford cohort. A project to evaluate differences in indicators of social and emotional wellbeing in the Born in Bradford cohort and their relationship to maternal age and maternal and neonatal outcomes would be useful to further explore their implications for the health of adolescent mothers and their babies. In addition to more detailed evaluation of differences between adolescent and adult women in the Born in Bradford cohort this study has identified the need for a more in depth evaluation of young women's diets during pregnancy in the UK compared to that of adult women. Research based on a comprehensive assessment of young women's diets during pregnancy would provide a more solid framework for the development of interventions to support young women to make changes.

Further work is also needed with this population to investigate young women's perspectives on what would help them to have healthier pregnancies more

holistically and to consult them on their views of the themes emerging from the data provided by health and social care professionals.

While this program of research has focused on adolescent women there is a significant body of evidence to suggest that maternal and neonatal outcomes among older women (aged 35 years or over) may also be of concern^{166,167,167,168,168}. Older women in the Born in Bradford cohort have not previously been the focus of any investigation into pregnancy and birth outcomes meaning this may be an important area for further research.

9.5.2 Practice recommendations

As previously discussed, recent work has been carried out with young women during or after pregnancy to ascertain their views on healthy eating and supplement use. Young women reported that they value the support of midwives and generally trust their advice, particularly on health related topics. Investigation of the feasibility of brief interventions which could be successfully delivered to adolescent pregnant women within existing midwifery appointments would help to capitalise on this relationship and maximise the effectiveness of the support young women already have during their pregnancies. Additional training for midwives both as continuing professional development and as part of midwifery education addressing the specific nutritional issues associated with adolescent pregnancies would be useful in giving midwives the tools to provide support in time pressured consultations.

Consultation regarding or, ideally, co-production of any interventions to improve pregnancy health for young women is essential in order to ensure that any interventions are useful and relevant to the population they aim to support. It is also important that those young women who may be particularly vulnerable within this population, for example due to very young maternal age, poverty and homelessness or mental health conditions, are considered specifically in the development of future work to ensure that those young women with the highest need are central to the development of interventions.

9.6 Summary

This section has discussed the key findings of this program of research and situated them within the existing academic literature. It has also considered the particular strengths and limitations of this study and highlighted the original contributions to knowledge made by this work. Recommendations for further research to build upon the findings have also been suggested.

Chapter 10 - Conclusions

10.1 Introduction

This thesis provides details of a program of mixed methods research presented in an article based format consisting of three papers published in reputable peer reviewed scientific journals and a further paper which is currently under review. In this chapter, the aims and objectives are reviewed and the contribution and impact of this work discussed.

10.2 Review of aims and objectives

The primary aim of this work was to answer the research question "*Is there a need for additional nutritional support in adolescent pregnancies?*". This question has been addressed through investigation of three research objectives:

Establish what is known in the current research literature regarding the nutritional status of pregnant adolescents

This objective was met through a systematic review of the available research literature presented in chapter 3. The review concluded that the nutritional status of adolescents during pregnancy is likely to be sub-optimal, particularly with reference to a number of key micronutrients.

Examine differences in dietary intake and birth outcomes between adolescent and adult mothers in a large cohort data set

This objective was addressed through two separate analyses of the Born in Bradford cohort data set. The first, presented in chapter 5, looks at differences in maternal and neonatal outcomes by maternal age. This study concluded that adolescent mothers had higher odds of delivering babies who were extremely preterm and/or of extremely low birthweight. This study also found that adolescent mothers had lower odds of instrumental deliveries, caesarean section and gestational diabetes. The second analysis of the Born in Bradford data is presented in chapter 6 and uses principal component analysis alongside assessment of questionnaire responses to consider differences in dietary pattern by maternal age. The study concludes that the adolescent women in the cohort had a poorer quality diet with lower intake of fruit and vegetables, lower intake of nutritional supplements and higher intake of sugar sweetened cola compared to adult women. *Explore the perspective of health and social care professionals of providing nutritional support for pregnant adolescents*

This objective was met through qualitative interviews with professionals, which were analysed thematically. This study, presented in chapter 7 in the format submitted for review, concluded that in order to improve the nutritional status of young women during pregnancy it is necessary to consider the impact of psychosocial factors such as access to services, social support, poverty and homelessness and mental health difficulties. Consistent, holistic support delivered by networks of professionals is needed in order to support young women to make changes that benefit their own health and that of their babies.

10.3 Impact to date

This program of work has produced three research papers which have been published in peer reviewed journals. At the time of printing this thesis on 20.01.19 the metrics for theses research outputs were:

Marvin-Dowle, K., Burley, V. J., & Soltani, H. (2016). Nutrient intakes and nutritional biomarkers in pregnant adolescents: a systematic review of studies in developed countries. *BMC pregnancy and childbirth*, *16*(1), 268.

Accessed 2,109 times, cited 6 times

Marvin-Dowle, K., Kilner, K., Burley, V. J., & Soltani, H. (2018). Impact of adolescent age on maternal and neonatal outcomes in the Born in Bradford cohort. *BMJ open*, *8*(3), e016258.

Accessed 1220 times, cited 3 times

Marvin-Dowle, K., Kilner, K., Burley, V., & Soltani, H. (2018). Differences in dietary pattern by maternal age in the Born in Bradford cohort: A comparative analysis. *PloS one*, *13*(12), e0208879.

Accessed 109 times

10.4 Summary

This thesis has presented a comprehensive program of mixed methods research which has produced an original contribution to knowledge in the fields of maternal and neonatal health, nutrition and adolescent health and wellbeing. The research has also identified a number of areas where additional research would be likely to further enhance understanding of the issues affecting young women's health and wellbeing and ultimately inform interventions to improve the health of young mothers and their children.

References

1. Cook SMC, Cameron ST. Social issues of teenage pregnancy. *Obstetrics, Gynaecology & Reproductive Medicine*. 2015;doi:10.1016/j.ogrm.2015.06.001.

2. Gibbs CM, Wendt A, Peters S, Hogue CJ. The impact of early age at first childbirth on maternal and infant health. *Paediatr Perinat Epidemiol*. 2012;26:259-284.

3. Burke BS, Harding VV, Stuart HC. Nutrition studies during pregnancy: IV. relation of protein content of mother's diet during pregnancy to birth length, birth weight, and condition of infant at birth. *J Pediatr*. 1943;23(5):506-515.

4. Roberts C, Steer T, Maplethorpe N, et al. National diet and nutrition survey: Results from years 7 and 8 (combined) of the rolling programme (2014/2015– 2015/2016). . 2018.

5. Scholl TO. Hediger ML. Weight gain, nutrition, and pregnancy outcome: Findings from the camden study of teenage and minority gravidas. *Semin Perinatol.* June 1995;19(3):171-181.

6. Jones RL, Cederberg HM, Wheeler SJ, et al. Relationship between maternal growth, infant birthweight and nutrient partitioning in teenage pregnancies. *BJOG*. 2010;117(2):200-211.

7. World Bank. Adolescent fertility rate.

https://data.worldbank.org/indicator/SP.ADO.TFRT. Updated 2018.

8. Eurostat. Fertility rates by age. <u>http://ec.europa.eu/eurostat/web/products-</u> <u>datasets/-/demo_frate</u>. Updated 2018. 9. Blum RW, Gates Sr W. Girlhood not motherhood. preventing adolescent pregnancy. . 2015.

10. New Policy Institute. Poverty among young people in the uk. . 2015.

11. Evans J, Slowley M. Not the end of the story: Supporting teenage mothers back into education. *Barnados: London*. 2010.

12. Reid V, Meadows-Oliver M. Postpartum depression in adolescent mothers: An integrative review of the literature. *J Pediatr Health Care*. 2007;21(5):289-298.

13. Office for National Statistics. Conception statistics, england and wales. <u>https://www.ons.gov.uk/peoplepopulationandcommunity/birthsdeathsandmarria</u> <u>ges/conceptionandfertilityrates/datasets/conceptionstatisticsenglandandwalesre</u> <u>ferencetables</u>. Updated 2018.

14. Wellings K, Jones KG, Mercer CH, et al. The prevalence of unplanned pregnancy and associated factors in britain: Findings from the third national survey of sexual attitudes and lifestyles (natsal-3). *The Lancet*. 2013;382(9907):1807-1816.

15. Social Exclusion Unit. Teenage pregnancy : Report by the social exclusion unit presented to parliament by the prime minister by command of her majesty. . 1999.

16. Hadley A, Ingham R, Chandra-Mouli V. Implementing the united kingdom's ten-year teenage pregnancy strategy for england (1999-2010): How was this done and what did it achieve? *Reproductive health*. 2016;13(1):139.

17. Unit TP. Teenage pregnancy strategy evaluation: Final report synthesis. *London: TPU*. 2005.

18. Saugstad OD, Aune D. Optimal oxygenation of extremely low birth weight infants: A meta-analysis and systematic review of the oxygen saturation target studies. *Neonatology*. 2014;105(1):55-63.

19. Tommy's. Premature birth statistics. <u>https://www.tommys.org/our-</u>organisation/why-we-exist/premature-birth-statistics. Updated 2017.

20. Aarnoudse-Moens CS, Weisglas-Kuperus N, van Goudoever JB, Oosterlaan J. Meta-analysis of neurobehavioral outcomes in very preterm and/or very low birth weight children. *Pediatrics*. 2009;124(2):717-728.

21. Valdez R, Athens M, Thompson G, Bradshaw B, Stern M. Birthweight and adult health outcomes in a biethnic population in the USA. *Diabetologia*. 1994;37(6):624-631.

22. Kawakita T, Wilson K, Grantz KL, Landy HJ, Huang C, Gomez-Lobo V. Adverse maternal and neonatal outcomes in adolescent pregnancy. *J Pediatr Adolesc Gynecol.* 2016;29(2):130-136.

23. Shrim A, Ates S, Mallozzi A, et al. Is young maternal age really a risk factor for adverse pregnancy outcome in a canadian tertiary referral hospital? *J Pediatr Adolesc Gynecol.* 2011;24(4):218-222.

24. Socolov D, lorga M, Carauleanu A, et al. Pregnancy during adolescence and associated risks: An 8-year hospital-based cohort study (2007–2014) in romania, the country with the highest rate of teenage pregnancy in europe. *BIOMED RES INT*. 2017:1-8. 25. Tempfer C, Schmid M, Bentz E, Huber J, Keck C, Hefler L. Outcome of delivery in 381 teenagers: A matched pair study. *Geburtshilfe Frauenheilkd*. 2007;67(09):978-981.

26. Beeckman K, van De Putte S, Putman K, Louckx F. Predictive social factors in relation to preterm birth in a metropolitan region. *Acta Obstet Gynecol Scand*. 2009;88(7):787-792.

27. Blomberg M, Birch Tyrberg R, Kjølhede P. Impact of maternal age on obstetric and neonatal outcome with emphasis on primiparous adolescents and older women: A swedish medical birth register study. *BMJ Open*. 2014;4(11):e005840-e005840.

28. El-Gilany A, Hammad S. Obstetric outcomes of teenagers and older mothers: Experience from saudi arabia. *International Journal of Collaborative Research on Internal Medicine & Public Health*. 2012;4(6):901.

29. Eure CR. Lindsay MK. Graves WL. Risk of adverse pregnancy outcomes in young adolescent parturients in an inner-city hospital. *Obstet Gynecol*. May 2002;186(5):918-920.

30. World Health Organisation. Maternal and perinatal health.

http://www.who.int/maternal_child_adolescent/topics/maternal/maternal_perinat al/en/. Updated 2018.

 Lewis LN, Hickey M, Doherty DA, Skinner SR. How do pregnancy outcomes differ in teenage mothers? A western australian study. *Med J Aust.* 2009;190(10):537-541. 32. Mohsin M, Bauman A, Jalaludin B. The influence of antenatal and maternal factors on stillbirths and neonatal deaths in new south wales, australia. *J Biosoc Sci*. 2006;38(05):643-657.

33. Wilson RE, Alio AP, Kirby RS, Salihu HM. Young maternal age and risk of intrapartum stillbirth. *Arch Gynecol Obstet*. 2008;278(3):231-236.

34. DuPlessis HM, Bell R, Richards T. Adolescent pregnancy: Understanding the impact of age and race on outcomes. *Journal of Adolescent health*. 1997;20(3):187-197.

35. Jivraj S. Nazzal Z. Davies P. Obstetric outcome of teenage pregnancies from 2002 to 2008: The sheffield experience. *Journal of Obstetrics and Gynaecology*. April 2010;30(3):253-256.

36. Otterblad Olausson PM. Cnattingius S. Goldenberg RL. Determinants of poor pregnancy outcomes among teenagers in sweden. *Obstet Gynecol*. March 1997;89(3):451-457.

37. Bateman BT, Simpson LL. Higher rate of stillbirth at the extremes of reproductive age: A large nationwide sample of deliveries in the united states. *Obstet Gynecol.* 2006;194(3):840-845.

38. Flenady V, Koopmans L, Middleton P, et al. Major risk factors for stillbirth in high-income countries: A systematic review and meta-analysis. *The lancet*. 2011;377(9774):1331-1340.

39. Chen C, Tsai C, Sung F, et al. Adverse birth outcomes among pregnancies of teen mothers: Age-specific analysis of national data in taiwan. *Child Care Health Dev.* 2010;36(2):232-240.

40. Apgar V. A proposal for a new method of evaluation of the newborn infant. *Anesthesia & Analgesia*. 2015;120(5):1056-1059.

41. Tyrberg RB, Blomberg M, Kjølhede P. Deliveries among teenage women– with emphasis on incidence and mode of delivery: A swedish national survey from 1973 to 2010. *BMC Pregnancy and childbirth*. 2013;13(1):204.

42. BAI, F. WONG, HELEN STEWART, JUN. The obstetric and neonatal performance of teenage mothers in an australian community. *Journal of Obstetrics and Gynaecology*. 1999;19(4):345-348.

43. O'leary CM, Bower C, Knuiman M, Stanley FJ. Changing risks of stillbirth and neonatal mortality associated with maternal age in western australia 1984– 2003. *Paediatr Perinat Epidemiol*. 2007;21(6):541-549.

44. Fayed AA, Wahabi H, Mamdouh H, Kotb R, Esmaeil S. Demographic profile and pregnancy outcomes of adolescents and older mothers in saudi arabia: Analysis from riyadh mother (RAHMA) and baby cohort study. *BMJ Open*. 2017;7(9):e016501-e016501.

45. Fleming N, Ng N, Osborne C, et al. Adolescent pregnancy outcomes in the province of ontario: A cohort study. *J Obstet Gynaecol Can*. 2013;35(3):234-245.

46. Knight M, Nair M, Tuffnell D, et al. On behalf of MBRRACE-UK.(2016) saving lives, improving mothers' Care-Surveillance of maternal deaths in the UK 2012-14 and lessons learned to inform maternity care from the UK and ireland confidential enquiries into maternal deaths and morbidity 2009-14. national

perinatal epidemiology unit, university of oxford. *National Perinatal Epidemiology Unit, University of Oxford, Oxford*. 2017.

47. Milne F, Redman C, Walker J, et al. The pre-eclampsia community guideline (PRECOG): How to screen for and detect onset of pre-eclampsia in the community. *BMJ*. 2005;330(7491):576-580.

48. Duley L. The global impact of pre-eclampsia and eclampsia. . 2009;33(3):130-137.

49. Bellamy L, Casas JP, Hingorani AD, Williams DJ. Pre-eclampsia and risk of cardiovascular disease and cancer in later life: Systematic review and metaanalysis. *BMJ*. 2007;335(7627):974.

50. de Vienne CM. Creveuil C. Dreyfus,M. Does young maternal age increase the risk of adverse obstetric, fetal and neonatal outcomes: A cohort study. *European Journal of Obstetrics and Gynecology and Reproductive Biology*. December 2009;147(2):151-156.

51. Crowther CA, Hiller JE, Moss JR, McPhee AJ, Jeffries WS, Robinson JS. Effect of treatment of gestational diabetes mellitus on pregnancy outcomes. *N Engl J Med.* 2005;352(24):2477-2486.

52. Collier A, Abraham EC, Armstrong J, Godwin J, Monteath K, Lindsay R. Reported prevalence of gestational diabetes in scotland: The relationship with obesity, age, socioeconomic status, smoking and macrosomia, and how many are we missing? *Journal of diabetes investigation*. 2017;8(2):161-167.

53. Gupta N. Kiran U. Bhal K. Teenage pregnancies: Obstetric characteristics and outcome. *European Journal of Obstetrics and Gynecology and Reproductive Biology*. April 2008;137(2):165-171.

54. van Ham MA, van Dongen PW, Mulder J. Maternal consequences of caesarean section. A retrospective study of intra-operative and postoperative maternal complications of caesarean section during a 10-year period. *Eur J Obstet Gynecol Reprod Biol.* 1997;74(1):1-6.

55. Lydon-Rochelle M, Holt VL, Martin DP, Easterling TR. Association between method of delivery and maternal rehospitalization. *JAMA*. 2000;283(18):2411-2416.

56. King JC. Physiology of pregnancy and nutrient metabolism–. *Am J Clin Nutr*. 2000;71(5):1218S-1225S.

57. Derbyshire E. Nutrition in the childbearing years. John Wiley & Sons; 2011.

 58. MRC Vitamin Study Research Group. Prevention of neural tube defects: Results of the medical research council vitamin study. *The lancet*.
 1991;338(8760):131-137.

59. Zachman RD. Role of vitamin A in lung development. *J Nutr*. 1995;125(suppl_6):1634S-1638S.

60. Al-Gubory KH, Fowler PA, Garrel C. The roles of cellular reactive oxygen species, oxidative stress and antioxidants in pregnancy outcomes. *Int J Biochem Cell Biol.* 2010;42(10):1634-1650.

61. Young BE, McNanley TJ, Cooper EM, et al. Maternal vitamin D status and calcium intake interact to affect fetal skeletal growth in utero in pregnant adolescents. *Am J Clin Nutr*. 2012;95(5):1103-1112.

62. Bucher HC, Guyatt GH, Cook RJ, et al. Effect of calcium supplementation on pregnancy-induced hypertension and preeclampsia: A meta-analysis of randomized controlled trials. *JAMA*. 1996;275(14):1113-1117.

63. Hypponen E, Cavadino A, Williams D, et al. Vitamin D and pre-eclampsia:
Original data, systematic review and meta-analysis. *Ann Nutr Metab*.
2013;63(4):331-340.

64. Poel Y, Hummel P, Lips P, Stam F, Van Der Ploeg T, Simsek S. Vitamin D and gestational diabetes: A systematic review and meta-analysis. *Eur J Intern Med*. 2012;23(5):465-469.

65. Allen LH. Anemia and iron deficiency: Effects on pregnancy outcome–. *Am J Clin Nutr.* 2000;71(5):1280S-1284S.

66. Barker DJ. Fetal origins of coronary heart disease. *BMJ*.1995;311(6998):171-174.

67. Barker DJ, Osmond C, Winter P, Margetts B, Simmonds SJ. Weight in infancy and death from ischaemic heart disease. *The Lancet*. 1989;334(8663):577-580.

Barker DJ. Fetal origins of cardiovascular disease. *Ann Med.* 1999;31(sup1):3-6.

69. Roseboom TJ, van der Meulen JH, Osmond C, et al. Coronary heart disease after prenatal exposure to the dutch famine, 1944-45. *Heart*. 2000;84(6):595-598.

70. Inskip HM, Godfrey KM, Robinson SM, Law CM, Barker DJ, Cooper C.
 Cohort profile: The southampton women's survey. *Int J Epidemiol.* 2005;35(1):42-48.

71. Haugen G, Hanson M, Kiserud T, Crozier S, Inskip H, Godfrey KM. Fetal liver-sparing cardiovascular adaptations linked to mother's slimness and diet. *Circ Res.* 2005;96(1):12-14.

72. Department of Health. Dietary reference values (DRVs) for food, energy and nutrients for the UK, report of the pannel on DVRs of the committee on medical aspects of food policy (COMA). 1991;41.

73. Elliott-Sale KJ, Graham A, Hanley SJ, Blumenthal S, Sale C. Modern dietary guidelines for healthy pregnancy; maximising maternal and foetal outcomes and limiting excessive gestational weight gain. *European journal of sport science*. 2018:1-9.

74. Scientific Advisory Committee for Nutrition. Dietary reference values for energy. . 2011.

75. Scientific Advisory Committee on Nutrition. Carbohydrates and health report. . 2015.

76. Murphy SP, Guenther PM, Kretsch MJ. Using the dietary reference intakes to assess intakes of groups: Pitfalls to avoid. *Journal of the American Dietetic Association*. 2006;106(10):1550-1553.

77. Barrie A, Coward W. A rapid analytical technique for the determination of energy expenditure by the doubly labelled water method. *Biol Mass Spectrom*. 1985;12(9):535-541.

 Bluck L. Doubly labelled water for the measurement of total energy expenditure in man–progress and applications in the last decade. *Nutr Bull*. 2008;33(2):80-90.

79. Archer E, Hand GA, Blair SN. Validity of US nutritional surveillance: National health and nutrition examination survey caloric energy intake data, 1971–2010. *PloS one*. 2013;8(10):e76632.

80. World Health Organization. Serum and red blood cell folate concentrations for assessing folate status in populations. . 2015.

81. Scientific Advisory Committee on Nutrition. SACN vitamin D and health report. . 2016.

82. WHO/UNICEF/ICCIDD. Assessment of iodine deficiency disorders and monitoring their elimination. *A guide for programme managers*. 2007;3:98.

83. Conolly A, Saunders C, Social N, Neave A. Health survey for england 2016: Adult overweight and obesity. *Health and Social Care Information Centre*. 2017.

84. Naeye RL. Teenaged and pre-teenaged pregnancies: Consequences of the fetal-maternal competition for nutrients. *Pediatrics*. 1981;67(1):146-150.

85. Frisancho AR, Matos J, Flegel P. Maternal nutritional status and adolescent pregnancy outcome. *Am J Clin Nutr*. 1983;38(5):739-746.

86. Scholl TO, Stein TP, Smith WK. Leptin and maternal growth during adolescent pregnancy–. *Am J Clin Nutr*. 2000;72(6):1542-1547.

87. Rosso P. Nutrition and maternal-fetal exchange. Am J Clin Nutr. 1981.

88. Berg BN. Dietary restriction and reproduction in the rat. *J Nutr*. 1965;87(3):344-348.

89. Nitzsche D. Effect of maternal feed restriction on prenatal development in rats and rabbits–A review of published data. *Regulatory Toxicology and Pharmacology*. 2017;90:95-103.

90. Hall Moran V. A systematic review of dietary assessments of pregnant adolescents in industrialised countries. *British journal of nutrition*. 2007;97(03):411-425.

91. Moran VH. Nutritional status in pregnant adolescents: A systematic review of biochemical markers. *Matern Child Nutr*. 2007;3(2):74-93.

92. Kuhn TS. The structure of scientific revolutions. . 1962.

93. Guba E. The alternative paradigm dialog. in. EG guba (ed.) the paradigm dialogue (p. 17-27). . 1990.

94. Yvonne Feilzer M. Doing mixed methods research pragmatically: Implications for the rediscovery of pragmatism as a research paradigm. *Journal of mixed methods research*. 2010;4(1):6-16.

95. Schwandt TA. Constructivist, interpretivist approaches to human inquiry. *Handbook of qualitative research*. 1994;1:118-137.

96. James W. *Pragmatism.* Vol 1. Harvard University Press; 1975. 127 97. Denzin NK. Post-pragmatism. Symbolic Interaction. 1996;19(1):61-75.

98. Dewey J. The theory of inquiry. New York: Holt, Rinehart & Wiston. 1938.

99. Tashakkori A, Teddlie C, Teddlie CB. *Mixed methodology: Combining qualitative and quantitative approaches.* Vol 46. Sage; 1998.

100. Plano Clark VL, Creswell JW. The mixed methods reader. . 2008.

101. O'Cathain A, Murphy E, Nicholl J. Three techniques for integrating data in mixed methods studies. *BMJ*. 2010;341:c4587.

102. Creswell JW, Plano Clark VL, Gutmann ML, Hanson WE. Advanced mixed methods research designs. *Handbook of mixed methods in social and behavioral research*. 2003;209:240.

103. Wright J, Small N, Raynor P, et al. Cohort profile: The born in bradford multi-ethnic family cohort study. *Int J Epidemiol*. 2012;42(4):978-991.

104. Fields A. Discovering statistics using IBM SPSS statistics. *Thousand Oaks, CA*. 2013.

105. Hoenig JM, Heisey DM. The abuse of power: The pervasive fallacy of power calculations for data analysis. *The American Statistician*. 2001;55(1):19-24.

106. Kaiser HF. An index of factorial simplicity. *Psychometrika*. 1974;39(1):31-36.

107. Howitt D.

INTRODUCTION TO

QUALITATIVE METHODS

IN PSYCHOLOGY 2nd ed. Essex, UK: Pearson education Limited; 2010.

108. Fielding NG, Fielding JL. Resistance and adaptation to criminal identity: Using secondary analysis to evaluate classic studies of crime and deviance. *Sociology*. 2000;34(4):671-689.

109. Rundle R, Soltani H, Duxbury A. Exploring the views of young women and their healthcare professionals on dietary habits and supplementation practices in adolescent pregnancy: A qualitative study. *BMC Nutrition*. 2018;4(1):45.

110. Public Health England. Public health outcomes framework - under 18 conceptions. <u>https://fingertips.phe.org.uk/profile/public-health-outcomes-framework/data#page/7/gid/1000042/pat/6/par/E12000004/ati/102/are/E060000</u> <u>15/iid/20401/age/173/sex/2</u>. Updated 2018.

111. McCall SJ, Bhattacharya S, Okpo E, Macfarlane GJ. Evaluating the social determinants of teenage pregnancy: A temporal analysis using a UK obstetric database from 1950 to 2010. *J Epidemiol Community Health*. 2015;69(1):49-54.

112. Kramer MS, Seguin L, Lydon J, Goulet L. Socio-economic disparities in pregnancy outcome: Why do the poor fare so poorly? *Paediatr Perinat Epidemiol.* 2000;14(3):194-210.

113. Smith GC, Pell JP. Teenage pregnancy and risk of adverse perinatal outcomes associated with first and second births: Population based retrospective cohort study. *BMJ*. 2001;323(7311):476.

114. Scholl TO, Hediger ML, Belsky DH. Prenatal care and maternal health during adolescent pregnancy: A review and meta-analysis. *Journal of Adolescent Health*. 1994;15(6):444-456.

115. Joseph NP, Hunkali KB, Wilson B, Morgan E, Cross M, Freund KM. Prepregnancy body mass index among pregnant adolescents: Gestational weight gain and long-term post partum weight retention. *J Pediatr Adolesc Gynecol.* 2008;21(4):195-200.

116. American College of Obstetricians and Gynecologists. Weight gain during pregnancy. committee opinion no. 548. *Obstet Gynecol.* 2013;121(1):210-212.

117. Goldstein RF, Abell SK, Ranasinha S, et al. Association of gestational weight gain with maternal and infant outcomes: A systematic review and metaanalysis. *JAMA*. 2017;317(21):2207-2225.

118. Sámano R, Chico-Barba G, Martínez-Rojano H, et al. Pre-pregnancy body mass index classification and gestational weight gain on neonatal outcomes in adolescent mothers: A follow-up study. *PloS one*. 2018;13(7):e0200361.

119. Han Z, Lutsiv O, Mulla S, Rosen A, Beyene J, McDonald SD. Low gestational weight gain and the risk of preterm birth and low birthweight: A systematic review and meta-analyses. *Acta Obstet Gynecol Scand*. 2011;90(9):935-954.

120. Nohr EA, Vaeth M, Baker JL, Sørensen TI, Olsen J, Rasmussen KM. Combined associations of prepregnancy body mass index and gestational weight gain with the outcome of pregnancy–. *Am J Clin Nutr*. 2008;87(6):1750-1759. 121. Heslehurst N, Rankin J, Wilkinson J, Summerbell C. A nationally representative study of maternal obesity in england, UK: Trends in incidence and demographic inequalities in 619 323 births, 1989–2007. *Int J Obes*. 2010;34(3):420.

122. Patel V, Flisher AJ, Hetrick S, McGorry P. Mental health of young people: A global public-health challenge. *The Lancet*. 2007;369(9569):1302-1313.

123. Xue Y, Leventhal T, Brooks-Gunn J, Earls FJ. Neighborhood residence and mental health problems of 5-to 11-year-olds. *Arch Gen Psychiatry*. 2005;62(5):554-563.

124. Hillis SD, Anda RF, Dube SR, Felitti VJ, Marchbanks PA, Marks JS. The association between adverse childhood experiences and adolescent pregnancy, long-term psychosocial consequences, and fetal death. *Pediatrics*. 2004;113(2):320-327.

125. Jacka FN, Rothon C, Taylor S, Berk M, Stansfeld SA. Diet quality and mental health problems in adolescents from east london: A prospective study. *Soc Psychiatry Psychiatr Epidemiol.* 2013;48(8):1297-1306.

126. Bodnar LM, Wisner KL. Nutrition and depression: Implications for improving mental health among childbearing-aged women. *Biol Psychiatry*. 2005;58(9):679-685.

127. Patrick H, Nicklas TA. A review of family and social determinants of children's eating patterns and diet quality. *J Am Coll Nutr*. 2005;24(2):83-92.

128. Mouratidou T, Ford F, Prountzou F, Fraser R. Dietary assessment of a population of pregnant women in sheffield, UK. *Br J Nutr*. 2006;96(5):929-935.
129. Vereecken C, Dohogne S, Covents M, Maes L. How accurate are adolescents in portion-size estimation using the computer tool young adolescents' nutrition assessment on computer (YANA-C)? *Br J Nutr*. 2010;103(12):1844-1850.

130. Burrows TL, Martin RJ, Collins CE. A systematic review of the validity of dietary assessment methods in children when compared with the method of doubly labeled water. *J Am Diet Assoc*. 2010;110(10):1501-1510.

131. Castillo-Duran C, Marin V, Alcazar L, Iturralde H, Ruz M. Controlled trial of zinc supplementation in chilean pregnant adolescents. *Nutr Res*. 2001;21(5):715-724.

132. Baker PN, Wheeler SJ, Sanders TA, et al. A prospective study of micronutrient status in adolescent pregnancy–. *Am J Clin Nutr*. 2009;89(4):1114-1124.

133. Job J, Capra S. Nutritional assessment of pregnant teenagers attending a metropolitan, public, maternity hospital.. *Australian Journal of Nutrition* & *Dietetics*. 1995;52(3):154.

134. Lee MC. Suhng LA. Lu TH. and others. Association of parental characteristics with adverse outcomes of adolescent pregnancy. *Fam Pract*.
1998;15(4):336-342.

135. Pobocik RS, Benavente JC, Boudreau NS, Spore CL. Pregnant adolescents in guam consume diets low in calcium and other micronutrients. *J Am Diet Assoc*. 2003;103(5):611-614.

136. Gutierrez YM. Cultural factors affecting diet and pregnancy outcome of mexican american adolescents. *Journal of Adolescent Health*. 1999;25(3):227-237.

137. Chan GM. McElligott K. McNaught T. Effects of dietary calcium intervention on adolescent mothers and newborns: A randomized controlled trial. *Obstet Gynecol.* September 2006;108(3, part 1):565-571.

138. Derbyshire E. Dietary habits of pregnant and non-pregnant adolescents. *Nutrition and Food Science*. 2009;39(5):490-498.

139. Giddens JB, Krug SK, Tsang RC, Guo S, Miodovnik M, PRADA JA. Pregnant adolescent and adult women have similarly low intakes of selected nutrients. *J Am Diet Assoc*. 2000;100(11):1334-1340.

140. Young M, Pressman E, Foehr M, et al. Impact of maternal and neonatal iron status on placental transferrin receptor expression in pregnant adolescents. *Placenta*. 2010;31(11):1010-1014.

141. Rycel M, Sobala W, Wilczynski J, Nowakowska D. Hemoglobin and hematocrit concentrations influence birth outcome in pregnant polish adolescents. *Archives of Perinatal Medicine*. 2009;15(2):101-105.

142. Mistry HD. Kurlak LO. Young SD. Maternal selenium, copper and zinc concentrations in pregnancy associated with small-for-gestational-age infants. *Maternal and Child Nutrition*. 1 July 2014;10(3):327-334.

143. Meier PR. Nickerson HJ. Olson KA. Prevention of iron deficiency anemia in adolescent and adult pregnancies. *Clinical Medicine Research*. January 2003;1(1):29-36.

144. Davis LM, Chang S, Mancini J, Nathanson MS, Witter FR, O'Brien KO. Vitamin D insufficiency is prevalent among pregnant african american adolescents. *J Pediatr Adolesc Gynecol*. 2010;23(1):45-52.

145. Ginde AA. Sullivan AF. Mansbach JM. Vitamin D insufficiency in pregnant and nonpregnant women of childbearing age in the united states. *Obstet Gynecol*. May 2010;202(5):436-438.

146. Gadowsky SL, Gale K, Wolfe SA, Jory J, Gibson R, O'Connor,D.L. Biochemical folate, B12, and iron status of a group of pregnant adolescents accessed through the public health system in southern ontario. *J Adolesc Health*. 1995;16(6):465-474.

147. Dawson EB, Albers JH, McGanity WJ. The apparent effect of iron supplementation on serum selenium levels in teenage pregnancy. *Biol Trace Elem Res.* 2000;77(3):209-217.

148. Nielsen JN, Gittelsohn J, Anliker J, O'Brien K. Interventions to improve diet and weight gain among pregnant adolescents and recommendations for future research. *J Am Diet Assoc*. 2006;106(11):1825-1840.

149. Salam RA, Hooda M, Das JK, et al. Interventions to improve adolescent nutrition: A systematic review and meta-analysis. *Journal of Adolescent Health*. 2016;59(4):S29-S39.

150. McFadden A, Green JM, Williams V, et al. Can food vouchers improve nutrition and reduce health inequalities in low-income mothers and young children: A multi-method evaluation of the experiences of beneficiaries and practitioners of the healthy start programme in england. *BMC Public Health*. 2014;14(1):148.

151. Soltani H, Duxbury A, Rundle R, Marvin-Dowle K. Dietary habits and supplementation practices of young women during pregnancy: An online cross-sectional survey of young mothers and health care professionals. *BMC Nutrition*. 2017;3(1):19.

152. Militello LK, Kelly SA, Melnyk BM. Systematic review of text-messaging interventions to promote healthy behaviors in pediatric and adolescent populations: Implications for clinical practice and research. *Worldviews on Evidence-Based Nursing*. 2012;9(2):66-77.

153. Bacigalupo R, Cudd P, Littlewood C, Bissell P, Hawley M, Buckley Woods H. Interventions employing mobile technology for overweight and obesity: An early systematic review of randomized controlled trials. *Obesity reviews*. 2013;14(4):279-291.

154. Soltani H, Duxbury AM, Arden MA, Dearden A, Furness PJ, Garland C. Maternal obesity management using mobile technology: A feasibility study to evaluate a text messaging based complex intervention during pregnancy. *J Obes*. 2015;2015:814830.

155. Williams J, Mai CT, Mulinare J, et al. Updated estimates of neural tube defects prevented by mandatory folic acid fortification-united states, 1995-2011. *MMWR.Morbidity and mortality weekly report*. 2015;64(1):1-5.

156. Department of Health and Social Care. **Fortifying flour with folic acid: Government to consult**. <u>https://www.gov.uk/government/news/fortifying-flour-</u> with-folic-acid-government-to-consult. Updated 2018.

157. Connolly A, Pietri G, Yu J, Humphreys S. Association between long-acting reversible contraceptive use, teenage pregnancy, and abortion rates in england. *Int J Womens Health.* 2014;6:961-974.

158. World Health Organisation. Adolescent pregnancy. . 2004.

159. Ojengbede OA, Otolorin EO, Fabanwo AO. Pregnancy performance of nigerian women aged 16 years and below, as seen in ibadan, nigeria. *Afr J Med Med Sci*. 1987;16(2):89-95.

160. Okonofua FE, Feyisetan BJ, Davies-Adetugbo A, Sanusi YO. Influence of socioeconomic factors on the treatment and prevention of malaria in pregnant and non-pregnant adolescent girls in nigeria. *J Trop Med Hyg*. 1992;95(5):309-315.

161. Sandall J, Soltani H, Gates S, Shennan A, Devane D. Midwife-led continuity models versus other models of care for childbearing women. *Cochrane database of systematic reviews*. 2016(4).

162. Kenyon S, Jolly K, Hemming K, et al. Lay support for pregnant women with social risk: A randomised controlled trial. *BMJ Open*. 2016;6(3):e009203-2015-009203.

163. Gale NK, Kenyon S, MacArthur C, Jolly K, Hope L. Synthetic social support: Theorizing lay health worker interventions. *Soc Sci Med*. 2018;196:96-105.

164. Kapaya H, Mercer E, Boffey F, Jones G, Mitchell C, Anumba D. Deprivation and poor psychosocial support are key determinants of late antenatal presentation and poor fetal outcomes-a combined retrospective and prospective study. *BMC pregnancy and childbirth*. 2015;15(1):309.

165. Merlo DF, Wild CP, Kogevinas M, Kyrtopoulos S, Kleinjans J, NewGeneris Consortium. NewGeneris: A european study on maternal diet during pregnancy and child health. *Cancer Epidemiol Biomarkers Prev.* 2009;18(1):5-10.

166. Jolly M, Sebire N, Harris J, Robinson S, Regan L. The risks associated with pregnancy in women aged 35 years or older. *Human reproduction*. 2000;15(11):2433-2437.

167. Jacobsson B, Ladfors L, Milsom I. Advanced maternal age and adverse perinatal outcome. *Obstetrics & Gynecology*. 2004;104(4):727-733.

168. Freinkel N, Metzger BE, Phelps RL, et al. Gestational diabetes mellitus: Heterogeneity of maternal age, weight, insulin secretion, HLA antigens, and islet cell antibodies and the impact of maternal metabolism on pancreatic B-cell and somatic development in the offspring. *Diabetes*. 1985;34(Supplement 2):1-7. Appendix 1.

Supplementary tables published alongside systematic review

Supplementary Table 1. Characteristics of Excluded Studies

Reference	Reason for exclusion					
Briggs, M. M., Hopman, W. M., & Jamieson, M. A. (2007). Comparing	Anaemia is the only relevant outcome					
pregnancy in adolescents and adults: obstetric outcomes and prevalence of	measure and is only reported as a binary					
anemia. Journal of obstetrics and gynaecology Canada: JOGC= Journal	measure					
d'obstetrique et gynecologie du Canada: JOGC, 29(7), 546-555.						
O'Brien, K. O., Donangelo, C. M., Ritchie, L. D., Gildengorin, G., Abrams, S.,	No baseline nutritional measures and no					
& King, J. C. (2012). Serum 1, 25-dihydroxyvitamin D and calcium intake	control group intakes					
affect rates of bone calcium deposition during pregnancy and the early						
postpartum period. The American journal of clinical nutrition, 96(1), 64-72.						
De Vienne, C. M., Creveuil, C., & Dreyfus, M. (2009). Does young maternal	Anaemia is the only relevant outcome					
age increase the risk of adverse obstetric, fetal and neonatal outcomes: a	measure and is only reported as a binary					
cohort study. European Journal of Obstetrics & Gynecology and Reproductive	measure					
<i>Biology</i> , <i>147</i> (2), 151-156.						
Dean, S. V., Lassi, Z. S., Imam, A. M., & Bhutta, Z. A. (2014). Preconception	Review - considers preconception nutritional					
care: nutritional risks and interventions. <i>Reprod Health</i> , 11(Suppl 3), S3.	interventions					
Garikapaty, V. P., Feyerharm, R., & Zhu, B. (2007). Folic acid consumption	Peri-conceptional period only					
among Missouri women in the periconceptional period. Missouri medicine,						
<i>105</i> (6), 504-509.						
Gupta, N., Kiran, U., & Bhal, K. (2008). Teenage Pregnancies: Obstetric	Anaemia is the only relevant outcome					
characteristics and outcome. European Journal of Obstetrics & Gynecology	measure and is only reported as a binary					

and Reproductive Biology, 137(2), 165-171.	measure
Jones, R. L., Cederberg, H. M. S., Wheeler, S. J., Poston, L., Hutchinson, C.	Data is split into 'growers' and 'non-growers',
J., Seed, P. T., & Baker, P. N. (2010). Relationship between maternal	same data as Baker 2009 which is included
growth, infant birthweight and nutrient partitioning in teenage pregnancies.	
BJOG: An International Journal of Obstetrics & Gynaecology, 117(2), 200-	
211.	
Oliveira, O. R., Santana, M. G., Santos, F. S., Conceição, F. D., Sardinha, F.	Country of origin excluded - Brazil
L., Veiga, G. V., & do Carmo, M. G. T. (2012). Composition of fatty acids in	
the maternal and umbilical cord plasma of adolescent and adult mothers:	
relationship with anthropometric parameters of newborn. Lipids in health and	
<i>disease</i> , <i>11</i> (1), 1.	
Tsikouras, P., Dafopoulos, A., Trypsianis, G., Vrachnis, N., Bouchlariotou, S.,	Anaemia is the only relevant outcome
Liatsikos, S. A., & Von Tempelhoff, G. F. (2012). Pregnancies and their	measure and is only reported as a binary
obstetric outcome in two selected age groups of teenage women in Greece.	measure
The Journal of Maternal-Fetal & Neonatal Medicine, 25(9), 1606-1611.	
Wheeler, S. J., Poston, L., Thomas, J. E., Seed, P. T., Baker, P. N., &	Oily fish is the only intake variable reported,
Sanders, T. A. (2011). Maternal plasma fatty acid composition and pregnancy	data is from the same study as Baker 2009
outcome in adolescents. British journal of nutrition, 105(04), 601-610.	which is included
Wise, N. J., & Arcamone, A. A. (2011). Survey of adolescent views of healthy	Does not report a valid comparable measure
eating during pregnancy. MCN: The American Journal of Maternal/Child	of dietary intake or nutritional status

Nursing, 36(6), 381-386.						
Young, B. E., McNanley, T. J., Cooper, E. M., McIntyre, A. W., Witter, F.,	Focus on effect of hormone changes, reports					
Harris, Z. L., & O'Brien, K. O. (2012). Vitamin D insufficiency is prevalent and	same data Young 2012 which is included					
vitamin D is inversely associated with parathyroid hormone and calcitriol in						
pregnant adolescents. Journal of Bone and Mineral Research, 27(1), 177-186.						
Finkelstein, J. L., Pressman, E. K., Cooper, E. M., Kent, T. R., Bar, H. Y., &	Reports same data as Young 2012 which is					
O'Brien, K. O. (2014). Vitamin D status affects serum metabolomic profiles in	included					
pregnant adolescents. Reproductive Sciences, 1933719114556477.						
Whisner, C. M., Young, B. E., Witter, F. R., Harris, Z. L., Queenan, R. A.,	Reports same data as Young 2012 which is					
Cooper, E. M., & O'Brien, K. O. (2014). Reductions in heel bone quality across	included					
gestation are attenuated in pregnant adolescents with higher prepregnancy						
weight and greater increases in PTH across gestation. Journal of Bone and						
Mineral Research, 29(9), 2109-2117.						
Lenders, C. M., McElrath, T. F., & Scholl, T. O. (2000). Nutrition in adolescent	Not a primary study, questionnaire/quiz to					
pregnancy. Current opinion in pediatrics, 12(3), 291-296.	determine knowledge					
Burchett, H., & Seeley, A. (2003). Short Report: Good Enough to Eat? The	Not a primary study, short review article					
Diet of Pregnant Teenagers. International Journal of Health Promotion and						
Education, 41(2), 59-61.						
COMA (2002) Scientific review of the Welfare Food Scheme, Reports on	Not a primary study, service review					
health and social subjects, 51, i						

Wallace, J. M., Luther, J. S., Milne, J. S., Aitken, R. P., Redmer, D. A.,	Not a primary study, news brief relating to an						
Reynolds, L. P., & Hay, W. W. (2006). Nutritional modulation of adolescent	included article						
pregnancy outcome-a review. Placenta, 27, 61-68.							
ANUMBA, D. (2009). SESSION 2-MATERNAL MEDICINE. Journal of	Conference papers, not relevant						
Obstetrics and Gynaecology.							
Anderson, N. E., Smiley, D. V., Flick, L. H., & Lewis, C. Y. (2000). Missouri	Does not report a valid comparable measure						
rural adolescent pregnancy project (MORAPP). Public Health Nursing, 17(5),	of dietary intake or nutritional status						
355-362.							
Aston, G. (1995). Building a healthy community. AHA News, 31(15), 7	Not a primary study, news article						
Barda, G., Arbel-Alon, S., Bernstein, D., Zakut, H., & Menczer, J. (1997).	Country of origin excluded - Israel						
Pregnancy and delivery in a group of Israeli teenagers. A case-controlled							
study. Clinical and experimental obstetrics & gynecology, 25(1-2), 32-35.							
Berenson, A. B., Wiemann, C. M., & McCombs, S. L. (1997). Adverse	Does not report a valid comparable measure						
perinatal outcomes in young adolescents. The Journal of reproductive	of dietary intake or nutritional status						
<i>medicine</i> , <i>4</i> 2(9), 559-564.							
Berenson, A. B., Wiemann, C. M., Rowe, T. F., & Rickert, V. I. (1997).	Does not report a valid comparable measure						
Inadequate weight gain among pregnant adolescents: risk factors and	of dietary intake or nutritional status						
relationship to infant birth weight. American journal of obstetrics and							
gynecology, 176(6), 1220-1227.							

Bergmann, R. L., Gravens-Müller, L., Hertwig, K., Hinkel, J., Andres, B.,	Wrong age group					
Bergmann, K. E., & Dudenhausen, J. W. (2002). Iron deficiency is prevalent in						
a sample of pregnant women at delivery in Germany. European Journal of						
Obstetrics & Gynecology and Reproductive Biology, 102(2), 155-160.						
Bonnette, R. E., Caudill, M. A., Boddie, A. M., Hutson, A. D., Kauwell, G. P., &	Intervention group given measured folate,					
Bailey, L. B. (1998). Plasma homocyst (e) ine concentrations in pregnant and	controls not pregnant					
nonpregnant women with controlled folate intake. Obstetrics & Gynecology,						
92(2), 167-170.						
Boult, B. E., & Cunningham, P. W. (1995). Some aspects of obstetrics in black	Does not report a valid comparable measure					
teenage pregnancy: a comparative analysis. <i>Med. & L., 14</i> , 93.	of dietary intake or nutritional status					
Cong, K., Chi, S., & Liu, G. (1995). Calcium supplementation during	Wrong age group					
pregnancy for reducing pregnancy induced hypertension. Chinese medical						
journal, 108(1), 57-59.						
Covington, D. L., Peoples-Sheps, M. D., Buescher, P. A., Bennett, T. A., &	Does not report a valid comparable measure					
Paul, M. V. (1998). An evaluation of an adolescent prenatal education	of dietary intake or nutritional status					
program. American Journal of Health Behavior.						
Cunnington, A. J. (2001). What's so bad about teenage pregnancy?. Journal	Not a primary study, review					
of Family Planning and Reproductive Health Care, 27(1), 36-41.						
Dunn, P. C., Kolasa, K., & McKee, A. (1998). Nutrition education materials for	Not a primary study, resource review					
pregnant teens. Journal of Nutrition Education, 30(6), 414-415.						

Erkkola, M., Karppinen, M., Järvinen, A., Knip, M., & Virtanen, S. M. (1998).	Wrong age group					
Folate, vitamin D, and iron intakes are low among pregnant Finnish women.						
European Journal of Clinical Nutrition, 52(10), 742-748.						
Goonewardene, I. M. R., & Waduge, R. P. K. (2009). Adverse effects of	Does not report a valid comparable measure					
teenage pregnancy. Ceylon Medical Journal, 50(3).	of dietary intake or nutritional status					
Howell, S. R., Barnett, A. G., & Underwood, M. R. (2001). The use of pre-	Supplement use only, does not report a valid					
conceptional folic acid as an indicator of uptake of a health message amongst	comparable measure of dietary intake or					
white and Bangladeshi women in Tower Hamlets, east London. Family	nutritional status					
<i>Practice</i> , <i>18</i> (3), 300-303.						
Howie, L. D., Parker, J. D., & Schoendorf, K. C. (2003). Excessive maternal	Does not report a valid comparable measure					
weight gain patterns in adolescents. Journal of the American Dietetic	of dietary intake or nutritional status					
Association, 103(12), 1653-1657.						
Hunt, D. J., Stoecker, B. J., Hermann, J. R., Kopel, B. L., WILLIAMS, G. S., &	Does not report a valid comparable measure					
Claypool, P. L. (2002). Effects of nutrition education programs on	of dietary intake or nutritional status					
anthropometric measurements and pregnancy outcomes of adolescents.						
Journal of the American Dietetic Association, 102(3), S100-S102.						
Koenig, J., & Elmadfa. (2000). Status of Calcium and VitaminD of Different	No pregnant adolescents included in the					
Population Groups in Austria. International journal for vitamin and nutrition	sample					
research, 70(5), 214-220.						

Kondo, A., Kamihira, O., Shimosuka, Y., Okai, I., Gotoh, M., & Ozawa, H.	Wrong age group					
(2005). Awareness of the role of folic acid, dietary folate intake and plasma						
folate concentration in Japan. Journal of Obstetrics and Gynaecology						
Research, 31(2), 172-177.						
Lake, R. S., & Drake, L. M. (1995). Nutrition Intervention Program for	Does not report a valid comparable measure					
Pregnant Girls [Abstract]. Public Health Reports (1974-), 208-208.	of dietary intake or nutritional status					
Lenders, C. M., McElrath, T. F., & Scholl, T. O. (2000). Nutrition in adolescent	Does not report a valid comparable measure					
pregnancy. Current opinion in pediatrics, 12(3), 291-296.	of dietary intake or nutritional status					
Loto, O. M., Ezechi, O. C., Kalu, B. K. E., Loto, A. B., Ezechi, L. O., &	Country of origin excluded -Nigeria					
Ogunniyi, S. O. (2004). Poor obstetric performance of teenagers: is it age-or						
quality of care-related?. Journal of obstetrics and Gynaecology, 24(4), 395-						
398.						
Mahfouz, A. A., El-Said, M. M., Al-Erian, R. A., & Hamid, A. M. (1995).	Does not report a valid comparable measure					
Teenage pregnancy: are teenagers a high risk group?. European Journal of	of dietary intake or nutritional status					
Obstetrics & Gynecology and Reproductive Biology, 59(1), 17-20.						
Malviya, M. K., Bhardwaj, V. K., Chansoria, M., & Khare, S. (2003).	Country of origin excluded - India					
Anthropometric profile and perinatal outcome of babies born to young women						
(< 18 years). Indian pediatrics, 40(10), 971-976.						
Mathews, F., Yudkin, P., & Neil, A. (1998). Folates in the periconceptional	Does not report a valid comparable measure					
period: are women getting enough?. BJOG: An International Journal of	of dietary intake or nutritional status					
Obstetrics & Gynaecology, 105(9), 954-959.						

Mathews, F., Yudkin, P., Smith, R. F., & Neil, A. (2000). Nutrient intakes	Wrong age group							
during pregnancy: the influence of smoking status and age. Journal of								
Epidemiology and Community Health, 54(1), 17-23.								
Perry, R. L., Mannino, B., Hediger, M. L., & Scholl, T. O. (1996). Pregnancy in	Does not report a valid comparable measure							
early adolescence: Are there obstetric risks?. Journal of Maternal-Fetal	of dietary intake or nutritional status							
Medicine, 5(6), 333-339.								
Ruppel, K. J. (2001). Health beliefs and health behaviors of pregnant	Not a primary study, dissertation							
adolescents.								
Stang, J., Story, M., & Feldman, S. (2005). Nutrition in adolescent pregnancy.	Not a primary study, review article							
International Journal of Childbirth Education, 20(2), 4.								
Story, M., & Alton, I. (1995). Nutrition Issues and Adolescent Pregnancya.	Not a primary study, review article							
Nutrition Today, 30(4), 142-151.								
Wrieden, W. L., & Symon, A. (2003). The development and pilot evaluation of	Does not report a valid comparable measure							
a nutrition education intervention programme for pregnant teenage women	of dietary intake or nutritional status							
(food for life). Journal of Human Nutrition and Dietetics, 16(2), 67-71.								
Whisner, C. M., Young, B. E., Pressman, E. K., Queenan, R. A., Cooper, E.	Reports same data as Young 2012 which is							
M., & O'Brien, K. O. (2015). Maternal diet but not gestational weight gain	included							
predicts central adiposity accretion in utero among pregnant adolescents.								
International Journal of Obesity, 39(4), 565-570.								
Symon, A. G., & Wrieden, W. L. (2003). A qualitative study of pregnant	Does not report a valid comparable measure							
teenagers' perceptions of the acceptability of a nutritional education	of dietary intake or nutritional status							

intervention. <i>Midwifery</i> , <i>19</i> (2), 140-147.						
Dubois, S., Coulombe, C., Pencharz, P., Pinsonneault, O., & Duquette, M. P.	Characteristics of participants at the time of					
(1997). Ability of the Higgins Nutrition Intervention Program to improve	data collection are not given. A paper where					
adolescent pregnancy outcome. Journal of the American Dietetic Association,	these can be found is referenced; however					
<i>97</i> (8), 871-873.	this is not available in English language.					
Pope, J. F., Skinner, J. D., & Carruth, B. R. (1997). Adolescents' self-reported	Does not report a valid comparable measure					
motivations for dietary changes during pregnancy. Journal of Nutrition	of dietary intake or nutritional status					
Education, 29(3), 137-144.						

Supplementary Table 2. Pooled, Weighted Mean Nutrient Intakes Expressed as Percentages of Dietary Reference Values by Stage of Pregnancy

	Percentage of UK Dietary Reference Value							Percentage of US Dietary Reference Value																
	Total			Se	Second Trimester			Third Trimester			in over pregn	ancy		Total		Second Trimester			Third Trimester			Mean over pregnancy		ancy
Nutrient	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval
Energy kcal/day ¹	9	91.0	30, 153	4	81.9	34.1, 129.8	7	83.0	37, 129	4	93.0	37, 149	9			4			7			4		
Protein grams/day ²	9	224.0	99.9, 348.1	4	126.4	44.2, 208.5	6	134.8	44.4, 225.1	4	174.1	137.9, 210.3	9	129.2	40.1, 218.4	4	90.8	31.8, 149.8	6	96.8	31.9, 161.7	4	125.1	17.0, 233.1
Calcium mg/day ³	8	107.3	0, 232	2	164.9	1.7, 328.1	5	117.6	9.9, 225.2	4	101.5	0, 233.7	8	66.0	0, 147.6	2	101.5	1, 201.9	5	72.4	6.1, 138.6	4	62.4	0, 194.7
Phosphorous mg/day ⁴	4	208.2	30.1, 386.2	1			2	195.1	47.2, 343	3	208.7	31.3, 386	4	104.1	30.7, 177.4	1			2	97.6	23.6, 171.5	3	104.3	15.6, 193
Iron mg/day ⁵	10	119.1	0, 259.9	4	111.0	29.7, 192.3	7	118.2	0, 268.3	4	125.1	0, 251	10	65.3	0, 139.5	4	60.8	16.3, 105.4	7	64.8	0, 147.1	4	68.6	0, 137.6
Magnesium mg/day ⁶	6	84.2	15.6, 152.8	1			4	79.4	27.3, 131.4	3	86.3	39.1, 133.6	6	63.2	20.1, 106.3	1			4	59.5	20.4, 98.6	3	64.7	7.6, 121.9
Potassium mg/day ⁷	2	83.2	43.6, 122.7	1			2	84.3	43.6, 125	0			2	61.9	32.6, 91.2	1			2	62.8	32.5, 93.1	0		0, 0
Zinc mg/day ⁸	10	145.7	0, 297.2	4	118.7	21.5, 215.9	7	128.4	21.9, 234.9	4	178.0	0, 385.6	10	85.0	0, 189.7	4	69.3	12.6, 126	7	74.9	12.8, 137	4	103.8	0, 224.9
Vitamin D µg/day ⁹	5	34.8	0, 83.1	1			4	31.2	0, 81.9	2	54.6	0, 114.3	5	23.2	0, 58.2	1			4	20.8	0, 54.6	2	36.4	0, 96.1
Vitamin E ¹⁰	5			1			4			2			5	55.0	0, 148.9	1			4	56.8	0, 129.2	2	51.4	0, 0
Vitamin C ¹¹	7	309.2	0, 894.9	1			4	297.3	0, 753.6	4	294.9	0, 916.8	7	193.2	0, 524.6	1			4	185.8	0, 471	4	184.3	0, 573
Folate ¹²	6	125.1	0, 487.3	1			4	146.9	0, 374	3	142.6	0, 580.1	6	62.6	0, 256	1			4	73.4	0, 187	3	71.3	0, 290
Riboflavin ¹³	4	200.3	0, 405.7	1			4	225.9	29.9, 421.9	4	199.1	0, 404.9	4	157.4	5.9, 308.9	1			4	177.5	23.5, 331.5	4	156.4	0, 319.3
B12 ¹⁴	5	224.0	0, 1011.1	1			3	162.9	0, 758.1	3	365.5	0, 1205.8	5	129.2	0, 488.1	1			3	207.3	0, 550.7	3	210.8	0, 685
Vitamin A ¹⁵	7	162.5	0, 583	1			4	141.5	0, 456.8	4	164.2	0, 578.3	7	130.0	0, 571.1	1			4	113.2	0, 365.5	4	131.4	0, 462.6

1.UK EARs, First and Second Trimester requirement (Average UK EAR for Females aged 11-19) 2355kcal/d, Third Trimester requirement 2546 kcal/d, Mean over pregnancy requirement (average of three trimester values) 2419kcal/d 2.UK RNI 51 g/day and US RDA 71g/day 3.UK RNI 800mg/day and US RDA 1300mg/day 4. UK RNI 625mg/day and US RDA 1250mg/day 5. UK RNI 14.8mg/day and US RDA 27mg/day 6. UK RNI 300mg/day and US RDA 400mg/day 7. UK RNI 3500mg/day and US RDA 4700mg/day 8. UK RNI 7mg/day and US RDA 12mg/day 9.UK RNI 10µg/day and US RDA 15µg/day 10. US RDA 15mg/day 11. UK RNI 40mg/day and US RDA 80mg/day 12.UK RNI 200µg/day and US RDA 600µg/day 13. UK RNI 1.1mg/day and US RDA 1.4mg/day 14. UK RNI 1.5µg/day and US RDA 2.6µg/day 15.UK RNI 600µg/day and US RDA 750µg/day

		Percer	ntage of UK Die	etary Refere	nce Val	ue	Percentage of US Dietary Reference Value						
		UK stud	ies		US stud	ies		UK stud	ies	US studies			
Nutrient	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies Mean		Confidence Interval	
Energy kcal/day ¹	2	89.1	39.2, 139.1	5	100.4	24.2, 176.6	2			5			
Calcium mg/day ²	2	106.3	20.8, 191.9	5	107.0	0, 239.9	2	65.4	12.8, 118.1	5	65.8	0, 147.6	
Iron mg/day ³	2	112.9	0, 331	5	131.9	0, 273.1	2	61.9	0, 181.5	5	72.3	0, 149.7	
Magnesium mg/day ⁴	2	78.8	30.3, 127.4	4	86.8	10.5, 163	2	59.1	22.7, 95.5	4	65.1	7.9, 122.3	
Zinc mg/day ⁵	2	115.8	36.2, 195.3	5	186.8	0, 399	2	67.5	21.1, 113.9	5	109.0	0, 232.8	
Vitamin D µg∕day ⁶	2	21.4	0, 54.4	3	53.2	0, 116.7	2	14.3	0, 36.3	3	35.4	0, 77.8	
Vitamin E ⁷	2			3			2	58.8	0, 124.5	3	53.1	0, 151.2	
Vitamin C ⁸	2	317.2	0, 791.3	4	309.6	0, 924.8	2	198.2	0, 494.6	4	193.5	0, 578	
Folate ⁹	2	93.8	4.3, 183.2	4	139.3	0, 459.1	2	46.9	2.1, 91.6	4	69.7	0, 229.5	
Vitamin A ¹⁰	2	124.3	0, 422.7	4	179.9	0, 658.3	2	99.4	0, 338.2	4	143.9	0, 526.7	

Supplementary Table 3. Pooled, Weighted Mean Nutrient Intakes Expressed as Percentages of Dietary Reference Values by Study Country of Origin

1.UK EARs, First and Second Trimester requirement (Average UK EAR for Females aged 11-19) 2355kcal/d, Third Trimester requirement 2546 kcal/d, Mean over pregnancy requirement (average of three trimester values) 2419kcal/d 2.UK RNI 800mg/day and US RDA 1300mg/day 3. UK RNI 14.8mg/day and US RDA 27mg/day 4. UK RNI 300mg/day and US RDA 400mg/day 5. UK RNI 7mg/day and US RDA 12mg/day 6.UK RNI 10µg/day and US RDA 15µg/day 7. US RDA 15mg/day 8. UK RNI 40mg/day and US RDA 80mg/day 9.UK RNI 200µg/day and US RDA 600µg/day 10.UK RNI 600µg/day and US RDA 750µg/day

		Total			First Trimeste	r	Sec	cond Trim	ester	Third Trimester			
Nutrient	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval	
Haemoglobin (g/dL) ¹	11	114.6	95.4, 133.8	2	121.8	100.2, 143.4	8	114.9	97.7, 132	11	109.2	90.3, 128.1	
Haematocrit (g/L) ²	4	33.7	28, 39.4	1			2	33.0	27.1, 38.9	4	32.3	26.5, 38	
Ferritin (µg/L) ³	5	18.2	0, 48.3	0			5	21.8	0, 60.8	4	16.5	0, 40.8	
Zinc (μ mol/L-1) ⁴	5	11.5	5.9, 17.1	2	11.8	5.1, 18.5	0			5	11.0	5.6, 16.5	
Selenium (µg/L)⁵	4	61.8	39.2, 84.4	0			2	46.8	29.1, 64.5	4	64.8	41.8, 87.7	
25_OH_D (nmol/L) ⁶	6	55.9	6.2, 105.7	0			1			5	55.9	8.9, 103	

Supplementary Table 4. Pooled, Weighted Mean Biological Markers of Nutritional Status by Stage of Pregnancy

1. Target value 110 g/L 2. Target value33 g/L 3. Target value 15 μg/L 4. Target value 6.12μmol/L-1 5. Target value 90 μg/L 6. Target value 25 nmol/L

Supplementary Table 5. Pooled, Weighted Mean Biological Markers of Nutritional Status by Study Country of Origin

	No	on-US Stud	lies		US studies	
Nutrient	Number of studies	Mean	Confidence Interval	Number of studies	Mean	Confidence Interval
Haemoglobin (g/dL) ¹	3	109.2	88.6, 129.8	8	114.3	96.1, 132.6
Haematocrit (g/L) ²	2	26.2	20.2, 32.2	2	33.6	27.7, 39.4

1. Target value 110 g/L 2. Target value33 g/L

Appendix 2.

Born in Bradford baseline questionnaire

Appendix 3.

Born in Bradford Data transfer agreement

32949	Study ID 1 2 0 0 1 P 1
Born in Bradford - Mothers' Questionnaire	
To be completed by interviewer:	
Interviewer's Number -	(2 initials - 2 numbers e.g. AN 01)
1. Date Completing this questionnaire?	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	
*2. What language(s) was used for administ	tering the questionnaire?
*3. Was an Interpreter used? No Hospital/Study Interpreter Fa (To be measured by interviewer)	amily Member/Friend
4. Height (Cms)	t (Kilos) (Grms)
*6. Triceps (Cms) .	☐ Not able to take
*7. Arm circumference (Cms)	☐ Not able to take
8. How old were you when you had your firs	st period?



Don't Know







9. Will this be your first child?

☐ Yes ☐ No

9a) If no:- what month and year were each of your previous children born in ? - starting with the eldest:

Month

First child	
Second child	
Third child	
Fourth child	
Fifth child	
(add birth dates of all other children)	









Section A - Where you live

These questions relate to where you are living at present	t				_
*A1. How long have you lived at your current address?					
	Y	Y -	m -	m	
*A2. In which of these ways does your household occupy (Cross ONE box ONLY)	this	add	lres	57	
If answers yes to any of the three * questions, please go to A2a). If no	it go t	o A3			
Buying it with the help of a mortgage or loan					
Owns outright					
T *Rents it					
*Lives here rent free (including rent free in relatives/friends prop	oerty	exclu	Iding	squa	(tting)
*Pays part rent and part mortgage (shared ownership)					
Don't know					
Squatting					
*A2a) If A2 was answered - Rents it: Lives rent free or part mortgage - ask who is your landlord? (Cross ONE box ONLY)	pays	s pai	rt re	nt a	nd
Private Landlord or Letting Agency, Another individual					
Housing Association, Housing Co-operative, Charitable Trust					
Local Authority/Council					
Relative or friend (before you lived here) of a household memb	er				
Employer (individual) of a household member					
Employer (organisation) of a household member					
Another Organisation					
Don't Know					
A3) How many bedrooms does your household have, in	clud	ing l	beds	sittir	g

rooms and spare bedrooms?

Enter number of bedrooms







Section B - Who you live with?

B1. What ages are those, including yourself, who live in your household or accommodation? [If age not known, please give best estimate]

Is there anybody:-

Age	Number of males	Number of females
Under 2 years		
between 2 -15 years		
between 16 - 17 years		
between 18 - 64 years		
65 years and over		

- **B2.** Are you: (Cross ONE box ONLY)
 - ☐ Married (first marriage)
 - Re-married
 - □ Single (never married)
 - Separated (but still legally married)
 - Divorced
 - U Widowed
- **B3.** Are you: (Cross ONE box ONLY)
 - Living with baby's father
 - Living with another partner
 - Not living with a partner but in a relationship (eg. partner living abroad or in another propery)
 - Not living with a partner and not in a relationship





Section C - About you, your family and your baby's father and his family

C1. What country were you and your baby's father born in?

(Cross ONE box ONLY in each column).

Country	You	Country	Baby's father
England		England	
Northern Ireland		Northern Ireland	
Scotland		Scotland	
Wales		Wales	
Channel Islands		Channel Islands	
Isle of Man		Isle of Man	
Republic of Ireland		Republic of Ireland	
Czech Republic		Czech Republic	
Poland		Poland	
Slovakia		Slovakia	
Bangladesh		Bangladesh	
India		India	
Pakistan		Pakistan	
Sri Lanka		Sri Lanka	
Philippines		Philippines	
Don't know		Don't know	

You - Other (Please write in) Baby's father - Other (Please write in)

	``		,									







*C2. To be asked if not born in the UK How old were you when you moved to the UK?

Age in Years



If answered Pakistan for you and/or baby's father in C1 go to C3. If not Pakistan for either person then go to C4.

C3) Were you and/or baby's father born in Mirpur District?

(Cross ONE box ONLY in each row)

You	🗌 Yes	🗌 No	🗌 Don't Know
Baby's father	🗌 Yes	🗌 No	🗌 Don't Know

*C3a) If yes, which town or village?

Yo	You (Please write in)													Don't Know														
Baby's father (Please write in)																												
C3	b) [י סס (Ir	yoı nterv	i kr ⁄iew	10v /er -	v tl · co	h e i nsu	nan It lis	ne st of	of Bir	yo ade	ur's eri if	ar neo	nd k cess	bab sary	y's).	fat	the	er's	Bi	rad	eri	?					

١	Υοι	J (F	lea	You (Please write in)														nov	V					

Baby's father (Please write in)													Don	't K	nov	V					







*C4. What country were your mother and father born in? (Cross ONE box ONLY in each column).

Country	Your Mother	Country	Your Father
England		England	
Northern Ireland		Northern Ireland	
Scotland		Scotland	
Wales		Wales	
Channel Islands		Channel Islands	
Isle of Man		Isle of Man	
Republic of Ireland		Republic of Ireland	
Czech Republic		Czech Republic	
Poland		Poland	
Slovakia		Slovakia	
Bangladesh		Bangladesh	
India		India	
Pakistan		Pakistan	
Sri Lanka		Sri Lanka	
Philippines		Philippines	
Don't know		Don't know	
Your mother - Other (Please wr	te in)		
Your father - Other (Please write	e in)		
If answered Pakistan for your	mother or fat	her in C4 go to C5 if not P	Pakistan then go to C6
C5) Were your mother and (Cross ONE box ONLY)	father born	in Mirpur district?	
Your mother Yes	□ No	Don't Know	
Your father	□ No □	Don't Know	
			32040



C5a) If yes, which town or village?

Yo	our	mo	the	r - (Plea	ase	writ	e ir	1)					Don	't Ki	now	/										
Yo	our	fath	ner	- (P	leas	se v	/rite	in)	_					Don	't Ki	าอพ	1		-	_	_				 		
C5	b) our	Do mo	yo the	u k r - (nov Plea	v t l	h e writ	na ı te ir	me 1)	of	yoı	ur ı	mot	t he Don	r's 't Ki	and now	d fa	ath	er':	s B	ira	de	ri?	_			
C5	b) bur	Do mo	yo the	u k r - (nov Plea	v ti ase	he writ	naı te ir	me	of	yo	ur i	mot	t he Don	r's 't Ki	now	d fa	ath	er':	s B	ira	de	ri?				
C5 Ya	5 b) our	Do mo	yo the	u k r - (- (Pl	nov Plea	v ti ase e w	h e writ	nai te ir in)	me	of	yo	ur 1		bon Don	r's 't Ki	and now	d f a /	ath	er'	s B	ira	de	ri?				







C6 What country were your grandparents born in?

(Cross ONE box ONLY in each column).

Country	Your mother's mother	Your mother's father	Your father's mother	Your father's father
England	🗆			
Northern Ireland				
Scotland	🗆			
Wales	🗆			
Channel Islands				
Isle of Man				
Republic of Ireland				
Czech Republic	🗆			
Poland	🗆			
Slovakia				
Bangladesh	🗆			
India				
Pakistan				
Sri Lanka	🗆			
Philippines	🗆			
Don't know				
Your mother's mother - Other (F	Please write in)			
Your mother's father - Other (PI	ease write			
Your father's mother - Other (Pl	ease write in)			
Your father's father - Other (Ple	ease write in)			







If answered Pakistan in C6 for where any of your grandparents were born ask C7: If not go to C8.

Answer this if your grandparents were born in Pakistan If not go to C8.

C7. Were your grandparents born in Mirpur district? (Cross ONE box ONLY in each row)

Your mothers mother	🗌 Yes	🗌 No	Don't Know
Your mother's father	🗌 Yes	🗌 No	Don't Know
Your father's mother	🗌 Yes	🗌 No	Don't Know
Your father's father	☐ Yes	🗌 No	Don't Know

C7a) If yes, which town or village ?

Yo	ur n	not	her'	s n	noth	ner	- (P	lea	se v	vrite	e in)			[)on'	t Kr	างพ									
Yo	ur n	not	her'	s fa	athe	er -	(Ple	ease	e wr	ite i	n)			[)on'	t Kr	างพ									
Yo	ur f	ath	er's	mo	othe	er -	(Ple	ease	e wr	ite i	n)			[)on'	t Kr	างพ									
Yo	ur f	ath	er's	fat	her	- (F	Plea	se	write	e in)			[)on'	t Kr	างพ									
Yo	ur n	DO not	yo her:	ик sm	nov oth	er -	ne · (Pl	na eas	me se w	or rite	yoı in)	ur ç	jra	nar I	ent)on'	t Kr	now	ade	S7			 		1	 	
Yo	ur n	not	her	s fa	the	r - (Ple	ase	wri	te ir	า)			I)on'	t Kr	NOW									
Yo	ur f	ath	ers	mo	the	r - (Ple	ase	wri	te ir	า)			[)on'	t Kr	างพ									
Yo	ur f	ath	ers	fatl	ner	- (P	lea	se v	write	e in))on'	t Kr	NOW									





	r _{enna}
1	32949

C8. What country were the parents of your baby's father born in?

(Cross ONE box ONLY in each column).

Country	Mother of baby's father	Country	Father of baby's father
England		England	
Northern Ireland		Northern Ireland	
Scotland		Scotland	
Wales		Wales	
Channel Islands		Channel Islands	
Isle of Man		Isle of Man	
Republic of Ireland		Republic of Ireland	
Czech Republic		Czech Republic	
Poland		Poland	
Slovakia		Slovakia	
Bangladesh		Bangladesh	
India		India	
Pakistan		Pakistan	
Sri Lanka		Sri Lanka	
Philippines		Philippines	
Don't know		Don't know	
Mother of baby's father - Other	ner (Please write in)	

Father of baby's father - Other (Please write in)







If answered Pakistan in C8 for where the mother of the baby's father or father of the baby's father was born ask C9: If not then go to C10.

Answer this if the mother of the baby's father or father of the baby's father was born in Pakistan

C9. Was the mother of the baby's father / father of the baby's father born in

Mirpur ? (Cross ONE box ONLY in each row)

Mother of baby's father	🗌 Yes	🗌 No	🗌 Don't Know
Father of baby's father	Yes	🗌 No	🗌 Don't Know

C9a) If yes, which town or village ?

Мс	othe	r of	r of baby's father - (Please write								te ir	า)]Dc	on't	Kno	W				
Fa	the	r of	ba	by's	s fat	the	r - (I	Plea	ase	writ	e ir	ı)] Do	on't	Kno	w				

C9b) Do you know the name of your baby's father's parents' Biraderies?

Мс	othe	r o	f ba	ıby'	s fa	the	r - (Ple	ase	writ	te ir	1)] Do	on't	Kno	w								
Fa	ther	r of	ba	by's	s fat	ther	[.] - (I	Plea	ase	write	e in)					Do	on't	Knc	w								
Re [•] C1 [•]	turi 0. 1] Wi] Mi	nin Fo hite xed	g t wh	o y ich	ou; of gro	th oup	ese	• gr	ouj] As] Ch	ps (ian iines	do y or A se	you Asia	n B	ons ritis	i de h	er y	ou	be	lor	ıg?	(C	ros	s Ol	NE	box	ON	LY)	I
C1	0a)] Bri y ot	itish	f A		vei Irisi bac	red h ckg	C1	0 a		Whi ase	te writ	wh te in	at (do	yoı		ons (Ci	ross	er y s Ol	you NE I		ulti ON	ura ILY)	l ba	ack	(gro	oun	d?
																									04	20.40		





C10b) If Answered C10 as Mixed ethnic group what do you consider your

		C	ult	ura	al b	ac	kgr	ou	nd	? (C	ros	s O	NE	box		ILY)											
		JW	/hite	e an	d B	lacł	k Ca	aribl	bea	n] WI	nite	and	d Ba	ngl	ade	shi										
	Γ	JW	/hite	e an	d B	lacł	k Af	rica	n] WI	nite	and	d Ind	diar	n Ca	aribb	bea	n								
	Γ	JW	/hite	e an	d Ir	ndia	n] WI	nite	and	d Af	rica	n-In	ndia	n									
	Ľ	JW	/hite	e an	d P	akis	stan	i																				
An	y of	ther	[,] mi	xed	l ba	ckg	rοι	und	(pl	ease	e w	rite	in)															
	<u> </u>		1	1	1	1	1	I	I		I	<u> </u>		I	<u> </u>				1			1	I	1	I			
C10)c)	lf a	ans	we	rec	I C	10	as	Bla	ack	or	Bla	ack	B	ritis	sh	wh	at e	do	γοι	I C	ons	side	er y	ou	r		
		С	ult	ura	l ba	ack	gr	oui	nd?	' (Cı	ross	3 O I	NE I	хос	ON	LY))											
]Ca	arib	bea	n] Af	rica	n																			
An	y ot	ther	Bla	ack	ba	ckg	rou	nd	(ple	ease	e wr	ite i	n)				_											
	•		•	•	•	•	•	•	•	•		•	•	•	•		•	•	•	•				•	•			
C1()d)	lf a	ans	we	erec	d C	10	as	As	ian	or	As	iar	n Bi	ritis	sh	wh	at e	do	γοι	I C	ons	side	er y	ou	r		
		С	ult	ura	l b	acl	kgr	ou	nd?	? (C	ros	s Ol	NE	box	ON	ILY)											
		ln	diar	۱				[lr	ndia	n C	arib	bea	an														
]Pa	akis	tani				[frica	an-I	ndia	an															
		Ba	angl	ade	shi																							
An	y of	ther	As	ian	ba	ckg	rou	nd	(ple	ease	e wr	ite i	n)															
	•	•	•	•	•	•		•	•	•			•	•	•		•	•						•				
C1	0e)	If	ans	SWe	ere	d C	10	as	Cł	ine	se	w	nat	do	yo	u c	on	sid	er	yoı	ur o	cult	tura	al				
		b	ac	kgı	rou	ndʻ	? (Cro	ss (ONE	bo	x O	NĽ	Y)														
		C	hin	ese			Jap	ane	se		_] F	ilipi	no		JV	ietn	am	ese										
An	y C	hine	ese	or	oth	er k	acl	kgro	oun	id (p	blea	se v	write	e in) I)		<u> </u>	1		r –	1	1	<u> </u>	1	1	1		
C 1	0f)	lf (nl	ans	5W(ere	d C	:10	as	0 1	the	r w	ha	t de	o yo	ou	COI	nsie	der	yo	ur	cu	tur	al	bac	:kg	rou	nd	?
		(pie		e wi		(11) 																						









Section D - Your Family

These questions are about you and your family and about baby's father and his family.

D1.	Are you related to the father of you example are you cousins? (Cross)	ur baby other than by marriage? For
	☐ Yes ☐ No ☐ Don't Know	
D1a)	If yes, how are you related to the f 2nd cousin (Cross ONE box ONLY)	ather of your baby? e.g. 1st cousin,
	1st Cousin	Other related by blood
	1st Cousin, once removed	Other related by marriage
	Second Cousin	Don't know
D2.	Were your parents related? For ex (Cross ONE box ONLY)	ample were they cousins?
	Yes No Don't Know	
D2a)) If yes, how were your parents rela	ted? (Cross ONE box ONLY)
	1st Cousins	Other related by blood
	☐ 1st Cousins, once removed	Other related by marriage
	Second Cousins	Don't know
D3.	Were the parents of the father of y they cousins? (Cross ONE box ONLY)	your baby related? For example were
	Yes No Don't Know	
D3a)	If yes, how were they related? (Cro	oss ONE box ONLY)
	1st Cousins	Other related by blood
	☐ 1st Cousins, once removed	Other related by marriage
	Second Cousins	Don't know
Interv	iewer: If answered yes in D1, please cor	nplete a family tree (on a separate form afte

Interviewer: If answered yes in D1, please complete a family tree (on a separate form after you have completed this section. Do not change questions D1 to D3 after the family tree is completed).

D4. Was a family tree completed?

Yes No







Section E Education

E1. What is the highest educational qualification you have? (Cross ONE box ONLY)

- 1 + 0 levels/CSEs/GCEs (any grades)
- 5 + 0 levels, 5+ CSEs (grade 1) 5 + GCSEs (grades A-C), School Certificate
- 1 + A levels/AS levels
- 2 + A levels, 4 + AS levels, Higher School Certificate
- NVQ Level 1, Foundation GNVQ
- NVQ Level 2, Intermediate GNVQ
- NVQ Level 3, Advanced GNVQ
- NVQ Levels 4-5, HNC, HND
- First Degree (e.g. BA, BSc)
- Higher Degree (e.g. MA, PhD, PGCE Post-graduate certificates/diplomas)
- Other qualifications (e.g. City and Guilds, RSA/OCR, BTEC/Edexcel)
- Overseas qualification (If obtained in Pakistan go to E1a, If obtained in another country go to E1b)
- No Qualifications
- Don't know

E1a) If your highest educational qualification was obtained in Pakistan please indicate: (Cross ONE box ONLY)

- Second School Certificate (SSC) Matriculation (Metric)
- Diploma in Commerce
- Higher Secondary (HSC) Cert/Intermediate Humanities, Pre-Eng or Pre-Medical/Science Streams
- Certificate from Board of Technical Education
- Diploma from Board of Technical Education
- Final Apprenticeship Certificate/Grade 2 Skilled
- Vocational Institute Diploma/Grade 3 Skilled Worker Certificate
- Batchelor Degree (4 year) in generally professional fields (excluding Bachelor of Education)
- Batchelor of Arts/Commerce/Engineering/Science/Technology (Pass and Honours)
- Postgraduate Eg Masters degree/PhD
- Don't know






E1b) If your highest educational qualification was not obtained in the UK or in Pakistan, please specify what that qualification was and which country it was obtained in.

Qualification

Со	unt	ry													

E2. What is baby's father's highest educational qualification? (Cross ONE box ONLY)

- 1 + 0 levels/CSEs/GCEs (any grades)
- 5 + 0 levels, 5+ CSEs (grade 1) 5 + GCSEs (grades A-C), School Certificate
- 1 + A levels/AS levels
- 2 + A levels, 4 + AS levels, Higher School Certificate
- NVQ Level 1, Foundation GNVQ
- NVQ Level 2, Intermediate GNVQ
- NVQ Level 3, Advanced GNVQ
- NVQ Levels 4-5, HNC, HND
- First Degree (e.g. BA, BSc)
- Higher Degree (e.g. MA, PhD, PGCE Post-graduate certificates/diplomas)
- Other qualifications (e.g. City and Guilds, RSA/OCR, BTEC/Edexcel)
- Overseas qualification (If obtained in Pakistan go to E2a, If obtained in another country go to E2b)
- No Qualifications
- Don't know







E2a)	If his highest educational qualification was obtained in Pakistan plea	ISe
	indicate (Cross ONE box ONLY)	

- Second School Certificate (SSC) Matriculation (Metric)
- Diploma in Commerce
- Higher Secondary (HSC) Cert/Intermediate Humanities, Pre-Eng or Pre-Medical/Science Streams
- Certificate from Board of Technical Education
- Diploma from Board of Technical Education
- Final Apprenticeship Certificate/Grade 2 Skilled
- Vocational Institute Diploma/Grade 3 Skilled Worker Certificate
- Batchelor Degree (4 year) in generally professional fields (excluding Bachelor of Education)
- Batchelor of Arts/Commerce/Engineering/Science/Technology (Pass and Honours)
- Postgraduate Eg Masters degree/PhD
- Don't know

E2b) If his highest educational qualification was not obtained in the UK or Pakistan, please specify what the qualification was and which country it was obtained in.

Qualification

Co	untr	у													







Se	ctio	on F	Yo	ur (Cu	rre	nt	Em	plo	ym	en	t														
F1.	Are	e yo	ou c	urr	en	tly	a f	ull	tim	e s	stu	deı	nt?			Yes	5	[c						
F2.	Are	e yo	ou c	urr	en	tly	en	nplo	oye	d?						Yes	5	[C	(If `	Yes	, go	to I	F3)	
F2a)	If P	lo -	Ha	ve	yo	u e	vei	r w	ork	ed	?					Yes	5	E)	(If I	No,	go t	o F	12)	
F2b)	lf y	/es	to I	F2a) - I	hov	v Io	ong	ag	0 0	did	yo	u s'	top	w	ork	ing	?								
	Ye	ars				ſ	Mor	nths													F			1		
lf stop neare	opec est w	l les eek	s tha)	an o	ne	mo	nth	ago	o rec	corc	d we	eks	s - (I	rour	nd u	ip to)	V	Veek	S						
F3. /	3. Are you currently on Maternity/Sick Leave? \[Yes \] No 4. Were/are you self employed? \[Yes \] No																									
F4. V	4. Were/are you self employed?																									
F4a)	 4. Were/are you self employed?																									
	for?																									
F5.	Wh	at	was	s/is	yo	ur	job	tit	le?																	
F5b)	Но	w r	nan	ур	eo	ple	w	ork	at	th	e p	lac	e t	hat	t yo	ou i	ısu	ally	y w	orl	k?					
		1-2] 3-2	24	[2	25-4	99	[]5	00+	-	(Cro	ss (ONE	E bo	ox Ol	NĽ	Y)					
F5c)	Ar	e/w	ere	yo	u a	h: (0	Cros	ss C	NE	bo	хO	NĽ	Y)													
		Man	ager	-		Sup	oer\	/iso	r		Oth	ner	Emp	oloy	ee											
F6. I	Plea	se	list	the	e 2	or	3 ו	mai	in t	as	ks	γοι	л ре	erfo	orm	/pe	erfo	orm	ed a	at	wo	ork'	?			
													-			-										
				\neg																						
⊢ F7.	Ho	wr	nan	v h	ou	rs (do/	did	vo	u v	NOI	rk i	n a	tvi	oic	al v	ve	ek?	I,						<u> </u>	
		1					J. 41]		-	ייייייייייייייייייייייייייייייייייייי			-71												
	Ind	icat	e nu	mbe	er o	of ho	ours	; [].[







Only answer F8-F11 if currently working or stopped working less than one year ago.

F8. How long have/had you done this job?

Years

Months

F9. Where is/was your main place of work? (Cross ONE box ONLY)

□ Work mainly at or from home □ No regular place of work

If neither of the above ask 10 and 10a. Everyone should be asked F11.

F10. What is/was your main place of work?

Street

Town	

Postcode

F10a. How do/did you usually travel to work?

Cross ONE box ONLY indicating what is/was the longest part, by distance, of your usual journey to work.

Work mainly at or from home	Passenger in a car or van
□ Train	□ Taxi
Bus, minibus or coach	Bicycle
Motor cycle, scooter or moped	On foot
Driving a car or van	Other

Other - (please write in)

F11. How many days in a typical week do/did you go to work?

(Enter 0 if works mainly at or from home)

Days









About baby's father

*F12. Which best describes the sort of work the baby's father does?

If not in work now, please **cross ONE box ONLY** to show what work he did in his last main job.

- Modern professional occupations
- Clerical and intermediate occupations
- Senior managers or administrators
- Technical and craft occupations
- Semi-routine manual and service occupations
- Routine manual and service occupations
- Middle or junior managers
- Traditional professional occupations
- Self Employed
- Student/in training
- Does not work long term unemployed/ill health (one year or over).
- Don't know







*F13. Can you tell me which of these credits/allowances/benefits you and your husband/partner receive?

(Please cross ALL that apply)

- Child Benefit
- Child Tax Credit
- Working Tax Credit (Formerly Working Family Tax Credit and Disabled Person's Tax Credit)
- Income Support
- Disability Living Allowance (including Disabled Person's Tax Credit)
- Income tested Job Seekers Allowance (Unemployment)
- Housing Benefit/Rent Rebate/Council Tax Benefit
- Incapacity Benefit (Replaces Invalidity and NI Sickness Benefit)
- Pension Credit
- Carer's allowance (was Invalid Care Allowance)
- None
- Don't know
- Does not wish to answer

Any Other State Benefit Please specify below







*F14. This table shows income in weekly, monthly and annual amounts.

Which of the amounts on this list represents you and your husband/partner's, total income from all jobs, (full and part time), all tax credits, all benefits and all other sources and earnings after tax when all income is added together. (Cross ONE box ONLY)

Weekly Income after Tax	Monthly Income after tax	Annual Income after Tax	
Less than £25	Less than £108	less than £1,299	
£25 - £39	£109 - £175	£1,300 - £2,099	
£40 - £59	£176 - £259	£2,100 - £3,099	
£60 - £79	£260 - £350	£3,100 - £4,199	
£80 - £99	£351 - £433	£4,200 - £5,199	
£100 - £124	£434 - £542	£5,200 - £6,499	
£125 - £149	£543 - £650	£6,500 - £7,799	
£150 - £179	£651 - £775	£7,800 - £9,299	
£180 - £209	£776 - £917	£9,300 - £10,999	
£210 - £259	£918 - £1,125	£11,000 - £13,499	
£260 - £299	£1,126 - £1,333	£13,500 - £15,999	
£300 - £379	£1,334 - £1,667	£16,000 - £19,999	
£380 - £479	£1,668 - £2,083	£20,000 - £24,999	
£480 - £577	£2,084 - £2,500	£25,000 – £29,999	
£578 - £769	£2,501 - £3,333	£30,000 - £39,999	
£770 - £962	£3,334 - £4,167	£40,000 - £49,999	
£963 - £1,154	£4,168 - £5,000	£50,000 - £59,999	
£1,155 - £1,346	£5,001 - £5,833	£60,000 - £69,999	
£1,347 - £1,538	£5,834 - £6,667	£70,000 - £79,999	
£1,539 or more	£6,668 or more	£80,000 or more	
Does not wish to an	swer 🗌	Don't know	







The next few questions are about the sorts of things that some people have but which many people have difficulty finding the money for.

*F15. Do you or you and your husband/partner have?

(Cross ONE box ONLY in each row)

	Yes	I/we would like this but can't afford it at this moment	I/we do not want/need this at the moment	Does not wish to answer	Don't know
a) A holiday from home for at least one week once a year (not including staying with relatives in their home)					
b) Friends or family who call for a drink or meal at your house at least once a month					
c) Two pairs of all weather shoes					
d) Enough money to keep your home in a decent state of decoration					
e) Household contents Insurance					
 f) Money to make regular savings of £10 a month or more for rainy days or retirement 					
g) Money to replace any worn out furniture					
h) Money to replace or repair major electrical goods such as a refrigerator or a washing machine when broken					
 i) A small amount of money to spend each week on yourself (not on your family) 					
j) A hobby or leisure activity					
k) In winter are you able to keep your home warm enough					

*F16. Sometimes people are not able to pay every bill when it falls due. May I ask, are you up to date with the bills on this list or are you behind with any of them?

Interviewer: Show card with list of bills

ΠNο

☐ Yes

F16a) Are you up to date with all these bills? (Cross ONE box ONLY) Don't Know

Does not wish to answer





F16b) If no, which ones are yo	u beh	ind with? (Cross	ALL that apply)	
Electricity Bill		Telephone Bill		
Gas		Television/video/E	OVD rental or hire p	ourchase
Other fuel bills like coal or oil		Other hire purcha	se payments	
Council tax		Water rates		
Insurance Policies				
*F17. These questions apply in now. (Cross ONE box ONL)	f you Y in ea Yes	have any children ach row) Would like to have this but cannot afford this at the moment	living in your he Children do not want/need this at the moment	Does not apply
 a) Are there enough bedrooms for every child of 10 or over of a different sex to have their own bedroom. 				
The following questions apply to you	r childr	ren living with you		
b) Does your child/children have leisure equipment or a bicycle				
c) Does your child/children have celebrations on special occasions such as birthdays, or religious festivals				
d) Does your child/do your children go swimming at least once a month				
e) Does your child/children do A hobby or leisure activity				
f) Does your child/children have friends round for tea or a snack once a fortnight				
F18. If you have any children	age u	Inder 6 who are no	t in School	
(Cross ONE box ONLY)	Yes	Would like to have this but cannot afford this at the moment	Children do not want/need this at the moment	Does not apply
Does your child/children go to a toddler group/nursery/playgroup at least once a week				
F19. If your child/children is/ar	e ove	r age 6 or in schoo	ol.	
(Cross ONE box ONLY)	Yes	Would like to have this but cannot afford this at the moment	Children do not want/need this at the moment	Does not apply
Does your child/children go on school trips				
Page 24 of 42 Mother's questionnaire Ve	ersion 4	1 dt 14-09-2007		32949

	32949
\	

F20.	For children of all ages (Cr	oss ONE box Ol	NLY)	Does not
	Does your child/children have an outdoor space or facilities nearby where they can play safely	Yes	No	apply
F21.	How well would you say y financially these days. W	/ou or you and /ould you say	d your husband you are? (Cros	d/partner are managing s ONE box ONLY)
	Living comfortably	🗌 Finding it q	uite difficult	
	Doing alright	Finding it v	ery difficult	
	☐ Just about getting by	Does not w	ish to answer	

F22. Compared to a year ago, how would you say you and your husband/partner are doing financially now? (Cross ONE box ONLY)

Better off	About the same
------------	----------------

Worse off Does not wish to answer







Section G - Smoking/Alcohol/Drug Use

We apologise if any questions in this section cause offence - we are asking everyone the same questions but we realise some religions do not permit certain things.

SMOKING

G1. Have you ever regularly smoked cigarettes; that is at least one cigarette a day? (Cross ONE box ONLY)

	Yes for more than 1 year Yes for less than 1 year No
	If NO, go to question G4
G1a)	How old were you when you started smoking cigarettes?
	Age: Years old Don't Remember
G2.	Do you smoke cigarettes nowadays? Yes No (Cross ONE box ONLY)
G2a)	If no, when did you stop smoking?
	Age: Years old Don't Remember
G3.	How many cigarettes do/did you smoke during pregnancy, or in the three months before pregnancy? (Cross ONE box ONLY in each row)

	None	1-5 a day	6-10 a day	11-20 a day	Over 20 a day
a) 3 months before					
b) First 3 months of pregnancy					
c) Since the beginning of 4th mon	th 🔲				

G4. During pregnancy have you been exposed to other peoples' cigarette smoke at work or at home and if Yes, for how many hours per day approx? (Cross ONE box ONLY)

☐ Yes ☐ No ☐ Less than 1 hour per day/occasionally

If yes - Hours







G5. Have you used any other tobacco products like Paan during pregnancy, or in the 3 months before pregnancy? Interviewer: please show list of possible products.

(Cross ONE box ONLY)

Yes No Don't Know

If No, Don't Know or you don't remember go to question G6

*G5a) If yes please identify which ones and how many you smoke/chew etc., (relevant to point in pregnancy)

3 Months before pregnancy	Daily	Weekly	Monthly	Rarely	how many pe week
First 3 months of pregnancy					
From beginning of 4th month to now					

G6. Have you used any drugs like marijuana or ecstasy during pregnancy or in the three months before pregnancy? (Cross ONE box ONLY)

🗌 Yes

No

🗌 Don't Know

If No, Don't Know or you don't remember go to question G7







*G6a) If yes please identify which ones and how often you have taken them (relevant to point in pregnancy)

3 M	onth	s bef	ore	oreq	nano	ev.						Daily	Weekly	Monthly	v Rarely	lf 1+ per week, how many per week
Firs	t 3 m	onth	ns of	preç	gnan	су										
From	n be	ginni	ing c	of 4th	n mo	onth t	to no	w								

ALCOHOL		
G7. Did you drink any alcohol during your pregnan before? (Cross ONE box ONLY)	cy or in	the 3 months
Yes No Don't Remember If NO or don't r	emember	go to Section H
G7a) Did you drink any alcohol in the 3 months befo (Cross ONE box ONLY)	ore preg	nancy?
Yes, Once per week or more Yes, occasionally	🗌 No	Don't remember
		3294







If NO or don't remember go to question G7d)

G7b) If once per week or more, what is the weekly average and maximum number of units in a week?

	Average number of units per week	Maximum units at one time
Beer/Lager		
Wine		
Spirits		
Other		
Don't remember		

G7c) In the 3 months before pregnancy how often did you consume 5 or more units of alcohol on one occasion? (Cross ONE box ONLY)

Everyday	☐ 1-3 times a month	1		
Nearly every day	Rarely			
1-4 times/week	Never			
G7d) Did you drink any (Cross ONE box ONLY)	alcohol in the first	t 3 months o	of pregnan	cy?
Yes, Once per wee	k or more 🛛 Yes, c	occasionally	🗌 No	Don't remember
If NO or don't remember go	to section G7g)			
G7e) If once per week units in a week?	or more, what is th	ne average :	and maxim	um number of
	Average number of units per week	Maximum ur one time	nits at	
Beer/Lager				
Wine				
Spirits				
Other				
Don't remember				
				32949





G7f) In the first 3 months of pregnancy how often did you consume 5 or more units of alcohol on one occasion? (Cross ONE box ONLY)

Every day or more often	☐ 1-3 times a month
Nearly every day	Rarely
1-4 times/week	Never

G7g) Did you drink any alcohol from the beginning of the 4th month until now of your pregnancy? (Cross ONE box ONLY)

☐ Yes, Once per week or more	🗌 Yes, occasionally	🗌 No	🗌 Don't remember
------------------------------	---------------------	------	------------------

If NO or don't remember go to section H

G7h) If once per week or more, what is the average and maximum number of units in a week?

	Average number of units per week	Maximum units at one time
Beer/Lager		
Wine		
Spirits		
Other		
Don't remember	r 🗆	

G7i) Since the beginning of the 4th month of your pregnancy how often did you consume 5 or more units of alcohol on one occasion?

(Cross ONE box ONLY)

Every day or more often	☐ 1-3 times a month
Nearly every day	Rarely

□ 1-4 times/week □ Never





BREADS AND BREAD PRODUCTS

H1. During the last 4 weeks, on average how many slices/pieces of the following did you eat per week?

	Total Slices/Pieces	No. eaten as toast
a) White bread incl baguette		
b) Brown bread incl granary, multiseed, best of both. 50/50.	t	
c) Baps/rolls/teacake		
d) Crumpets		
e) Pizza (1 slice)		
f) Roti/Chappatis		
g) Naan, pitta bread, bagel		
h) Paratha		
Other bread products e.g. wraps, croissants, pancakes, flatbreads, English muffin		

Other please write in







CAFFEINATED DRINKS

H2. During the last 4 weeks of pregnancy, on average, how many cups or mugs of the following drinks would you have per day or per week?

(Glass is 200 ml Cup is 200 ml 1 Mug = 2 cups. If less than 1 per day enter weekly average)

How many cups of: ?	Per day	Per Week
a) Instant coffee (Caffeinated)		
b) Instant coffee (De-caffeinated)		
c) Filter/cafetiere coffee (Caffeinated)		
d) Filter/cafetiere coffee (De-caffeinated)		
e) Tea (Caffeinated)		
f) Tea (De-caffeinated)		
g) Kashmiri tea (Caffeinated)		
h) Kashmiri tea (De-caffeinated)		
i) Herbal/fruit teas (Caffeinated)		
j) Herbal/fruit teas (De-caffeinated)		
k) Cola (regular, with sugar Caffeinated)		
I) Cola (regular, with sugar De-caffeinated)		
m) Diet or sugar free cola (Caffeinated)		
n) Diet or sugar-free cola (De-caffeinated)		







SUPPLEMENTS/VITAMINS

- H3. Have you taken any dietary supplements including vitamins or iron tablets in the last 4 weeks of pregnancy? (Cross ONE box ONLY)
 - Yes No Don't Remember

H3a)	lf	Yes,	which:	
------	----	------	--------	--

	Daily	5-6 per week	2-4 per week	Once a week	Less Often
Vitamin C					
Vitamin D					
Vitamin E					
Iron					

Other vitamins and dietary supplements, please write in below:

		Do	on't l	Kno	W		Ľ														
lf n	nult	ivita	ami	ns:																	
	F	Preg	Inac	are	•		Ľ]			Ľ]					
	S	Sana	atog	jen	pre	nata	al []			Ľ]					
Otł	ner	mu	ltivi	tam	nins	s, pl	eas	e w	rite	e in	bel	ow									
	-	Dor	n't K	۲	N	-							 		 				 	 	







Section I Water Consumption

*I1. On a typical day how much of the following do you drink?

	At home	At work/study	Elsewhere
a) Tap water	Glasses per day:	Glasses per day:	Glasses per day:
b) Bottled water (Includes water cooler)	Glasses per day:	Glasses per day:	Glasses per day:
c) Tea (any sort)	Cups per day:	Cups per day:	Cups per day:
d) Coffee	Cups per day:	Cups per day:	Cups per day:
e) Squash (Including any othe made with tap wate	Glasses per day: er drinks er)	Glasses per day:	Glasses per day:

I2. Do you filter the water you drink at home? (Cross ONE box ONLY)

Yes No Don't Know

I3. Do you filter the water you drink at work? (Cross ONE box ONLY)

☐ Yes ☐ No ☐ Don't Know ☐ N/A

I4. In a typical week while you have been pregnant how often and for how long do you undertake the following?

(if you do not do any then fill in 0)









32949

Section J General Health

Interviewer to give questionnaire for this section to be self-completed.

We should like to know if you have had any medical complaints and how your health has been in general, over the past few weeks. Please answer ALL the questions on the following pages simply by putting a cross by the answer which you think most nearly applies to you. Remember that we want to know about present and recent complaints, not those that you had in the past. It is important that you try to answer ALL the questions.

Cross ONE box ONLY for each question - have you:

J1a. Been fee	usual Same as usual	good health? Worse than usual	☐ Much worse than usua
J1b. Been fee	ling in need of a good ton	ic?	Much more than usual
J1c. Been fee ☐ Not at all	ling run down and out of s ☐ No more than usual	sorts?	☐ Much more than usual
J1d. Felt that	you are ill?	_	
Not at all	No more than usual	Rather more than usual	Much more than usual
Not at all	No more than usual	Rather more than usual	Much more than usual
J1f. Been get	ting a feeling of tightness	or pressure in your head?	Much more than usual
J1g. Been have Not at all	ving hot or cold spells?	Rather more than usual	Much more than usual
J2a. Lost mud	ch sleep over worry? □ No more than usual	Rather more than usual	Much more than usual
J2b. Had diffi	culty in staying asleep on	ce you are off?	
☐ Not at all	No more than usual	Rather more than usual	Much more than usual
J2c. Felt cons	stantly under strain?	Rather more than usual	Much more than usual
J2d. Been get	ting edgy and bad- tempe	red?	
☐ Not at all	☐ No more than usual	Rather more than usual	Much more than usual
J2e. Been get	ting scared or panicky for	no good reason?	☐ Much more than usual



	₽₽₽₽┥
/	32949

J2f. Found everything getting on top of you?
□ Not at all □ No more than usual □ Rather more than usual □ Much more than usual
J3a. Been managing to keep yourself busy and occupied?
J3b. Been taking longer over the things you do?
🗌 Quicker than usual 🛛 🗌 Same as usual 📄 Longer than usual 📄 Much longer than usual
J3c. Felt on the whole you were doing things well?
J3d. Been satisfied with the way you've carried out your tasks?
☐ More satisfied ☐ About the same as usual ☐ Less satisfied than usual ☐ Much less satisfied
J3e. Felt that you are playing a useful part in things?
☐ More so than usual
J3f. Felt capable of making decisions about things?
☐ More so than usual ☐ Same as usual ☐ Rather less so than usual ☐ Much less capable
J3g. Been able to enjoy your normal day-to-day activities?
J4a. Been thinking of yourself as a worthless person?
□ Not at all □ No more than usual □ Rather more than usual □ Much more than usual
J4b. Felt that life is entirely hopeless?
J4c. Felt that life isn't worth living?
□ Not at all □ No more than usual □ Rather more than usual □ Much more than usual
J4d. Thought of the possibility that you might make away with yourself?
J4e. Found at times you couldn't do anything because your nerves were too bad?
☐ Not at all ☐ No more than usual ☐ Rather more than usual ☐ Much more than usual
J4f. Found yourself wishing you were dead and away from it all?
J4g. Found that the idea of taking your own life kept coming into your mind?
Page 36 of 42 Mother's questionnaire Version 41 dt 14-09-2007



Section K Exercise

Interviewer to give questionnaire for this section to be self-completed

K1. Please tell us about the type and amount of physical activity involved in your paid work. (Cross ONE box ONLY)

I am not in paid employment
I spend most of my time at work sitting (such as in an office)
I spend most of my time at work standing or walking. However my work does not require much intense physical effort (e.g. shop assistant; hairdresser; childminder)
My work involves definite physical effort including handling of heavy objects and use of tools (e.g. cleaner; hospital nurse; gardener, postal delivery worker)
My work involves vigorous physical activity including handling of very heavy objects.

K2. During the last week how many hours did you spend on each of the following activities?

(Cross only one box in each row)

	None	Some but less than one hour	1 hour but less than 3 hours	3 hours or more
a) Physical exercise such as swimming, jogging, aerobics, tennis, gym workout etc				
b) Cycling, including cycling to work and during leisure time				
c) Walking, including walking to work, shopping, for pleasure etc				
d) Housework/childcare				
e) Gardening/DIY (Do it Yourself	f) 🗌			

K3. How would you describe your usual walking pace?

Slow pace	Steady average pace
Brisk pace	Fast pace

Please return to the interviewer' - 'Thank you for completing this questionnaire





32949
Section Interviewer's feedback
L1. Was anyone present with Mother during the interview? (Cross ONE box ONLY)
Yes No Part of interview
L1a) If yes or part of interview: who was present? (Cross ALL that apply)
Baby's father Mother's friend
Mother's mother Relative
Mother's father
Other (please write in)
L2 Was a transliteration used to administer the questionnaire? Yes No Partially
L3 Were there any problems in completing this interview? \Box Yes \Box No
L3a) If yes, what were the problems
L 4. De veu feel confident with the ensurer provided 2 Ves No
L4 Do you feel confident with the answers provided?
L4a) If no, why are you not confident?
L4 Do you feel confident with the answers provided?
L4 Do you feel confident with the answers provided? L4a) If no, why are you not confident?
L4 Do you feel confident with the answers provided? Ves No L4a) If no, why are you not confident?
L4 Do you feel confident with the answers provided? L4a) If no, why are you not confident?
L4 Do you feel confident with the answers provided? Yes L4a) If no, why are you not confident? COMPLETED QUESTIONNAIRE - CHECKED BY STUDY ADMINISTRATOR
L4 Do you feel confident with the answers provided? Yes No L4a) If no, why are you not confident?
L4 Do you feel confident with the answers provided? Yes No L4a) If no, why are you not confident?
L4 Do you feel confident with the answers provided? Yes L4a) If no, why are you not confident? Completed QUESTIONNAIRE - CHECKED BY STUDY ADMINISTRATOR Yes ALSO CHECKED: M Diet J General Health K Exercise Interviewer Number (if applicable) -
L4 Do you feel confident with the answers provided? Yes L4a) If no, why are you not confident? L4a) If no, why are you not confident? COMPLETED QUESTIONNAIRE - CHECKED BY STUDY ADMINISTRATOR Yes ALSO CHECKED: M Diet J General Health K Exercise Interviewer Number (if applicable) -





Questionnaire about your diet

This short questionnaire asks you about the food you have eaten over the last four weeks of your pregnancy. You may not eat all the foods given or you may find that some of the foods you eat are not included – please do not worry but complete all of the question asked.

Please do not leave any of the lines blank and answer every question even if you are uncertain.

INSTRUCTIONS ABOUT HOW TO COMPLETE THE QUESTIONS

Please put a cross in each box to show how often you have eaten each food item. E.g. if you eat 4 slices of white bread a day – cross the box as shown below

FOOD ITEM	HOW	HOW OFTEN HAVE YOU EATEN THIS IN THE LAST 4 WEEKS?						
	Rarely or never	Less than 1 a Week	Once a Week	2-3 times a Week	4-6 times a Week	1-2 times a Day	3-4 times a Day	5+ a Day
White bread	0	1	2	3	4	5	X_{6}	7

If you make a mistake and cross the wrong box, just cross out and enter the cross in the correct box.

E.g. If you cross you had fruit juice 3 times a day when you meant 3 times a week just cross out the '3-4 times a day' answer and cross the '2-3 times a week' box.

FOOD ITEM	HOW	HOW OFTEN HAVE YOU EATEN THIS IN THE LAST 4 WEEKS?							
	Rarely or never	Less than 1 a Week	Once a Week	2-3 times a Week	4-6 times a Week	1-2 times a Day	3-4 times a Day	5+ a Day	
Fruit juice (not cordial or squash)	0	1	2	X ₃	4	5	X	7	





Ρ



M1. The following questions ask about some food and drinks you might have consumed during the last 4 weeks of your pregnancy. Do not be concerned if some things you eat or drink are not mentioned.

Please cross how often you eat at least ONE portion of the following foods & drinks: (a portion includes: a packet of crisps, a serving of chips, one bowl of cereal). (Please cross ONE box ONLY, but answer EVERY line even if you don't eat that food)

	Rarely or never	Less than 1 a Week	Once a Week	2-3 times a Week	4-6 times a Week	1-2 times a Day	3-4 times a Day	5+ Times a Day
a) Chips								
b) Roast or fried potatoes, hash browns or potato waffles								
c) Fibre or bran-rich wheat breakfast cereal, like Weetabix, Fruit 'n Fibre, Bran flakes								
d) Oat cereals including muesli, porridge, crunchy oats, instant hot oats								
e) Other breakfast cereals like cornflakes, rice krispies, Cheerios								
f) Crispbread, like Ryvita								
g) Pasta or noodles (also pot noodles, tinned spaghetti)								
 h) Savouries like Yorkshire puddings, dumplings, pakoras or bhajia 								
i) Potato crisps								
j) Other salted savoury snacks like tortilla chips, Wotsits, Quavers, Bombay mix								
k) Cakes, buns, gateaux, doughnuts, muffins								
I) Sweet pastries like fruit pies, Danish pastries, custard/curd tarts, croissants								
m) Chocolate bars and chocolate coated biscuits e.g. Twix, Kit-Kat, Dairy milk bar								
n) Sweet biscuits like digestive, custard creams, gingernut, shortbread								







M2. The following questions ask about types of meat and fish you might have consumed over the last 4 weeks of your pregnancy. Please cross how often you eat at least ONE portion of the following:

	Rarely or never	Less than 1 a Week	2-3 times a Week	4-6 times a Week	7+ times a week
Whole meats					
a) Beef - steaks, roasts, joints, or chops (not in sauce)					
b) Pork - steaks, roasts, joints, or chops (not in sauce)					
c) Lamb, mutton or goat					
d) Chicken or Turkey - steaks, roasts, joints, portions (not in batter, sauce or breadcrumbs)					
Processed meats/meat					
e) Meat sausages e.g. Walls or chipolata					
f) Beef burgers, either home cooked or takeaway					
g) Kebabs					
h) Hot dog, frankfurter or saveloy					
i) Bacon rashers					
 j) Meat pies and pastries (sausage roll, pasties, meat samosa, steak/meat pie) 					
k) Chicken/turkey nuggets, Kiev, turkey or chicken burgers, chicken pies, or in batter or breadcrumbs					
I) Ham					
m) Cured/dried sausage e.g. Chorizo, Salami					
Meat dishes					
n) Chicken or turkey with sauce e.g. curry, stir-fry, casserole					
o) Beef, lamb or goat in sauce e.g. curry, stew, Shepherd's pie, Bolognese sauce, Chilli con carne, Lasagne					
p) Pork in sauce e.g. stew, casserole or stir-fry					
q) Gravy made with pan or meat juices (not instant)					
Fish					
r) White fish in batter or breadcrumbs, like 'fish 'n chips'					
s) White fish not in batter or breadcrumbs e.g. cod in parsley sauce, fish curry (marsala fish), fish pie					
t) Tinned tuna					
 u) Fresh or tinned oily fish like sardines, mackerel, salmon, trout (not tuna) 					
 v) Smoked fish, like smoked mackerel, kippers or smoked salmon 					
w) Salted/dried fish e.g. 'Bombay duck'/bummalo					







M3. If eaten in the last 4 weeks of pregnancy how did you mainly cook the following?

Please enter only one cross on each line for cooking method. Cross yes if mainly eaten very well done, crispy or heavily browned as shown.

	Did not eat	Don't know or take-away	Grill	Fry	Roast	BBQ	Well d Yes	ione? No
a) Beef - steaks, roasts, joints, or chops (not in sauce)								
b) Beef burgers, either home cooked or takeaway								
c) Pork - steaks, roasts, joints, or chops (not in sauce)								
d) Lamb, mutton or goat - steaks, roasts, joints, or chops (not in sauce)								
e) Chicken or Turkey-steaks, roasts, joints, portions (not in batter, sauce or breadcrumbs)								
f) Meat Sausages e.g. Walls or chipolata								
g) Bacon rashers, chops or bacon ribs								
h) White fish fillets or steaks e.g. cod or haddock NOT in batter								
 i) Oily fish fillets or steaks e.g. salmon, mackerel, trout 								

M4. a. Are you familiar with the "5 a day" recommendations for fruit and vegetables?

☐ Yes ☐ No

b. Do you consume 5 portions of fruit and vegetables per day? (Please place a CROSS in ONE box ONLY)

Always Sometimes Never

(Please place a CROSS in ONE box ONLY)

M5. Where does most of your advice about healthy eating during pregnancy come from?

Family members	GPs/Doctors
----------------	-------------

🗌 Friends	Midwife/Health Visitor
-----------	------------------------

Magazines/Newspapers	Other
----------------------	-------

Books

Thank you for completing this questionnaire - please leave it in the place indicated.







Born in Bradford Data Transfer Agreement

This agreement is between the Born in Bradford study and

Recipient of data	Katie Marvin-Dowle
Institution	Sheffield Hallam University
Data request number/ study number	119
Study title	A comparative study of demographic and nutritional factors in relation to birth outcomes among pregnant adolescents and adult pregnant women

1. All investigators and other relevant employees involved in the research have read and will abide by the "Born in Bradford Guidance for Collaborators"

2. The data remains the property of the Born in Bradford study. This agreement does not restrict the rights of Born in Bradford to distribute the data to other institutions or to publish any document relating to the data.

3. The recipient will retain the data in a secure location on its premises and will not permit the data or any part of it to come into the possession or control of any other organisation or any individual other than those employees who are involved in the research study specified above under direct supervision of the recipient.

4. The recipient will not transfer the data in whole or in part to third parties without the relevant third party entering into a separate data transfer agreement with Born in Bradford.

5. The recipient will use the data only to carry out the research described in approved proforma relating to the study specified above and only for research that has appropriate ethnical approval. The recipient will not use the data or any parts thereof for any commercial purposes or any purpose that is subject to consulting or licensing obligations to third parties.

6. The recipient will use all reasonable endeavours to ensure that the data in its possession or under the control of the recipient shall as soon as possible be returned or destroyed upon (i) the reasonable request of Born in Bradford, (ii) on termination of this agreement, (iii) in the event that the recipient is in breach of any of the conditions of this agreement or (iv) the withdrawal of consent of the relevant study participant. If the recipient is required to destroy the data then it will confirm in writing to Born in Bradford that the data has been destroyed.

7. All information (including the results of chemical and biological analyses and cleaned or derived variables) relating directly to study participants will be incorporated into Born in Bradford study and shall be owned by the Born in Bradford study.

8. The recipient will provide Born in Bradford with a fully documented electronic copy of the full results before its publication in any form or within 6 months of the completion of the research whichever is sooner.

9. The recipient will keep the data confidential and will not try to identify study participants.

10. The recipients will not try to link the data provided by Born in Bradford to other Born in Bradford data held by different recipients or by the same recipient for different projects.

Signed for on behalf of the Born in Bradford study Signature:

Name:

Date:

Signed on behalf of the recipient

Signature: K.M.

Name: Katie Marvin-Dowle

Date: 20.04.2015

We require two signed original copies (not photocopies) one for our records and the other we will return to you

Please return to Born in Bradford Data and Research Team Bradford Institute for Health Research Bradford Royal Infirmary Duckworth Lane Bradford BD9 6RJ Appendix 4.

Born in Bradford letter of ethics approval



National Research Ethics Service

Bradford Research Ethics Committee Top Floor Extension Block St Lukes Hospital Little Horton Lane Bradford BD5 0NA

Chairman: Professor Alan C Roberts OBE TD DL MPhil PhD DSc DTech LLD FLS FIBiol Administrator: Susan Jude

Tel: 01274 365508 **Fax:** 01274 365509 **Email:** susan.jude@bradfordhospitals.nhs.uk **Email:** alan.roberts@bradfordhositals.nhs.uk

01 April 2008

Professor John Wright Director of Bradford Institute for Health Research Bradford Teaching Hospitals NHS Foundation Trust Bradford Institute for Health Research Bradford Royal Infirmary

Dear Professor Wright

Full title of study:

REC reference number:

Born in Bradford: A longitudinal cohort study of babies born in Bradford and their mothers and fathers 07/H1302/112

Thank you for your letter of 30 January 2008, responding to the Committee's request for further information on the above research and submitting revised documentation.

The further information was considered at the meeting of the Sub-Committee of the REC held on 10 March 2008. A list of the members who were present at the meeting is attached.

Confirmation of ethical opinion

On behalf of the Committee, I am pleased to confirm a favourable ethical opinion for the above research on the basis described in the application form, protocol and supporting documentation.

Ethical review of research sites

The favourable opinion applies to the research sites listed on the attached form.

Conditions of approval

The favourable opinion is given provided that you comply with the conditions set out in the attached document. You are advised to study the conditions carefully.

Approved documents

The final list of documents reviewed and approved by the Committee is as follows:

Document	Version	Date
Application		04 December 2007
Investigator CV		

This Research Ethics Committee is an advisory committee to Yorkshire and The Humber Strategic Health Authority The National Research Ethics Service (NRES) represents the NRES Directorate within the National Patient Safety Agency and Research Ethics Committees in England.

Protocol	4	01 December 2007
Covering Letter		07 December 2007
Letter from Sponsor		29 March 2007
Questionnaire: Mothers	41	14 September 2007
GP/Consultant Information Sheets	1	19 April 2006
Participant Information Sheet: Part 2	1	27 July 2006
Participant Information Sheet: Parents	3	24 July 2007
Participant Consent Form: Mothers	6	10 July 2007
Participant Consent Form: Fathers	6	10 July 2007
Response to Request for Further Information		30 January 2008
Data Collection Schedule	9	12 January 2007
DVD		(L)
Recruitment Flow Chart	6	26 October 2007
Husband/Partner Registration/Sample	1.1	11 September 2007
DVD Script		
Advertising material		

R&D approval

All researchers and research collaborators who will be participating in the research at NHS sites should apply for R&D approval from the relevant care organisation, if they have not yet done so. R&D approval is required, whether or not the study is exempt from SSA. You should advise researchers and local collaborators accordingly.

Guidance on applying for R&D approval is available from http://www.rdforum.nhs.uk/rdform.htm.

Statement of compliance

The Committee is constituted in accordance with the Governance Arrangements for Research Ethics Committees (July 2001) and complies fully with the Standard Operating Procedures for Research Ethics Committees in the UK.

After ethical review

Now that you have completed the application process please visit the National Research Ethics Website > After Review

Here you will find links to the following

- a) Providing feedback. You are invited to give your view of the service that you have received from the National Research Ethics Service on the application procedure. If you wish to make your views known please use the feedback form available on the website.
- b) Progress Reports. Please refer to the attached Standard conditions of approval by Research Ethics Committees.
- c) Safety Reports. Please refer to the attached Standard conditions of approval by Research Ethics Committees.
- d) Amendments. Please refer to the attached Standard conditions of approval by Research Ethics Committees.
- e) End of Study/Project. Please refer to the attached Standard conditions of approval by Research Ethics Committees.

We would also like to inform you that we consult regularly with stakeholders to improve our service. If you would like to join our Reference Group please email <u>referencegroup@nationalres.org.uk</u>.

With the Committee's best wishes for the success of this project

Yours sincerely Professor A C Roberts

Chairman

Enclosures:

Standard approval conditions Site approval form

e-mailed copy to:

R & D Department, BRI

Bradford Research Ethics Committee

Attendance at Sub-Committee of the REC meeting on 10 March 2008

Committee Members:

Name	Profession	Present	Notes
Dr MI Busby	Consultant Neurologist Bradford Teaching Hospital Foundation	Yes	
Dr Caroline Plews	Research Nurse	Yes	
Professor Alan Roberts	Chairman	Yes	

07/H1302/112

		Bradford Research	Ethics Committee		
	LIS	IT OF SITES WITH A FAVO	JURABLE ETHICAL OPINIC	N	
For all studies requiring sit following subsequent notifi	e-specific assessment, this t cations from site assessors.	orm is issued by the main F For issue 2 onwards, all si	EC to the Chief Investigator tes with a favourable opinion	and sponsor with the favou are listed, adding the new .	rable opinion letter and sites approved.
REC reference number:	07/H1302/112	Issue number:	0	Date of issue:	01 April 2008
Chief Investigator:	Professor John Wright				
Full title of study:	Born in Bradford:A longituo	linal cohort study of babies	born in Bradford and their m	others and fathers	
This study was given a fav sites listed below. The res	ourable ethical opinion by B earch may commence at ea	radford Research Ethics Co ch NHS site when manager	mmittee on 10 March 2008. nent approval from the relev	The favourable opinion is e) ant NHS care organisation I	tended to each of the nas been confirmed.
Principal Investigator	Post	Research site	Site assessor	Date of favourable opinion for this site	Notes ⁽¹⁾
Professor John Wright	Consultant in Clinical Epidemiology and Public Health	Research and Development Unit, Bradford Teaching Hospitals NHS Foundation Trust	Bradford Research Ethics Committee	01/04/2008	
Approved by the Chair on I (delete as applicable)	oehalf of the REC: 	chair/Co-ordinator)			

لىلى F



(Name)

(1) The notes column may be used by the main REC to record the early closure or withdrawal of a site (where notified by the Chief Investigator or sponsor), the suspension of termination of the favourable opinion for an individual site, or any other relevant development. The date should be recorded.
Appendix 5.

Ethical approval for Born in Bradford analysis from Sheffield Hallam University



Date 01042015

Research proposal number: HWB-HSC-29

Dear Katie Marvin-Dowle

This letter relates to your research proposal:

A comparative study of demographic and nutritional factors in relation to birth outcomes among pregnant adolescents and adult pregnant women

This proposal was submitted to the Faculty Research Ethics Committee with a standard SHREC 1 form. It has been added to the register of projects and given a reference number. You do not need any further review from the Ethics Committee. You will need to ensure you have all other necessary permission in place before proceeding, for example, from the Research Governance office of any sites outside the University where your research will take place. This letter can be used as evidence that the proposal has been registered within Sheffield Hallam University.

The documents reviewed were:

MARVIN DOWLE 2015Binder1.pdf

Good luck with your project.

Yours sincerely

Potor Allmark

Peter Allmark Chair Faculty Research Ethics Committee Faculty of Health and Wellbeing Sheffield Hallam University 32 Collegiate Crescent Sheffield S10 2BP 0114 224 5727 p.allmark@shu.ac.uk Appendix 6.

Participant information sheet

Version 2 November 2016



Participant information sheet

Study title:	Exploring the Need for Support in Adolescent
	Pregnancies
Chief investigator	Katie Marvin-Dowle
Telephone number	07816424950

Study Sponsor: Sheffield Hallam University

We would like to invite you to take part in our research study which forms part of a PhD project. Before you decide we would like you to understand why the research is being done and what it would involve for you. Talk to others about the study if you wish. Ask us if there is anything that is not clear. The purpose of this research is to find out whether young women need extra support to have healthy pregnancies, and if so gets some ideas about what the best ways to offer this might be.

Participant name:

You will be given a copy of this information sheet to keep

Exploring the Need for Support in Adolescent Pregnancies Version 2 November 2016

1. What is the purpose of this study?

The purpose of this study is to find out if young women (under 20 years old) need, or would like, extra support with having a healthy pregnancy. We are also interested in finding out how the professionals who support them might be able to help with this.

2. Why have I been invited?

You have been asked to take part in this study because your job role includes supporting the health and wellbeing of young people. You should also have some experience of working with young women during pregnancy.

3. Do I have to take part?

Your decision to take part in this study is entirely voluntary. You may refuse to participate or you can withdraw from the study at any time. Your refusal to participate or wish to withdraw would not have any consequences for you.

4. What will happen to me if I take part?

If you participate in the study you will be either asked to attend a focus group with other professionals whose job role is similar to yours or a one to one interview with a researcher. You will be asked to discuss questions relating to how you see the role of your profession in supporting young women to have healthy pregnancies.

5. Expenses and payments

You will not be paid for taking part in this study. Travelling expenses to and from the research venue can be reimbursed.

6. What are the possible disadvantages and risks of taking part?

There are no physical risks from taking part in this study; the main disadvantage is that we are asking you to give up some of your time. Sometimes people might find talking about health or pregnancy upsetting. We will try to make sure this doesn't happen by only asking for your opinions about things and not asking for any personal details. We will also make sure there is plenty of time for you to ask questions both before and after the focus group or interview and make sure more support is available if needed.

7. What are the possible benefits of taking part?

The main benefit to taking part in this research is being able to put forward your views and help inform the development of strategies or interventions to improve pregnancy health in young women. You might also find having time to discuss and reflect upon these issues with other professionals useful for your practice or professional development.

8. What if there is a problem or I want to complain?

Exploring the Need for Support in Adolescent Pregnancies Version 2 November 2016 If you have any queries or questions please contact:

Principal investigator:

Katie Marvin-Dowle

<u>katie.marvin-</u> <u>dowle@student.shu.ac.uk</u>

07816424950

Sheffield Hallam University, Faculty of Health and Wellbeing

Alternatively, you can contact my supervisor:

Prof. Hora Soltani

h.soltani@shu.ac.uk

If you would rather contact an independent person, you can contact Peter Allmark (Chair Faculty Research Ethics

Committee) <u>p.allmark@shu.ac.uk;</u> 0114 225 5727

9. Will my taking part in this study be kept confidential?

The focus group or interview will be recorded and then written up word for word. The researcher will check that the recording and the written transcript are the same. She will then erase the recording. The transcript will be kept on a password-protected computer. Identifying details will be taken out of any final report and any publication so people reading these will not be able to identify you. The written transcripts will have all links to you removed at the end of the study and will then be kept for as long as they might be useful in future research.

It might be that in the interviews something of concern arises relating to young women's care. If that happens, the researcher will consult with her supervisor to discuss what to

Exploring the Need for Support in Adolescent Pregnancies Version 2 November 2016

do. She will act in accordance with her professional Code of Conduct.

The documents relating to the administration of this research, such as the consent form you sign to take part, will be kept in a folder called a site file or project file. This is locked away securely. The folder might be checked by people in authority who want to make sure that researchers are following the correct procedures. These people will not pass on your details to anyone else. The documents will be destroyed seven years after the end of the study.

13. What will happen to the results of the research study?

The results of the study will be written up and included in the researchers doctoral thesis, which may then be available in the University library. The results may also be written up as a research article and published in an academic journal.

14. Who is sponsoring the study?

The sponsor of the study has the duty to ensure that it runs properly and that it is insured. In this study, the sponsor is Sheffield Hallam University.

15. Who has reviewed this study?

All research based at Sheffield Hallam University is looked at by a group of people called a Research Ethics Committee. This Committee is run by Sheffield Hallam University but its members are not connected to the research they examine. The Research Version 2 November 2016

Ethics Committee has reviewed this study and given a favourable opinion.

16. Further information and contact details

If you have any queries or questions please contact:

Principal investigator:

Katie Marvin-Dowle

<u>katie.marvin-</u> <u>dowle@student.shu.ac.uk</u>

07816424950

Sheffield Hallam University, Faculty of Health and Wellbeing

Alternatively, you can contact my supervisor:

Prof. Hora Soltani

h.soltani@shu.ac.uk

If you would rather contact an independent person, you can contact Peter Allmark (Chair Faculty Research Ethics

Committee) <u>p.allmark@shu.ac.uk;</u> 0114 225 5727 Appendix 7.

Participant consent form

Version 2 November 2016

Sheffield Hallam University

Participant consent form

Study title:	Exploring the Need for Support in Adolescent	
	Pregnancies	
Chief investigator	Katie Marvin-Dowle	
Telephone number	07816424950	

Dontinin out in one o	
Participant name	

	Please read the following statements and put your initials in the box to show that you have read and understood them and that you agree with them	Please initial each box
1	I confirm that I have read and understood the information sheet dated November 2016 for the above study. I have had the opportunity to consider the information, ask questions and have had these answered satisfactorily.	
2	I understand that my involvement in this study is voluntary and that I am free to withdraw at any time, without give any reason and without my legal rights being affected.	
3	I understand that data collected during the study may be looked at by responsible individuals from the Sponsor and the Research Ethics Committee, where it is relevant to this research. I give permission for these individuals to have access to my records.	
4	I agree to take part in this study	

Version 2 November 2016

To be filled in by the participant			
I agree to take part in the above study			
Your name	Date	Signature	

To be filled in by the person obtaining consent

I confirm that I have explained the nature, purposes and possible effects of this research study to the person whose name is printed above.

Name of investigator	Date	Signature

Filing instructions	
 copy to the participant original in the Project or Site file copy in the medical notes (if applicable) 	

Appendix 8

Ethical approval for qualitative interviews from

Sheffield Hallam University



Sheffield Hallam Centre for Health and Social Care

19 July 2016

Katie Marvin-Dowle Sheffield Hallam University Centre for Health and Social Care Research

Research proposal number: 2015-16/HWB/HSC/35

Dear

This letter relates to your research proposal:

Professional perspectives on supporting adolescent pregnancies

This proposal was submitted to the Faculty Research Ethics Committee with a standard SHUREC1 form. This indicates that your project does not require formal ethics and scientific review. As such, it has been added to the register of projects and given a reference number. You do not need any further review from the Ethics Committee. You will need to ensure you have all other necessary permission in place before proceeding, for example, from the Research Governance office of any sites outside the University where your research will take place. This letter can be used as evidence that the proposal has been registered within Sheffield Hallam University.

The documents reviewed were:

Binder1.pdf

Good luck with your project.

Yours sincerely

Potor Allmark

Peter Allmark Chair Faculty Research Ethics Committee Faculty of Health and Wellbeing Sheffield Hallam University 32 Collegiate Crescent Sheffield S10 2BP 0114 224 5727 p.allmark@shu.ac.uk

Centre for Health and Social Care Research Faculty of Health and Wellbeing | Montgomery House | 32 Collegiate Crescent | Sheffield | S10 2BP | UK Telephone +44 (0) 114 225 5854 | Fax +44 (0) 114 225 4377 chscr@shu.ac.uk | www.shu.ac.uk/chscr

Appendix 9.

Interview Topic Guide

What do you think are the main issues for pregnant young women in relation to their health?

Describe how you see your role in supporting young women to have healthy pregnancies

What do you think are the key barriers for young women? What do you think are motivating factors?

How important do you think nutrition is to pregnancy health?

How important do you think young women feel what they eat is?

How confident do you feel giving advice to young women on health topics? On nutrition?

Are there any areas you would like additional training/information on to improve your confidence?

What format would you like to receive information in?

Do you have anything to add?

Appendix 10

Dietary habits and supplementation practices of young women during pregnancy: an online cross-sectional survey of young mothers and health care professionals.

Open Access



Dietary habits and supplementation practices of young women during pregnancy: an online cross-sectional survey of young mothers and health care professionals

Hora Soltani^{1*}, Alexandra Duxbury¹, Rachel Rundle² and Katie Marvin-Dowle³

Abstract

Background: Nutrition is a modifiable factor affecting birth outcomes, particularly in adolescent pregnancies. This study explores diet and supplementation practices, information and advice before, during and after pregnancy from the perspectives of pregnant or new young mothers and healthcare professionals.

Methods: Two cross-sectional surveys used online questionnaires for young women who were currently pregnant or who had recently given birth, and health care professionals providing antenatal care. The surveys utilised a combination of question types including free text and multiple choice. Recruitment was conducted via the Tommy's website, online forums for young mothers and professional networks.

Results: A total of 205 young women and 146 health care professionals were included in the study. Most young women reported taking supplements at some stage of pregnancy (93.2%), with 54.6% taking it on a daily basis. Those who reported taking supplements less than 7 days a week stated it was mainly due to forgetting. Health care professionals however reported that some young women had difficulties accessing healthy start supplements. Young women reported positive dietary changes; however a significant proportion of participants indicated that they avoided some foods unnecessarily. Avoiding or reducing foods such as red meat (22.7%), eggs (40.6%), oily fish (60.4%) and soft cheese (36.2%) is of concern. Midwife/family nurse (38.0%) was young women's current favourite information source; smartphone applications (apps) and recipe booklets were suggested by over 50% of participants as a new addition to existing services. Health care professionals reported they included nutritional information and support as part of their role; however they felt there were some gaps in knowledge and confidence. Midwives in particular suggested a lack of sufficient time and resources as a main challenge in providing adequate support. (Continued on next page)

* Correspondence: h.soltani@shu.ac.uk

¹Maternal and Infant Health, Centre for Health and Social Care Research, Sheffield Hallam University, Sheffield, UK

Full list of author information is available at the end of the article



© The Author(s). 2017 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

(Continued from previous page)

Conclusions: Avoiding or reducing major food groups was reported which can indicate a poor dietary pattern. A positive change in dietary intake reported by the majority of young women in this survey indicates willingness to adopt a healthy lifestyle. This, in addition to their trust in health professionals particularly midwives, provides an opportunity for health interventions which support behaviour change to improve birth outcomes. Identified gaps in knowledge and confidence by health professionals in providing dietary advice highlight the need for some additional training for health professionals in delivering dietary and lifestyle behaviour change interventions. Independent and trustworthy online resources for women and their health professionals which can be accessed at any time to provide up to date information in between appointments are also required.

Keywords: Nutrition, Supplements, Adolescent pregnancy, Midwives, Healthy eating information sources, Young mother, Digital

Background

A healthy diet and lifestyle during pregnancy is essential to ensure fetal growth and development is optimised. This is of a particular importance for young adolescent mothers in whom poor nutrition [1] and adverse outcomes such as stillbirth, preterm birth and low birthweight have been frequently reported [2–4]. Adolescent pregnancy (birth among women aged 15–19 when the pregnancy ended) has been considered a major contributor to maternal and child mortality worldwide [5] and is a major public health concern in United Kingdom (UK) with a rate of 19.7 births per 1000 women [6].

Young or adolescent mothers have specific requirements for macro- and micronutrients to support fetal development in addition to their own growing needs [7]. Systematic evidence [1] in addition to the recent UK 'National Diet and Nutrition Survey' (NDNS) [8] highlight that many adolescents' diets do not provide the recommended intakes for iron, zinc, calcium and folate and that those with the lowest incomes had significantly lower mean intakes than those from the highest income groups. Spending money on healthy food is often a lower priority compared to other fixed financial pressures such as housing, transport and utility bills [9]. With many competing priorities, it can be a challenge for health professionals to support women in making positive dietary changes and providing practical advice such as guiding on recipes that are suitable for circumstances with limited budget, cooking skills or facilities.

The Department of Health guidance and National Health Service (NHS) Choices [10] website provide standard healthy eating and food safety information for pregnant women in the UK. However most of the routinely provided information is not tailored to the specific nutritional requirements of adolescent mothers or in an adolescent friendly format. Midwives traditionally give every pregnant woman 'The NHS Pregnancy Guide', but the 200 page printed book can now only be read online so is likely to be only read by the most motivated of mothers with internet access. Tommy's, a UK baby charity, has produced the 'Young Woman's Guide to Pregnancy' which provides dietary advice in a booklet format more accessible to adolescents, but distribution of this free resource by midwives is variable [11]. It is therefore paramount to explore where and how young women access information and their preferences so that future resources are more appropriate to their needs.

Young mothers in the UK are entitled to standard antenatal care, although they are more likely to present for their first appointment later and miss appointments [12]. Various models of care are provided for adolescent pregnant mothers. In some areas, specialist teenage pregnancy midwives provide all or some of their care which may involve slightly longer appointment times and signposting to other services such as parenting classes. The Family Nurse Partnership (FNP) scheme provides intensive support for the young pregnant mother from 14 weeks and family for up to 2 years postnatal but only the most vulnerable are eligible and it is not available nationally. Understanding how young women are supported and guided regarding nutrition and healthy eating during pregnancy, and how they would like to be supported in future is essential to optimise care and enhance birth outcomes for this vulnerable group of women.

A previous study surveying adolescent's views of healthy eating during pregnancy in the USA [13] found that while young women reported that they considered healthy eating to be important the majority of their snacks consisted of unhealthy foods. The results from this study are however unclear in terms of the foods eaten at meal times and the frequency of consuming fruit and vegetables. There are no previous studies of this type undertaken in the UK, nor have any studies been conducted which include exploring the views of health professionals' dietary advice and support for this particular vulnerable group of women.

In an in-depth qualitative study [11] we have explored dietary patterns and supplementation adherence, and the facilitators and barriers in achieving a healthy diet and lifestyle during adolescent pregnancy. This qualitative study suggested that many young mothers have a dietary pattern consisting of high fat snacks, carbonated drinks and occasional meals, but their diet often depends on their living arrangements with those still at home more likely to have more traditional eating habits. A larger study was required to verify these provisional findings with a specific focus on what, if any, changes young women made to their diets during pregnancy; what motivated them to make such changes; where they obtained information and how in future they would like to receive information and support regarding diet and supplements during pregnancy. This study therefore aimed to explore diet and supplementation practices, information and advice during pregnancy from the perspectives of adolescent pregnant or new mothers and their health care professionals.

Methods

Two cross-sectional surveys were developed, one using a questionnaire which was specifically designed, piloted and administered for young women and the other for health care professionals. The questionnaires were informed by the findings from our previous qualitative study on exploring views of adolescent pregnant women and their health care professionals on pregnant adolescents' dietary pattern and, facilitators and barriers for improvement [11].

Ethical approval was obtained from Sheffield Hallam University Research Ethics Committee. An information sheet was provided at the beginning of the questionnaires, providing information about the study, contact details for further information. Young women were invited to participate if they were aged between 16–20 years of age at the time of their pregnancy, with a recent pregnancy experience. Health care professionals were invited to participate if their role included providing antenatal care to adolescent pregnant women. Consent was assumed inherent for the respondents who completed the questionnaire voluntarily.

Survey design: Questions were drafted and formatted in SurveyMonkey and results downloaded for analysis into SPSS (Version 23). The young women's questionnaire was designed to gather data from young women who were currently pregnant or those who had been pregnant recently, so care was taken to word the questions and question logic in order to capture all possibilities. There were a total of 26 questions in the young women's survey and 25 in the survey for health care professionals both using a mixture of question types, including some free text questions and some where participants selected one from a list of options and matrices. Demographic data was collected from the young women including the first part of their postcode, age and number of pregnancies and, job role, grade and year of qualification were collected from health care professionals.

Questions for young women focussed around changes to their diet and supplement use, beliefs about why they made changes, current information sources and suggestions for future resources or services. Questions for health care professionals focused on when and how information and advice on diet, nutrition and supplements was offered to young women and their confidence in giving this advice.

Piloting

The surveys were developed in consultation with the young women at the local Young Men Christian Association (YMCA: a global organisation aiming to put principals into practice in order to promote healthy body, mind and spirit) training centre and health professionals at their place of work. The questionnaire was also sent to the project steering group including experts in the field whose comments and suggestions were used to further refine the survey. The survey was updated based on the feedback received such as providing examples where terms identified as complex (e.g. for the term red meat we provided examples for clarity) and reduced the length of time required for questionnaire completion to 10-15 min.

Sampling/Recruitment

Tommy's (the UK based baby charity) facilitated dissemination of the online survey through its well-established networking systems with a wide range of organisations and support groups for young women and their professionals in the UK. The survey was distributed by email and website links on the 23rd of November 2013 from which participants were invited to the study. Implicit consent was presumed by participants following a link and completing the questionnaire.

Data analysis

Descriptive statistics are reported for all demographic data and for closed answer questions including proportions, means, standard deviation (SD) and ranges as appropriate. Average score or proportions in percent are presented for individual survey items. A simple content analysis was used for the open ended questions to establish categories.

Results

Young Women's Questionnaire.

Respondent characteristics

The sample consisted of 212 young women who started the survey with 162 completing it (76%), most within 48 h of it being posted online. Of 212 original participants, five were excluded as they indicated they have never been pregnant and two because they were under 16 years of age. Characteristics of participants in the survey are presented in Table 1. About 74% of the respondents were over 20 years and 26% aged 16–20 years.

The majority of respondents (94.6%) described their ethnicity as White British and a small proportion identified themselves as Black, Asian or Mixed race (Table 1).

About 60% of participants had only one pregnancy and the remaining had two or more pregnancies. Fiftyeight women were currently pregnant, ranging from 3 to 40 weeks (mean 25.6, SD10.2) with 30 as their first pregnancy.

Valid postcode district data was available for 180 participants, of whom 141 were resident in England, 20 in Scotland, 15 in Wales and 4 in Northern Ireland [14].

Use of supplements

Young women were asked if they had taken any supplements before, during and/or after their pregnancy, with 93.2% reporting taking supplements at any stage of pregnancy and 5.8% stated they have never taken

Table 1 Characteristics of young women participating in the study (N = 205)

	Number	Percent
Age		
16–19 years	23	11.2
20 years	31	15.1
Over 20 years	151	73.7
Ethnicity		
White	194	94.6
Black	2	1.0
Asian	4	2.0
Chinese/other	0	0.0
Mixed	2	1.0
Not stated	3	1.5
Gravida		
Gravida 1	124	60.5
Gravida 2	50	24.4
Gravida 3 or more	29	14.1
Missing	2	1.0
Living arrangements		
Living with partner	131	63.9
Living with my family	36	17.6
Living with my partner's family	8	3.9
Living alone	26	12.7
Living in shared accommodation/hostel	2	1.0
Missing	2	1.0

supplements during pregnancy. Folic acid was the most commonly used supplement with 75.1% stated having taken it at some stage in their pregnancy, followed by pregnancy multivitamins (42.9%) and iron supplements (40.0%) also being prevalent (Fig. 1).

A total of 84.5% took a folic acid supplement before or during their pregnancy, although the remainder might be getting their folic acid from Healthy Start (18.6%), a combined form of iron and folic acid (19.0%) or pregnancy multivitamins (48.3%) in this period.

Out of 205 participants, 54.6% stated that on average they took supplements daily, 24.9% took it 5–6 days a week, 12.2% 1–4 days a week, 4.4% never and 1% didn't know. Of 87 (out of 205) who reported taking supplements less than 7 days a week, 88.5% said it was mainly due to 'forgetting' and only a few stated it to be because of other reasons such as making them feel unwell, being too expensive, not liking to take tablets or not knowing what they are for.

Food choices and dietary changes

When young women were asked if they made changes to their diet during pregnancy 78.0% indicated yes, 17.6% said no, 0.5% (1 participant) said don't know and the remaining (3.9%) did not answer this question. As presented in Fig. 2, the respondents reported consumption of more fruit (53.2%), vegetables (51.2%), milk (41.5%) and breakfast cereal (33.7%). However a considerable proportion of participants indicated that they ate less or never ate red meat (22.9%), eggs (52.2%), oily fish (61.0%) and soft cheese (68.8%).

Table 2 shows that for many young mothers, the main contributing factor for changing to a healthier diet and supplement intake during pregnancy is the connection between healthy diet, supplementation and its impact on foetal growth and development.

Sources of information and support

As presented in Fig. 3, a great majority of young women remembered receiving information on food to avoid (76.6%) but only around a third or less of them stated that they received information on increasing calcium intake, safe intake of oily fish or weight gain. When they were asked about the main current sources of healthy eating information, they identified midwives (68%), Bounty pack (50%), family (39%) and websites (e.g. Baby Centre, YouTube) (37%) which reflected the state of practice.

When asked to rank their favourite sources of information from a list of 6 (Fig. 4), midwives and family nurse practitioners were the most popular with 38.0% selecting it as their favourite and parenting classes were the least popular (2.9%).

The average rating scores are also calculated from the proportion of respondents who ordered the 6 information





Table 2 Participants who agree to each statement regarding their reasons for eating a healthy diet or supplement use during pregnancy (N = 205)

	Eating well during pregnancy will	Taking supplements during pregnancy will
Help me to gain less weight	51.7%	3.9%
Help my baby grow properly	77.6%	71.7%
Help my baby to be healthier after it is born	76.1%	69.8%
Help my baby be more intelligent	20.5%	22.9%
Please my midwife/family nurse	51.2%	59.5%
Please my partner/family	39.0%	45.4%
Cost a fortune	25.9%	31.7%
Be hard work	23.4%	14.6%
Prepare my body for breastfeeding	62.4%	56.1%

source options based on their preferences (Fig. 5), confirming midwives/FNPs as the highly popular source of information (2.1), theoretical sources of information (Bounty pack, leaflets and websites) in the middle with an average score around 3, followed by partner/family and antenatal classes (scored around 4) to be the least favourite options for this group of women.

The free text responses about participants' views on favourite information sources are summarised in Table 3 under major categories relating to two major themes of advantages and disadvantages. The main advantages of receiving information from health care professionals were indicated to be related to its trustworthiness, reliability, being face to face and that it is through building relationship. The respondents however, highlighted issues such as lack of sufficient information or time as disadvantages for health care professionals to be able to communicate information effectively. There were also those who liked books or websites where they could access information in their own time or Bounty pack free samples. Some had concerns about the accuracy of information on websites or lack of a universal access to internet. Least popular information sources included







Sources of information	Advantages	Disadvantages
Health care professionals: Midwife or Family Nurse Partnership	 Trust in Professionals "My midwife has been doing her job for a long time and I trust her, her knowledge and her judgement" "Midwives are qualified and know exactly what they're talking about." Accurate and reliable advice "Midwife know everything what is needed for a healthy pregnancy, she was easy to talk to and very informative" "Midwife/Family nurse will have knowledge from training, be more understanding of personal circumstance." Face to Face information/questions "Face to face. Factual and can ask questions and get answers immediately!" "The midwife discusses it in more detail." Continuous relationship through pregnancy "My first two pregnancies I had the same midwife, I developed a bond with her in this time and felt I could trust the information that has been given to me." "More personal; through experience; they know you better" 	 Not getting sufficient information "I want information off the midwifes but they haven't given me any" Appointment length short "Pre booked appointment to talk more than seven minutes".
Self- Study: Leaflet/Booklet/Tommy's Guide	• Easy to read in own time "I can read through in my own time" "Can keep for reference"	• Lose leaflets "It's easy to lose bits of paper like booklets and guides" "They are easy to misplace and lose"
Bounty pack	 Good timing and free goodies "They are given to you at an early stage" "Creative and lots of goodies with it" Universal "Everyone gets a bounty pack so everyone gets same info." 	 Too much information/reading "I would not read a whole book on how to do something" Advertising & Commercial (Bounty) Bounty is a marketing scheme so sometimes sceptical on information as they may be trying to sell things to you
Websites/Social media/ Apps	 Accessible anytime "I can gain the information straight away on my phone instead of having to wait to see your midwife to hand you tonnes of stupid pointless leaflets!" "Its easy to get hold of the info 24/7" "It's easy access, for instance if I'm not sure wether a certain food is safe to eat during pregnancy I try to look online for the answer." "As my social media sites are always on so I can easily access it wherever I am" "You can access it anytime without feeling you are constantly hassling someone" 	 Unreliable and contradictory information "Would have liked more information from midwife as you feel it's trusted information coming from a healthcare provider, whereas on the internet you can receive conflicting information" "May not be a reliable source" "Cannot always trust the Internet as some sites give opposite advice" "Some websites can contradict what you have read on other websites" Not accessible to all "Not everybody has access to them" "My phone and tablet don't work very well so I was unable to use apps"
Social support: Partner/Family/Friends	 Available from day 1 Family cos you change your habits of eating unhealthy the day you find out your pregnant Having experience and knowing you "family told me information because they have loads of experi with preganices and what to do before getting pregnant with vitamins and eating and drinking after your body and baby" 	 Interfering and patronising "I couldn't have family advising me, I would find it patronizing." "Gets frustrating with family interfering" "Don't like being told what to do" Outdated/inaccurate advice "What family friends and partners tell you is not always true and is mainly their opinion" "They all have different views", "A lot of advice you get giving my family, friends etc. are normally 'old wives tales' that are untrue."

Table 3 Young women's views about sources of information in pregnancy

Table 3 Young women's views about sources of information in pregnancy (Continued)

Parenting Classes • Social way to meet other mums • Con	ntent too general er told us about healthy eating focus was on labour deliver."
and a Feat sessic "Wen and a "Othe have • Tim "I did "Not "Dislik	ar/experience of being judged by other mums and on leaders it once felt inadequate, woman was too patronizing didn't return" er mums and leader judging different lifestyles, wouldnt t the confidence to speak in one of those classes" ne to attend dn't have time to attend parenting classes when pregnant" everyone goes to classes and theres alot to take in" ike the parenting class atmosphere."

parenting classes which some respondents thought were patronising, hard to access or felt judged.

Finally, young women were asked to choose three options from a list of 9 suggestions for developing a new supporting tool to provide information about eating a healthy diet and supplements during pregnancy. These suggestions came from the earlier qualitative study and were included for verification in this study with a larger sample. Table 4, shows that the apps (52.7%) and recipe booklets (51.2%) were the most popular options for future developments.

Health care professionals questionnaire

Respondent characteristics

The 146 respondents were mainly midwives (hospital based, community, specialist teenage pregnancy), FNPs (with health visitor, nursing or midwifery background) or health support workers, qualifying between 1972 and 2012 (Table 5).

Table 4 Responses to the question "*If we could develop* something new to provide information about eating a healthy diet and supplements to support young pregnant women, what do you think would be best" (N = 205)

	Young women's choices
App containing healthy eating information	52.7%
Recipe booklet showing healthy meals and snacks to be eaten during pregnancy	51.2%
Education session to be delivered at antenatal/parenting classes	31.2%
Young woman specific booklet	30.2%
Education session to be delivered in secondary schools/colleges	22.4%
Wallet sized fact sheet/reference card	17.6%
Young woman specific online forum	15.1%
Poster summarising key facts	11.2%
YouTube videos about healthy eating during pregnancy	6.8%
Telephone or email helpline	6.8%

FNPs were mainly Band 7/8 and midwives were mainly 6/7. Geographical location was not explicitly asked but qualitative answers suggest one respondent works in the Channel Islands (where Healthy Start is not applicable), one near Castleford and others mentioning schemes in London boroughs.

This paper focuses on midwives and FNPs (90% of respondents) as they provide different models of care to young mothers during the antenatal period, and can be related back to findings from the interviews.

Use of supplements

Nearly all (98.4%) of HCPs provided information about pregnancy supplements for young women, however there were differences in how this information was delivered depending on job role. Midwives were more likely than FNPs to discuss current supplement use and provide a bounty pack whereas FNPs were more likely to provide young women with a copy of the Tommy's 'Young Woman's Guide to a Healthy Pregnancy and other leaflets and booklets and were more likely to help with goal setting or provide worksheets. Differences in how professions provided information on supplements are summarised in Fig. 6.

When asked if they had any comments on young women accessing Healthy Start vouchers and vitamins both midwives and FNPs reported issues, particularly with application, redemption of vouchers, eligibility criteria being too restrictive, lack of supply in local area and access to collect. In some areas they could hand bottles directly to women and this was seen as a good

Table 5 Job role of health care professionals responding to the questionnaire (N = 146)

	Number	Percent
Any midwife	46	32
Family Nurse Practitioner (FNP)	84	58
Health visitor	2	1
Other professional	1	1
Any support worker	12	8





system. Others noted that it is those clients who are most motivated that take supplements and so they are likely to buy them from a supermarket to have immediately, rather than apply and wait for their vouchers to come through. Discussions regarding supplements and healthy eating were more likely to be prompted by documentation earlier in the pregnancy with midwives, usually at the booking visit (their longest appointment). Young women are not referred to FNPs until later in the pregnancy (approximately 14 weeks), however prompts to discuss diet and nutrition then occurred at most visits as shown in Fig. 7.

Food choices and dietary advice

Information about healthy eating during pregnancy was provided by 98% of midwives and 100% of FNPs, but how this information was provided differed. The FNPs have more time compared to midwives for goal setting (87% vs 14%) and provided more physical resources such as Tommy's Young Woman's Guide to Pregnancy (96% vs 57%) providing other leaflets (94% vs 71%) and worksheets (88% vs 1%) which form a major part of the FNP programme. Bounty packs were handed out by 48% of midwives but no FNPs.

The food topics discussed with young women by health care professionals are shown in Fig. 8. All 13 food topics were covered by at least 70% of FNPs, but only six (healthy eating, regular meals, iron rich foods, food safety, foods to avoid and caffeine) were covered by at least 70% of midwives. Those less often covered by midwives were energy drinks, healthy snacks, 5 A Day, fortified foods, calcium foods, oily fish and weight gain.

Confidence and training

Health professionals claimed they felt confident discussing healthy eating and most supplements, however were less confident discussing omega 3, multi vitamins and



gestational weight gain. This was particularly true for FNPs and is reflected in the topics that they would like further training on. Overall the demand for additional training was relatively low however the topics which professionals in different roles would like training in differed substantially. There was a higher demand for training among FNPs, particularly on Omega-3 fatty acids and gestational weight gain, whereas midwives most commonly requested training on vitamin D as shown in Table 6.

All FNPs and 85% of midwives feel they adequately cover nutrition during pregnancy. The midwives who thought they didn't adequately cover it stated time pressure (4) and lack of clear guidelines as reasons.

When asked where they accessed information on diet and supplements, they reported accessing dietary advice from most sources listed (90–100%) including dieticians, internal and external training, journals, media and slimming clubs; while less sources were used to access supplement information (17–55%). Only 38% of midwives and 55% of FNPs learnt about supplements in prequalification training, however this may be due to some qualifying decades ago when the recommendations were different.

Discussion

This cross-sectional study of young mothers and their health care providers gives an insight into the pattern of dietary changes, supplement intake, current sources of information and preferred information sources in addition to the reasons behind their preferences for information support. Addressing eating behaviour and nutritional requirements of adolescent pregnant women in order to enhance their pregnancy and birth experiences is important as an association between poor nutritional status and adverse pregnancy outcomes is well documented [2–4, 15, 16]. This survey allowed a wider verification of the identified issues in a series of in-depth explorative interviews of young women and health care professionals from Doncaster, Manchester and London, details of which are reported elsewhere [11].



These surveys which included responses from all regions in England as well as Scotland, Wales and Northern Ireland showed that young mothers are motivated and keen in adopting healthy dietary changes as indicated in their responses for making positive changes during pregnancy. However it was important that a considerable proportion of responding women were not consuming red meat, eggs, oily fish and soft cheese (such as cream cheese or spreadable cheese). It was encouraging to see a high level of interest and enthusiasm for making positive changes however such a high proportion of adolescents indicating avoiding foods which are potentially valuable sources of vitamins and minerals, such as B vitamins and iron (red meat and eggs) or vitamin D and calcium (oily fish and soft cheese) [17] are of concern. Attention to adequate nutrition in supplying sufficient amount of nutrients are increasingly realised due to its impact on healthy pregnancy and birth outcomes [18].

The poor dietary habit identified in this study is in line with other investigators' findings showing a poor nutritional intake in young mothers [1, 8]. The low intake of foods such as soft cheese and eggs may suggest inadequate communication with young women and a state of confusion regarding prohibiting raw/runny eggs or unpasteurised mould ripened soft cheeses rather than avoiding all types of eggs or pasteurised soft cheeses. It is likely that food avoidance be due to personal taste or cultural restrictions in which case health professionals' role in guiding for alternative sources becomes essential for all particularly this group of women.

It is therefore of prime importance to provide sufficient support and appropriate education for health professionals who have direct contact with young pregnant mothers to enhance their knowledge and dietary behaviour with the overall aim of optimising pregnancy and birth outcomes. This is particularly noteworthy, considering the level of awareness and appreciation of the role, and the trust these young women place in their health professionals, especially in midwives, to obtain nutritional information. This ultimately affirms the need for equipping health professionals with appropriate support, skills and knowledge for effective communication of nutritional messages during pregnancy and postpartum.

The majority of midwives and family nurse practitioners felt that they were providing nutritional support to young women during pregnancy as part of their role but acknowledged there were some areas where they lacked confidence and would like further training. The role of midwives in promoting dietary changes has been explored in previous studies and it has been shown to be central in supporting women to make positive changes [19, 20]. The qualitative interviews with young women and health professionals undertaken as part of the earlier stages of this study however found that while most health care professionals provided some nutritional

Table (6 Responses to	the guestion	"How confident do	vou feel	discussing the	following topics wit	h young women?"	(N = 108)
						<u> </u>	,,	(,

	Midwives			FNPs		
	Confident	Not confident	Would like training	Confident	Not confident	Would like training
Folic Acid supplements	97%	3%	0%	88%	3%	7%
Good dietary sources of Folic Acid/Folates	89%	9%	3%	87%	8%	4%
Vitamin D supplements	86%	6%	9%	89%	8%	5%
Other sources of Vitamin D	71%	21%	9%	91%	7%	4%
Iron supplements	97%	0%	3%	92%	7%	3%
Good dietary sources of Iron	97%	0%	3%	99%	0%	1%
Healthy Start	94%	6%	0%	97%	0%	3%
Pregnancy multi-vitamins	83%	14%	3%	75%	18%	8%
Omega 3 (fish oils)	56%	35%	9%	62%	25%	15%
General healthy eating guidelines for pregnancy	100%	0%	0%	97%	3%	1%
Foods to avoid during pregnancy	100%	0%	0%	93%	7%	4%
Caffeine	100%	0%	0%	96%	4%	3%
Alcohol	100%	3%	0%	99%	4%	1%
Gestational weight gain	88%	12%	3%	72%	17%	11%

support and information, the extent to which this was tailored to young women's individual needs varied considerably [11]. There are significant barriers to providing individualised nutritional support particularly in terms of time pressures and a lack of appropriate resources [21]. It is therefore important that the need for post-registration training, both relating to the specific nutritional issues identified in this study and wider skills regarding how behaviour change techniques can be applied in this context [22].

There were differences in the resources provided by midwives and FNPs to young women and also in the resources used for their own information. These differences however largely reflect the different models of support and care offered by the different professions. This is further supported by open answer comments from young women, although indicating a high satisfaction and trust in midwifery care, expressing a sense of time limitation or insufficient resources to provide adequate and appropriate support during pregnancy. Young women were aware of risks and biases associated with accessing information via websites or commercial sources due to a lack of consistency and trustworthiness. The findings however were indicative of their desire to receive specific, robust, trustworthy and standardised information from authoritative sources; the potential for alternative digital formats is also indicated via mobile technology (e.g. apps) or interactive websites (e.g. easy to follow recipes, healthy eating guidance). Although they acknowledged mobile technology and digital sources should not replace face to face contacts, they appreciated additional support to allow continuous access, facilitate healthy dietary change and maximise benefits in between appointments. A lack of using online technologies either as tools to facilitate discussion or for helping young women to access further support was evident in the data collected from health care professionals. Young women have been found to routinely access online information [23] and support [24] during pregnancy; however research has suggested that health care professionals lack confidence in using these technologies effectively [25]. This suggests that further investigation to develop and evaluate dietary and lifestyle interventions delivered through mobile technology and interactive websites as an adjunct to care provided by health care professionals may be advantageous.

Limitations and strengths

The majority of young women respondents were aged over 20 years at the time of the survey; however young women were invited to participate if they aged between 16 and 20 during their pregnancies.

Young women were asked their current age and number of pregnancies, but were not asked to specify their age when they had their first child, so those '20 or over' at the time of the survey, may have had their first child as an adolescent.

Although we asked participants to respond if they were pregnant or recently have been pregnant, the time since the pregnancy may also have varied between participants meaning that some young women may be less likely to remember details about their diet or supplement use accurately. However, the overall dietary habit rather than details of dietary intake was explored which may have less of an impact on the availability of the information. Any pregnancy complications experienced may also have affected the young women's ability to recall information accurately however pregnancy complications were not recorded. With the exception that none of our participants were from Chinese origin, the participants' ethnic backgrounds were mainly in line with the national data as a great majority have identified themselves as white and a much smaller proportions were from black, mixed and Asian ethnic minorities (national data includes 2.0, 84.4, 2.7, 2.0, 2.1% respectively) [26].

Our study included a self-selected group of young pregnant mothers or those who recently gave birth and this limits the representativeness of our findings; this may have been a more motivated and interested group of young women to respond to such a survey. However, the variation in the included ethnicities in line with the national data regarding ethnicities and postcode data indicating inclusivity of all geographical regions within the UK is encouraging. This as well as the nature of provided information in identifying the poor quality of nutritional behaviour of a considerable proportion of young mothers and their desire for making healthy changes confirms the need for developing appropriate supportive interventions delivered through health care professionals.

Similarly with respect to the health care professionals' survey it was not possible to calculate a response rate or record the number or characteristics of non-responders to determine if they differed in any way from respondents. This may also have had an impact in that those professionals least comfortable with using online resources may have been less likely to complete the questionnaire. However using online networks allows a remote access with a possibility of open and honest answers without interference or the pressures sometimes felt in the face to face or research situations with close contacts [27].

Conclusions

A stated positive change in dietary intake by majority of young women in this survey indicates willingness to adapt a healthy lifestyle. This in addition to their trust in health professionals provides an opportunity for health interventions to improve birth outcomes. Avoiding or reducing foods such as red meat, eggs, oily fish or soft cheeses as major sources of required vitamins and minerals is of concern. Improving confidence among health professionals in using a range of resources, including online technologies, to support young women would be advantageous.

Implications for practice

Providing support for communicating appropriate knowledge and information regarding diet and supplement use to improve nutritional health of this vulnerable group of women is essential. Independent and trustworthy resources for women and professionals, available in a range of formats, providing up to date information which is accessible in between appointments would also be beneficial.

Abbreviations

FNP: Family nurse practitioner; NDNS: National diet and nutrition survey; NHS: National health service; SD: Standard deviation; UK: United Kingdom; YMCA: The young women at the local young men Christian association

Acknowledgements

First of all, we would like to thank all the young women who completed the online survey.

We would like to thank Joan Currie (Training Advisor) and colleagues and their Young Parents Programme learners at YMCA Training-Doncaster as well as Patricia Hurley (Teenage Pregnancy Specialist Midwife), Roz Morris (Family Nurse Partnership Supervisor) and her team from Doncaster and Bassetlaw NHS Foundations Hospitals for their input, in developing and piloting the online young women's questionnaires. Our special thanks to Katie Marvin-Dowle for her help on postcode data analysis and description of participants' geographical areas.

Our gratitude also extends to the steering group members for their support and encouragement during the project development: Becky Lang, Alison Hadley, Emma Sharp, Jacqueline Clinton, Mary Barker, Melissa Whitworth, Natalie Walker, Regina Steegers, Rachel Coss, Rebecca Jones, Simon Wheeler, Jane Brewin and Lucilla Poston.

We would also like to thank Tommy's and NIHR CLAHRC SY for supporting this project. This article presents independent research by the National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care for South Yorkshire (NIHR CLAHRC SY) a pilot which ended in 2013. Further details about the new NIHR CLAHRC Yorkshire and Humber can be found at www.clahrc-yh.nihr.ac.uk. The views and opinions expressed are those of the authors, and not necessarily those of the NHS, the NIHR or the Department of Health.

Funding

This project was funded jointly by Tommy's and the National Institute of Health Research Collaboration for Leadership in Applied Health Research and Care for South Yorkshire [NIHR CLAHRC SY].

Availability of data and materials

Data will be available through Sheffield Hallam University website.

Authors' contributions

HS was the project lead and managed the overall study progression and research governance approval. The questionnaire was designed initially by RR & AD and confirmed by HS, and data was collected by AD and RR. Qualitative data was predominantly analysed by RR and quantitative data predominantly analysed by AD and HS and all cross-validated by HS. All authors contributed to the paper writing, HS and KMD completed the paper draft, it was commented by AD and confirmed by RR too. All authors read and approved the final manuscript.

Competing interests

The authors declare that they have no competing interests.

Consent for publication

The data will be reported and publications will be in line with Data Protection Act, respecting full anonymity and confidentiality of information.

Ethics approval and consent to participate

This project was approved by Sheffield Hallam University Ethics Committee.

Author details

¹Maternal and Infant Health, Centre for Health and Social Care Research, Sheffield Hallam University, Sheffield, UK. ²Public Health Nutrition, Food and Nutrition Group, Sheffield Business School, Sheffield Hallam University, Sheffield, UK. ³Centre for Health and Social Care Research, Sheffield Hallam University, Sheffield, UK.

Received: 8 October 2016 Accepted: 14 February 2017 Published online: 01 March 2017

References

- Hall Moran VH. A systematic review of dietary assessments of pregnant adolescents in industrialised countries. Br J Nutr. 2007;97:411–25.
- Gilbert W, Jandial D, Field N, Bigelow P, Danielsen B. Birth outcomes in teenage pregnancies. J Matern Fetal Neonatal Med. 2004;16(5):265–70.
- Baker PN, Wheeler SJ, Sanders TA, Thomas JE, Hutchinson CJ, Clarke K, Poston L. A prospective study of micronutrient status in adolescent pregnancy. Am J Clin Nutr. 2009;89(4):1114–24.
- Chen XK, Wen SW, Fleming N, Demissie K, Rhoads GG, Walker M. Teenage pregnancy and adverse birth outcomes: a large population based retrospective cohort study. Int J Epidemiol. 2007;36(2):368–73.
- World Health Organisation. Adolescent pregnancy. Fact sheet N°364, September 2014. 2014. Available from: http://www.who.int/mediacentre/ factsheets/fs364/en/index.html. Accessed 05 Aug 2015
- Office for National Statistics. ONS analyses the latest teenage birth rates how does the UK compare?. 2014. Available from: https://www.ons.gov.uk/ peoplepopulationandcommunity/birthsdeathsandmarriages/livebirths/ datasets/birthsbyareaofusualresidenceofmotheruk. Accessed 31 July 2015
- Scholl TO, Hediger ML, Schall JI, Khoo CS, Fischer RL. Maternal growth during pregnancy and the competition for nutrients. Am J Clin Nutr. 1994; 60(2):183–8.
- Bates B, Lennox A, Prentice A, Bates C, Page P, Nicholson S, Swan G. National Diet and Nutrition Survey: Results from Years 1–4 (combined) of the Rolling Programme (2008/2009 – 2011/12). Public health England, PHE publications gateway number: 2014051. 2014. Available from: https://www. gov.uk/government/uploads/system/uploads/attachment_data/file/310997/ NDNS_Y1_to_4_UK_report_Executive_summary.pdf. Accessed 05 Aug 2015
- Burchett H, Seeley A. Good enough to eat: The diet of pregnant teenagers. London: The Maternity Alliance/The food Commission; 2003.
- 10. NHS Choices. Your health Your Choices. 2015. Available from: http://www. nhs.uk/Livewell/Goodfood/Pages/eatwell-plate.aspx. Accessed 05 Aug 2015
- 11. Rundle R, Duxbury A, Soltani H. An exploratory study of young women's food choices and compliance with supplementation recommendations during pregnancy. Sheffield: Sheffield Hallam University; 2014.
- Kupeka E, Petroub S, Vausec S, Maresh M. Clinical, provider and sociodemographic predictors of late initiation of antenatal care in England and Wales. Br J Obstet Gynaecol. 2002;109:265–73. Available from: http:// onlinelibrary.wiley.com/doi/10.1111/j.1471-0528.2002.00524.x/pdf. Accessed 04 Aug 2015
- Wise NJ, Arcamone AA. Survey of adolescent views of healthy eating during pregnancy. MCN: The American Journal of Maternal/Child Nursing, 36(6), 381–386.
- Office for National Statistics (2013) enumeration postcodes to 2011 output areas (OA), lower layer super output areas (LSOA), middle layer super output areas (MSOA) and local authority districts (LAD) (with split postcode indicator).
 2011. Available from http://www.ons.gov.uk/ons/guide-method/geography/ products/census/lookup/other/index.html. Accessed 05 Aug 2015
- 15. Barker DJP. In utero programming of chronic disease. Clin Sci. 1998;95(2): 115–28.
- Leppalahti S, Gissler M, Mentula M, Heikinheimo O. Is teenage pregnancy an obstetric risk in a welfare society? A population-based study in Finland, from 2006 to 2011. BMJ Open. 2013;3(8):e003225.
- 17. British Dietetic Association (BDA). Food Fact Sheets. 2014. Available from: https://www.bda.uk.com/foodfacts/home#nutrients. Accessed 04 Aug 2015
- Hovdenak N, Kjell Haram K. Influence of mineral and vitamin supplements on pregnancy outcome. Eur J Obstet Gynecol Reprod Biol. 2012;164:127–32.
- Arrish J, Yeatman H, Williamson M. Midwives and nutrition education during pregnancy: a literature review. Women Birth. 2014;27(1):2–8.
- Wennberg AL, Lundqvist A, Hogberg U, et al. Women's experiences of dietary advice and dietary changes during pregnancy. Midwifery. 2013;29(9): 1027–34.
- 21. Lucas C, Charlton KE, Yeatman H. Nutrition advice during pregnancy: do women receive it and can health professionals provide it? Matern Child Health J. 2014;18(10):2465–78.
- 22. Michie S, Abraham C, Whittington C, et al. Effective techniques in healthy eating and physical activity interventions: A meta-regression. Health Psychol. 2009;28:690–701.

- Bert F, Gualano MR, Brusaferro S, et al. Pregnancy e-health: a multi-center Italian cross-sectional study on internet use and decision-making among pregnant women. J Epidemiol Community Health. 2013;67(12):1013–8.
- Nolan S, Hendricks J, Trowell A. Social networking sites (SNS); exploring their uses and associated value for adolescent mothers in Western Australia in terms of social support provision and building social capital. Midwifery. 2015;31(9):912–9.
- Dalton JA, Rodger DL, Wilmore M, et al. "Who's afraid?": attitudes of midwives to the use of information and communication technologies (ICTs) for delivery of pregnancy-related health information. Women Birth. 2014; 27(3):168–73.
- Health and Social Care Information Centre, Hospital Episode Statistics: NHS Maternity Statistics - England, 2013–14. Table 6.b: Ethnic group by age of mother, 2013–14. 2015. Available from:http://www.hscic.gov.uk/ searchcatalogue?productid=17198&q=%22nhs+maternity +statistics%22&topics=0%2fHospital+care&sort=Relevance&size=10&page= 1#top. Accessed 04 Aug 2015
- 27. Van Gelder MM, Bretveld RW, Roeleveld N. Web-based questionnaires: the future in epidemiology? American journal of epidemiology, kwq291. 2010.

Appendix 11

Exploring the views of young women and their healthcare professionals on dietary habits and supplementation practices in adolescent pregnancy: a qualitative study.

Open Access



Exploring the views of young women and their healthcare professionals on dietary habits and supplementation practices in adolescent pregnancy: a qualitative study

Rachel Rundle^{1*}, Hora Soltani² and Alexandra Duxbury²

Abstract

Background: Nutrition is a modifiable factor affecting foetal growth and pregnancy outcomes. Inadequate nutrition is of particular concern in adolescent pregnancies with poor quality diet and competing demands for nutrients. The aim of this study was to explore knowledge and understanding of nutrition advice during adolescent pregnancy, and identify barriers and facilitators to dietary change and supplementation use in this vulnerable population.

Methods: Semi-structured interviews were conducted with young women and key antenatal healthcare providers: midwives, family nurses and obstetricians. Doncaster, Manchester and London were chosen as sites offering different models of midwifery care alongside referral to the Family Nurse Partnership programme.

Results: A total of 34 young women (adolescents aged 16–19 years) and 20 health professionals were interviewed. Young women made small changes to their dietary intake despite limited knowledge and social constraints. Supplementation use varied; the tablet format was identified by few participants as a barrier but forgetting to take them was the main reason for poor adherence. Health professionals provided nutrition information but often lack the time and resources to tailor this appropriately. Young women's prime motivator was a desire to have a healthy baby; they wanted to understand the benefits of supplementation and dietary change in those terms.

Conclusion: Pregnancy is a window of opportunity for improving nutrition but often constrained by social circumstances. Health professionals should be supported in their role to access education, training and resources which build their self-efficacy to facilitate change in this vulnerable population group beyond the routine care they provide.

Keywords: Adolescent pregnancy, Dietary habits, Nutrition supplementation, Behaviour change

Background

The increased risk of poor birth outcomes and adverse health consequences for mother and child are well documented for adolescent pregnancy [1-4]. Nutrition is a potentially modifiable risk factor but one that is challenging to address [5, 6] given the poor dietary patterns associated with adolescent young women, particularly those from socially disadvantaged groups, which experience higher rates of adolescent conception [7, 8]. Young women with

habitually poor dietary intake are at increased risk of nutritional deficiencies when entering pregnancy. Without dietary modification or supplemental nutrition interventions during pregnancy this may decrease the likelihood of their baby achieving an adequate birth weight [1].

Addressing diet and nutrition is central to improving birth outcomes and maintaining well-being throughout pregnancy. In the UK routine antenatal care for adolescent pregnancy prompts discussion of diet and nutrition at key points within the care pathway [9-11]; this covers nutrition supplementation, food-acquired infection (foods to avoid) and the provision of "*information on the benefits of a healthy diet and practical advice on how to eat healthily*



^{*} Correspondence: r.rundle@shu.ac.uk

¹Food and Nutrition Group, Sheffield Business School, Sheffield Hallam University, City Campus, Howard Street, Sheffield S1 1WB, UK Full list of author information is available at the end of the article

throughout pregnancy... tailored to the woman's circumstances" [12]. In the face of complex, competing social and emotional issues associated with adolescent pregnancy [8] the time available and approaches taken to addressing these factors effectively are less well known. Therefore, whilst the majority of young women will receive some nutrition guidance during their pregnancy it is unclear how much they understand, whether they make any changes to their diet as a result and the extent to which they adhere to recommendations for nutrition supplementation. For such a nutritionally vulnerable population it is important to understand how existing practices impact on their choices and the additional support required, for both young women and their ante-natal care teams.

Therefore the objectives for this study were; (i) to explore young women's knowledge and understanding of diet and nutrition advice during pregnancy; (ii) to identify whether they make any changes to their dietary intake and what prompts these changes; (iii) to understand young women's use of nutrition supplements before and during pregnancy; and (iv) to explore sources of information and support regarding diet and nutrition.

Methods

Study design

In this multicentre explorative study, the views of key antenatal care providers and a diverse population of pregnant adolescents were sought. In the UK midwives provide the majority of antenatal care with the youngest and most vulnerable being referred for obstetric support. In some areas the Family Nurse Partnership [13], provides additional care and support to young women assigning them to a family nurse practitioner for intensive intervention until their child is 2 years of age. Therefore, views of obstetricians, midwives, and family nurse practitioners were sought alongside young women. The data collection strategy and interview guide were developed in consultation with a research user group (comprising five young women (antenatal and postnatal) one midwife and one family nurse). The user group made valuable suggestions in terms of terminology (preferring terms such as "food you choose to eat" rather than "your diet", "healthy eating" rather than "nutrition" and "young women" or "young mums" rather than "teens" or "teenagers") interview structure and incentives which informed recruitment and data collection.

Settings

Doncaster, Manchester and the London boroughs of Lambeth and Southwark were chosen as research sites offering different models of service provision for pregnant young women. During the study period, Doncaster and Manchester offered community midwifery led services with one specialist midwife in young women's pregnancies for the service, locally referred to as the "teen pregnancy midwife" by other health professionals; this is noteworthy given the user group's preference in terminology for "young mums". London offered a specialist case-loading midwifery led continuity care model where young women were assigned to one key specialist midwife for their pregnancy with support from other members of the midwifery team. At the time of study the family nurse partnership followed the national model at all three research sites [13]. Obstetricians across the three sites provided specialist support for high risk pregnancies in young women.

Participants and recruitment

A purposive sampling strategy was employed for data collection. Selection criteria for young women were based on age and pregnancy status; 16-19 years whilst pregnant, currently pregnant or had given birth less than 6 months previous. Young women were provided with an information sheet and referred to the study by their midwife or family nurse practitioner based on our selection criteria; these details were confirmed whilst written consent was obtained. All young women were offered a £20 high street voucher in appreciation of their time. Health professionals were also purposefully sampled to include obstetricians and family nurse practitioners with the majority being midwives (specialists in young women, community and hospital). Interviews were conducted in quiet spaces within the ante-natal setting, in community venues or in young women's homes.

Data collection

Semi-structured interviews were designed to explore diet, supplementation use and sources of information and support during pregnancy. Interviews were conducted independently by two researchers and lasted between 10 and 42 min (mean time 25 min). The researchers were experienced in the collection of interview data and had prior experience working with young women from more socially disadvantaged backgrounds, both having previously worked in the National Health Service in community outreach programmes. A visual interview guide was used to keep conversations on topic and act as a focal point e.g. to limit distraction from mobile phones (see Additional file 1 Young Women's Interview Guide). The guide included photos of routinely used information sources, different types of nutrition supplements and tablet formats to demonstrate variations in tablet size; the same guide with minor changes to wording was used with the health professionals. During the interviews with young women the researchers sought to put them at ease and establish a rapport, reassuring them that their individual experiences were important and that there were no "right or wrong" answers. The majority of young women talked freely but some of the shorter interviews (<15mins duration) were a result of young women giving very brief responses. Young women that were pregnant at the time of interview were asked to focus on their experiences up to that point and postnatal young women were asked to think back to their pregnancy. Health professionals were asked to focus on their experiences and perceptions working with adolescent young women in their current role; this was particularly relevant for family nurse practitioners that had previously worked as midwives or health visitors. Interviews were audio recorded and professionally transcribed verbatim. The initial audio recordings and transcriptions were reviewed by both researchers for consistency and completeness.

Data analysis

The data was analysed thematically using the approach advocated by Braun and Clarke [14]. Through careful reading, data coding, categorisation of codes and constant comparative analysis, similarities and differences were explored in order to develop the overarching themes and subthemes. The main data analysis was managed by two researchers independently and verified by a third researcher (not one of the interviewers) to reduce the potential for bias and "sense check" during theme/sub-theme development thus reducing ambiguity of meaning. It was not the purpose of the analysis to compare different models of service provision at each research site rather highlight the aspects of care that were barriers to or facilitated young women to make healthier food choices and take their supplements.

Results

Fifty four in-depth interviews were conducted across the three study sites; young women (n = 34) and health professionals (n = 20). The characteristics for young women are presented in Table 1. All health professionals were involved in some aspect of antenatal care. Midwives made up the greatest proportion of the sample (n = 12) followed by family nurse practitioners (n = 5) and obstetricians (n = 3); of the midwife interviewed five had specialist adolescent pregnancy roles, two as *"teen pregnancy"* specialist for their locality (Doncaster and Manchester) and three with adolescent pregnancy caseloads (London).

Theme 1: A time for dietary change amidst confusion

(See Additional file 2: Table S2 of results for "Diet" theme and sub-themes).

Pregnancy was acknowledged by the majority of interviewees as a time for dietary change yet the opportunity to make healthier dietary choices was often hampered by misconceptions and entrenched food preferences. A *"typical teenage diet"* was how many health professionals referred to young women's food choices with lack of

Manchester and London)				
Variable	n (%)	mean		
Age (years)		17.62 years		
16	4 (12)			
17	13 (38)			
18	9 (26)			
19	8 (24)			
First pregnancy	33			
Second pregnancy	1			
Stage of pregnancy at interview				
Pregnant	24 (70)			

Gestational age (weeks)

Alone, own home (rented)

With family, at home

With partner, own home (rented)

With partner and family, at home

Shared house, hostel accommodation

Postnatal

Weeks

Living situation

understanding regarding the impact of dietary intake affecting their motivation to change. Midwives and family nurse practitioners often spoke about developing young women's "connection with their baby" as a means of initiating dietary change. Where young women made this connection, "All the way through, like, I just eat healthier because then your baby will come out healthier", more regular eating patterns and healthier food choices were reported. Conversely misaligned beliefs about what they ate and their baby's development were a barrier to positive change. Young women reported restricting food intake, motivated by a desire for a "smaller baby" in the belief that it would ease labour; health professionals highlighted the importance dispelling of these misconceptions.

Food safety advice often prompted dietary changes but young women could recall few specifics when asked; they admitted to feeling confused which led them to exclude certain foods such as tinned tuna, eggs, nuts and cheese, in particular processed soft cheeses,. Midwives expressed concern that some of these foods (e.g. cheese, eggs and tuna) could provide potentially cheap and useful sources of nutrients (e.g. calcium and protein). Yet despite their confusion young women's motivation was primarily to protect their baby; they believed consuming these foods may be harmful, even though they did not understand why, and they erred on the side of caution by excluding them completely from their diet.

28 weeks

14 weeks

10 (30)

3 (8.5)

10 (30)

12 (17)

6 (17)

3 (8.5)

Table 1 Participant characteristics for young women (aged 16–19 years) recruited across the three interview sites (Doncaster,
Manchester and London)

Health professionals often described young women's dietary choices as being of poor nutritional quality, expressing concern about missed meals and long periods of time between eating. Where young women reported making healthier choices these were not in the absence of junk food, crisps and confectionary but small positive changes to an overall poor diet; for example, one portion of fruit per day where none had been consumed prior to pregnancy, swapping carbonated sugary drinks for fruit juice, milk or water or eating breakfast.

Young women's living arrangements, access to cooking facilities and financial constraints all influenced food choice. Cooking was not a regular feature of their day and they relied on others cooking for them (mother, grandmother or partner's family). Some health professionals felt that living with family meant young women had a more balanced and nutritious diet but others felt this was not the case where the habitual family diet was reliant on take-away and convenience foods. Young women living alone or in temporary accommodation were least likely to cook for themselves. A few young women, particularly those with a supportive partner, spoke about taking more interest in cooking and trying new foods during pregnancy.

Young women's use of the Healthy Start voucher scheme [15] which provides monetary assistance for purchasing fruit, vegetables and milk during pregnancy was also explored during the interviews. They reported using the vouchers for fruit and milk or as contribution towards the household food shopping. The application process for the vouchers was a barrier and decreased motivation for some young women. One family nurse highlighted misuse of the vouchers as an issue in her locality although this was attributed to small local retailers. Whilst this was corroborated by some young women they were not concerned when they were able to exchange the vouchers for ineligible foods such as bread. Health professionals also felt that supporting development of shopping and cooking skills alongside use of Healthy Start vouchers could improve confidence to make small dietary changes in young women that in many cases are fending for themselves for the first time.

Theme 2: Erratic adherence to supplementation despite uncertainty

(See Additional file 3: Table S3 of results for "Supplementation" theme and sub-themes).

Use of nutritional supplements during pregnancy varied amongst the young women and was often challenging to explore when they were confused about what they were taking and why. Some young women denied taking any pregnancy supplements, associating them with *"medicine"* which they had been told to avoid. Often as interviews progressed they would talk about *"starter vits"* or *"folic something"* and by sensitive exploration the interviewer was able to elicit what supplements they were taking (if any), when and why.

Whilst the majority of young women reported taking a nutritional supplement at some stage of their pregnancy they often took them erratically and started later, towards the end of their first trimester. Later presentation of pregnancies was commonly reported by midwives and they were concerned the delayed initiation of folic acid supplementation resulted in the critical first trimester being missed. They felt being able to provide folic acid or Healthy Start vitamins directly to young women at first contact (8–10 weeks), rather than a prescription or application for Healthy Start, would provide an opportunity to reinforce the importance of taking supplements. One obstetrician expressed a desire to be more proactive in their approach to identifying those at risk of nutritional deficiencies earlier in pregnancy.

Young women were most knowledgeable of the recommendation for folic acid; prior to their initial contact with a midwife or GP, a supportive partner, family or friend was more likely to prompt taking folic acid than other types of supplement. Some young women did not realise the Healthy Start vitamins contained folic acid and were safe to use throughout pregnancy. They considered them "not safe after 12 weeks" and were concerned about "overdosing" or "harming my baby" if they continued. Young women described iron as being important for "blood or brain" but had poorer knowledge of vitamin D. Midwives described the routine haemoglobin test at 20 weeks as an opportunity for discussing dietary modification and supplementation where necessary. The use of multi-vitamins and fish oil supplements was less common but these were sometimes purchased by a well-meaning partner or relative. Young women considered them expensive but the premium packaging and marketing alongside these products made them wonder whether they were better than folic acid or Healthy Start vitamins alone. Some midwives also reported recommending multi-vitamin supplements in circumstances where they were particularly concerned about poor habitual dietary intake.

Health professionals frequently reported to young women's reluctance to take supplements in the tablet format but the most common reason for young women not taking them was simply forgetting. Some young women reported feeling nauseous when the tablet size was too large, (more commonly with folic acid/iron combinations and vitamin D) or when taking them on an empty stomach or with sugary carbonated beverages. Young women that were motivated to take their supplements set their own prompts such as putting the tablets in a memorable place (e.g. by their toothbrush, kettle or in their purse), setting an alarm on their mobile phone or enrolling a willing partner or mother to remind them. Midwives used results from routine blood tests to prompt young women to take their supplements and tried to incentivise them by promoting baby's healthy development and fewer pregnancy complications.

Alternative supplement formats were only made by two young women; these were for fortified products such as "healthy chocolate" or fruit smoothies. Few health professionals suggested alternatives, usually a fortified biscuit, but they acknowledged that it would be challenging to find an acceptable alternative with young women's varied food preferences. It was also noted that where healthcare trusts experienced difficulties with the supply and distribution of Healthy Start vitamins this would not be eased by a more costly and bulkier product to provide the necessary supplementation. In addition, it would be counter-intuitive to promote "healthy chocolate" or "healthy biscuits" when these are the foods young women try to reduce or substitute in an attempt to make small positive dietary changes.

Theme 3: Seeking relevant and reliable information and support

(See Additional file 4: Table S4 of results for "Information & Support" theme and sub-themes).

Midwives and family nurses felt that providing information and support on diet and nutrition was part of their role. Obstetricians described supporting the most vulnerable young women but assumed discussions about diet and supplementation would be provided by the midwifery teams. Partner, family and friends were highlighted as influential sources but the accuracy and consistency of their advice was questioned by both young women and health professionals. Midwives and family nurses recognised the trust that develops between themselves and the young women they care for. Yet the challenge of tailoring information to meet young women's needs, given the pressured environment in which they work was acknowledged by all health professionals. The majority of midwives were confident providing routine information but admitted that time constraints and other priorities meant conversations were brief and not always relevant to young women's situation. Midwives were frustrated they could not do more, particularly specialist "teen pregnancy" midwives that were the only person in that role for a particular locality. Health professionals identified a lack of confidence and practical skills as key barriers for young women putting dietary advice into practice. Family nurses acknowledged their unique position in having more time to spend with young women, which enabled them to tailor information and provide practical and emotional support. Family nurses spoke about their training and specially developed programme resources designed to help young women make the connection with their baby and offer a more holistic view of diet, supplementation and pregnancy.

Leaflets and booklets were routinely provided during antenatal appointments but there was mixed response to these. Some young women valued a resource they could keep and refer to after the appointment whilst others expressed apathy unless they had been developed specifically for adolescent pregnancy (e.g. Tommy's Young Women's Guide to Pregnancy). Midwives acknowledged their limitations and some reported to using them sparingly. However, young women's experience of receiving leaflets with little opportunity for discussion often left them feeling overwhelmed and confused. Alternative sources of information were explored with websites, apps for mobile devices, You Tube (video) clips and online discussion forums being the most accessible. Social media was frequently used by young women for seeking information and sharing their experiences. They valued the experiences of other young women, particularly when it came to practical tips and ideas for making changes, which were seen as more relevant to their needs. Health professionals acknowledged young women's prolific use of such social media platforms but were wary of user-generated content. Young women expressed concern about the reliability and accuracy of some online information. Midwives used online resources as part of their routine care but whilst most were familiar with branded sites such as NHS Choices [16] they were less familiar with other sources. Some midwives reported using You Tube videos, to deliver key messages in a more engaging manner.

Young women's understanding and interpretation of the information received during pregnancy had the potential to influence the changes they made. Making the connection with their baby appeared to influence young women's motivation to change; for some this was a natural result of being pregnant but others struggled to make the connection due to lack of physical changes, heightened concern with their own bodies changing (weight gain) or other competing priorities. Some health professionals felt that over and above providing information about healthy eating and supplements their responsibility lay in helping young women foster their connection with baby by developing an awareness of the impact their choices on their baby's health.

Discussion

This multi-centre qualitative study provides an in depth insight into young women's dietary behaviours and supplementation use during pregnancy. The findings from our study illustrate that pregnancy is a window of opportunity when young women are willing to change albeit constrained by social circumstance and lack of intrinsic motivation. Evidence of what works to improve the nutritional status for this vulnerable population group is
limited but through our analysis we have attempted to identify existing good practice, highlight the challenges and propose mechanisms for facilitating change; both behaviour change for young women and changes in practice for healthcare teams and the support they provide.

Making the connection with baby and enabling dietary change

Studies have shown that pregnancy is a time when women are motivated to make small changes to health behaviours, but that changes to dietary patterns are limited and the extent to which they impact on nutritional status unknown [17, 18]. A recent survey of dietary beliefs and behaviours of pregnant adolescents [19] found that whilst participants recognised the importance of dietary intake this knowledge did not always translate into behaviour. Yet our study recorded accounts of dietary change, albeit small, that were made despite lack of understanding, social support and financial constraints. Our young women expressed a desire to do the best for their baby, attempted and appeared more willing to change than health professionals gave them credit for. The extent to which our young women's dietary changes impacted on their nutritional status is unknown but reinforcing small steps in the right direction and increasing young women's self-efficacy to change would be a valuable approach for health professionals.

Studies have acknowledged midwives central role in the provision of nutrition education and counselling during pregnancy [20] and their role in motivating and effecting dietary change [21]. However, we found that whilst the majority of health professionals provided nutrition information the extent to which this was tailored to young women's needs and enabled change varied considerably. The barriers to providing adequate, timely and tailored nutrition support are acknowledged by ours and other studies; a directive, risk-reduction approach to providing dietary information [22] plus lack of time and appropriate resources [23] impact on women's experiences and motivation. Information overload without opportunity for discussion can result in young women feeling confused and overwhelmed. Health professional's nutrition knowledge and self-efficacy in tailoring information and support were also identified as important in our study. The need for improved post registration nutrition training and skills development for midwives [20, 24] plus a better understanding of how this can impact on practice and pregnancy outcomes [25, 26] has been identified. Understanding of specific techniques and approaches with regards to dietary behaviour change, particularly within disadvantaged and vulnerable population groups is still limited. Supporting health professionals to effectively direct the nutrition education and support they provide is essential to enable sustained dietary changes. Therefore, drawing on wider understanding of how different behaviour change techniques can be employed to effect dietary change [27, 28] alongside improved nutrition education would be worthy of exploration.

Pregnancy resources as an adjunct to care and in an accessible format

It was acknowledged in our study that there already exists a wealth of resources to support healthy choices in pregnancy but these are not always tailored to young women's needs or available in accessible formats. Information that is representative of young women's experiences and acts to complement health professionals in their role is needed. We found that leaflets were routinely used to provide information on diet and supplementation with mixed response from young women; whilst some young women welcomed a resource they could take home and read at a later time, often, especially where these were provided without opportunity for discussion, young women were left feeling overwhelmed and confused, which mirrors findings from other studies [21, 23, 29]. Studies have shown that where time constraints and other pressures result in leaflets being provided without opportunity for discussion this reinforces hierarchical relationships [30] which could limit young women's motivation and self-efficacy to change. Supporting health professionals in their role by utilising new communication networks, such as web-based and mobile technology, could be a means of easing time constraints by increasing accessibility of tailored information beyond routine antenatal contact.

The use of the internet, particularly by young women, is widely accepted as a means of seeking information during pregnancy [31, 32] and has been shown to be effective as a means of providing emotional support for isolated young mothers [33]. Similar to the health professionals in our study other research has found that health professionals are more likely to express reservations and lack of skills in using technologies effectively [32]. Drawing on existing evidence of how adolescents use social media to access health information [34] and how health literacy skills can impact on accessibility and use of online information [35] are all worthy of further exploration within pregnant adolescents. The potential for digital and social media opportunities to support dietary behaviour change have been acknowledged [36] although specific approaches and methodologies are still developing.

Supplementation – Understanding benefits, barriers and access to pregnancy vitamins

Poor adherence to recommendations for pregnancy supplementation is not unique to adolescents and studies have highlighted women's negative attitudes and ambivalence

[37–39]. Yet our findings show that where women understand supplementation recommendations in terms of their baby's health and development, and identify practical strategies to prompt use, these barriers may be overcome. Our findings also echo those found in a study exploring the uptake of Healthy Start vitamins and vouchers in England [40] where health professionals experienced difficulties with the supply and access to vitamins which reduced uptake of vitamins in those eligible for the scheme. Universal provision of Healthy Start vitamins to all pregnant women has been proposed [40] but this was not considered cost effective in a recent review by NICE [41]. Other studies have found that direct provision of folic acid to women, in addition to counselling and reminder phone call, increased weekly intake of folic acid [37]. More research is needed to establish whether the mode of accessing vitamins (direct provision from health professionals rather than the existing application and assessment process) plus behavioural counselling can increase uptake and use, particularly in nutritionally vulnerable populations.

Strengthening the evidence base for effective dietary intervention

Whilst there is evidence that intervention to improve nutrition through foods, fortified foods and nutrition counselling may have benefits in terms of birth outcome, particularly for low income and nutritionally vulnerable populations [42, 43], there is a dearth of evidence for effective interventions in pregnant young women and adolescents in particular. A recent systematic review of nutritional interventions and birth outcomes with this population group identified only five studies eligible for inclusion and only two of the five reported significant effect on birth outcome [44]; one study using zinc supplementation reducing the likelihood of low birth weight and the other increasing average birth weight with four servings of dairy each day. The challenge of implementing interventions to change dietary behaviour in a particular group has been acknowledged and recognises the need to also change the behaviour of those delivering the intervention [45]. More research is needed which combines improving knowledge, skills and self-efficacy in health professionals to facilitate behaviour change with targeted dietary outcomes based on the available evidence.

Strengths and limitations of the study

The strength of this research is the combined exploration of young women's and health professional's experiences of dietary change and supplementation use in adolescent pregnancy; we have also considered their different perspectives in terms of access to information and support and their relevance to young women's needs. We have considered how the synergies and differences between young women's and health professional's views can provide insights into changes in practice and where additional research is needed. In qualitative studies such as this expectations of the researchers prior to the data collection should always be acknowledged. In this instance, based on literature reviews and anecdotal evidence from health professionals, young women were assumed to have limited interest in nutrition and resistance to dietary change during pregnancy. However, the researchers approached the interviews with an open mind and a willingness to explore young women's experiences and they found their initial pre-conceptions challenged. Adolescent pregnancies, as represented by the young women interviewed, should not be viewed as a homogenous group whose nutritional needs can be met by a simple one-size fits all intervention. The diversity of opinion and willingness to change within the young women interviewed reinforced the researcher's belief that their experiences should be heard and better ways of supporting them, during pregnancy and beyond, should be explored.

Limitations of the research primarily concern the age and representativeness of the young women in the interview sample; the youngest (< 16 years) and potentially most vulnerable (those not accessing routine antenatal care) may be under-represented. However, the method of recruitment enabled midwives and family nurse practitioners to refer some of their most vulnerable young women to the research team and the use of high street vouchers as an incentive may have provided encouragement to participate irrespective of individual's prior knowledge or interest in nutrition.

To ensure reliability of the methods for data collection the researchers (interviewers) worked closely together checking the questioning style and topic coverage within the interview guide. However, there will be inevitable differences in style and rapport of the interviewer with the study participants and it is acknowledged that this may have had an impact of the reliability of data collected. Reliability of the data analysis and interpretation of codes into themes and sub-themes was aided by the oversight of the third researcher to ensure the decisions made were clear and transparent. Throughout the analysis process discussions were had to ensure that the personal viewpoints and experiences of the researchers were recognised but did not unduly impact on the impartial interpretation of the findings. We have sought to represent the experiences and viewpoints of all participants in this research as truthfully as possible to ensure the validity of our findings; whilst they may not be generalisable across all adolescent pregnancies and approaches to healthcare in the ante-natal setting we believe they offer valuable insights.

Conclusion

Supporting young women to improve dietary quality and adhere to recommendations for pregnancy supplementation

is complex. The habitual diet of young women and their families, combined with a lack of knowledge, confidence, skills and encouragement contribute to entrenched behaviours that are difficult to change. Diet may be low on the list of priorities for young women and their health professionals, particularly when faced with more immediate health and social concerns. Yet, as so many of the young women in our study reported, pregnancy is a time when they are prepared, willing and motivated to change, however small those changes may be. Whilst acknowledging that young women face numerous challenges, their prime motivator is a desire to have a healthy baby; they need to understand the benefits of supplementation and how healthier dietary choices can enhance nutrition and outcomes for their baby in those terms. Health professionals should be supported in their role to access education, training and resources which build their knowledge, skills and confidence to facilitate change in this vulnerable population group beyond the routine care they provide. This is an opportunity not to be missed as the potential benefit for mother and baby are lifelong, including improved pregnancy outcome and healthier habits into childhood and beyond.

Additional files

Additional file 1: Young Women's Interview Guide. (PDF 1208 kb) Additional file 2: Table S2. Diet theme, sub-themes and coding with illustrative quotes from young women and health professionals. (DOCX 22 kb)

Additional file 3: Table S3. Supplementation theme, sub-themes and coding with illustrative quotes from young women and health professionals. (DOCX 20 kb)

Additional file 4: Table S4. Information and support theme, subthemes and coding with illustrative quotes from young women and health professionals. (DOCX 20 kb)

Acknowledgments

The authors would like to thank all the young women, midwives, family nurse practitioners and obstetricians who gave up their time to participate in the interviews. We would like to thank YMCA Training-Doncaster and their Young Parents Programme learners, as well as Teenage Pregnancy Specialist Midwife and Family Nurse Partnership supervisor for their input, in developing and piloting the interview guide. Our gratitude also extends to Tommy's charity Diet & Teenage Pregnancy steering group members for their support and encouragement during the project development.

Thanks also to Katie Marvin-Dowle, PhD student in the Centre for Health and Social Care Research, Sheffield Hallam University for the final formatting of this document.

Funding

This research was funded jointly by Tommy's charity and the National Institute of Health Research Collaboration for Leadership in Applied Health Research and Care for South Yorkshire (NIHR CLAHRC SY). The views expressed are those of the author(s), and not necessarily those of the Tommy's charity, NHS, the NIHR or the Department of Health and Social Care. Further details about the new NIHR CLAHRC Yorkshire and Humber can be found at www.clahrc-yh.nihr.ac.uk.

Availability of data and materials

The anonymised datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

Authors' contributions

RR, HS and AD developed the study design. All interviews were conducted by RR & AD; also reviewed transcripts. Data was analysed by RR and AD; themes, sub-themes and meaning were verified by HS. The manuscript was written and prepared by RR with contributions and revisions of successive drafts by HS. All authors read and approved the final manuscript.

Ethics approval and consent to participate

Ethical approval to conduct the interviews was sought and granted by the NRES Committee East Midlands-Derby (13/EM/0082), in addition to full compliance with local R&D frameworks at each research site. Written informed consent was obtained from all participants.

Consent for publication

Consent to use anonymised data in publication of this research was obtained alongside consent for participation.

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Food and Nutrition Group, Sheffield Business School, Sheffield Hallam University, City Campus, Howard Street, Sheffield S1 1WB, UK. ²Centre for Health and Social Care Research, Faculty of Health and Wellbeing, Sheffield Hallam University, Sheffield, UK.

Received: 30 May 2018 Accepted: 25 October 2018 Published online: 12 November 2018

References

- Lechtig A, Habicht JP, Delgado H, et al. Influence of maternal nutrition on birth weight. Am J Clin Nutr. 1975;28(11):1223–33.
- Barker DJP. In utero programming of chronic disease. Clin Sci. 1998;95:115–28.
 Baker PN, Wheeler SJ, Sanders TA, et al. A prospective study of
- micronutrient status in adolescent pregnancy. Am J Clin Nutr. 2009;89(4): 1114–24.
- Gilbert W, Jandial D, Field N, et al. Birth outcomes in teenage pregnancies. J Matern Fetal Neonatal Med. 2004;16(5):265–70.
- Moran VH. A systematic review of dietary assessments of pregnant adolescents in industrialised countries. Br J Nutr. 2007;97(3):411–25.
- Abu-Saad K, Fraser D. Maternal nutrition and birth outcomes. Epidemiol Rev. 2010;32:5–25.
- Whitton C, Nicholson SK, Roberts C, et al. National Diet and nutrition survey: UK food consumption and nutrient intakes from the first year of the rolling programme and comparisons with previous surveys. Br J Nutr. 2011;106(12): 1899–914.
- McCarthy FP, O'Brien U, Kenny LC. The management of teenage pregnancy. BMJ. 2014;349:g5887.
- National Institute for Health and Care Excellence (NICE). Maternal and child nutrition NICE Guidelines PH11 2008 – updated. 2014. https://www.nice.org. uk/guidance/ph11. Accessed 21 May 2018.
- National Institute for Health and Care Excellence (NICE). Antenatal care for uncomplicated pregnancies. NICE Guidelines CG62. 2008 – updated. 2016 https://www.nice.org.uk/guidance/cg62. Accessed 21 May 2018.
- National Institute for Health and Care Excellence (NICE). Pregnancy and complex social factors: a model for service provision for pregnant women with complex social factors. NICE Guidelines CG110. 2010. https://www.nice. org.uk/guidance/cg110. Accessed 21 May 2018.
- National Institute for Health and Care Excellence (NICE). Lifestyle advice on diet and physical activity: advice on diet and use of supplements before and during pregnancy. NICE Pathway - Diet. https://pathways.nice.org.uk/ pathways/diet. Accessed 21 May 2018.
- 13. Family Nurse Partnership (FNP). http://fnp.nhs.uk/. Accessed 21 May 2018.
- Braun V, Clarke V. Using thematic analysis in psychology. Qual Res Psychol. 2006;3(2):77–101.
- Healthy Start (NHS). https://www.healthystart.nhs.uk/. Accessed 21 May 2018.

- NHS Choices Healthy diet in pregnancy pregnancy and baby guide. http:// www.nhs.uk/Conditions/pregnancy-and-baby/pages/healthy-pregnancy-diet. aspx. Accessed 21 May 2018.
- 17. Crozier SR, Robinson SM, Borland SE, et al. Do women change their health behaviours during pregnancy? Findings from the Southampton Women's study. Paediatr Perinat Epidemiol. 2009;23(5):446–53.
- Crozier SR, Robinson SM, Borland SE, et al. Women's dietary patterns change little from before to during pregnancy. J Nutr. 2009;139(10):1956–63.
- Whisner CM, Bruening M, O'Brien KO. A brief survey of dietary beliefs and behaviors of pregnant adolescents. J Pediatr Adolesc Gynecol. 2016;29:476–81.
- Arrish J, Yeatman H, Williamson M. Midwives and nutrition education during pregnancy: a literature review. Women Birth. 2014;27(1):2–8.
- Wennberg AL, Lundqvist A, Hogberg U, et al. Women's experiences of dietary advice and dietary changes during pregnancy. Midwifery. 2013;29(9): 1027–34.
- Wennberg AL, Hörnsten Å, Hamberg K. A questioned authority meets wellinformed pregnant women – a qualitative study examining how midwives perceive their role in dietary counselling. BMC Pregnancy Childbirth. 2015; 15:88. https://doi.org/10.1186/s12884-015-0523-2.
- Lucas C, Charlton KE, Yeatman H. Nutrition advice during pregnancy: do women receive it and can health professionals provide it? Matern Child Health J. 2014;18(10):2465–78.
- 24. Beldon A, Crozier S. Health promotion in pregnancy: the role of the midwife. Perspect Public Health. 2005;125(5):216–20.
- Gijbels H, O'Connell R, Dalton-O'Connor C, et al. A systematic review evaluating the impact of post-registration nursing and midwifery education on practice. Nurse Educ Pract. 2010;10(2):64–9.
- Girard AW, Olude O. Nutrition education and counselling provided during pregnancy: effects on maternal, neonatal and child health outcomes. Paediatr Perinat Epidemiol. 2012;26(suppl 1):191–204.
- Michie S, Abraham C, Whittington C, et al. Effective techniques in healthy eating and physical activity interventions: a meta-regression. Health Psychol. 2009;28:690–701.
- Abraham C, Michie S. A taxonomy of behavior change techniques used in interventions. Health Psychol. 2008;27:379–87.
- Ferrari RM, Siega-Riz AM, Evenson KR, et al. A qualitative study of women's perceptions of provider advice about diet and physical activity during pregnancy. Patient Educ Couns. 2013;91:372–7.
- Stapleton H, Kirkham M, Thomas G. Qualitative study of evidence based leaflets in maternity care. BMJ. 2002;324:639.
- Bert F, Gualano MR, Brusaferro S, et al. Pregnancy e-health: a multi-center Italian cross-sectional study on internet use and decision-making among pregnant women. J Epidemiol Community Health. 2013;67(12):1013–8.
- Dalton JA, Rodger DL, Wilmore M, et al. "Who's afraid?": attitudes of midwives to the use of information and communication technologies (ICTs) for delivery of pregnancy-related health information. Women Birth. 2014; 27(3):168–73.
- Nolan S, Hendricks J, Trowell A. Social networking sites (SNS); exploring their uses and associated value for adolescent mothers in Western Australia in terms of social support provision and building social capital. Midwifery. 2015;31(9):912–9.
- Yonker LM, Zan S, Scirica CV et al. "Friending" teens: systematic review of social media in adolescent and young adult healthcare. J Med Internet Res Published online 2015 Jan 5. doi: https://doi.org/10.2196/jmir.3692.
- 35. Gray NJ, Klein JD, Noyce PR, et al. The internet: a window on adolescent health literacy. J Adolesc Health. 2005;37(3):243.e1–7.
- McGloin AF, Eslami S. Digital and social media opportunities for dietary behaviour change. Proc Nutr Soc. 2014;74(2):139–48.
- Robbins JM, Cleves MA, Collins HB, et al. Randomized trial of a physicianbased intervention to increase the use of folic acid supplements among women. Am J Obstet Gynecol. 2005;192(4):1126–32.
- Barbour RS, Macleod M, Mires G, et al. Uptake of folic acid supplements before and during pregnancy: focus group analysis of women's views and experiences. J Hum Nutr Diet. 2012;25:140–7.
- Barrett G, Shawe J, Howden B, et al. Why do women invest in prepregnancy health and care? A qualitative investigation with women attending maternity services. BMC Pregnancy Childbirth. 2015;15(236).
- 40. Jessiman T, Cameron A, Wiggins M, et al. A qualitative study of uptake of free vitamins in England. Arch Dis Child. 2013;98:587–91.
- 41. National Institute for Health and Care Excellence (NICE). Healthy Start vitamins: special report on cost effectiveness. NICE corporate document

ECD5. 2015. https://www.nice.org.uk/corporate/ecd5/chapter/executivesummary. Accessed 21 May 2018.

- 42. Hollowell J, Oakley L, Kurinczuk JJ, et al. The effectiveness of antenatal care programmes to reduce infant mortality and preterm birth in socially disadvantaged and vulnerable women in high income countries: a systematic review. BMC Pregnancy Childbirth. 2011;11:13.
- Gresham E, Byles JE, Bisquera A, et al. Effects of dietary interventions on neonatal and infant outcomes: a systematic review and meta-analysis. Am J Clin Nutr. 2014;100:1298–321.
- Soltani H, Duxbury A, Rundle R, et al. A systematic review of dietary interventions on neonatal outcomes in adolescent pregnancy. Evid Based Midwifery. 2014;13(1):29–34.
- Atkins L, Michie S. Designing interventions to change eating behaviours. Proc Nutr Soc. 2015;74:164–70.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

